

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Kersten, Wolfgang (Ed.); Blecker, Thorsten (Ed.); Ringle, Christian M. (Ed.)

Proceedings

The Road to a Digitalized Supply Chain Management: Smart and Digital Solutions for Supply Chain Management

Proceedings of the Hamburg International Conference of Logistics (HICL), No. 25

Provided in Cooperation with:

Hamburg University of Technology (TUHH), Institute of Business Logistics and General Management

Suggested Citation: Kersten, Wolfgang (Ed.); Blecker, Thorsten (Ed.); Ringle, Christian M. (Ed.) (2018) : The Road to a Digitalized Supply Chain Management: Smart and Digital Solutions for Supply Chain Management, Proceedings of the Hamburg International Conference of Logistics (HICL), No. 25, ISBN 978-3-7467-6535-8, epubli GmbH, Berlin, https://doi.org/10.15480/882.1777

This Version is available at: https://hdl.handle.net/10419/209194

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by-sa/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





Proceedings of the Hamburg International Conference of Logistics (HICL) – 25

Wolfgang Kersten, Thorsten Blecker and Christian M. Ringle (Eds.)

The Road to a Digitalized Supply **Chain Management**





HICL Proceedings

Editors: Kersten, W., Blecker, T., Ringle, C.M. and Jahn, C. 2018 **The Road to a Digitalized Supply Chain Management Logistics 4.0 and Sustainable Supply Chain Management**

Editors: Kersten, W., Blecker, T., Ringle, C.M. and Jahn, C. 2017 Digitalization in Supply Chain Management and Logistics ISBN 978-3-7450-4328-0 Digitalization in Maritime Logistics and Sustainable Supply Chain Management ISBN 978-3-7450-4332-7

Editors: Kersten, W., Blecker, T. and Ringle, C.M. 2015

Innovations and Strategies for Logistics and Supply Chains ISBN 978-3-7375-6206-5 Sustainability in Logistics and Supply Chain Management ISBN 978-3-7375-4057-5 Operational Excellence in Logistics and Supply Chains ISBN 978-3-7375-4056-8

Editors: Kersten, W., Blecker, T. and Ringle, C.M. 2014

Innovative Methods in Logistics and Supply Chain Management ISBN 978-3-7375-0341-9 Next Generation Supply Chains *ISBN* 978-3-7375-0339-6

Editors: Kersten, W., Blecker, T. and Ringle, C.M. 2013

Pioneering Solutions in Supply Chain Performance Management ISBN 978-3-8441-0267-3 **Sustainability and Collaboration in Supply Chain Management** ISBN 978-3-8441-0266-6

...find more proceedings on https://hicl.org/publications

Proceedings of the Hamburg International Conference of Logistics (HICL) 25

The Road to a Digitalized Supply Chain Management

Smart and Digital Solutions for Supply Chain Management

Prof. Dr. Dr. h. c. Wolfgang Kersten Prof. Dr. Thorsten Blecker Prof. Dr. Christian M. Ringle *(Editors)* This work is licensed under the Creative Commons Attribution-ShareAlike 4.0 International License. To view a copy of this license, visit https://creativecommons. org/licenses/by-sa/4.0/ or send a letter to Creative Commons, PO Box 1866, Mountain View, CA 94042, USA.

Edition	1 st edition, September 2018
Publisher	epubli GmbH, Berlin, www.epubli.de
Editors	Wolfgang Kersten, Thorsten Blecker, Christian M. Ringle
Cover (Photo)	Christoph Mahlstedt – unsplash.com/@_senator_
Layout	Luis Brunswick Franco, Francisco Williams Riquer
ISBN	978-3-746765-35-8
ISSN (print)	2635-4430
ISSN (online)	2365-5070

Contents

Preface	vii			
I Advanced Manufacturing and Industry 4.0	1			
Running the Machine Faster: Acceleration, Humans and Warehousing 3 Tony Cragg, Thierry Sauvage, Mohammed Haouari, Sarrah Chraibi, Oussama El Khalil Houssaini				
Supplier Integration in Industry 4.0 – Requirements and Strategies 23 Julian Marius Müller, Johannes W. Veile, Kai-Ingo Volgt				
Scope for Industry 4.0 in Agri-food Supply Chains				
Advanced Scientific Algorithms in Digital Factory Design Applications 57 Jacob Lohmer, Armin Klausnitzer, Rainer Lasch				
Additive Manufacturing in Supply Chains – The Future of Purchasing Processes 79 Stephanie Niehues, Laura Berger, Michael Henke				
II Innovation and Technology Management	97			
Efficiency Analysis for Digitalised Working Systems of Truck Drivers 99 Dominic Loske, Matthias Klumpp				
Alternative Development Paths for Supply Chains in 2030 121 Denis Daus, Ana Cristina Barros, Dimitra Kalaitzi, Victoria Muerza, Irene Marchiori				
About the Future Role of Software in the Product				

Contents

Environmental Innovation of Transportation Sector in OECD Countries Duygu Şahan, Okan Tuna	157
Design of an Added Plan with Social Responsibility	171
III Supply Chain Analytics	195
Multi-Method Decision Support Framework for Supply Network Design Giuseppe Timperio, Robert de Souza, Boy Panjaitan Bernado, Sumit Sakhuja, Yoseph Sunardhi	197 i
Business Model of Aircraft Fleet Planning using ANN	221
IV Risk and Security Management	249
Adequate Flexibility Potential to handle Supply Chain Uncertainties \ldots .	251
Remote Sensing in Humanitarian Logistics: An Integrative Approach Christian Hein, Henning Hünemohr, Rainer Lasch	271
Impact of Managerial Risk-taking and IRM on Innovation	291
Enhanced FMEA for Supply Chain Risk Identification	311

Preface

Digitalization is changing the way organizations manage their supply chain and their daily logistical processes. The development of digitalized solutions and industry 4.0 have created a completely new business ecosystem. Additionally, customers are demanding more innovative, more diverse and greener products. This creates numerous challenges for all actors in the supply chain; yet, they also present an opportunity to create solutions and practices that improve performance and productivity.

This year's edition of the HICL proceedings complements the last years' volume: Digitalization in Supply Chain Management and Logistics. All entities along the supply chain are challenged to adapt new business models, techniques and processes to enable a smooth transition into a digitalized supply chain management.

This book focuses on core topics of digitalization in the supply chain. It contains manuscripts by international authors providing comprehensive insights into topics such as industrial internet of things, digital factory design, risk management or aircraft fleet planning and provide future research opportunities in the field of supply chain management. All manuscripts contribute to theory development and verification in their respective research area.

We would like to thank the authors for their excellent contributions, which advance the logistics research process. Without their support and hard work, the creation of this volume would not have been possible.

Hamburg, September 2018

Prof. Dr. Dr. h. c. Wolfgang Kersten Prof. Dr. Thorsten Blecker Prof. Dr. Christian M. Ringle

Part I

Advanced Manufacturing and Industry 4.0

Running the Machine Faster: Acceleration, Humans and Warehousing

Tony Cragg¹, Thierry Sauvage¹, Mohammed Haouari¹, Sarrah Chraibi¹, Oussama El Khalil Houssaini¹

1 - Ecole Supérieure de Logistique Industrielle

To reduce the lead-time, modern logistics seeks to respond faster by accelerating physical and information flows. However, what are the impacts on logistics workers of an ever-faster logic? The purpose of this paper is to examine the relationship between process acceleration and the autonomy of order pickers. The method is to use exploratory qualitative research, based on fifteen visits to different regional distribution centers (RDCs) in the retail supermarket sector. The contribution of this paper is to apply Rosa's (2013) social acceleration theory to the specific context of logistics warehousing and to demonstrate how speeding up order picking systems is a key driver of change that has an impact on worker autonomy.

Keywords: Acceleration; autonomy; warehousing; order picking

First recieved: 28.May.2018 Revised: 19.Jun.2018 Accepted: 03.Jul.2018

1 Introduction

In France, the warehousing and transport sector employs 1.37 million people (Stratégies Logistique 2017) and the surface of the regional distribution centers (RDCs) of supermarkets now doubles in size every two years. Three reasons can explain this rapid growth in the sector: supply, internal capacity and demand. Firstly, on the supply side, the growth in world trade, maritime shipping and the trend towards global sourcing in the last forty years has resulted in more and more goods circulating that require an effective logistic industry to make them available to the final customer.

Secondly, technological advances enable the automation of processes and mechanization in RDCs. The demand for automated warehouse systems is worldwide. In the USA, forecasters expect the robotics market in warehousing and logistics to have increased more than tenfold to \$22.4 billion by 2021, up from \$1.9 billion in 2016 (Yale Materials Handling 2017).

Thirdly, on the demand side, the advent of e-commerce and omni-channel distribution has led to the further expansion of the sector, making the modern warehouse the place "where the virtual becomes physical" (Moore, 2018). Information technology now instantaneously relays point of sales information to the distribution center, requiring demand-pull systems to be more and more responsive. Indeed, following the acceleration of information flows, the acceleration of physical flows becomes essential in order to fulfil the demand promise.

Owing to earlier advances in communications technology and digitalization, accelerated information flows predate accelerated physical flows and this has impacted the order picking process in the intervening period. Pick-by-voice systems, introduced in the late 2000's, were symptomatic of speeding up information flow technology, while we are only now seeing the installation of fully automated and mechanized warehousing systems more widely in the sector. The consequence of this desynchronization of the two flows has been borne by warehouse order pickers. Before the introduction of pick-by-voice systems, order pickers used their knowledge and skills to plan a route around the warehouse and stack their pallet in an efficient manner. With its introduction, algorithm-based software instruct workers via headsets which product to pick next: "the savoir-faire of order pickers has been reduced to a physical engagement" (Gaborieau 2012, p.1). While process acceleration increases productivity, it can also have consequences for human operators. Therefore, the research question is as follows: what is the relationship between the acceleration of warehouse processes in RDCs and the autonomy of order pickers? Fig.1 presents the research focus. We choose to focus on order picking, because it is an important and expensive warehouse operation that is either labor or capital intensive (Gu et al. 2007). As such, from a technical perspective it has been the subject of performance evaluation studies with a view to optimization (Gu et al. 2010). A research gap exists because, although studies into the role of humans in warehouses can be found from a sociological perspective (Gaborieau, 2012; 2016), logistics research into this subject is rare. One exception is a literature review by Grosse et al. (2015) which found that researchers' order picking planning models have focused on cost efficiency rather than on human operators. They describe the human factor as the "missing link" in order picking system design.

To explore this missing link and its relationship with process acceleration, we use exploratory qualitative research methods, involving 15 visits to warehouse sites managed by four brand name RDCs and two leading third party logistics providers. A questionnaire about order picking was administered at one of the sites and discussions were held with managers. We structure this paper as follows. Firstly we examine the literature related to acceleration theory, warehousing, desynchronization, dynamic capabilities and the notion of worker autonomy within the context of social sustainability. Then we explain the research methodology and present the findings and propositions. This leads to a discussion and conclusion.



Research focus

Figure 1: Schematic illustration of subject studied

2 Literature Review

2.1 Acceleration Theory

Writers have focused on speeding up movement as a defining characteristic of modern capitalism: "by far the greatest effect of industrialization...was to speed up a society's entire material processing system" (Beniger 1986, p.427) and "everything that requires a long time lasts too long and everything that asks for time asks for too much time" (Rosa 2013, p.155). Underlying this drive for speed is a systems theoretical approach that concerns itself with "the securing of a cease-less renewal of the elements of the system [...] not static, but dynamic stability" (Luhmann 1996, p.79).

From the perspective of critical theory, Rosa (2013) has developed a theory of social acceleration that relates to three domains. Firstly, technological accel-

eration is found in production and transport and is defined as "the intentional acceleration of goal-directed processes" (2013, p.74). Secondly, the acceleration of social change is defined as a contraction of the present in all areas of life and a growing instability of our time horizons and expectations. Finally, the pace of life speeds up despite the increased free time that technological acceleration should enable, as a result of a scarcity of time resources. Importantly, he asserts that technological acceleration not only alters our experience of space and time (put simply, things seem to be happening more quickly), it also changes the quality and quantity of our social relationships. Acceleration can be viewed as the antonym of 'depth' in relationships. For our purposes, this analysis is interesting because it relates technological acceleration, on the one hand, to its possible impacts on humans, on the other. Before we explore this connection in more detail, it is necessary to consider acceleration in the context of warehousing.

2.2 Acceleration and Warehousing

Acceleration is an imperative of the modern RDC for three main reasons. Firstly, because the quantity of goods moving through a given site is steadily increasing, due to rising demand and the advent of multiple distribution channels. In order to maintain performance levels, there is no choice other than to speed up the order fulfillment process. Secondly, commodities (and above all perishable commodities) progressively lose economic value for the producer the longer they are in storage. The role of the logistic warehouse is therefore to minimize the time goods spend immobile and to speed up processing time, thereby reducing inventory costs through faster rotation. Finally, margins are tight and competition is in the supermarket sector, so advantage can be gained through investing in speeding up processes and replenishing supermarket shelves with the right products rapidly.

For these three reasons of growth, value and competition, the notion of stocks/stores/storage as something stable or fixed, or as provisions set aside until need arises, is now outmoded. Gu et al. (2006) define the major roles of warehousing as buffering and consolidation. We can add that the underlying logic of RDCs is movement, not immobility. In a picker-to-parts order picking system minimizing the order retrieval time is the main priority, since it has been estimated that order picking comprises as much as 55% of warehouse operating costs (de Koster et al. 2007). The sooner an order is ready for shipping the better. In order to speed up manual order picking, travel time and therefore travel distance around the warehouse has to be reduced to a minimum. This can be achieved by layout, grouping and storage assignment practices and by augmenting the work of the order picker by linking him/her to IT systems via headsets and microphones.

2.3 Desynchronization

However, warehouse acceleration risks the desynchronization of processes and functions (Rosa 2013) – a serious risk, given the importance of synchronizing flows for coordination, as advocated in the logistics literature (Simatupang et al. 2002). For example, speeding up warehouse materials handling will have an impact on inbound and outbound transport flows. Queuing and bottlenecks can occur because of improvements to one flow. Desynchronization, not only applies to material and data flows, but also to organizational functions: introducing an Enterprise Resource Planning (ERP) system, for example, will not produce the required results if the warehouse operations department is not appropriately trained in the new software. A reconfigured supply chain only operates at the speed of its slowest sub system with potentially dysfunctional consequences for the system as a whole (Rosa 2013).

Attempts to accelerate through automation and improved labor productivity always face risks, which are both technical and human. Because data processing and transmission now outpace material handling systems, the relative slowness of the latter becomes the weak link in the chain, demanding attention. Desynchronization and non-compatibility are an inevitable consequence of an acceleration of one part of the system, necessitating a holistic upstream and downstream vision of the whole system. For warehouse management, piloting parallel merging flows that function at different speeds becomes essential to avoid zero benefit from accelerating one of the flows.

2.4 Dynamic Capabilities

Efficient logistics increases the volume of transactions and availability of goods by managing time and eliminating barriers to circulation. The objective of a logistics warehouse is to minimize the lead time (the time taken from order reception to product delivery) by accelerating processing time. In the academic literature, a company's capacity to accelerate its supply chain is presented as "dynamic capabilities" (Teece et al. 1997; Beske et al. 2014). Specifically, Eisenhardt and

Martin (2000) define dynamic capabilities as "organizational and strategic routines by which firms achieve new resource configurations". Logistics capabilities help to build competitiveness for organizations. Therefore, in the highly competitive retail sector, where margins are tight, the capability to accelerate processing time enables a firm to gain a competitive advantage over another.

Helfat and Peteraf (2003) introduced the concept of the capability lifecycle to develop further this notion of dynamic capabilities. In the same way that products have a lifecycle of growth, maturity and decline, so too do capabilities. Therefore, logistics processes are capabilities that develop and eventually lose their ability to provide a competitive supply chain advantage. Acceleration theory, postulating that society is in an incessant cycle of speeding up processes, suggests that the lifecycle of logistic capabilities in the modern RDC is becoming shorter and shorter, as the organizational environment becomes increasingly turbulent and complex, requiring adaptability and regular reconfiguration.

Reconfiguring resources in the food industry is seen as essential, given the constant changes in consumer demands (Wiengarten et al. 2011; Trienekens et al. 2012) and the need to respond to them. Faster communications technology cycles and big data (Waller and Fawcett 2013) mean that firms seeking to maintain or gain market share, have to constantly monitor, evaluate and reconfigure their resources. The pull flow logic of demand chain management places the final customer as the driving force that the modern logistics warehouse aims to satisfy through product availability via multiple delivery channels:

"The implication of today's turbulent and unpredictable business environment is that demand chain solutions are required. That is, we need solutions that are flexible and capable of responding rapidly to structural change on both the supply side and the demand side of the business" (Christopher and Ryals 2014, p.29).

'Responding rapidly to structural change' involves accelerating warehouse processes. Yet the 'ever-faster' logic raises important questions about the social sustainability of the modern warehousing sector and the impact on the people who work in it.

2.5 Social sustainability and autonomy

This article considers the relationship between the acceleration of warehouse processes and the autonomy of workers. Of the three sustainability dimensions

(social, environmental and economic), Ahmadi et al. (2017) show that social sustainability in supply chains has been under-researched, compared with environmental and economic sustainability. They conclude that this is "a research topic that will only gain in importance in years to come" (p.105). The importance of the human dimension as a research agenda is confirmed by Wieland et al. (2016) in data collected from 141 SCM researchers. After analyzing the difference between what should and what will become important, the people dimension of SCM was ranked the most underestimated research theme out of 35 subjects, followed in second place by ethical issues. They write: "Feedback from participants notes that supply chains are not "soulless machines," but complex socio-technical systems involving cognitive elements and impacted by face-to-face negotiations and conversations" (Wieland et al. 2016, p.207).

In a study of ten cases of sustainable supply chain management exemplars Pagell and Wu (2009) found that sustainable firms invest in human capital, aim to increase employee wellbeing, enhance organizational commitment and maintain a culture that values people and the environment. Workers in these organizations described their employers as thoughtful, caring, and committed. Varsei et al. (2014) evaluated the social performance of partners in a global supply chain. They focused on the four primary social dimensions specified in the Global Reporting Initiative (GRI 2012), namely, labor practices and decent work conditions, human rights, society and product responsibility.

2.6 Defining autonomy

In the warehousing context in France, where arguably, labor rights and systems of social protection exist, the focus for researchers into social sustainability is primarily on the experience of working conditions (Gaborieau 2012; 2016) and in the case of this research, on job autonomy. This has been defined as the degree of control that workers have over their own work situation (Brey 1999) and as spheres of independence that are directly or indirectly delegated by the organization to employees (Katz 1966).

In seeking to explain the paradox of disempowered industrial employees collaborating and engaging in a firm's activities, Katz (1966) argued that it was the undefined time left to workers within work time, to bring their culture into the bureaucratic workplace, which rendered the work tolerable for employees. In other words, worker autonomy engenders integration into an organization, through allowing a continuity between non-work life and the working life. Therefore, reducing worker autonomy in a tightly controlled work environment, with little time for association, could negatively affect worker commitment to the organization.

Importantly, Brey (1999) noted that even if organizations limit goal setting by workers, deciding on the means to achieve those goals can still provide autonomous spaces for meaningful and rewarding work. However, Brey describes three ways in which autonomy can be compromised. Firstly, monitoring and constant surveillance, enhanced by digitalization, remove moral autonomy and cause a loss of a sense of dignity and a perception of outside judgement. Similarly, task prestructuring by computer-defined systems imposes conformity on the employee and reduces his/her scope for freedom of action and decision-making. Finally, new computer systems create dependency on third parties, such as managers or system operators, who possess the necessary skills to install and maintain the technology, whereas the worker does not.

Vidal (2013) argues that there is a connection between an acceleration in the pace of work and low-autonomy work particularly in highly competitive sectors: the faster the process, the less time the worker has to decide what action to take or to communicate with colleagues. Where, as in the case of order pickers, a firm emphasizes and rewards the speed of a worker to complete a task accurately, little value is seen in allowing autonomous worker input.

This review of the literature relating to acceleration, autonomy and warehousing is summarized in Fig.2 below:



Figure 2: Proposed conceptual model for RDC acceleration

3 Methodology

Warehousing academic literature considers design, operation and performance evaluation, rather than understanding and contextualizing working conditions or the transformation of the modern warehouse. In a literature review of warehousing research, Davarzani and Norrman (2015, p.15) find that the "results of this study reveal a lack of reality-based investigations. Most of the scholars focus on quantitative research methods and mathematical modeling without any examples from real cases". They conclude: "that more empirical investigations should be conducted to understand and capture complexities of the real environment". Warehousing literature review articles (Gu et al. 2007; 2010; Davarzani and Norrman 2015) reveal an absence of theoretical frameworks and an emphasis on operational and technical solutions without reference to sustainability issues.

A systems approach rests on positivist assumptions of objective reality and independence from context. Such an approach, aimed at process optimization and improving productivity, encourages the progressive introduction of technological solutions, seen as neutral. Alternatively, complexity theory (Nilsson and Gamelgaard 2012) aims to take into account the diversity of human involvement in the organization of logistic processes.

The difference between the two approaches is made clear if we consider the question of self-organization or autonomy – to what extent is a worker able to plan his/her workload his/herself. For a systems approach, self-organization brings uncertainty and needs to be minimized. On the other hand, complexity theory recognizes that human intervention is an integral and inevitable part of the logistics process.

Tuble 1. Visits to wateriouse sites during research				
Firm	Sites and visits	Particularity of site		
RDC A	3 sites and 4 visits	Ambient products only. Mecha- nized zone in each site.		
RDC B	I site and 2 visits	6 warehouses for fresh, frozen and ambient products on one site, including large fully mechanized warehouse.		
RDC C	I site and 2 visits	Site handling all product types, but shortly due for closure due to re- gional reorganization.		
RDC D	I site and 2 visits	Ambient and fresh products. Site shortly due for closure due to regional reorganization. Order picking questionnaire adminis- tered here.		
3PL A	3 sites visited once	3PL specializing in fresh and frozen products. Clients include leading supermarket brands.		
3PL B	2 sites visited once	Client is a leading supermarket brands. One site due to close shortly due to contract termina- tion.		

Table 1: Visits to warehouse sites during research

The primary objective of this exploratory research is to consider the consequences of acceleration for the autonomy of order pickers. The research is based on 15 visits to different warehouse sites managed by six different firms (see Table 1) in France in 2017 and 2018, connected with the supervision of logistics students on internships. The selection of sites is random, since the logistics school receives offers of internships from firms and the researcher is allocated to supervise a certain number of students. Each visit to a site lasted between two and four hours and included a tour of the warehouse itself, detailed explanations of site operations and discussions with managers. After each visit, notes were made to keep a record of the principal observations. The research has been supplemented by

discussions with and reports from logistics students on internships. Additionally a questionnaire was administered at RDC D that focused on managers' and order pickers' evaluations of pick-by-voice. The warehouses visited either handled frozen, fresh or ambient goods or in some cases all three types. The sites were also at different stages of automation and mechanization and had different strategies for their implementation. Supermarkets ran the majority of the sites visited, while specialist third party logistics service providers (3PLs) ran the minority.

4 Research findings

This chapter sets out four findings from this initial exploratory research, which are pertinent to the question of RDCs and acceleration and lead onto propositions, intended as possible future research directions. Table 2 sets out these propositions. The research initially focused on the impact of flow acceleration on worker autonomy. It has led to extra findings illustrating further consequences of acceleration on warehouse processes and organization.

Concepts	Linked Propositions			
Acceleration and autonomy	P1: Accelerating processes by intensifying an order picker's work rate reduces worker autonomy.			
Desynchronization	P2: Desynchronization in warehouse processes is an inevitable consequence of speeding up flows, making global flow coordination essential.			
Human Resources	P3: Process acceleration makes human resource management in RDCs more not less important.			
Supply Chain	P4: Acceleration accentuates the role of RDCs as inward-looking, performance-focused and constrained logistics operations.			

Table 2: Summary of propositions

4.1 Accelerating order picking and worker autonomy

While other types of order picking exist (Richards 2011), the sites visited presented two types of acceleration of the order picking process. To start with the more recent, in France mechanized zones have been introduced in warehouses in the current decade, either as specific enclosed zones in a part of an existing site or as a whole building unit. These are defined as zones where all processes are mechanical and automated without human intervention in the sorting and picking process, except in a maintenance role. An enclosed zone handles heavier packages. It accepts full pallet loads, separates them and then prepares full pallet loads as ordered. While these specific zones have a high productivity rate, this is a capital-intensive solution to order picking and the return on investment is estimated at five years or more.

The second solution dates from the 2000's and is known as pick-by-voice software. A headset and microphone link order pickers to computer software, which determines the order of pallet preparation. BCP software estimate an increase in order picker productivity of 15% with pick-by-voice technology. However, it increases the workload, leading to concerns about risks to health (Anact 2010).

To ascertain the appreciation of pick-by-voice technology, a questionnaire was administered to 8 managers and 24 order pickers at RDC D. This revealed different evaluations of the pick-by-voice technology by the two groups. Managers' average score out of ten was 7.4, while workers gave a score of 5.1. Managers appreciate the technology because it enables tracking of activity, optimizes picking routes, reduces picking errors and improves productivity. From a human resource management perspective, it enables an accurate planning of the number of pickers needed each day. It also leads to better ergonomics for the worker, who now has his/her hands and eyes free.

The order pickers, on the other hand, found that the computer voice leads to a sentiment of dehumanization. During a visit to this site by the author, the manager asked a worker to explain how the headset and picking process works. He replied: "I just follow orders like an idiot". Order pickers using pick-by-voice also reported a feeling of being monitored and controlled; an increased workload leading to tiredness; limited possibility of communication between colleagues; a diminution of collective working; no global visibility of an order, making the job less interesting. The technology itself was criticized for frequent malfunctions, failures of the network, headaches and lack of comfort caused by wearing the headset all day and the regular repetitions needed to communicate with the software.

These findings confirm those of Gaborieau (2012; 2016) that pick-by-voice accelerates the pace of work, renders the work repetitive, reduces the opportunity for socializing and increases the weight carried per day. The first proposition relates to the human consequences of acceleration.

P1: : Accelerating processes by intensifying an order picker's work rate reduces worker autonomy.

4.2 Merging flows and desynchronization

At a site managed by RDC A, a manager presented the following problem of four different types of flow, operating at different speeds, both push and pull, some predictable and others not. These flows have to merge to be loaded onto the same truck, requiring piloting to minimize delays. The first flow is that of order

pickers in pull flow using the pick-by-voice technology to stack pallets and deliver to the loading bays. The second flow is cross docking, where goods arrive from other RDCs of the same retailer for immediate dispatch to stores in the region. The third flow concerns special offers, launched by the centralized marketing department and operating in push flow.

Finally, the newly constructed mechanized zone, operating in pull flow, adds complexity to the site, for three main reasons. Firstly because there has to be a very careful selection of references that are suitable for this zone, which must respect both the specific materials handling criteria and the required pace of entry of goods into the zone. Secondly, because the zone handles 30% of references and the number of full pallet loads entering and exiting the zone is high, extra flows circulate within the same warehouse space. Finally, because the transfer of completed pallets from the mechanized zone to the loading bays is carried out by automatic guided vehicles (AGVs or robots). However, the route of AGVs from the mechanized zone to the loading bays at the site and impedes the progress of order pickers and forklift drivers, whose pay is performance linked. (At another site of this group with the same configuration, this had led to incidences of sabotage of the AGVs by workers.)

Speeding up one flow or creating a new one and finding a solution to one problem - in this case that of heavy loads of 10 to 15 kilos, which can now be handled by the mechanized zone and not order pickers manually – has a knock on effect and sets up new challenges to be resolved. The shortening of dynamic capabilities lifecycles (Helfat and Peteraf 2003) suggests that the desynchronization of flows and the need to audit and pilot flows effectively are set to recur more frequently. The second proposition is about desynchronization. P2: Desynchronization in warehouse processes is an inevitable consequence of speeding up flows, making global flow coordination essential.

4.3 The roles of humans in warehousing

Changing flows in RDCs has an impact on the organization of work tasks. Because management realizes that order picking is an unattractive task, many of the sites visited had moved towards greater flexibility or polyvalence. In general, while order pickers were content to be trained to take on the role of forklift drivers, the inverse was not the case. One manager at RDC site D acknowledged this problem and refused to accept that staff could choose not to do order picking. It was clear

that within the site requiring staff to do order picking as part of their different tasks had become a delicate issue.

Sites visited had different levels of temporary staff, reflecting the recruitment difficulties that affect the sector. At RDC site C, due to close in 2018 and be relocated, the percentage of temporary staff had reached 50% and was making the task of management in planning and organizing a workforce difficult. The best run sites also had the lowest levels of temporary staff. The overall impression from the visits to the 15 different sites was one of a sector undergoing rapid change. Some sites were due to close as part of a national restructuring programme. Other sites were in the process of introducing mechanized zones and robotic systems. RDC site B had recently completed a fully mechanized warehouse for full pallet loads, where the only humans are those in the truck loading and unloading areas and maintenance workers. Some managers were aware of the impact of these changes on workers and stressed the need for retraining, upskilling and recruiting more highly qualified staff. This leads to the third proposition that concerns human resources. P3: Process acceleration makes human resource management in RDCs more not less important.

4.4 The bow tie metaphor

The structure of a supply chain, in which RDCs operate, can be likened to a bow tie, as they are at the center of high volume inbound and outbound flows. (Most RDCs stock around 10.000 different product references.) Here, the main feature of the bow-tie metaphor is that there are complex and variable inputs or inbound flows, that a compact core accepts, then recomposes and distributes what has been stored to a wide variety of destinations. Two observations can be made about the impact of acceleration on the organization of the supply chain. Firstly, due to the need to accelerate and reconfigure processes, the pressure to reach performance targets, the difficulty in recruiting and retaining gualified staff and their position at the center of massive inbound and outbound flows, RDCs focus purely on the management of internal flows and arriving and departing transport. A 50,000m2 warehouse that serves as a conduit for many suppliers and customers only has contact with them in matters directly related to flow management, such as packaging problems. In concerning themselves primarily with their own logistics, RDCs exemplify the strict division of labor and functions along the supply chain and the rigid boundaries that characterize this sector of activity.

The second observation relates to what Carter et al. (2015), in their development of a theory of the supply chain, refer to as the "horizon or visibility boundary". These authors suggest as a formal premise that: "the supply chain is bounded by the visible horizon of the focal agent". Although RDCs are central nodes in the supply chain, they are more hidden from view than visible, bounded more by confidentiality and security than openness. Moore (2018) comments: "It is tempting to say that these buildings make the internet visible, except that their visibility is strictly limited". It would be interesting to ascertain suppliers' and customers' level of knowledge of RDC operations, since they represent the next node in the supply chain. Furthermore, has the acceleration of processes in RDCs led to greater or lesser visibility of them by suppliers and customers? If the latter is the case and there is less visibility, following Carter et al.'s premise, can RDCs be viewed as part of supply chains or are they more accurately described as specialized constrained logistics operations. This leads us to the fourth proposition that relates to supply chain structure. P4: Acceleration accentuates the role of RDCs as inward-looking, performance-focused and constrained logistics operations.

5 Discussion and Conclusion

This exploratory research has shown that speeding up information flows through technology in labor-intensive order picking processes reduces worker autonomy and that process acceleration is an underlying logic of the modern RDC. In this final chapter, we discuss further the propositions made and consider where this might lead a warehousing research agenda.

To optimize information and physical flows the boundaries or borders, in the widest meaning of the terms, between and within firms have to be managed – boundaries between buyers and suppliers or between different departments within the same firm or between different zones in the same warehouse. For flows of goods and information to operate efficiently, boundaries have to be almost invisible or frictionless, soft rather than hard. The management of logistic processes across and within organizations aims to be seamless and the boundaries blurred.

Boundaries reflect the constraints imposed in the functioning of organizations and exist for purposes of control, channeling or connecting (Mezzadra and Nielsen 2012). However, process acceleration puts these boundaries under stress, disturbs established configurations, provokes desynchronization (P2) and requires the reorganization of different flows. The disruptive power of acceleration sets in motion a series of impacts, both positive and negative, anticipated and unforeseen, that bring into question the stability of existing boundaries (Hernes and Paulsen 2003). As Vakkayil (2012, p.206) has observed: "In constantly changing environments it is impossible to draw permanent lines of demarcation".

We have noted that one of the imperatives driving acceleration in warehousing is economic value loss. Rosa (2013, p.163) describes the time goods spend in storage and distribution as time when "the realization of created surplus value is delayed". He sees one of the basic systemic problems of capitalist economies as the need to maintain accelerated circulation to avoid such value loss. He argues that it is for this reason that logistics has to be more technically advanced than production – to ensure that the sphere of potential value loss does not negatively affect the whole value creation process and eventually, value capture.

Through applying Rosa's theory of social acceleration to RDCs, the contribution of this exploratory research is to demonstrate that there is an ongoing tension between the systemic need for acceleration in logistics warehousing and the existing boundaries or constraints that have been negotiated and established in the supply chain and in the workplace. An example of these tensions, presented in this paper, is the autonomy of order pickers (P1), defined either as control over the work situation (Brey 1999) or as spheres of independence (Katz 1966). A future case study research agenda could examine in more detail how this tension between the established boundaries, which allow a degree of worker autonomy in warehousing and the imperative to accelerate is evolving. Additionally the nature of relationships between RDCs and suppliers/customers, and the changes to these relationships and supply chain structure associated with RDC process acceleration (P4) could be studied. Finally, by moving away from technical performancebased optimization of order picking systems, as suggested by Grosse et al. (2015), research could consider the future role of human resources in warehouses (P3) and provide exemplars of valuing human input.

References

(Anact), A. nationale pour l amelioration des conditions de travail (2010). "Preparateurs de commandes quelles pistes de prevention".

(gri), G. R. I. (2012). The GRI Annual Report 2011/2012. Available at.

Ahmadi, H. B., S. Kusi-Sarpong, and J. Rezaei (2017). "Assessing the social sustainability of supply chains using Best Worst Method". In: Resources, Conservation & Recycling 126, pp. 99–106.

- Beniger, J. (2007). "The control revolution". In: Communication in history: Technology, culture, society, pp. 301–311.
- Beske, P., A. Land, and S. Seuring (2014). "Sustainable supply chain management practices and dynamic capabilities in the food industry: A critical analysis of the literature". In: International Journal of Production Economics 152, pp. 131–143.
- Brey, P. (1999). "Worker Autonomy and the Drama of Digital Networks in Organizations". In: Journal of Business Ethics 22, pp. 15–25.
- Carter, C. R., D. S. Rogers, and T. Y. Choi (2015). "Toward the Theory of the Supply Chain". In: Journal of Supply Chain Management 51.2, pp. 89–97.
- Christopher, M. and L. J. Ryals (2014). "The Supply Chain Becomes the Demand Chain". In: Journal of Business Logistics 35.1, pp. 29–35.
- Davarzani, H. and A. Norrman (2015). "Toward a relevant agenda for warehousing research: literature review and practitioners' input". In: Logistics Research 8.1, pp. 1–18.
- De Koster, R., T. Le-Duc, and K. J. Roodbergen (2007). "Design and control of warehouse order picking: a literature review". In: European Journal of Operational Research 182.2, pp. 481–501.
- Eisenhardt, K. M. and J. A. Martin (2000). "Dynamic capabilities: what are they?" In: Strategic management journal 21.10-11, pp. 1105–1121.
- Gaborieau, D. (2012). "«Le nez dans le micro». Répercussions du travail sous commande vocale dans les entrepôts de la grande distribution alimentaire". In: La nouvelle revue du travail 1.
- Gaborieau, D. (2016). "Des usines à colis: trajectoire ouvrière des entrepôts de la grande distribution". PhD thesis. Paris 1.
- Grosse, E. H., C. H. Glock, and W. P. Neumann (2015). "Human factors in order picking system design: a content analysis". In: *IFAC-PapersOnLine* 48.3, pp. 320–325.
- Gu, J., M. Goetschalckx, and L. F. McGinnis (2007). "Research on warehouse operation: a comprehensive review". In: European Journal of Operations Research 177.1, pp. 1–21.
- Gu, J., M. Goetschalckx, and L. F. McGinnis (2010). "Research on warehouse design and performance evaluation: a comprehensive review". In: *European Journal of Operations Research* 203.3, pp. 539–549.
- Handling, Y. M. (2017). The ROI of Robotics: Financial and Operational Sense.
- Helfat, C. E. and M. A. Peteraf (2003). "The dynamic resource based view: capability lifecycles". In: Strategic Management Journal 24, pp. 997–1010.
- Hernes, T. and N. Paulsen (2003). "Introduction: boundaries and organization". In: Managing boundaries in organizations: multiple perspectives. Springer, pp. 1–13.
- Katz, F. E. (1965). "Explaining informal work groups in complex organizations: the case for autonomy in structure". In: Administrative Science Quarterly 10.2, pp. 204–223.

 $Luhmann, N. (1996). ``Soziale Systeme". In: {\it Grundriss einer all gemeinen Theorie}. Frankfurt: Suhrkamp.$

- Mezzadra, S. and B. Neilson (2012). "Between inclusion and exclusion: On the topology of global space and borders". In: *Theory, Culture & Society* 29.4-5, pp. 58–75.
- Moore, R. (2018). A shed the size of a town:? what Britain's giant distribution centres? tell? us about modern life.
- Nilsson, F. and B. Gammelgaard (2012). "Moving beyond the systems approach in SCM and logistics research". In: International Journal of Physical Distribution and Logistics Management 42.8/9, pp. 764–783.
- Pagell, M. and Z. Wu (2009). "Building a more complete theory of sustainable supply chain management using case studies of 10 exemplars". In: *Journal of Supply Chain Management* 45.2, pp. 37–56.

Running the Machine Faster: Acceleration, Humans and Warehousing

- Richards, G. (2017). Warehouse management: a complete guide to improving efficiency and minimizing costs in the modern warehouse. Kogan Page Publishers.
- Rosa, H. (2013). Social Acceleration: A new Theory of Modernity. New York: Columbia University Press.
- Simatupang, T. M., A. C. Wright, and R. Sridharan (2002). "The knowledge of coordination for supply chain integration". In: Business Process Management Journal 8.3, pp. 289–308.
- Teece, D. J., G. Pisano, and A. Shuen (1997). "Dynamic capabilities and strategic management". In: Strategic Management Journal 18.7, pp. 509–533.
- Trienekens, J. H., P. M. Wognum, A. J. M. Beulens, and J. G. A. J. van der Vorst (2012). "Transparency in complex dynamic food supply chains". In: Advanced Engineering Informatics 26, pp. 55–65.
- Vakkayil, J. D. (2012). "Boundaries and organizations: a few considerations for research". In: International Journal of Organizational Analysis 20.2, pp. 203–220.
- Varsei, M., C. Soosay, B. Fahimnia, and J. Sarkis (2014). "Framing sustainability performance of Supply Chains with multidimensional indicators". In: Supply Chain Management: An International Journal 19.3, pp. 242–257.
- Vidal, M. (2013). "Low-autonomy work and bad jobs in postfordist capitalism". In: *Human Relations* 66.4, pp. 587–612.
- Waller, M. A. and S. E. Fawcett (2013). "Data science, predictive analytics, and big data: a revolution that will transform supply chain design and management". In: *Journal of Business Logistics* 34.2, pp. 77–84.
- Wieland, A., R. B. Handfield, and C. F. Durach (2016). "Mapping the landscape of future research themes in supply chain management". In: Journal of Business Logistics 37.3, pp. 205–212.
- Wiengarten, F., M. Pagell, and B. Fynes (2011). "Supply chain environmental investments in dynamic industries: Comparing investment and performance differences with static industries". In: International Journal of Production Economics 135.2, pp. 541–551.

Supplier Integration in Industry 4.0 – Requirements and Strategies

Julian Marius Müller¹, Johannes W. Veile², Kai-Ingo Volgt²

1 - Salzburg University of Applied Sciences

2 - University of Erlangen-Nürnberg

Cross-company networking is essential to successfully implement Industry 4.0. In this context, numerous new demands on suppliers arise leading to integration challenges that require specific integration strategies. While these topics are important in business practice, an aggregated holistic overview is still missing. Therefore, this article examines new demands on suppliers, challenges in the implementation process, and integration strategies for supplier integration in the context of Industry 4.0. Expert interviews with 15 different industrial companies from the sectors mechanical and plant engineering, electronics and electrical engineering, automotive, and information and communication technology serve as an empirical basis. This study provides insights into the challenges and strategies of supplier integration, helps academia to understand this topic, and indicates need for future research. Furthermore, this paper develops implications for corporate practice in the area of supplier management.

Keywords: Industry 4.0; Industrial Internet of Things; Supplier Integration; Integration Strategies

First recieved: 17.May.2018 Revised: 07.Jun.2018 Accepted: 26.Jun.2018

1 Introduction

Industry 4.0 aims to establish intelligent, self-managing, and interconnected industrial value creation to ensure future competitiveness of the manufacturing industry (Kagermann et al., 2013; Kang et al., 2016; Lasi et al., 2014). Both research and practice have mainly focused on technological developments and the technical implications for value creation so far (Brettel et al., 2014; Emmrich et al., 2015; Kagermann et al., 2013; Kans & Ingwald, 2016; Kowalkowski et al., 2013; Liao et al., 2017; Rennung et al., 2016). Furthermore, almost exclusively large companies have been the focus of academia and corporate practice (Kowalkowsi et al., 2013; Radziwon et al., 2014).

Up to the present day, Industry 4.0 is primarily thought within the boundaries of a company and consequently present efforts aim at implementing the concept in a company. Yet, the predicted potential of intelligent interconnected value-added processes can only be exploited in its entirety by interconnecting companies and value chains resulting in networks (Kagermann et al., 2013; Lasi et al., 2014). Therefore, the implementation of the holistic concept Industry 4.0 also requires a holistic approach.

However, integration processes across company boundaries pose numerous challenges. First, such integration requires openness, willingness to cooperate, and compatible technologies on all sides (Kiel et al., 2017; Müller et al., 2018a). Second, the integration of suppliers requires, e.g., them to have the necessary infrastructure and drive forward the implementation of Industry 4.0. As suppliers are often small and medium-sized enterprises (SMEs), some might have neither the necessary resources nor access to the required knowledge (Icks et al., 2017, Müller et al., 2018b). Therefore, especially SMEs need to find adequate partners for cooperation (Müller et al., 2017).

The aim of this study is to analyze how to integrate suppliers in the context of Industry 4.0. The following research questions are proposed:

- (1) Which requirements need to be met when integrating suppliers?
- (2) Which challenges arise when integrating suppliers?
- (3) Which strategies can be used to integrate suppliers?

Shedding light into the research object is important for several reasons. The way suppliers are integrated into the value creation process of a company has an impact on the extent to which possibilities Industry 4.0 can be used at all (Siepmann, 2016).

Supplier integration represents a source of differentiation and can therefore help to create sustainable competitive advantages. Last but not least, versatile, realtime-optimized, and autonomous cross-company value creation networks can only be established through adequate supplier integration, which is the overriding goal of Industry 4.0 (Bauernhansl, 2014).

2 Theoretical background

2.1 Industry 4.0

The term "Industry 4.0" refers to a paradigm shift in industrial value creation. It is particularly widespread in the German-speaking world (Burmeister et al., 2015; Lasi et al., 2014), while the term "Industrial Internet of Things" is particularly used in the Anglophonic world (Hartmann & Halecker, 2015; IIC, 2017). The origin of Industry 4.0 dates back to the year 2011 and was significantly influenced by the work of Kagermann et al. (2011) in the context of the Hanover Fair. Furthermore, they published implementation recommendations in 2013 in a final report of the Working Group Industry 4.0 (Kagermann et al., 2013).

In the age of industrialization, technical innovations repeatedly led to paradigm shifts that are called "industrial revolutions" ex-post (Lasi et al., 2014). The first industrial revolution began at the end of the 18th century and was characterized by the mechanization of the value creation process and the use of water and steam power. Dated back to the beginning of the 20th century, the second industrial revolution was characterized by mass production through assembly line production, the application of the Taylor principle of division of labor, and the use of electrical energy. The use of electronics and information technology to automate and digitize production heralded the third industrial revolution in the 1970s. All industrial revolutions have led to an increasing degree of complexity of the production systems (Bauernhansl, 2014; Kagermann et al., 2013; Kelkar et al., 2014; Lasi et al., 2014). It is predicted that the present economy is at the beginning of a fourth industrial revolution, summarized by the term Industry 4.0. This new paradigm shift is characterized by a digital interconnection and virtualization of the industrial value creation process (Bauernhansl, 2014; Kagermann et al., 2013; Kelkar et al., 2014; Lasi et al., 2014). For the first time in history, a change of paradigm is announced a priori (Drath & Horch, 2014). For this reason, Industry 4.0 is to be understood as a vision whose potential can be realized in the future (Drath & Horch, 2014, Lasi et al., 2014). However, as the technical foundations have existed for some time, while the practical implementation is only gradually developing, some scientists perceive Industry 4.0 more as an evolution than a revolution (Kagermann, 2014; Sendler, 2013).

Industry 4.0 is controversially discussed in science, hence no common understanding has emerged so far (Bauer et al., 2014). According to Bauer et al. (2014, p. 18), Industry 4.0 is a "real-time capable, intelligent, horizontal, and vertical networking of people, machines, objects" and information and communication technology systems. Based on intelligent, digitally interconnected systems, people, machines, plants, logistic processes, and products can communicate and cooperate in real-time with each other (Platform Industrie 4.0, 2017). Industry 4.0 is "a new level of organization and control of the entire value chain across the life cycle of products" (Platform Industry 4.0, 2017). The interconnection of the operational value creation process takes place across corporate functions, companies, and entire value creation chains (Kagermann et al., 2013). Therefore, a high level of standardization of interfaces between companies is required (Müller & Voigt, 2017).

Using new technologies enables the development of an intelligent value-added system within the framework of Industry 4.0. First, cyber-physical systems (CPS) result from the interconnection of embedded systems and link information technologies with mechanical and electrical components (Becker, 2015; Kagermann et al., 2013; Zhou et al., 2015). In addition, the collection, analysis, and use of large amounts of data play a decisive role and is subsumed among the term "big data". Finally, cloud solutions for storing and transmitting data via stable networks are used (Rüßmann et al., 2015; Bauer et al., 2014).

Research on the subject of Industry 4.0 is generally still in its infancy, which is particular true for the implementation across company borders and value creation chains.
2.2 Supplier Management and supplier integration

Supplier management is "the design, management, and development of a company's supplier portfolio and supplier relationships" (Wagner, 2002, p. 11). The aim of supplier management is to secure a company's demand through an efficient supplier network and thereby contribute to the value creation (Helmold & Terry, 2016).

Supplier integration is a sub-process of supplier management (Helmold & Terry, 2016) representing a form of vertical cooperation (Möller, 2002). This implies close strategic cooperation with both key suppliers and customers in order to generate advantages (Schoenherr & Swink, 2012; Thun, 2010; Wiengarten et al., 2016). The goal of supplier integration is to design integration strategies, practices, processes, and behaviors in a collaborative, synchronized, and well-controllable manner (Zhao et al., 2015). Combining a company's resources with the capabilities of its supplier and realizing joint activities can help to generate sustainable competitive advantages (Rink & Wagner, 2007). Shortening product development and product life cycles and the associated fast, flexible, and efficient production processes in the context of Industry 4.0 increases the importance of supplier integration (Hofbauer et al., 2016).

3 Methodology

3.1 Research design

The study follows a qualitative research design to answer the research questions (Gläser & Laudel, 2010). A qualitative design is characterized by considering and analyzing different perspectives and integrating the interviewees' and researchers' views (Flick, 1995). This design is particularly suitable for the research at hand because supplier integration in the context of Industry 4.0 is a very topical issue and little comprehensive knowledge is available (Kaiser, 2014).

Semi-standardized in-depth expert interviews serve as a data basis (Gläser & Laudel, 2010). All interviews followed an interview guideline and were conducted via telephone in German. On the one hand, the interview guideline was designed so that the interviewee was able to openly respond to questions and present his or her subjective perspective. The result was a natural course of conversation

in which the interviewee could answer freely. In addition, the interviewer could follow up on certain questions or adapt his questions. On the other hand, a partial standardization of the interviews allowed to compare and evaluate the interviews (Mayring, 2015; Gläser & Laudel, 2010). After general questions, the experts were asked about their opinion on the topic of supplier integration in the context of Industry 4.0. The main part contained specific questions on the sub-topics (1) challenges with suppliers in the context of Industry 4.0, (2) expectations of supplier ers as for Industry 4.0, (3) and adequate integration strategies. All interviews were audio-recorded and subsequently transcribed with the permission of the interview partners leading to more than 200 pages of text material. For confidentiality reasons, all interviews were anonymized.

3.2 Data sample

The data sample comprises 15 semi-standardized in-depth expert interviews. Originally, 46 companies were surveyed, and thereof experts from 15 companies were recruited for an interview, corresponding to a response rate of approximately 33 percent.

The surveyed experts come from a heterogeneous sample of companies, headquartered in the Federal Republic of Germany. The sample includes companies from the sectors electrical engineering and electronics, mechanical and plant engineering, automotive, and information and communication technology. These sectors were chosen because they are representative for the most important sectors in Germany, and as they are in particular, driving forward Industry 4.0 (Kagermann et al., 2013). The following figure shows the distribution of enterprises within the sample.

The size of the sample companies is heterogeneous. It varies in terms of turnover from approximately 150 million to approximately 80 billion euros and in terms of employees from approximately 3,000 to approximately 400,000 employees. All experts hold a position in either medium or upper management and have several years of business experience. The interviewed positions include nine representatives from purchasing departments, of which six are head of purchasing, two Chief Digital Officers, one Chief Executive Officer, one Chief Technology Officer, one head of external cooperation, and one head of supplier management.



Figure 1: Sector distribution of sample companies

All interviews were conducted from April to June 2017 via telephone. In accordance with the interviewees, the interviews were audio-recorded and transcribed.

3.3 Data analysis

The interviews were examined using a qualitative content analysis according to Mayring (2015). Therein, the experts' statements are reduced to their core statements and, later on, paraphrased and subsequently generalized (Mayring, 2015). The paraphrases, which contain similar content, can be divided into few statements, which Mayring calls categories. The categories are formed inductively using a keyword analysis (Gläser & Laudel, 2010; Kaiser, 2014). The frequency analysis of individual nominations allows an interpretation and analysis of expert statements from which relevant research results can be extracted (Bogner et al., 2014; Gläser & Laudel, 2010; Mayring, 2015).

4 Analysis and results

4.1 New requirements for suppliers

The increasingly demanded ability to collect, store, and profitably evaluate data is mentioned by ten respondents to be new requirements for suppliers in the context of Industry 4.0. At the same time, the company representatives emphasize that many suppliers have paid little attention to this so far. Therefore, suppliers are demanded expand these abilities by the interviewees, even if those are just supplying parts or raw materials.

Eight respondents considered the creation of interfaces and the implementation of standards to be important requirements. Smooth interfaces are required especially for digital real-time data exchange, as mentioned by the respondents. From their perspective, suppliers need to be willing to adapt to customer-specific standards and interfaces, even if different standards exist among their customers.

The willingness to provide data as a requirement for their own suppliers in the context of Industry 4.0 was mentioned by six respondents. So far, data have often been published only if necessary and hesitantly, which should be done proactively in the future, according to the interviewed experts.

Five respondents named a cultural shift towards common, collaborative value creation within the Industry 4.0 concept as important. In the future, suppliers should no longer respect their own company boundaries as borders, but they should think and act beyond these borders, increasing partnership-based exchange. According to the interviewees, partnership and cooperation needs to be extended, replacing predominant competitive thinking.

The understanding of the shortening innovation cycles by Industry 4.0 to be an essential requirement was named by four respondents. Traditional industrial sectors, such as mechanical engineering, would have to rethink their way of creating value, in order to approach short innovation cycles like those of, e.g., the

software industry. The interviewees regard this as an essential prerequisite in order to, e.g., produce cyber-physical systems or offer platform-based business models.

Three respondents described an increased orientation towards the common end customer of different supply chain partners. Consequently, the primary goal of value creation should focus on the common end customer and his needs, following the opinion of the interviewed experts.

4.2 Challenges of supplier integration

Nine surveyed company representatives named a high degree of complexity in supplier integration in the context of Industry 4.0. Inhomogeneous standards and differing requirements across industries hinder the implementation of Industry 4.0 across value chains. For example, different Enterprise-Resource-Planning systems must be harmonized in order to enable a global network.

The lack of resources on the part of suppliers as a hindrance was mentioned by eight respondents. These statements do not only refer to a lack of financial resources, but also to a lack of knowledge and work force.

Another eight respondents stated inadequate structures and interfaces as a challenge. In many companies, different departments work with different standards and a company-wide coordination does not take place.

The suppliers' uncertainty about the expected developments was described by five interviewees. Many suppliers do have a wait-and-see-attitude and, as for now, do not address the issue of Industry 4.0 proactively. This leads to a lack of competence creation instead of using time to gather information and define strategies for Industry 4.0.

Five respondents cited a lack of understanding, particularly as for the urgency of implementing Industry 4.0. This is especially true for SMEs. Instead, sectors would pursue the typical goals, e.g., mechanical engineering strives rather for improving mechanical quality, than preparing for future topics.

Another five respondents described issues of data security and data protection. Until now, issues of data security are not completely clarified from a legal perspective and the question of data ownership remains unclear. Further, concerns such as hacker attacks and espionage should be taken care of and secure encryption methods should be applied.

Challenges resulting from a changing balance of power were mentioned by five respondents. Suppliers would be afraid of losing importance in the future, as their competence is no longer of central importance in Industry 4.0. In addition, customers are concerned because suppliers could put more pressure on them and even completely bypass or replace them.

Four respondents described possible disruptions because of new competitors as a challenge. There are concerns in particular with regard to platform providers or data-based business models. These could create the core benefit for the customer in the future and thus degrade existing companies as suppliers or completely displace their business model.

A lack of willingness to exchange data on the part of suppliers to be an obstacle was also mentioned by four respondents. For example, there is a lack of understanding that data can only generate value in the future if it is exchanged across the entire value chain. At present, data is considered to be a trade secret and most firms disclose it whenever possible. However, a compromise must be found in order to create advantages for all partners within a value chain.

4.3 Integration strategies

Eight respondents described precise and comprehensive communication of common standards as an appropriate approach to supplier integration in Industry 4.0. Many suppliers did not know, which standards were required and which type of cooperation asks for which specific standard. It is therefore essential to clarify issues of standards and to make it easy for the suppliers to understand the requirements.

The establishment of digital platforms as a strategy for integrating suppliers is also mentioned by eight experts. The platforms could be used in different contexts, for example, in procurement, supply chain management, joint tool management, and product development. It is important to increase potentials through joint networking and to make them accessible to all partners in the value chain.

Six of the company representatives cited transparency in communication with suppliers as an essential approach. Thus, honesty and transparency are central

points here. A clear presentation of the consequences following from a lack of cooperation and an honest communication are indispensable in order to cooperate with suppliers in the context of Industry 4.0.

Further, contractual security is an important issue described by two respondents. Suppliers need to have a reliable basis with a long-term perspective in order to be able to make the corresponding investments for the integration process. Since suppliers have to meet different standards for their numerous customers, they aim for keeping their customers for a long period of time, as otherwise specific investments would not payoff. Establishing long-term relationships is the only way to encourage the supplier's willingness to invest in Industry 4.0 and the integration process.

The creation joint business models was mentioned by two respondents. As suppliers can play a key role in new, joint business models in the context of Industry 4.0, this is a way to motivate the suppliers. For example, there are data-driven business models in which the supplier is responsible for data evaluation and is thereby offered an incentive to make investments.

Two of the respondents also mentioned common research and development activities. For example, joint development of Industry 4.0 components and software could help to qualify suppliers for Industry 4.0. It is in the hands of customers to help their suppliers to implement Industry 4.0 and to accelerate their efforts through corresponding demand.

The provision of resources was mentioned by a single respondent. Nevertheless, this integration strategy represents an interesting approach to support suppliers. For instance, specialists can be lent out for consultation and technical components can be made available at low cost or free of charge to speed up the implementation process.

5 Conclusion

This article presents the results of 15 in-depth expert interviews as for supplier requirements, challenges during the implementation process, and the adequate integration strategies for supplier integration in the context of Industry 4.0.

According to this study's results, key requirements include developing the ability to collect, store, and evaluate data, creating smooth interfaces, implementing

standards, and creating a willingness to exchange data. In the course of the supplier integration, various challenges arise. The most important ones comprise coping with the high complexity, lacking in resource base, and having no adequate structures and interfaces. The new demands on suppliers and the challenges that arise in supplier integration can be classified into different fields of action that require specific integration strategies for companies.

In the future, important strategical issues are, among others, undertaking a comprehensive communication, establishing common standards, and maintaining long-term contractual security. These provide the basis for establishing strategies to face the challenges. Key strategies include creating digital platforms, carrying our joint business models, and undertaking common research and development activities.

It should be noted that the integration strategies presented by the interview partners reflect the status quo and include the strategic environment of the companies, which represents a limitation of the study at hand. Therefore, companies should review their own strategic options and their individual strategic environment to develop their own supplier integration strategies.

The study aggregates key information that is relevant to practice, as companies can use it as a starting point to develop successful supplier integration strategies in the context of Industry 4.0. Research can also use the results to develop the theoretical basis for future changes in supplier management caused by Industry 4.0.

References

- Bauer, W., S. Schlund, D. Marrenbach, and O. Ganschar (2014). Industrie 4.0 Volks-wirtschaftliches Potenzial für Deutschland. Berlin: Bundesverband Infor-mationswirtschaft, Telekommunikation und neue Medien e. V., Fraunhofer-Institut für Arbeitswirtschaft und Organisation.
- Bauernhansl, T. (2014). "Die Vierte Industrielle Revolution Der Weg in ein wertschaffendes Produktionsparadigma". In: T. Ed. by M. t. H. Bauernhansl and B. Vo-gel-Heuser. Industrie 4.0 in Produktion, Automatisierung und Logistik, pp. 5-35, Wiesbaden: Springer Vieweg.
- Becker, K.-d. (2015). "Arbeit in der Industrie 4.0 Erwartungen des Instituts fur angewandte Arbeitswissenschaft e.V". In: Zukunft der Arbeit in Industrie 4. Ed. by A. Botthof and E. A. Hartmann. Berlin: Springer, pp. 23-29.
- Bogner, A., B. Littig, and W. Menz (2014). Interviews mit Experten Eine praxisorien-tierte Einfuhrung. Wiesbaden: Springer.

Brettel, M., N. Friederichsen, M. Keller, and M. Rosenberg (2014). "How Virtualization, Decentralization and Network Building Change the Manufacturing Land-scape An Industry 4.0 Perspective". In: International Journal of Mechanical, Aerospace, Industrial, Mechatronic and Manufacturing Engineering 8, pp. 37-44.

Burmeister, C., D. Luttgens, and F. T. Piller (2015). "Business Model Innovation for Industrie 4.0 Why the Industrial Internet Mandates a New Perspective on Innovation". In: *Die Unternehmung Swiss Journal of Business Research and Practice* 70, pp. 124-152.

- Drath, R. and A. Horch (2014). "Industrie 4.0". In: *Hit or Hype IEEE Industrial Electron-ics Magazine* 8, pp. 56-58.
- Emmrich, V., M. Dobele, T. Bauernhansl, D. Paulus-Rohmer, A. Schatz, and M. Weskamp (2015). Geschaftsmodell-Innovation durch Industrie 4.0: Chancen und Risiken fur den Maschinen- und Anlagenbau. Munchen: Dr. Wieselhuber & Partner.
- Flick, U. (1995). Qualitative Forschung: Theorie, Methoden, Anwendung in Psychologie und Sozialwissenschaften. Reinbek bei Hamburg: Rowohlt.
- Glaser, J. and G. Laudel (2010). Experteninterviews und qualitative Inhaltsanalyse: Als Instrumente rekonstruierender Untersuchungen (4. Auflage. Wiesbaden: Springer VS.
- Hartmann, M. and B. Halecker (2015). "Management of Innovation in the Industrial Internet of Things". In: Proceedings of the 26th International Society for Profes-sional Innovation Management Conference (ISPIM). Budapest, pp. 421-439.
- Helmold, M. and B. Terry (2016). Lieferantenmanagement 2030: Wertschopfung und Sicherung der Wettbewerbsfahigkeit in digitalen und globalen Markten. Wiesbaden: Springer Gabler.
- Hofbauer, G., T. Mashhour, and M. Fischer (2016). Lieferantenmanagement: Die wert-orientierte Gestaltung der Lieferbeziehung (3rd Edition). Berlin/Boston: De Gruyter Oldenbourg.
- Icks, A., C. Schroder, S. Brink, C. Dienes, and S. Schneck (2017). Digitalisierungspro-zesse von KMU im Verarbeitenden Gewerbe. Bonn: IfM Bonn.
- Kagermann, H., W. Lukas, and W. Wahlster (2011). "Industrie 4.0: Mit dem Internet der Dinge auf dem Weg zur 4. industriellen Revolution". In: VDI nachrichten, No 13, p. 2.
- Kagermann, H., W. Wahlster, and J. Helbig (May 2013). In: Umsatzempfehlungen fur das Zukunftsprojekt Industrie 4.0 8, p. 2018.
- Kaiser, R. (2014). Qualitative Experteninterviews: Konzeptionelle Grundlagen und praktische Durchfuhrung. Wiesbaden: Springer VS.
- Kang, H. S., J. Y. Lee, S. Choi, H. Kim, J. H. Park, and J. Y. Son (2016). "Smart manufacturing: Past research, present findings, and future directions". In: *International Journal of Precision Engineering and Manufacturing-Green Technology* 3, pp. 111-128.
- Kans, M. and A. Ingwald (2016). "Business Model Development: Towards Service Man-agement 4.0". In: *Procedia CIRP* 47, pp. 489-494.
- Kelkar, O., R. Heger, and D. Dao (2014). Studie Industrie 4.0 Eine Standortbestim-mung der Automobilund Fertigungsindustrie. Reutlingen: Mieschke, Hof-mann und Partner.
- Kiel, D., J. M. Müller, C. Arnold, and K.-i. Voigt (2017). "Sustainable Industrial Value Creation: Benefits and Challenges of Industry 4.0". In: *International Journal of Innovation Management* 21.8, p. 1740015.
- Kowalkowski, C., L. Witell, and A. Gustafsson (2013). "Any way goes: Identifying value constellations for service infusion in SMEs". In: *Industrial Marketing Manage-ment* 42, pp. 18-30.
- Lasi, H., P. Fettke, H. G. Kemper, T. Feld, and M. Hoffmann (2014). "Industry 4.0". In: Busi-ness & Information Systems Engineering 6, pp. 239-242.

Liao, Y. and L. Pierin Ramos (n.d.). "2017. Past, present and future of Industry 4.0 A systematic literature review and re-search agenda proposal. International Journal of Production Research,)," in: 55 (), pp. 3609-3629.

Mayring, P. (2015). Qualitative Inhaltsanalyse: Grundlagen und Techniken (12th Edition). Weinheim: Beltz.

- Moller, K. (2002). Zuliefererintegration in das Target Costing: Auf Basis der Transak-tionskostentheorie. München: Vahlen.
- Müller, J. M., O. Buliga, and K.-i. Voigt (2018). "Fortune favors the prepared: How SMEs approach business model innovations in Industry 4.0". In: *Technological Forecasting and Social Change* 127, pp. 2–17.
- Müller, J. M. and V. K.-i. (2017). "Standardisierung fur Industrie 4.0 in KMU Welche Herausforderungen werden von kleinen und mittleren Unternehmen durch Standardisierung hinsichtlich Industrie 4.0 gesehen?" In: *Industrie 4.0 Management* 4, pp. 25–28.
- Müller, J. M., D. Kiel, and V. K.-i. (2018). "What Drives the Implementation of Indus-try 4.0?" In: The Role of Opportunities and Challenges in the Context of Sustaina-bility. Sustainability 10, p. 1.
- Müller, J. M., L. Maier, J. Veile, and K.-i. Voigt (2017). "Cooperation strategies among SMEs for implementing Industry 4.0". In: *Hamburg International Conference on Logistics (HICL*.
- Plattform Industrie 4.0 (2017). "Was ist Industrie 4.0?". In: https://www.plattform-i40.de/I40/ Navigation/DE/Industrie40/WasIndustrie40/was-ist-industrie-40.html
- Radziwon, A., A. Bilberg, M. Bogers, and E. S. Madsen (2014). "The Smart Factory: Exploring Adaptive and Flexible Manufacturing Solutions". In: Procedia Engi-neering 69, pp. 1184-1190. Rennung,
- F., C. T. Luminosu, and A. Draghici (2016). "Service Provision in the Frame-work of Industry 4.0". In: Procedia Social and Behavioral Sciences 221, pp. 372-377.
- Rink, C. and S. M. Wagner (2007). "Lieferantenmanagement: Strategien, Prozesse und systemtechnische Unterstutzung". In: *Elektronische Beschaffung: Stand und Entwicklungstendenzen, pp.* Ed. by W. Brenner and R. Wenger. Berlin/Heidelberg: Springer, pp. 39–62.
- Russmann, M., M. Lorenz, P. Gerbert, M. Waldner, J. Justus, P. Engel, and M. Harnisch (2015). Industry 4.0: The future of productivity and growth in manufacturing industries. Boston: Consulting Group.
- Schoenherr, T. and M. Swink (2012). "Revisiting the arcs of integration: Cross-validations and extensions". In: Journal of Operations Management 30, pp. 99-115.
- Sendler, U., ed. (2013). Industrie 4.0. Beherrschung der industriellen Komplexitat mit SysLM. Berlin: Springer Vieweg.
- Siepmann, D. (2016). "Industrie 4.0 Funf zentrale Paradigmen". In: Einfuhrung und Umsetzung von Industrie 4. Ed. by A. Roth. 0. Grundlagen, Vorgehensmodell und Use Cases aus der Praxis. Berlin/Heidelberg: Springer Gabler.
- Thun, J. H. (2010). "Angles of Integration: An Empirical Analysis of the Alignment of internet-based Information Technology and global Supply Chain Integra-tion". In: *Journal of Supply Chain Management* 46, pp. 30-44.

Wagner, S. M. (2002). Lieferantenmanagement. Munchen/Wien: Hanser.

- Wiengarten, F., P. Humphreys, C. Gimenez, and R. McIvor (2016). "Risk, risk manage-ment practices, and the success of supply chain integration". In: *International Journal of Production Economics* 171, pp. 361-370.
- Zhao, G., T. Feng, and D. Wang (2015). "Is more supply chain integration always benefi-cial to financial performance?" In: *Industrial Marketing Management* 45, pp. 162-172.
- Zhou, K., T. Liu, and L. Zhou (2015). "Industry 4.0: Towards future industrial oppor-tunities and challenges". In: Proceedings of the 12th International Conference on Fuzzy Systems and Knowledge Discovery (FSKD). Zhang-jiajie, pp. 2147–2152.

Scope for Industry 4.0 in Agri-food Supply Chains

Claudine Soosay¹, Raja Kannusamy¹

1 - University of South Australia

This study investigates the current capabilities and technologies adopted in the agri-food industry in Australia and the scope for transition to Industry 4.0. Data were collected from 360 firms representing suppliers, producers, manufacturers, wholesalers, logistics providers and retailers to represent a supply chain perspective. The technologies and strategies were grouped based on the various supply chain players against the maturity stages of Industry 4.0 as prescribed by Schuh et al (2017) in order to discern the integration efforts and degree of interoperability in the supply chain. We establish that upstream players tend to adopt technology mainly for internal operational efficiencies and B2B transactions. We propose that the individual orientation, interoperability and capabilities of these firms will need to be reassessed to derive a systematic plan for progression into a technology architecture for the overall supply chain. Despite the fragmented adoption of advance technologies evident at various points of value creation in the supply chain, we recognize and highlight the vulnerability of many small businesses and upstream players in the food industry who appear to be lagging behind in the fourth industrial revolution as well as the disruptive changes entailed to keep up and compete in the digital age.

Keywords: Agri-food supply chain; Industry 4.0; Maturity stages; Survey methodology

First recieved: 28.May.2018 Revised: 11.Jul.2018 Accepted: 16.Jul.2018

1 Introduction

In the agri-food industry, logistics and supply chain management activities are deemed more challenging due to the need for time-based approaches, integral quality control and associated tracking and tracing systems of food products along the chain. These include meeting temperature control requirements, product perishability issues and the variability in agri-food quality (Soosay 2008, Sahin et al 2007). These products include grain, livestock (beef and dairy) and horticulture. Inter-organizational technology systems offer the potential to improve operational efficiency, responsiveness and traceability by supporting transactions and material flow more efficiently (Kim et al 2018, Johnston & Vitale, 1988). Consequently, integrative efforts are needed between supply chain members to reach new levels of competencies where firms can willingly share data and business applications.

Industry 4.0 requires supply chains to not only adopt modern technologies and engage in capability development, but also to transform their business models and network structures to achieve coherent vertical integration. This is likely to change traditional relationships between suppliers, manufacturers, wholesalers, retailers and customers. There are already some industries such as electronics and fast moving consumer goods (FMCG) adopting state of the art technologies and moving along the supply chain 4.0 continuum at present; while other industries are still lagging behind. There is concern that the diffusion rate of modern technologies in the agri-food industry has been much lower than anticipated, suggesting that their implementation may involve challenges or lack of collaboration (Costa et al 2013, Vlachos 2004, Salin 1998).

There are various empirical studies on supply chain technology adoption in the literature (Aydin & Parker 2018, Krishnan et al 2015). While the authors have established reasons why firms and some industries struggle with technology implementation, the theories, concepts and practices developed in the context of food supply chains need greater understanding as this industry faces distinct economic settings, such as market imperfections, heterogeneity of actors, information asymmetries, technology infrastructure, immature supply chain networks and the perishable nature of food products (Solanki & Brewster 2014, Xu et al 2004). As a result, this study examines the current capabilities, technologies and interoperability strategies adopted in agri-food supply chains and the scope for transition to Industry 4.0. The central research question is 'How is Industry 4.0

distributed across the stages of value creation in the agri-food industry in Australia?'

2 Theoretical Framework

Industry 4.0, also known as the fourth industrial revolution, is a collective approach to digitalization, interconnectedness and new technologies. It is increasingly gaining consideration from policymakers, businesses and academia worldwide. The term 'Industrie 4.0' became popular in 2011 with a Working Group offering the German federal government a vision for the future of industrial manufacturing. It subsequently formed part of the government's 'Action Plan High-Tech Strategy 2020' to ensure technological leadership with digitalization, smart factories and Internet of Things (IoT) (Klitou et al 2017). Industry 4.0 is expected to result in four long-term relationship paradigm shifts and changes to the landscape of European manufacturing: Factory and nature, Factory and local community, Factory and value chains (distributed and responsive manufacturing through collaborative processes, enabling mass customization of products and services); and Factory and humans. These will have impacts on technology implementation in a wider manufacturing and distribution environment (Santos et al 2017).

Schuh et al (2017) articulate that the fourth industrial revolution extends beyond ICT integration in industrial manufacturing to include transformations in organizations and their culture. Businesses will need to become more agile and adapt to changing environments. These authors prescribe the acatech Industrie 4.0 Maturity Index model which delineates the successive maturity stages for Industry 4.0. This model assists companies identify which stage they are currently at and their potential capability development and transformation to fully implementing Industry 4.0. The stages span from 'Computerization', 'Connectivity', 'Visibility', 'Transparency', 'Predictive Capability' to 'Adaptability'.

Industry 4.0 should not only be discerned at the organizational level, but will also revolutionize manufacturing supply chains with new products, services and business models through IoT from product design right through to customer delivery (Roblek et al 2016). For instance, it emphasizes the global network of machines in a manufacturing environment capable of exchanging information, knowing variations to be made to the product and being able to control each other. This is possible with collaboration between suppliers, manufacturers and customers to

increase the transparency from when the order is initiated, manufactured and dispatched until the end of the product's life cycle. Hence, it is important to analyze how supply chains are impacted by Industry 4.0.

Supply chain networks today depend on a number of key technologies that enable integrated planning and execution systems, logistics visibility, autonomous logistics, smart procurement, smart warehousing, spare part management and advanced analytics (Schrauf & Berttram 2016). Lee et al (2014) highlight how technology is the key to 21st century global supply chain management for operational competitiveness. By tracking the evolution of supply chain technologies in the textile and apparel industry, these authors classify the technologies adopted in achieving superior supply chain performance and competitive advantage.

It is evident that digital networks offer higher levels of resilience and responsiveness with more efficient and transparent service delivery. Mussomeli et al (2017) report the shift from linear, sequential supply chain operations to interconnected, dynamic and integrated networks. Predictive shipping is another emerging concept, where according to Alicke et al (2017), a shipment which is already in the logistics network is matched with customer order at a later stage. Resultantly, demand management will need to be implemented at a more granular level using techniques such as micro segmentation, mass customization, innovative distribution concepts and more sophisticated scheduling practices. The emerging trends in Industry 4.0 can pertain to warehouse robotics, autonomous road transportation, logistics and technology services, supply chain social responsibility, the race for the last mile, and the rise of the virtual logistics team (O'Byrne 2017). Hence, businesses will need to reframe their business models and invest in how they can digitize their products and systems, starting with the supply chain. They will also need to reassess their capabilities, technologies and interoperability strategies in order to transition to Industry 4.0.

In this paper, we examine Industry 4.0 in the context of agri-food supply chains in Australia using the maturity phases from Schuh et al (2017) to understand the extent of technologies used by firms throughout the value creation. These phases are grouped as 'Computerization & Connectivity', 'Visibility & Transparency', 'Predictive capability', and 'Adaptability & Self-learning'.

3 Methods

Data were collected in three stages over a 12-month period as part of a larger study on supply chain integration. An online survey was sent to over 2,000 organizations obtained from various databases and industry associations. During the first phase. we collected data from suppliers, producers, manufacturers, wholesalers and retailers associated with the agri-food industry. Subsequently, we implemented a second phase of data collection after identifying the secondary players in the supply chain comprising input suppliers, packaging suppliers and third party logistics providers who also service the food industry. Input suppliers include firms who provide equipment, machinery, feedstock, fertilizers and other related products to farmers, growers and agri-food producers. Packaging suppliers include an array of firms who produce and supply paper, plastic, fibre containers, foam food trays, cartons, boxes, glass, closures, foil, film and other products used in the food industry. Logistics providers were largely transport companies, although some provided warehousing, light assembly and secondary packaging services. The third phase of data collection were reminders sent to firms in order to increase the response rate. As a result, a total of 360 usable guestionnaires were received, constituting a response rate of 18% and where these firms represented a whole of chain perspective. These firms are illustrated in the following table showing their position in the supply chain as well as the firm size as prescribed by the Australian Bureau of Statistics classification. Majority of the firms were small and medium sized accounting for 74% of the sample population.

		n	%
Position in the Chain			
	Input suppliers	30	8.3
	Growers/Agri-producers	68	18.8
	Packaging suppliers	20	5.6
	Food Manufacturers	54	15.0
	Wholesalers	67	18.6
	Logistics providers	63	17.5
	Retailers	58	16.1
	Total	360	100
Firm size			
	Small (5-19)	81	22
	Medium (20-199)	186	52
	Large (>200)	93	26
	Total	360	100

Table 1: Profile of firms

In the survey, firms were required to rate the extent of usage for a list of technologies using a 7-point Likert scale (7 = to a very great extent; 6 = to a great extent; 5 = to a fairly great extent; 4 = to a moderate extent; 3 = to a small extent; 2 = to a very small extent; 1 = not at all). In this paper, we have categorized these technologies into four main maturity stages of Industry 4.0 using Schuh et al's (2017) framework. Table 2 depicts the means and standard deviations of the technologies. The reliability is reported using Cronbach's alpha which ranged from 0.76 to 0.93, indicating moderate to excellent reliability.

Items	Description	М	SD	α			
Compute	Computerization & Connectivity						
		0.86					
CC1	Barcoding systems	5.31	1.14				
CC2	Customer Relationship Management	5.68	0.95				
CC3	E-business/ e-marketplace	5.76	0.98				
CC4	Electronic Data Interchange	5.76	1.00				
CC5	Electronic Point of Sale	5.73	0.97				
CC6	E-procurement	5.54	1.08				
Visibility	Visibility & Transparency						
		0.93					
VT1	Global Positioning Systems	5.61	1.31				
VT2	Time Temperature Integrators	5.74	1.24				
VT3	Data Loggers	5.68	1.20				
VT4	Transport Management Systems	5.76	1.23				
VT5	Warehouse Management Systems	5.72	1.26				
Predictiv	Predictive Capability						
		0.87					
PC1	Enterprise Resource Planning Systems	3.85	1.42				
PC2	Manufacturing Execution Systems	4.12	1.32				
PC3	Radio Frequency Identification Systems	4.03	1.61				
Adaptabi	Adaptability & Self-learning						
		0.76					
ASL1	Collaborative Planning, Forecasting and Replenishment	2.19	0.77				
ASL2	Efficient Consumer Response	2.07	0.78				
ASL3	Vendor Managed Inventory	1.99	0.82				

Table 2: Extent of technologies used

5	Input isuppliers	Agri-food producers	► Packaging Suppliers	Food Manufacturers/ Processors	Vhole Logistics providers	esalers	Retailers
	Input suppliers	Agri-food producers	Packaging suppliers	Food manufacturers	Logistics providers	Wholesalers	Retailers
Computerization & co	onnectivity						
Barcode	53.3%	29.4%	80.0%	24.1%	91.0%	76.2%	100.0%
CRM	20.0%	2.9%	60.0%	64.8%	34.3%	17.5%	75.9%
E-business	83.3%	86.7%	45.0%	77.8%	65.7%	12.7%	70.7%
EDI	0%	4.4%	0%	79.6%	28.4%	30.2%	81.0%
EPOS	30.0%	0%	0%	0%	0%	3.2%	100.0%
E-procurement	43.3%	41.1%	80.0%	90.7%	80.6%	15.9%	89.7%
Mean	38.3%	27.4%	44.2%	56.2%	50.0%	26.0%	86.2%
Visibility & transpare	ncy						
GPS	0%	0%	0%	0%	80.6%	49.2%	3.4%
TTI	0%	7.4%	0%	0%	44.8%	46.0%	12.0%
Data loggers	0%	16.2%	0%	13.0%	52.2%	30.2%	3.4%
TMS	0%	0%	15.0%	9.3%	80.6%	7.9%	0%
WMS	13.3%	0%	70.0%	9.3%	10.4%	39.7%	75.9%
Mean	2.7%	4.7%	17.0%	6.32%	53.7%	34.6%	18.9%
Predictive capability							
ERP	0%	1.5%	55.0%	79.6%	70.1%	66.7%	75.9%
MES	0%	0%	75.0%	77.8%	0%	0%	0%
RFID	10.0%	0%	40.0%	27.8%	80.6%	36.5%	79.3%
Mean	3.3%	0.5%	56.7%	61.7%	50.2%	34.4%	51.7%
Adaptability & self-learning							
CPFR	10.0%	0%	0%	44.4%	28.4%	3.2%	58.6%
ECR	10.0%	0%	0%	29.6%	0%	6.3%	60.3%
VMI	0%	0%	10.0%	44.4%	4.5%	0%	29.3%
Mean	6.7%	0%	3.3%	39.5%	11.0%	3.2%	49.4%

Figure 1: Technologies used in the agri-food supply chain

4 Findings and discussion

The range of technologies and collaborative strategies adopted depict varying levels of value co-creation, supply chain visibility and traceability. It is apparent that these organizations employ technologies and strategies to improve business processes and decision-making. In food supply chain operations, technologies are predominantly used as a means for communication and information exchange between partners, facilitating uninterrupted flows. Prior to information sharing, the technology used needs to be linked and integrated between their upstream and downstream members in supply chains (Bhatt et al 2017). For this study, we investigated the strategies, technologies and integration efforts between two or more companies to understand the degree of interoperability in the supply chain. Figure 1 presents a full picture of the technologies used and percentage of firms at the value creation stages in the supply chain.

Additionally, in order to further discern the realization of Industry 4.0 maturity stages for each group, we obtained the mean scores for the various technologies and initiatives based on the number of firms that adopted them.

4.1 Computerization and connectivity

Computerization is the basis for digitalization and encompasses the deployment of information technologies. This is the first step and basic prerequisite for Industry 4.0. At this stage, many of the technologies appear to be used in isolation either within the organization or to assist in digitally supported B2B processes and transactions. When firms replace their isolated technology usage with connected business applications, a shift to embedded systems is facilitated (Schuh et al 2017).

Our findings depict that barcoding identification technology was used at all stages in the supply chain. There is a lesser extent of usage by upstream growers, farmers and other food producers; particularly those dealing with fresh fruit, vegetables, meat and seafood which are generally sold by weight. This technology facilitates inventory control and information sharing with suppliers upstream through electronic data interchange (EDI) (Kinsey and Ashman 2000) and often integrated with Warehouse Management Systems (WMS) (Patterson et al 2004).

Electronic point of sale (EPOS) was found to be only used by input suppliers (30%), wholesalers to a small extent (3%) and all retailers (100%) in our sample. This technology is usually incorporated with barcoding, and allows food retailers, manufacturers and packaging companies to collaboratively discern purchasing patterns and develop new products (Cox & Mowatt 2004). Customer relationship management (CRM) appeared to be used throughout the chain as a strategy. It is believed that firms utilize data warehousing and mining techniques to segment customers for improved customers service and retention (Chen & Chen 2004, Cox & Mowatt 2004), which is particularly critical given the nature of the food industry.

The e-business enabled supply or commonly known as e-marketplace is becoming an increasingly popular business model for firms to source, trade and collaborate with chain partners (Howard et al 2006, Le 2005). Our findings depict that 87% of agri-food producers, farmers and growers were using this, arguably to diversify their business opportunities and enhance profitability. Such platforms allow firms to expand globally and enter new markets that were previously inaccessible due to geographical barriers (White et al 2007).

Electronic data interchange (EDI) is most used in manufacturers and retailers with 80% and 81% respectively. This computer-to-computer exchange of business documents in a standard format between supply chain partners enables firms

to reduce transaction costs, improve information transparency, efficiency and achieve integrated efforts (Leonard & Davis 2006, Hill & Scudder 2002). EDI in the food industry is also known to reduce the bullwhip effect, although being a complex and expensive system to implement especially for many small businesses (Vlachos 2004, Kinsey & Ashman 2000).

E-procurement technologies include e-sourcing, e-auctions and EDI to trade with suppliers online. Our findings depict that this technology is apparent in all stages in the supply chain, and more so among packaging suppliers (80%), food manufacturers (91%) and logistics providers (81%). Firms can respond more effectively using real-time information to meet demand patterns and supplier prices, while reducing delivery times and procurement costs (Chibani et al 2018, William 2003). Many organizations operate e-procurement in the food industry to assist with JIT strategies.

Based on our findings, we draw inference that majority of the firms in our sample have embedded digitally supported processes and transactions for computerization and connectivity. Based on the mean scores, this is more apparent in retailers, followed by food manufacturers and logistics providers. We had expected growers and food producers to be less likely to be digitally connected owing to the nature of this industry; however, discovered that the wholesalers in our sample were the least connected.

4.2 Visibility and transparency

The second phase of maturity in Industry 4.0 is visibility and transparency. In the context of supply chains, we argue that this should extend beyond B2B transactions and result in end-to-end visibility. The real-time capture of events from several data points in the supply chain enhances visibility, based on information availability and quality. Schuh et al (2017) postulate that by aggregating this information and a corresponding contextualization, transparency is enhanced for faster decision-making. We identified four main types of technologies used in the food supply chain for this phase which are primarily in the area of logistics activities.

Global positioning system (GPS) was found to be used by 81% of third party logistics providers and 49% of wholesalers in our study. The use of radio signals from satellites is widely implemented to manage logistics and transportation activities. Firms can locate and monitor the direction of fleet operators and vehicles, further allowing for optimized vehicle scheduling and routing (Theiss et al 2005). This is also prevalent upstream in the food industry where food producers can collect data on crops, soils, environmental and climate data, and monitor the movement of livestock in paddocks, facilitating precision agriculture and farming practice (Opara 2003). However none of the upstream players in our sample adopted this technology.

Transport management system (TMS) on the other hand, was used to a greater extent by firms in the chain including packaging suppliers, manufacturers and retailers. As expected, a large proportion of logistics providers (81%) adopted this technology which is closely related with GPS. TMS optimizes distribution activities, particularly in the area of fleet planning, truck scheduling and vehicle routing (Pokharel 2005). The real-time information enables firms to monitor turnaround time, driver productivity and fleet utilization. It could further facilitate cross-docking in warehouses, which is highly pertinent in the food and grocery sector, where the timely flow of perishable products is critical (Apte & Viswanathan 2000).

Warehouse management system (WMS) is also another technology adopted throughout the chain except for growers, farmers and food producers. We establish that the sale of agri-food products occurs almost immediately or using JIT after harvest for optimum freshness and quality. Most growers and farmers do not have adequate storage facilities or keep much inventory. Retailers (76%) and packaging suppliers (70%) adopted WMS to a greater extent given the volume and variety of products they deal with. Surprisingly, WMS was only used in 25 out of the 63 wholesalers (40%). It equips firms with the capabilities in monitoring stock levels, the inbound and outbound flows of inventory and exact location of items stored in the warehouse. WMS captures data on product dimensions and characteristics to optimize warehouse space, personnel and material handling equipment (Patterson et al 2004). It is often interfaced with TMS, RFID and barcoding technologies to enhance logistics operations (Mason et al 2003). We argue that most food products tend to be highly perishable; hence their inventory management is critical at all stages.

Our findings highlight that time-temperature integrators (TTI) and data loggers are being used predominantly by logistics providers (45% & 52%) and wholesalers (46% & 30%). These devices or tags are placed within a storage container to record the temperature of food products throughout the distribution process. Data are used to monitor temperature stability, product quality or identify breaches at various points in the supply chain. This is critical for temperature sensitive

or perishable food products. Such technology enables optimum quality, shelflife and handling throughout the supply chain (Soosay 2008, Sahin et al 2007, Koutsoumanis et al 2005). However, only 7 out of 58 retailers (12%) in our sample had adopted this technology.

Overall, information integration is required for tracking and tracing, ensuring visibility and transparency in the chain to meet food safety and regulatory requirements. The mean scores shows a varied extent of visibility and transparency among various supply chain players. It appears that logistics providers adopt these technologies the most, followed by wholesalers, manufacturers and retailers. We discern that upstream players (farmers, growers and input suppliers) tend to adopt technology to a lesser extent as compared to downstream players; and that these were used mainly for internal operational efficiencies and B2B transactions. Our finding is also in line with Solanki and Brewster's (2014 p. 46) study on agri-food supply chains; where they confirm and highlight how the "flow of data is restricted based on a very conservative 'need-to-know' attitude such that information flows only 'one up, one down'". Moreover, due to cultural barriers and despite technological solutions, the firms do not appear to use or integrate data across the supply chain, "thereby greatly increasing the possibility of interoperability issues arising between supporting applications". Technology should be viewed as the key for SCM development and transformation of the food industry 'from dyadic, material management oriented relationships into complex, collaborative, networked, web-enabled, extended architectures' to enable visibility, transparency and food integrity throughout the chain (Akyuz & Gursoy 2013).

4.3 Predictive capability

The third phase of maturity lies in the ability of firms to use shared data from end to end supply chain partners to prepare for future scenarios. At a network level, the visibility and transparency of operational data, inventory levels and production plans provide focal firms with relevant and real-time information for supply chain decision-making. Unexpected events such as the bullwhip effect, forecasting errors, production variations or delays in distribution could be minimized to a large extent. These result in time-based approaches to reduce costs, waste, inventory and other supply chain inefficiencies. We identify three types of technologies in the food industry that are used to attain this predictive capability. Enterprise Resource Planning (ERP) systems were evident in 188 firms or over half the sample surveyed. These consist of integrated modules that support not only various functions within the organization, but also across supply chain operations. The IT systems of suppliers and customers could be interfaced for seamless operations. Alongside ERP, there are also manufacturing execution systems (MES) used to manage factory operations and production systems. 78% of food manufacturers and 75% of packaging suppliers adopted this technology. MES controls the movement of materials from point-to-point, assigns and schedules resources, and tracks the costs and status of materials being processed (Beavers 2001). When production plans are shared with suppliers, retailers and wholesalers, the flow of food products in the supply chain could be better streamlined.

Radio Frequency Identification (RFID) technology was adopted throughout the supply chain in our sample except for agri-food producers. RFID in food supply chains can determine the location and history of products, which help to maintain quality and prevent breaches in food integrity (Van Der Vorst et al 2007, Opara 2003). When combined with other technologies, RFID equips firms with predictive capability to achieve reliable and accurate data forecasts, reduced inventory and labor costs (Wu et al 2006).

From the mean scores, the area of predictive capability shows a stark difference between the upstream and downstream activities. Manufacturers, retailers and logistics providers generally possess the capacity to engage in big data and supply chain analytics, given the extent of technologies adopted, scale and scope of operations. These firms are generally larger in size and have better resources to obtain value from large scale data to gain competitiveness in terms of demand volatility, cost fluctuations and inventory management. We argue that the utilization of supply chain analytics is still at an early stage in the food industry currently; but has the potential to assist in strategic demand planning, sourcing, manufacturing, logistics activities, inventory management, and overall network configuration to advance into a supply chain 4.0 continuum. The use of time temperature sensors, RFID, tracking devices and ERP systems generates big data for the supply chain, which serve as a new frontier for process improvement, demand management and decision-making.

Our findings show that upstream players in the supply chain are generally lagging in this maturity stage of Industry 4.0. Hence it is proposed that the individual orientation, interoperability and capabilities of upstream players will need to be reassessed to derive a systematic plan for progression into a technology architecture for the overall supply chain. Bryceson and Yaseen (2018) illustrate how various disruptive technologies have been changing the landscape of the agrifood industry including upstream farming and production activities. IoT systems are conducive to agri-food firms in environmental monitoring, precision farming, precision livestock and cold chain logistics. For example in livestock management, IoT technology coupled with RFID can be used for real-time monitoring of cattle health and behavior. These authors also emphasize how big data analytics can assist businesses overcome environmental degradation, food safety and food security. Smart farming is increasingly being used with tailored seed varieties, crop nutrients, pathogen monitoring, and improving crop yields and addressing food safety concerns. New types of data can be used to measure biophysical characteristics (such as climate, soil alkalinity, seed cultivation, fertilizer and pest control) from the paddocks and farms upstream through to food processing, facilitated storage and food retail downstream (Bryceson & Yaseen 2018). Additionally, smart packaging is gaining prominence in food supply chains today. They not only help to meet HACCP and QACCP requirements, but also detect real-time biochemical changes occurring in the food and its environment to help extend shelf life. Another feature of Industry 4.0 is blockchain technology which offers huge potential for food traceability solutions. This technology works as a technical schema of databases which could contain information of all activities, processes and transactions in the chain forming a digital form of fingerprinting; and through interoperability mechanisms, this could ultimately provide visibility and traceability of the product's entire life cycle.

4.4 Adaptability and self-learning

The fourth maturity stage occurs when firms are able to take corresponding measures autonomously based on shared data. According to Schuh et al (2017), this delegation of decisions allows for rapid adaptation to changing business environments. Applied at a broader level, we establish that the combination of technologies achieves supply chain agility for customization and responsiveness to end consumers. We considered various supply chain strategies and classify three initiatives for this category.

Vendor Managed Inventory (VMI) was the least adopted technology in 13% of firms only. It was apparent mainly in manufacturers (44%) and retailers (29%). VMI is a powerful initiative based on technology integration and established transparency and visibility. Upstream suppliers or manufacturers are authorized to monitor retailers' real-time inventory levels and replenish accordingly (Waller et al 1999). Its low uptake in the food industry could be attributed to the effects of inventory holding costs or high shortage penalty costs (Ru et al 2018). Closely linked to this initiative is efficient consumer response (ECR) applied in the food industry to alleviate supply chain inefficiencies. Once again, this was apparent mostly in food manufacturers (30%) and retailers (60%). This initiative requires retailers to share point-of-sales data with supply chain partners for not only better store assortment, promotion, product development and replenishment (Harris et al 1999), but also better financial and operational performance (Martens & Dooley 2010).

Collaborative Planning Forecasting and Replenishment (CPFR) is another shared IT system to project demand patterns. These were adopted by over half the retailers (59%), followed by manufacturers (44%) and logistics providers (28%). By using real-time data in the supply chain, this initiative facilitates demand forecasting, inventory management, production and replenishment planning, and order fulfilment (Hill et al 2018). CPFR is critical given the price points of food commodities, time-sensitivity, perishability and shelf-life. In this context, firms make autonomous decisions on a breadth of areas including timely supply, food production, processing, wholesale, distribution, marketing and retail. Our findings suggest that these factors are highly considered by downstream players due to their proximity and reliance on end customers and food consumers' demand preferences. This is further confirmed by the mean scores, showing the disparate heterogeneity in the supply chain for adaptability and self-learning. The other stages in the supply chain (input suppliers, agri-food producers, packaging suppliers, logistics providers and wholesalers) had very low values for this maturity stage.

These three powerful initiatives applicable in the food industry require the establishment and mastery of the first three maturity stages. This fourth maturity stage of adaptability and self-learning represents an autonomous supply chain. Characterized by robotics and automated networks, the concept of autonomously delivering products is becoming a reality. Heard et al (2018) claim that connected and autonomous vehicles (CAV) will be embraced first in the food distribution industry, given the interplay between transportation logistics, food perishability and cost. Enabled by logistics innovations and e-commerce, these authors state that autonomous trucks could one day displace grocery stores, and change the business model of distribution by delivering ordered foods directly to consumers instead. Similarly, MES and ERP systems have been around for a while; and possess the intelligence to organize and control manufacturing processes and job sequencing. They mandate using appropriate technologies, achieving streamlined visibility and transparency across supply chain partners and facilitating the autonomous and continuous adaptation to changes individually at the firm level as well as collectively as a network of partners.

5 Conclusions

Considering an end-to-end supply chain model, we deem the autonomous supply chain to be far from achievable in the food industry at this stage, given the presence of traditional labor intensive methods in some farming and agri-food production activities, especially for fresh and perishable food produce. Despite the fragmented adoption of advance technologies at various points in the supply chain (such as in manufacturing and logistics), we establish that agri-food supply chains still have a long way in reaching this milestone for Industry 4.0. This is because a true autonomous supply chain does not require human intervention (River Logic 2018).

Industry 4.0 requires firms to develop agile capabilities for rapid adaptation to changing business environments where customization and responsiveness come into play to fulfil end consumers' needs. We argue that the technology infrastructure provides integrated, empowered and more responsive decision-making; hence enabling a new frontier for customer receptiveness and market changes. The evolving and dynamic nature of food consumption means that customers expect high quality sensory properties, more variety and better food security (Castro & Jaimes 2017). Additionally the perishability of food products requires more advanced handling, transportation and storage solutions that requires firms to be flexible, efficient and responsive. Today's supply chain transformation involves the use of digital technologies, predictive analytics and artificial intelligence. To compete and survive, firms must embrace new business models, digital transformation and interoperability aspects. With this in mind, supply chains of the future should be viewed as an autonomous ecosystem of firms rather than traditional linear structures.

6 Limitations and further research

This study has several limitations, which must be taken into account when interpreting the results and their implications. Firstly, it presents preliminary findings from a larger study on supply chain integration in the agri-food industry. Notwithstanding that, it extends our knowledge on the issues relating to the impact of relevant technologies with respect to different stages of the supply chain and their activities. It uncovers the breadth of technologies used and the impacts of technology interface in agri-food supply chains. Secondly, we acknowledge the reliance on cross-sectional data, which measured technology adopted at one point in time that has weaknesses in establishing any causality inferences. Also market factors and stakeholder pressure may differ for various food groups. Further research could be more informative by investigating market configurations pertaining to the types of food products (e.g. fruit, vegetables, meat, seafood) and proximity of chain members to end consumers. Additionally, by employing a larger dataset, there is scope to perform measurement invariance across supply chain nodes and/or multi-level analyses. These could then provide better insights into the technology interface at specific points in value creation in the supply chain in more detail. Additionally, we recognize that our sample comprises a significant proportion (74%) of small and medium enterprises which is characterized by the food industry predominantly. We do not intend to generalize our findings, but highlight the vulnerability of many small businesses and up-stream players in our sample who appear to be lagging behind in the fourth industrial revolution as well as the disruptive changes entailed to keep up and compete in the digital age due to their size, resource constraints, scope and scale of operations. Despite these limitations, this paper offers discernment to supply chain researchers and practitioners by illustrating the maturity phases of Industry 4.0 in the context of agri-food supply chains in Australia.

References

- Akyuz, G. A. and G. Gursoy (2013). "Paradigm shift in supply chain management (SCM)". In: Proceedings of the international annual conference of the American Society for Engineering Management 2013. ASEM 2013, pp. 291–302.
- Albert, L. (2017). "Industry 4.0 and the Transformation of the Supply Chain".
- Alicke, K., D. Rexhausen, and A. Seyfert (May 2017). "Supply chain 4.0 in consumer goods". In: McKinseny and Company 15, p. 2018.
- Apte, U. M. and S. Viswanathan (2000). "Effective cross docking for improving distribution efficiencies". In: International journal of logistics: Research and applications 3.3, pp. 291–302.
- Aydin, A. and R. Parker (2018). "Innovation and technology diffusion in competitive supply chains". In: European journal of operational research 265, pp. 1102–1114.
- Beavers, A. N. (2001). Roadmap to the e-factory. United States: CRC Press.

- Bhatt, J., M. Gooch, B. Dent, and G. Sylvia (2017). "Implementing interoperability in the seafood industry: Learning from experiences in other sectors". In: Journal of food science 82, A22–A44.
- Bollen, F., C. Kissling, J. P. Emond, R. Crompton, C. Nunes, A. Metz, K. Duval, M. Laniel, and J. Ye (2004). Sea and air container track and trace technologies: Analysis and case studies. Lincoln ventures limited supply chain systems research, New Zealand: APEC project.
- Bryceson, K. P. and A. Yaseen (2018). "Smart technologies: Disruptive technologies and the agrifood industry – fit for practice?" In: *E-business and supply chain integration: Strategies and case studies from industry*, ed. by O. Bak. Kogan Page, pp. 27–50.
- Castro, J. A. O. and W. A. Jaimes (2017). "Dynamic impact of the structure of the supply chain of perishable foods on logistics performance and food security". In: *Journal of industrial engineering* and management 10.4, pp. 687–710.
- Chen, Q. and H. M. Chen (2004). "Exploring the success factors of eCRM strategies in practice". In: Database marketing and customer strategy management 11.4, pp. 333–343.
- Chibani, A., X. Delorme, A. Dolgui, and H. Pierreval (2018). Dynamic optimisation for highly agile supply chains in e-procurement context. International journal of production research, In Press.
- Corrado, C., F. Antonucci, F. Pallottino, J. Aguzzi, D. Sarria, and P. Menesatti (2012). "A review on agri-food supply chain traceability by means of RFID technology". In: *Food and bioprocess* technology 6.2, pp. 353–366.
- Cox, H. and S. Mowatt (2004). "Consumer-driven innovation networks and e-business management systems". In: Qualitative market research: An international journal 7.1, pp. 9–19.
- Harris, J. K., P. M. C. Swatman, and S. Kurnia (1999). "Efficient consumer response (ECR): A survey of the Australian grocery industry". In: Supply chain management: An international journal 4.1, pp. 35–42.
- Heard, B. R., M. Taiebat, M. Xu, and S. A. Miller (2018). "Sustainability implications of connected and autonomous vehicles for the food supply chain". In: *Resources, conservation and recycling* 128, pp. 22–24.
- Hill, C. A. and G. D. Scudder (2002). "The use of electronic data interchange for supply chain coordination in the food industry". In: Journal of operations management 20.4, pp. 375–387.
- Hill, C. A., G. P. Zhang, and K. E. Miller (2018). "Collaborative planning, forecasting and replenishment and firm performance: An empirical evaluation". In: *International journal of production* economics 196, pp. 12–23.
- Howard, M., R. Vidgen, and P. Powell (2006). "Automotive e-hubs: Exploring motivations and barriers to collaboration and interaction". In: *Journal of strategic information systems* 15.1, pp. 51–75.
- Inc., R. L. (Aug. 2018). "The autonomous supply chain: Possible or impossible?" In: The River Logic Blog 17, p. 2017.
- Johnston, H. R. and M. R. Vitale (1988). "Creating competitive advantage with inter-organizational systems". In: MIS quarterly 12.2, pp. 153–165.
- Kim, D., R. J. B. Jean, and R. R. Sinkovics (2018). "Drivers of virtual interfirm integration and its impact on performance in international customer-supplier relationships". In: *Management international review*, pp, pp. 1–28.
- Kim, H. J. (2017). "Information technology and firm performance: the role of supply chain integration". In: Operations management research 10.1, pp. 1–9.
- Kinsey, J. and S. Ashman (2000). "Information technology in the retail food industry". In: *Technology in society* 22.2, pp. 83–96.

- Klitou, F., J. Conrads, M. Rasmussen, L. Probst, and B. Pedersen (June 2017). "Germany: Industrie 4.0". In: European Commission, European Union 17, p. 2018.
- Koutsoumanis, K., P. S. Taoukis, and G. J. E. Nychas (2005). "Development of a safety monitoring and assurance system for chilled food products". In: *International journal of food microbiology* 100.1, pp. 253–260.
- Krishnan, V., O. Mnyshenko, and H. Shin (2015). "Inclusive innovation: Broader market coverage for innovative products with deliberate supply chain leadership". San Diego: University of California. Working paper.
- Le, T. (2005). "Business-to-business electronic marketplaces: Evolving business models and competitive landscapes". In: International journal of services technology and management 6.1, pp. 40–53.
- Lee, K. L., Z. M. Udin, and M. G. Hassan (2014). "Supply chain technology adoption: Its clarification, evolution, classification and particularly in textile and apparel industry". In: *International journal* of business and economics research 3.6, pp. 15–21.
- Leonard, L. N. K. and C. C. Davis (2006). "Supply chain replenishment: before-and-after EDI implementation". In: Supply chain management: An international journal 11.3, pp. 225–235.
- Martens, B. J. and F. J. Dooley (2010). "Food and grocery supply chains: A reappraisal of ECR performance". In: *International journal of physical distribution and logistics management* 40.7, pp. 534–549.
- Mason, S. J., P. M. Ribera, J. A. Farris, and R. G. Kirk (2003). "Integrating the warehouse and transportation functions of the supply chains". In: *Transportation research* 39.2, pp. 141–159.
- McLaren, T., M. Head, and Y. Yuan (2002). "Supply chain collaboration alternatives: Understanding the expected costs and benefits". In: *Internet research: Electronic networking applications and policy* 12.4, pp. 348–364.
- Mussomeli, A., D. Gish, and S. Laaper (May 2017). "The rise of the digital supply network". In: Supply Chain, Deloitte [online] < [accessed 15, p. 2108.</p>
- O'Byrne, R. (May 2017). "6 Key supply chain and logistics trends to watch in 2017". In: *Logistics Bureau* 15, p. 2018.
- Opara, L. U. (2003). "Traceability in agriculture and food supply chain: A review of basic concepts, technological implications, and future prospects". In: *Food, agriculture and environment* 1.1, pp. 101–106.
- Patterson, K. A., C. M. Grimm, and T. M. Corsi (2004). "Diffusion of supply chain technologies". In: *Transportation journal* 43.30, pp. 5–23.
- Pokharel, S. (2005). "Perception on information and communication technology perspectives in logistics: A study of transportation and warehouses sector in Singapore". In: *The journal of enterprise information management* 18.2, pp. 136–149.
- Ramamurthy, K. and G. Premkumar (1995). "Determinants and outcomes of electronic data interchange diffusion". In: IEEE Transactions on engineering management 42.4, pp. 332–351.
- Roblek, V., M. Mesko, and A. Krapez (2016). "A complex view of Industry 4.0, SAGE Open". In: p, pp. 1–11.
- Ru, J., R. Shi, and J. Zhang (2018). "When does a supply chain member benefit from vendormanaged inventory?" In: *Production and operations management* 27.5, pp. 807–821.
- Sahin, E., M. Z. Babai, and Y. Dallery (2007). "Ensuring supply chain safety through time temperature integrators". In: *The international journal of logistics management* 18.1, pp. 102–124.
- Salin, V. (1998). "Information technology in agri-food supply chains". In: International food and agribusiness management review 1.3, pp. 329–334.

- Santos, C., A. Mehrsai, A. C. Barros, M. Araujo, and E. Ares (2017). "Towards Industry 4.0: an overview of European strategic roadmaps". In: *Procedia manufacturing* 13, pp. 972–979.
- Schrauf, S. and P. Berttram (May 2016). "Industry 4.0: How digitization makes the supply chain more efficient, agile and customer focused". In: Strategy [online] < [accessed 15, p. 2018.</p>
- Schuh, G., R. Anderl, J. Gausemeir, M. T. Hompel, and W. Wahlster (2017). Industrie 4.0 maturity index: Managing the digital transformation of companies. Acatech Study, Munich: Herbert Utz Verlag.
- Solanki, M. and C. Brewster (2014). "Enhancing visibility in EPCIS governing agri-food supply chains via linked pedigrees". In: *International journal on semantic web and information systems* 10.3, pp. 45–73.
- Soosay, C. (2008). "Operational issues and challenges in cold chain logistics management". In: Proceedings of the 6th Australia and New Zealand Academy of Management: Operations. Queensland, Australia: supply chain and services management symposium, pp. 1–14.
- Theiss, A., D. C. Yen, and C. Y. Ku (2005). "Global positioning systems: An analysis of applications, current development and future implementations". In: *Computer standards and interfaces* 27.2, pp. 89–100.
- Vlachos, I. P. (2004). "Adoption of electronic data interchange by agribusiness organizations". In: *Journal of international food and agribusiness marketing* 16.1, pp. 19–42.
- Vorst, V. D. and D. S. J. G. A. J. (n.d.). "C.A. and Trienekens, J.H". In: 2007 ().
- Waller, M., E. M. Johnson, and T. Davis (1999). "Vendor managed inventory in the retail supply chain". In: Journal of business logistics 45.1, pp. 63–68.
- White, A., E. Daniel, J. Ward, and H. Wilson (2007). "The adoption of consortium B2B e-marketplace: An exploratory study". In: *The journal of strategic information systems* 16.1, pp. 71–102.
- William, D. P. J. (2003). "Supply management and e-procurement: Creating value added in the supply chain". In: Industrial marketing management 32.3, pp. 219–226.
- Wu, N. C., M. A. Nystrom, T. R. Lin, and H. C. Yu (2006). "Challenges to global RFID adoption". In: Technovation 26.12, pp. 1317–1323.
- Xu, S., K. Zhu, and J. L. Gibbs (2004). "Global technology, local adoption: A cross-country investigation of internet adoption by companies in the United States and China". In: *Electronic markets* 14.1, pp. 13–24.
- Zhu, K., K. L. Kraemer, and S. Xu. (2003). "E-business adoption by European firms: A cross country assessment of the facilitators and inhibitors". In: *European journal of information systems* 12.4, pp. 251–277.

Advanced Scientific Algorithms in Digital Factory Design Applications

Jacob Lohmer¹, Armin Klausnitzer¹, Rainer Lasch¹

1 - Technische Universität Dresden

In contrast to manual procedures, research constantly develops modern computerized factory design approaches applying concepts of Industry 4.0. Nevertheless, literature hardly reveals their practical application. A market review analyzing recent software applications shows that they still focus on simple and outdated algorithms, resulting in limited support for facility arrangement. This study further analyzes the use of scientific approaches implemented in software applications for factory design and suggests new research directions to increase their practical applicability. Therefore, we conducted a semi-structured interview study investigating four layout design software developers. Applying an iterative approach, semi-structured and guided interviews were performed, coded, grouped into cateaories referring to similar phenomena, and interpreted to provide proposals for future research. The results indicate that innovative, automated methods are met with skepticism, although their performance is considerably superior to traditional approaches. The interviewed developers for factory design software reveal certain features they expect from modern approaches emerging from research to ensure successful implementation.

Keywords: Factory layout design; Software applications; Metaheuristics; Interview study

First recieved: 28.May.2018 Revised: 22.Jun.2018 Accepted: 05.Jul.2018

1 Introduction

Manufacturing companies are constantly faced with major challenges shaping their requirements for success, such as new production technologies, shorter product life cycles, as well as frequent changes in market dynamics. Certain situations may compel companies to make major changes to their manufacturing sites, affecting the arrangement of facilities and equipment in the plants. This common problem is referred to as the "facility layout problem" (FLP) in literature. There is substantial prior research starting from the 1960s (Anjos and Vieira, 2007).

During the last decades, a trend towards metaheuristics and algorithm-based optimization of the arrangement problem has evolved (Ahmadi et al., 2017). Nevertheless, the practical application of scientific methods, particularly their software implementation, within design projects is scarcely mentioned. However, in the context of Industry 4.0 and Smart Manufacturing, advanced software applications are gaining importance throughout all subject areas in manufacturing companies and constitute a worthy area for optimization and further research (Bracht et al., 2018).

In a recent paper, Klausnitzer et al. (2017) conducted a case study to investigate the application of scientific algorithmic methods in practice. Users of software applications, consulting engineers in this case, and a scientific institute specialized in factory design were interviewed. Their key findings are that advanced mathematical methods are not applied to factory design projects due to the users' unawareness of extant superior methods and lack of algorithmic support regarding the applied software solutions. These findings add impetus to the research of this paper.

Whereas Klausnitzer et al. (2017) focused on users of factory design applications, we suggest that investigating developers of factory design applications might produce interesting findings. It can be safely presumed that the developers of software applications in digital factory design have a profound knowledge of methods and trends in their field of interest and have a high affinity for scientific methods. To the best of our knowledge, there is no empirical research on the approach applied by software engineers to develop the applications a well as the impact of scientific research on these development procedures. Furthermore, the evolving trend towards metaheuristics has produced various scientific algorithms for the FLP that produce promising results but little is known about the value the software developers for factory design applications give these algorithms.

Therefore, this paper aims to answer the following research questions using a market review followed by a multiple semi-structured interview study:

- How do modern software applications methodically support the factory design process?

- How does scientific research affect developers of those applications?

- Why are advanced scientific algorithms adopted by software developers for factory design?

Our contribution is two-fold: From a theoretical standpoint, we provide insights from practice, in particular the needs and wishes that software developers have regarding scientific methods to encourage their implementation in factory design applications. Researchers are able to focus on specific aspects that will make their methods usable outside of the field of research. From a practical standpoint, we aggregate the recent developments in science and highlight important future developments as noted by leading representatives of software developing companies.

The remainder of this paper is organized as follows. Section 2 provides a brief overview of the literature concerning solution methods for the facility arrangement and digital factory design applications. Section 3 describes the methodology adopted for the empirical research. Section 4 presents the findings from the market review as well as the interview study. Section 5 provides recommendations for practice, section 6 our conclusions. The limitations of the study and opportunities for future research are discussed in section 7.

2 State of Research

The basic process of factory design as defined in VDI (2011) or Grundig (2015) consists of the following planning phases: Target planning, concept planning, detailed planning, execution planning, and implementation. A large amount of research considering the FLP focuses on the concept and detailed planning phases, where several layout variants are generated to minimize the material handling effort as a function of travel distances. The first steps include determining the functions of facilities, dimensioning of required areas, and the arrangement of the facilities within the area. Subsequently, the block layout variants need to be refined with real life requirements considering structural conditions like walls and

pillars, as well as ergonomic and logistic requirements. Favored layout variants are then designed according to specific conditions and restrictions such as facility details and orientation.

The step of arranging the facilities can be reduced to an algorithmic process that solves an optimization problem as it depends on material flow intensities and closeness ratings (Drira et al., 2007; Tompkins et al., 2010). Popular procedural approaches for factory planning and solving the FLP (e.g. Schenk et al., 2014; Grundig, 2015) still focus on outdated and simple algorithmic methods, such as the approaches of Schwerdtfeger (Kettner et al., 1984) and Schmigalla (Schmigalla, 1968). These methods illustrate flow relationships between functional areas to arrange them according to a determined pattern. Specific facility areas and real-life requirements are taken into account subsequently. The solution quality largely depends on the experience of the layout planner as well as trial-and-error.

Other approaches to solve the FLP are mathematical methods that lead to optimal solutions. However, the FLP is known to be NP-hard (Garey and Johnson, 1979), resulting in computational intractability for most of the practical problems, due to the large number of facilities to be arranged. This led to the emergence of heuristics and, more recently, metaheuristics, such as genetic algorithms, tabu searches, or simulated annealing. They are suitable for large problems that can be solved in a short time with solution qualities near the optimum, although the optimum cannot be guaranteed (Drira et al., 2007). In contrast to the traditional methods, modern approaches consider a variety of requirements like area constraints, material handling points, path design, and multiple floors in one single step. A concurrent approach to the arrangement of facilities and the design of real life requirements can significantly improve the quality of the layout solutions and the productivity of the user. Within a short time, several layout variants can be generated and evaluated without the follow-up work or a costly replanning of infeasible layouts obtained during a sequential process.

To support the design process for the FLP there are several factory design software applications. The selection of the suitable software application depends on the specific problems, data, information and defined goals (Schenk et al., 2014). In the context of Industry 4.0/Smart Manufacturing, advanced applications supporting the "digital factory" are entering the market. Digital factory is a network of digital models, methods, and tools integrating a continuous data management system that aims for holistic planning, evaluation, and improvement of structures, processes, and resources of a factory.

It allows for collaborative planning to consider the experience of various users, which further contributes to an improvement of solution quality and approval (VDI, 2008). The digital factory applications can be categorized into three main groups that are briefly presented in the following:

Tools for visualization: CAD (Computer Aided Design) tools are used to visualize facility layouts and draft well-detailed drawings that are easy to create and to change in a short time. CAD tools are widely utilized in all kinds of manufacturing companies worldwide. However, standard applications like AutoCAD rarely provide scientific methods to support the layout design procedure (Grundig, 2015).

Simulation tools: Simulation tools are used to study the dynamic interrelations of manufacturing systems. The manufacturing system can be optimized by creating and analyzing different variants to select the best alternative (Arnold and Furmans, 2009). In recent years, CAD tools have been coupled with simulation tools to create layouts based on CAD data directly using standard 3D components. The complexity of the software applications and the needed expertise to generate a coherent representation of the real process flow from a static visualization is still one of the main challenges of simulative factory design (Schenk et al., 2014).

Virtual reality tools: Intuitive applications like virtual reality and augmented reality tools are receiving an increased interest regarding layout design. They are able to construct a direct or indirect view of real-world environments augmented with additional haptic or visual information (Schenk et al., 2014). 3D modeling and optimization of whole factory floors is simplified. So called "participative planning tools" combine two-dimensional representations as in CAD tools with three-dimensional representations for a better visual experience and cooperative collaboration in project teams. In addition, users are provided with evaluation and optimization components (Jiang and Nee, 2013).

In summary, new and innovative mathematical methods like metaheuristics pose several improvements and advantages over more simplistic methods from a scientific perspective. They are able to solve large problems quickly with solution qualities near the optimum and concurrently integrate several design aspects that have been solved in a sequential process so far. However, the software applications are still focusing on features that support the manual design process without the use of advanced algorithms. This gap motivates the investigation of the development procedures of software applications, the underlying intentions, used scientific methods and the assumed but unclear scientific impact on the procedures.

3 Research Methodology

As there is no comparable research so far, we used an approach of rather explorative nature. First, we conducted a market review to identify promising software applications for facility layout design that use scientific methods to arrange facilities in the manufacturing sites. Therefore, a systematic search was conducted using web databases, project descriptions and scientific articles. We then classified the identified software applications after a thorough review and analysis (see section 4).

Second, we conducted a multiple semi-structured interview study with experts from a subset of the identified software engineering companies. The semi-structured interview design allows flexibility to adapt to specific but initially unknown circumstances in practice (Yin, 2009). To analyze a representative sample of cases for the area of interest, the concept of theoretical sampling was used (Eisenhardt and Graebner, 2007). Therefore, four software engineering companies took part in the interview study. The cases develop and distribute software applications with implemented scientific methods, especially in factory design. Although the number of interviews is relatively small, in the limited field of facility design software solutions the selected cases and the findings are representative and can be theoretically generalized (Meredith, 1998). The organizational characteristics of the examined software companies are shown in Table 1.

Cases A, B and C represent software engineering companies of different size. Cases A and B are small companies operating in Germany with less than 15 employees. In contrast, case C constitutes a large company with headquarters in Germany and the USA and employing over 9,000 people. In order to investigate the layout design methods applied by a scientific organization, we also interviewed an institute of a university (case D). The interview partner of case D has developed a design software application and distributes it as a freelancer.
ase Em Nur 11	ployee	S				
11 Nur	mber		Customers		contact with Science	Interviewee
11		Degree course	Sectors	Regions		
		ME/IE,CS	AII	Worldwide	Software used in university courses	Managing Partner
12-	15	ME/IE,	Automotive	Worldwide	Practice journals and associations (VDI)	Director
9.0	00	ME/IE,CS, E	All	Worldwide	University work- shops, confer- ences	Technical Specialist
20		ME/IE	AII	Germany	Software used in university courses	Research Associate, Freelancer

* ME/IE =mechanical / industrial engineering, CS = computer science, E = economics

Besides the distribution of software, all cases provide workshops and training courses to teach their customers as well as consulting services around layout design. The interviewed cases employ a highly skilled workforce with university degrees mainly in the fields of mechanical and industrial engineering. The customers of cases A to D operate in all sectors and industries around the world, though most are manufacturing companies. A comparison of project figures is complex since all cases stated that project durations and budgets differ depending on the specific customer needs. However, most projects range between a few days for basic training workshops to about three months for additional consultancy and may be extended by several months for major planning support.

Only case A actively cooperates with universities and scientific institutions regarding facility design projects. The interview partner of case B sporadically reads journals and uses the mutual exchange of information in associations like the Verein Deutscher Ingenieure (VDI) to stay up-to-date. There is no active collaboration considering the development of the software application with scientific institutions. This also applies for case C. Nevertheless, the interview partner takes part in workshops at universities and is a speaker at certain professional events. The academic institution of case D collaborates with colleagues from the same university in projects where they often utilize the software application. Further feedback comes from several other universities that use the software application for teaching and project purposes.

After the investigation of general characteristics, we applied an iterative approach for data collection, coding, and analysis (Eisenhardt, 1989). For preparation purposes and the appropriate selection of interview partners in the case companies, we provided the interview partners with a questionnaire prior to the interview as an extract from the guidance document. During the interview, we used the guidance document to focus on all relevant aspects. The data from the transcribed interviews was coded using MAXQDA 2018 to detect important aspects from the transcript.

All statements were grouped into categories referring to similar phenomena. According to a deductive-inductive approach, main categories were established prior to the interviews whereas subcategories were created from the investigation of the first transcript. In a first step, the text passages were grouped into the main categories. Organizational aspects like staff structure, company size, and customer features were grouped in "Organization". The second main category "Software" covered aspects like application range and development procedures. "Layout Planning" included the implemented scientific methods for facility design and all comments regarding advanced mathematical methods.

The "Link to Science" was further analyzed in a fourth main category including the information approach towards current development and an assessment of the mutual exchange with scientific institutions. After the inductive determination of the subcategories, the whole document was grouped according to a differentiated system of categories (Kuckartz, 2014).

The analysis was carried out case-by-case first and completed with a comparative analysis. The resulting conceptual model represents a common understanding of phenomena regarding the cases. In addition to the interviews, we used company websites and provided material by the interview partners (brochures) to triangulate the interview data. The final reports were sent to the interview partners for validation purposes and to maintain participant engagement.

4 Findings

4.1 Market review on software applications

Table 2 illustrates the identified software applications and respective methods, features, and limitations. The software applications were then grouped into four categories and visualized in Figure 1. The two aspects of interest for this classification are the degree of integration of all planning tasks into the basic planning process of factory design (e.g. VDI, 2011) and the provided scope of functionalities for the arrangement of objects in the factory layout.

Table 2	: Identified soft	ware applications	and their respective scientifi	c methods, features, and limitations
	Tool	Scientific method for fa- cility design	Features	Limitations
	AutoCAD	none	Widely used in the in- dustry; enables a digital representation of possi- ble real world systems	Only supports trial-and-er- ror for facility arrange- ment without any quanti- fied evaluation
	G-MAFLAD	Branch-and- Bound (exact)	Considers facility di- mensions for a first block layout variant	Users need to provide cer- tain feasible arrangement sites for each facility; no material paths considered; limited import/export fea- tures
	FACOPT	Simulated An- nealing / Ge- netic Algo- rithm	Considers facility and factory (incl. gates) di- mensions for a first block layout variant	Provides block layouts without considering mate- rial paths or real life re- quirements; not commer- cially available
	VIP- PLANOPT	Simulated An- nealing + Local Search	Considers facility di- mensions; extensive configuration options (dimensions, shape flex- ibility, factory floor)	Considers predefined ma- terial paths via blocked ar- eas in the layout; scientific method not fully disclosed

Advanced Scientific Algorithms in Digital Factory Design Applications

SIMOGGA	Genetic Algorithm	Considers various input data (factory floor, ma- terial paths, facility di- mensions)	Only usable for cellular layouts with redundant re- sources; needs an initial solution to start upon
MALAGA	Greedy Heuristic	Integrated plant model; connection to simula- tion software (Plant Simulation by Siemens)	Facility dimensions ne- glected for first steps of factory design, dimensions need to be manually ap- plied subsequently
Tecnomatix Factory De- sign	Graphical Method	Connection to various related software mod- ules (Plant Simulation by Siemens)	Needs an initial solution to start with an optimization; no consideration of mate- rial paths and input/out- put points
SketchUp MF-VO	Graphical Method	Plugin for 3D software "SketchUp"; presenta- tion, analysis, and opti- mization of material flow; considers paths	Solution method provides recommendations for ar- rangement that need to be applied manually
visTA- BLE®touch	Greedy Heuristic	Combined tool for visu- alization and layout de- sign; considers paths; quantified evaluation	Solution method provides recommendations for ar- rangement that need to be applied manually

Although three identified applications used metaheuristics to arrange objects in the factory layout (VIP-PLANOPT, FACOPT and SIMOGGA), these applications are unable to connect with other commercial applications and therefore cannot be widely used. They only offer partial support for the factory design phases providing first block layout variants that need further fine-tuning, and were therefore entitled 'specific tools' in Figure 1. Comprehensive tools with a modern, user-friendly graphical user interfaces like visTABLE®touch or the SketchUp MF-VO Plugin still use simple but outdated methods, such as the method by Schmigalla (1968) or the method by Schwerdtfeger (Kettner et al., 1984).

4.2 Interview Study

Cases A, B, and D provide a single software application while case C provides several applications specified for different sectors. The mechanisms that led to the development of the software applications are similar in all cases. At the time of development, no product was available on the market which offered the functionalities that the interview partners needed for their projects. This led to in-house developments of software applications appropriate for the specified design projects after which cases A to D started offering their applications for sale to other engineering companies. The software was then increasingly enhanced in order to improve customer-orientation and user-friendliness. However, hardly any developments considering advanced scientific algorithms were adopted from contemporary scientific contributions. On the contrary, it is apparent that software developers are familiar with the general layout planning procedure of dividing the design task into several feasible smaller design steps (e.g. Schenk, 2014; Klausnitzer et al., 2017).

All cases use simple methods in their software applications to arrange objects in the sites. Case A uses greedy heuristics (related to the method by Schmigalla (1968) and the method by Martin (1976)) to optimize existing or initial layouts. Case B uses a modified approach of the method by Schmigalla (1968), while case C is currently testing an algorithm for a generative design of layout variants to generate a feasible layout variant. Case D uses the graphical method by Schwerdtfeger (Kettner et al., 1984) to arrange facilities. The interviewees are consistent in their skepticism towards the use of modern algorithmic approaches and stated that advanced algorithmic approaches are too difficult to apply in factory layout design. Moreover, the additional effort of implementing and applying the metaheuristics in comparison to simple heuristics does not represent a favorable



Figure 1: Classification of reviewed software applications for facility layout planning

cost-benefit ratio. Interviewees A and D further criticize that metaheuristics still only produce an initial solution for the layout problem. This solution has to be subsequently adjusted to real life requirements that were not considered by the algorithm.

The general skepticism about novel approaches suggests a lack of knowledge considering the existence of advanced algorithms. In all cases, the employees are usually mechanical and computer engineers. Advanced algorithms regarding the FLP as a long-time challenging discipline of operations research are usually not focused on these fields of study. Therefore, the interviewees are not aware of the considerable benefit in the quality of the layout solution and the potential of optimization of the overall planning. Since the traditional planning procedure suggests a top-down approach where restrictions are considered step by step, the planning process might run up against infeasible intermediate layout solutions

several times that require multiple setbacks and replications of previous planning phases. In contrast, advanced computer supported layout design approaches can integrate several design elements and restrictions simultaneously within a single planning step. This leads to feasible layouts with respect to the considered restrictions and planning details. If such an algorithm is implemented within a software application, it is able to produce several layout variants within a few minutes that can be qualitatively and quantitatively evaluated and modified by the user. As discovered in this study, current software applications often only automate the relative arrangement of facilities using the approaches of Martin (1976) and Schmigalla (1968). Both approaches do not consider any restrictions, not even the areas of facilities and the planning areas. Therefore, solutions obtained with current applications frequently get stuck in infeasible intermediate solutions that require time-consuming replanning.

As regards the scientific education and the lack of implementability of advanced methods, interviewees A and C noted that the general quality of education has decreased and that it is difficult to find qualified employees capable of implementing advanced algorithms. This is enormously important for software developers like cases A to D since the software as well as the algorithms evolve continuously according to the design tasks. Moreover, interviewee B noted that there is a general lack of time to proactively enhance the software application. All interviewees are still occupied by the daily business of layout design projects due to their origins as software providers.

With regard to the customer's capabilities in factory design and planning, all interviewees are consistent in their observation that the majority of customers do not grasp the relevance of factory design. Particularly interviewees B and C noted that the customers are not aware of the overarching objective of minimizing the material handling effort in the layout design as a considerable contribution for long-term efficiency and success.

Instead, objectives like a resemblance to existing testimonial factory layouts or low alteration costs are prioritized, resulting in low layout performance and a redesign early on. In addition, only a few users of software applications use factory design methods correctly and consistently due to a lack of expertise and constant time pressure in the planning process. This difficulty could also be overcome by the use of advanced approaches in software applications for digital factory design. Interviewee A believes that with new Big Data analytics there is an opportunity to gain a deeper understanding of the impact of factory design methods on decision makers. Interviewee D foresees a shifting focus from simulation to Big Data applications. The digital factory and the associated trend towards constant redesigns caused by the volatile markets, new technologies, and changing legal conditions must therefore trigger a rethinking of the users towards a consistent and transparent planning approach.

Moreover, interviewee B sees a nascent conflict between the planning department and the IT department. An adoption of advanced algorithms that integrate design aspects of several planning phases would result in major changes in the well-established planning procedure in all cases. Thus, the IT department is increasingly seizing more planning responsibility and decision-making power from the factory designers. Additionally, all interviewees hinted that layout designers want a transparent and comprehensive layout planning procedure. They are skeptical about layout solutions that are generated automatically by pushing a button. These statements can be summarized as general acceptance problems concerning advanced algorithms.

Despite the general wariness about advanced layout design approaches, particularly interviewees A and B showed strong interest in these methods and in a close cooperation with specialized scientific institutes. Especially interviewees A and C assume that metaheuristic solutions will play a greater role in future developments, which is closely related to the further increasing computing capacity.

5 Recommendations for Practice

As a result of the conducted study, it is clear that the software developers are currently not focusing on advanced algorithmic approaches. They are not aware of their existence, possible benefits and implementation and showed therefore a general suspicion. A closer collaboration with universities would pave a direct way for the implementation of promising methods and would facilitate superior software applications. Therefore, scientific institutions like universities have the task to change and modernize their factory design courses. They should enhance the strong engineering view with findings from operations research that has focused on the layout design problem for more than 60 years as a separate stream of research (Koopmans and Beckmann, 1957). Students need to be better prepared for the future challenges of factory design resulting from digitization. This includes training in software applications in support of the entire planning process. Additionally, the focus should shift from CAD systems to specialized

factory design applications that consider the overall production planning process and the connected logistic processes.

Software providers often serve as consultants in layout design projects and therefore focus on employees with engineering skills. They should also consider employing specialists skilled in operations research to continuously develop the design software and to implement advanced approaches.

In order to solve the acceptance problems, interviewees A, B, and D noted that automated layout design solutions could be valuable if they are used for orientation purposes and in comparison to manually designed layout solutions. Hence, the customers' acceptance will increase and advanced scientific methods can be positioned and marketed as a key feature of the software application. Interviewee C states that a complete replacement of previous simple, greedy methods by metaheuristics is conceivable. Moreover, interviewee C noted that software solutions are generally acceptable if they are user-friendly without time-consuming training and difficult settings.

Interviewee B further recommends well-prepared key users for the usage of advanced solutions in software to ensure a correct use of the generated layouts. The key users should be guided through the design process while automatically focusing on the overarching aim to minimize the material handling effort. Simple and intuitive software usability as well as certain adjustment possibilities would further improve customer acceptance.

According to all interviewees, future software applications should support the entire design process. This also requires interface functions to export the solution into common data formats for further visualization and simulation. Interviewee D noted that integrating the algorithms into a three-dimensional planning procedure is of further importance. Since the interview partners were concerned about the decreasing influence of the users in case of implementing advanced algorithms in the software, literature suggests implementing expert systems (García-Hernandez et al., 2014). These advanced approaches allow for the consideration of the user's experience at several planning steps.

6 Conclusions

There are a significant number of publications in the scientific area of facility layout design, with an apparent trend towards metaheuristics and algorithm-based optimization of the arrangement problem. Although these advanced methods offer several advantages over traditional methods from a scientific perspective, our market review indicated that current design software still uses these simple and outdated methods in the applications. The prevailing part of the design task remains with the designer without any software support. This answers our first research question regarding the methodical support of factory design processes in software applications.

In order to investigate the reason for this practice-research gap and to identify future research direction, we further conducted a multiple interview study with experts from software engineering companies and a scientific institution that distributes a self-developed software.

As regards our second and third research question, our study showed that all investigated cases are connected to science to some extent and apply a similar general layout planning procedure that is well-known in manufacturing engineering research. Current advances in advanced algorithmic approaches from the scientific area of operations research are not affecting the interviewed software developers due to the fact that the case companies

- lack knowledge regarding the existence of advanced algorithms, caused by the scarcity of employees familiar with operations research;
- lack knowledge regarding the implementation, due to the decreased quality of the staff's education;
- lack knowledge regarding the usage of advanced algorithms, due to the use of the software by customers unskilled in layout design;
- expect problems of acceptance, since advanced algorithms that automatically solve design problems would shift reasonability and decisionmaking power from the designer to the IT department.

Even the scientific institute, with an anticipated deeper understanding of scientific advanced methods, still uses simple algorithms and reveals the same findings as the other cases. This suggests that the research communities (engineering and operations research) are segregated, although they focus on the same topic from different viewpoints.

This study further derived recommendations for practice in light of Industry 4.0, besides the motivation for an intensified cooperation of software developers and scientific institutes from different disciplines:

- software engineering companies should employ staff with specific operations research skills;
- the education of mechanical engineering in the field of manufacturing design should integrate specific operations research issues;
- advanced algorithms should be made user-friendly and be implemented as complementary features at first, until the customers have gained trust in these features.

7 Limitations and further Research

The results of this study are analytically generalizable in terms of correlative validity; however, there are some limitations. The validity of this interview study is within a clearly defined scope:

- Software manufacturers with branches in Germany, specialized in layout design software and services;
- Customers from the manufacturing industry (SMEs to multinationals);
- Individual consultation and fulfillment of customer requests within the projects.

Future research using a larger sample size with diverse industries and firm characteristics would improve reliability of this interview study. An approach which involves interviewing software manufacturers with an international focus should also be expedited in order to validate the findings.

Additionally, we indicate some recommendations for research in order to focus the research for a better applicability of findings in practice in the imminent trend to Smart Manufacturing and Industry 4.0:

The interview partners described several features that they expect from an advanced approach to be implemented in their software. Interviewees A and B demand real-time capabilities of the solution method to make use of the benefits of collaborative planning. Interviewees A and C consider several minutes as a time requirement for a sophisticated solution to be acceptable in terms of collaborative planning.

This allows for a discussion of often uncertain input data and the following solution evaluation in collaboration. Moreover, if the algorithm produces several layout variants, interviewee C expects a recommendation for a preferred layout variant. Interviewee D emphasizes the relevance of a comparison value with traditionally obtained solutions to classify the solution quality. Therefore, the software should be able to calculate the material handling effort expected from the obtained layout variant and should further enable the user to define customized evaluation criteria.

Besides the objective function aiming to minimize the material handling effort, several main restrictions were identified in the interview study that should be taken into account by researchers when developing a metaheuristic:

- Facility shapes
- Facility orientations
- Dimensions
- Path and aisle design
- Fixable functional areas
- Structural conditions, e.g. walls and pillars

Most scientific approaches already satisfy several of these requirements since they are often rather fast, evaluate an adjustable quantitative objective function value and satisfy most of the required restrictions. However, there are only a few approaches that consider three-dimensional layout information (Barbosa-Póvoa et al., 2002) and the material path system (Friedrich et al., 2018). To the best of our knowledge, there is still a lack of scientific approaches that explicitly consider the aisle design. In order to make their findings better accessible for practitioners, scientific institutes should proactively support the comprehensibility throughout the research process. Researchers are moreover advised to publish their findings considering layout design in practice-relevant journals in addition to scientific journals.

References

- Ahmadi, A., M. S. Pishvaee, and M. R. Akbari Jokar (2017). "A survey on multi-floor facility layout problems". In: Computers & Industrial Engineering 107, pp. 158–170.
- Anjos, M. F. and M. V. C. Vieira (2017). "Mathematical optimization approaches for facility layout problems: The state-of-the-art and future research directions". In: *European Journal of Operational Research* 261.1, pp. 1–16.
- Arnold, D. and K. Furmans (2009). Materialfluss in Logistiksystemen. Springer Berlin.
- Barbosa-Povoa, A. P., R. Mateus, and A. Q. Novais (2002). "Optimal 3D layout of industrial facilities". In: International Journal of Production Research 40.7, pp. 1669–1698.
- Bracht, U., D. Geckler, and S. Wenzel (2018). Digitale Fabrik Methoden und Praxisbeispiele. Springer Berlin.
- Drira, A., H. Pierreval, and S. Hajri-Gabouj (2007). "Facility layout problems: A survey". In: Annual Reviews in Control 31.2, pp. 255–267.
- Eisenhardt, K. M. (1989). "Building Theories from Case Study Research". In: The Academy of Management Review 14.4, pp. 532–550.
- Eisenhardt, K. M. and M. E. Graebner (2007). "Theory building from cases: Opportunities and challenges". In: Academy of Management Journal 50.1, pp. 25–32.
- Friedrich, C., A. Klausnitzer, and R. Lasch (2018). "Integrated slicing tree approach for solving the facility layout problem with input and output locations based on contour distance". In: *European Journal of Operational Research* 0, pp. 1–15.
- García-Hernández, L., M. Pérez-Ortiz, A. Araúzo-Azofra, L. Salas-Morera, and C. Hervás-Martínez (2014). "An evolutionary neural system for incorporating expert knowledge into the UA-FLP". In: *Neurocomputing* 135, pp. 69–78.
- Garey, M. and D. Johnson (1979). Computers and intractability: a guide to the theory of NP-completeness. New York: A series of books in the mathematical sciences.
- Grundig, C.-G. (2015). Fabrikplanung : Planungssystematik, Methoden, Anwendungen. München: Hanser Verlag.
- Jiang, S. and A. Y. Nee (2013). "A novel facility layout planning and optimization methodology". In: CIRP Annals - Manufacturing Technology 62.1, pp. 483–486.
- Kettner, H., J. Schmidt, and H.-R. Greim (1984). Leitfaden der systematischen Fabrikplanung. München: Hanser Verlag.
- Klausnitzer, A., S. Stürzel, and R. Lasch (2017). "Facility Layout Planning Investigating the gap between research and practice". In: *Logistikmanagement - Beiträge zur LM 2017*. Ed. by R. Large, N. Kramer, A.-K. Radig, M. Schäfer, and A. Sulzbach, pp. 267–276.
- Koopmans, T. and M. Beckmann (1957). "Assignment Problems and the Location of Economic Activities". In: *Econometrica* 25.1, pp. 53–76.

- Kuckartz, U. (2014). Qualitative Inhaltsanalyse : Methoden, Praxis, Computerunterstützung. Weinheim: Beltz Juventa Verlag.
- Martin, H. (1976). "Eine Methode zur integrierten Betriebsmittelanordnung und Transportplanung". Dissertation. TU Berlin.
- Meredith, J. (1998). "Building operations management theory through case and field research". In: Journal of Operations Management 16.4, pp. 441–454.

Schenk, M., E. Müller, and S. Wirth (2014). Fabrikplanung und Fabrikbetrieb. Springer Berlin.

- Schmigalla, H. (1968). Methoden zur optimalen Maschinenanordnung. Berlin: Verl. Technik.
- Tao, J., P. Wang, H. Qiao, and Z. Tang (2012). "Facility layouts based on intelligent optimization approaches". In: 2012 IEEE 5th International Conference on Advanced Computational Intelligence, ICACI 2012, pp. 502–508.
- Toksari, M. D. (2016). "A hybrid algorithm of Ant Colony Optimization (ACO) and Iterated Local Search (ILS) for estimating electricity domestic consumption: Case of Turkey". In: *International Journal of Electrical Power and Energy Systems* 78, pp. 776–782.

Tompinks, J. A. (2010). Facilities planning. Wiley.

VDI (2008). VDI 4499 - Digitale Fabrik - Grundlagen. Düsseldorf: VDI-Verlag.

VDI (2011). VDI 5200 - Fabrikplanung - Planungsvorgehen. Düsseldorf: VDI-Verlag.

Yin, R. K. (2009). Case study research : design and methods. Los Angeles: Sage.

Additive Manufacturing in Supply Chains – The Future of Purchasing Processes

Stephanie Niehues¹, Laura Berger¹, Michael Henke¹

1 - TU Dortmund

Additive manufacturing is one of the leading production technologies when it comes to the efficient production of individual parts. This paper explains how additive manufacturing will influence purchasing processes and network structures of producing companies in future supply chain networks. Therefore, an exploratory research of relevant literature and recent studies in a systematic literature review is conducted, giving an overview of how additive manufacturing will change the processes of purchasing. Traditional purchasing and modern processes affected by the implementation of additive manufacturing are compared, in order to point out differences and new requirements. In the future purchasers have to extend their knowledge of additive manufacturing, to integrate this production technology into their sourcing strategy in an effective way. Technological knowledge and capabilities, supplier relations and internal integration of purchasers need to be improved, to make use of benefits e.g. reduced inventory. This exploration of the impact of additive manufacturing is focused on purchasing processes. Therefore, this paper investigates one special field of the supply chain layer, which is the unique characteristic of this submission. The findings are based on a literature review of studies and specialized literature, an analysis in collaboration with companies was not executed. Additive manufacturing has great advantages in many fields, but it cannot be integrated in supply chain processes with a significant leap forward. The sourcing strategy has to be changed; employees have to be trained, to interact with the new supply chain members and to learn which components can be manufactured additively.

First recieved: 28.May.2018 Revised: 06.Jul.2018 Accepted: 13.Jul.2018

Keywords: Additive Manufacturing; Purchasing; Supply Chains; Additive Manufacturing in Supply Chains

1 Introduction

Additive Manufacturing is going its way from a pacemaker to a key-manufacturing technology. A lot of high technology companies spot the benefits of additive manufactured parts. At the moment, most of these parts are prototypes, spare parts or individualized consumer goods, but in the near future, taking the evolution of markets into consideration, more and more individual additive manufactured parts will be produced to enable individualized mass production (Zeyn, 2017). In this scenario, the decision to print additively or to produce a part conventionally has to be made during the purchasing process, preferably only in a matter of seconds. The velocity and the quality of this decision process will be a commercial advantage of companies acting on a globalized market. (Johannknecht and Lippert, 2015; Gress and Kalafsky, 2015)

A number of up-to-date studies and articles describe the trends and current developments in procurement. They do all point out digitalization as one of the main aspects influencing the future position of Purchasing & Supply Management (PSM) and the roles of purchasers (e.g. Kushner, 2015; Nowosel, Terrill and Timmermans, 2015; Pellengahr et al., 2016; PricewaterhouseCoopers, 2014). Digitalization, individualization and increasing autonomy as basic principles of Industry 4.0 also have an extensive impact on Purchasing and Supply Management (PSM) (Pellengahr, Schulte, Richard and Berg, 2016). Due to these facts and also because of the ongoing digitalization in production and administrative processes, the purchasing processes in companies will reach a new technological level.

On the one hand, digitalization is represented by new categories. Firms purchase digital products, services, and hybrid components. In order to acquire the demanded technical expertise, it is even more important to collaborate with other company functions. Purchasers have to be involved in product development processes on an early stage. New categories, e.g. related to additive manufacturing, will have to be procured with new sourcing strategies while others might lose importance or even disappear from the purchasing portfolio. (Arnolds, Heege, Röh and Tussing, 2016; Pellengahr et al., 2016; PricewaterhouseCoopers, 2014; Schuh, Aghassi, Bremer and Graw, 2014)

On the other hand, digital transformation causes not only technological advances and new requirements but also organizational and process-related changes. The administrative purchase-to-pay process will mostly be automated, as digital technologies help in the facilitation process. Further automation of transactional procurement decreases involvement of employees in purchase-to-pay and releases more personal resources for strategic tasks (Kushner, 2015; Pellengahr et al., 2016; Scharlach, Schuh and Strohmer, 2014; Drake, 2012). The purchasing staff members have to develop from former administrators to innovation managers, controlling data analysts and engineers in one person.

The network structures of future supply chains containing additive manufacturing as a production technology will refine and even different network partners will change their position and scope of duties. The value of data and their transfer through the supply chain network will be a new key performance indicator for effective supply chains. (Mellor et al., 2014; Zeyn, 2017)

Data for additive manufacturing processes will contain all information which is needed for an implementation in the production process. They will not only contain construction data but also parametric data and specialized settings for the building process in 3D-printers e.g. temperatures, feeding velocity and material advices. Taking this into consideration, one scenario could be that network partners that were not considered to produce goods can be enabled to execute additive manufacturing processes. (Zeyn, 2017)

A great example for this are forwarding companies that are able to become a producing network partner. The storing and distribution of goods will not be their main business area. The producing of parts on their 3D printers will become an important function for the supply chain network. All the parts that caused high storing costs for companies and that were no particular fast sellers can then be produced additively. As an example, these are parts that are only needed sporadically and were stored for the customer satisfaction, so that they have spare parts for a long period but also parts that had to be stored for legal regulation (e.g. aviation industry). (Mellor et al., 2014)

This paper will give an overview of how processes, especially in purchasing areas and network structures of supply chain, will change and develop because of additive manufacturing making its way to an important manufacturing technology. Based on that, recommended actions can be derived for companies, e.g. regarding know-how and competencies for procurement staff necessary for additive manufacturing, that is, amongst others, process knowledge and optimization, technical knowledge about the new products which need to be sourced.

2 Methodology

A literature review is a key method to overcome and manage the broad diversity of knowledge in management research. The main reason for conducting a literature review is to collect and assess already existing literature in order to specify a research question. With respect to the research question, the existing base of knowledge is about to be developed further. (Tranfield et al., 2003)

In addition to the advantages of literature review papers, such as giving a wellstructured and up-to-date overview of a specific topic, it adds value by revealing knowledge gaps to the researcher. (Van Wee and Banister, 2016)

Newer technologies such as additive manufacturing that have not been adopted by a lot of companies yet, need a lot of guidance through its management. The difficulty is that there are only a few, if any, experienced employees or management staff because of the circumstances and challenges arising with such evolving technologies. Decision making can be really tough without any or only a little knowledge in this area. Literature reviews can provide the company with a broad base of information and knowledge. From this standpoint, it is easier for management and employees to handle decision making around additive manufacturing and develop its own experience and know-how.

The methodology for developing evidence-informed management knowledge provides three steps that have to be passed through for a systematic literature review. (Tranfield, et al., 2003) These steps are deployed in this paper to provide a desirable methodological approach: planning, conducting and evaluate the review.

2.1 Planning the review

The first step contains the systematic planning of the review. In this phase the guiding research question was developed, which is phrased as follows:

How does the implementation of additive manufacturing as a production technology influence future purchasing processes and the needed qualification of employees working in those processes?

The research question was developed upon the fact that an earlier literature review on the influence of additive manufacturing on company processes, in general,

showed that those examinations mostly focused on production processes, value processes or supply chain processes. Due to the fact that purchasing processes get more and more important for companies, because of a globalized market and a reduction of manufacturing depths, also the impact of additive manufacturing on those processes has to be evaluated.

The database, which was used for the review research, was "google scholar". In a second stage, the searching process was also conducted on the database of "Scopus", but in this case it did not provide further relevant papers, and so no additional paper could be added to the detailed review.

In this database, the keywords "additive manufacturing", "3D-Printing", "procurement", and "supply chain management" were combined and in the following the most corresponding papers were selected.

The following combinations of keywords were entered, the number of results for every combination is specified behind:

"additive manufacturing" AND procurement (2.610 results), "3D-printing" AND procurement (2.390 results), "additive manufacturing" AND "supply chain management" (1.630 results), "3D-printing" AND "supply chain management" (2.370 results)

Because of the two very similar keywords "additive manufacturing" and "3D-Printing" a lot of papers appeared in both searches and made the review process less elaborately.

Statistically, most of the publications on this set of issues were published in the years 2016 and 2017 and were classified as an Engineering or Business, Management and Accounting topic.

2.2 Conducting the review

From the first collection of papers, the relevant section was selected to execute the literature review.

For this, a selection of papers and studies was developed, based on a predetermined research strategy. The criteria for the selection was the conceptual link of the papers to both of the topics "Additive manufacturing" AND "Procurement". Only those papers that had a connection to both of these fields and treated the impacts in those fields were chosen for the literature review. This very restrictive search criterion has the advantage of avoiding issues that only mention the procurement process in additive manufacturing as an adjunct.

Beside this criteria, only papers which appeared in the period from 2006 to 2018, were collected in order to develop a current picture of the research field. The search yielded the 11 papers listed in table 1 and represent the final database for our analysis. A second search on the database Scopus was not able to identify additional papers concerning additive manufacturing and procurement.

The selection of papers is summarized in the following table.

Author	Title	Year	Key words
Baldinger M.	Supply Chain Manage- ment für Additive Man- ufacturing: Konzepte, Werkzeuge und Prozesse für die Zusammenarbeit mit Dienstleistern zur Reduktion der Risiken beim Einstieg in Additive Manufacturing	2016	Additive manufacturing cost, cost estimates, make-or-buy, service providers
Bogers M., Hadar R. Bilbergb A.	Additive manufacturing for consumer-centric business models: Im- plications for supply chains in consumer goods manufacturing	2015	3D printing, Additive manufacturing, Busi- ness models, Digital fabrication, Glocalized production, Rapid manufacturing, Rapid prototyping, Supply chains
Conner BP., Manogharan GP., Martof AN., Rodom- sky LM., Rodomsky CM., Jor- dan DC., Limperos JW.	Making sense of 3-D printing: Creating a map of additive manu- facturing products and services	2014	Additive manufacturing; 3D printing; Product de- velopment; Complexity; Customization; Volume; Complexity factor; STL; Surface area; Features; Part geometry; Product mapping; Strategy
Huang SH., Liu P., Mokas- dar A., Hou L.	Additive manufacturing and its societal impact: a literature review	2012	Additive manufacturing. Environmental impact. Energy consumption. Supply chain. Health and wellbeing

Table 1: Selected Papers

Jiang R., Kleer R., Piller F.T.	Predicting the future of additive manufacturing: A Delphi study on eco- nomic and societal impli- cations of 3D printing in 2030	2017	Additive manufacturing, 3D printing, Delphi, Fore- casting, Scenario devel- opment
Liu P., Huang SH., Mokas- dar A., Zhou H., Hou L.	The impact of additive manufacturing in the air- craft spare parts supply chain: supply chain op- eration reference (scor) model based analysis	2014	additive manufacturing; supply chain operations reference (SCOR) model; aircraft spare parts; safety inventory
Pellengahr K., Schulte KT., Richard J., Berg M.	Pilot Study Procurement 4.0 The Digitalisation of Procurement	2016	Procurement 4.0, Digital- ization
Stucker B.	Additive Manufacturing Technologies: and Busi- ness Implications	2011	Additive manufacturing, Business Implications
Zhang Q., Vonderembse MA., Cao M.	Achieving flexible man- ufacturing competence: The role of advanced manufacturing technol- ogy and operations im- provement practices	2006	Flexible manufacturing systems, Advanced man- ufacturing technologies, Operations manage- ment, Competences
Weller, C.; Kleer, R.; Piller, T.	Economic implications of 3D printing: Market structure models in light of additive manufactur- ing revisited	2015	3D printing, Additive manufacturing, Market structure, Flexible man- ufacturing, Economic modelling
Campbell, T.; Williams, C.; Ivanova, O.; Garret, B.	Could 3D Printing Change the World? Technologies, Potential, and Implications of Additive Manufacturing	2011	3D-Printing, Platform, Additive manufacturing, Potentials

For the actual review the following impact areas, which were developed in the planning process, were evaluated, Influence of additive manufacturing on the purchasing portfolio, Influence of additive manufacturing on procurement organization, Impact of additive manufacturing on procurement processes, Influence of additive manufacturing on buyer-supplier relationships and the Influence of additive manufacturing on competence requirements in procurement.

3 Findings

For a first brief dissemination of this review a diagram was developed, which illustrates the main findings of the literature review on a numerical basis. The number of references of the impact in context with the changes of additive manufacturing on the different areas of procurement were counted and applied to create a descriptive diagram. This diagram shows in which of the above mentioned categories the selected authors of the papers see the broadest number of changes, or rather which ones are most important for the future procurement processes and in this connection also for future supply chains. It also shows which category is yet not mentioned that often but can also be an important impact in the future.

The diagram points out that the changing in procurement organization caused by additive manufacturing is most mentioned in the selected papers. The need of new qualification for especially procurement employees is not yet mentioned in many papers

In the following section the most striking findings about the influence of additive manufacturing on procurement are summarized based on the information from the papers listed in (Table 1: Selected papers). The results are divided into the mentioned categories purchasing portfolio, organization, processes, buyer supplier relationships and competence requirements.

3.1 Influence of additive manufacturing on the purchasing portfolio

With the installation of additive manufacturing in a company, new products and services are included in the purchasing portfolio that are 3D printers, raw materials, construction data, additional devices and equipment for post processing, or even additive manufacturing services, like design or production services. (Pellengahr at al., 2016)



Additive Manufacturing in Supply Chains - The Future of Purchasing Processes

Figure 1: Number of references in papers

As the technological development progresses, innovative products must be sourced to stay competitive by keeping up with higher customer demands and changes in demand and price structures. (Pellengahr et al., 2016) To face the increasingly shorter product lifecycle Weller et al. point out that additive manufacturing is the appropriate manufacturing technology to produce initial prototypes. (Weller et al., 2005) That means the purchasing portfolio of industrial companies does not longer contain the prior number of tools or molds for conventional prototype production. This procurement of tools or molds is replaced by the purchasing of various raw material for the 3D-printers.

The increase of the integration of added manufacturing over different supply chains also influences the market prices that are volatile due to the current trends. It must be taken into account whether to make or buy additive manufacturing and which specific parts would be needed along the supply chain of additive manufactured goods. (Baldinger, 2016)

Conner discusses the nature of additive manufactured parts and necessary inputs like geometric complexity, customization, volumes, and implications for the interaction of product development, procurement and manufacturing further. (Conner, 2014)

3.2 Influence of additive manufacturing on the procurement organization

According to Bogers, due to a high customer focus and individualization, supply chains become more distributed on a decentralized basis. Production activities shift from manufacturer closer to customer. (Bogers, 2015)

In contrast, Liu describes two approaches for the organizational anchoring of additive manufacturing capacity: The capacity can either be centralized to replace inventories or distributed close to the customers. (Liu, 2014)

Additive manufacturing enables direct production of objects and enables the consumers to design own products. As an emerging market online databases offer product designs for additive manufacturing purposes (Jiang, 2017). Customercentric focus on business models gives indications for operations along the supply chain. Flexibility and responsiveness need to be ensured by procurement organizations. (Bogers, 2015)

Also the integration of additive manufacturing platforms plays a significant role for the procurement organization. The option to use a network of companies as a supplier by using digital purchasing platforms can be one of the solutions for a not existing know-how of the production technology additive manufacturing. (Campbell et al., 2011)

Procurement along with other involved company functions faces the challenge that regulatory frameworks for additive manufacturing are still premature. That includes design and liability regulations, taking into consideration an outsourced additive manufacturing including digital platforms as service providers for data storage and print. (Gress and Kalafsky, 2015)

3.3 Influence of additive manufacturing on the procurement processes

Additive manufacturing influences supply chains and several enterprise processes. Additive manufacturing methods are used to reduce small batches. It has the potential to eliminate stocking activities for certain items and enable on-demand purchasing and production instead. Rapid prototyping becomes part of the strategic sourcing process. Supply logistics processes change as on-shoring is standard for additive manufacturing, different materials have to be procured and the procurement of spare part products decreases. (Stucker, 2011)

Simplified additive manufacturing increases efficiency and responsiveness in supply chains. (Huang, 2012) One advantage that takes effect in that case is that



Figure 2: Effects on the PSM process phases. (Adapted from: Van Weele, 2014)

additive manufacturing enables the production of functionally integrated products. That means downstream production processes of several products can be eliminated using additive manufacturing. For example the assembly steps of different product parts can be reduced regarding the fact that with additive manufacturing these different parts can be integrated in just one component. For the procurement that means that the "scale-scope dilemma" of higher procurement costs links to a higher complexity of products can be solved by an supplier using additive manufacturing as one of his production technologies. (Weller et al., 2005; Campbell et al., 2011.)

Also the inclusion of the costumer into the product design process will play a big role for the changes in procurement processes. Additive manufacturing enables them to become a co-designer of their individual product. For procurement the sourcing of different modular parts for the individualization of products will decrease. (Weller et al., 2005)

With the procurement process structure by van Weele at hand, it becomes visible that all the following process steps are affected by changes through additive manufacturing:

In the process area of sourcing, the processes specification and selecting are most affected by additive manufacturing. The selecting process as a strategic procurement process contains the selection of a suitable supplier. Integrating additive manufacturing into the supply chain changes the supplier portfolio for this process. Now one service provider for additive manufacturing can take over the role of several modular materials suppliers.

In the process area of supply the ordering process can be simplified. Through functional integration, the complexity of products, which are produced additively,

can be reduced. That means that the supply risk along the supply chain decreases, as these products can be produced in a decentral manner at the point of use and by the lower number of suppliers involved in one product engineering process.

3.4 Influence of additive manufacturing on buyer supplier relationships

As additive manufacturing opportunities increase, procurement is more and more obliged to look at the costs and risks for make-or-buy-decisions and the choice of suppliers. (Baldinger, 2016)

Different and new suppliers are needed triggered by changes in purchasing portfolios. On the other hand, established suppliers might become obsolete. Procurement and manufacturing will be challenged to identify which items currently purchased from a third party can be developed internally. Companies can make use of offered additive manufacturing services or, with enough demand, buy their own printers. As material requirements continue to evolve, procurement continuously faces new challenges in identifying and developing strategically relevant suppliers and network partners. (Connor, 2014)

In the case of the special field of spare part supply, additive manufacturing can replace the buyer supplier process. With additive manufacturing it is now possible to produce spare parts for products or machines on demand at the point of use. In this case the supplier has to share the design file of the part only and the production takes place at the factory site. (Campbell et al., 2011)

3.5 Influence of additive manufacturing on competence requirements in procurement

PSM can stress its strategic role by gaining additional knowledge and networking as an equal partner within and outside the company. Besides technical expertise, which ensures that the correct new types of products can be bought, the networking as well as innovation scouting and sourcing role of the modern purchasers is fostered by further implementation of additive manufacturing. (Pellengahr et al., 2016)

Internal capacity for additive manufacturing projects is built up. This increases the knowledge about the technology and its potential applications for the focal firm. For successful implementation, personnel along the whole supply chain and network of which procurement is one of the essential functions connecting the involved companies need to be qualified accordingly. Expertise is required for the procurement of the right products and technologies. Therefore, PSM needs to network with other departments, with suitable suppliers and stakeholders which calls for interdisciplinary know-how and interpersonal skills. In addition, entrepreneurial approaches and innovation abilities of the workforce are fostered. Buyers must have reliable product knowledge about their new digital procurement portfolio, e.g. about quality and pricing. As knowledge about current supply chain fundamentals and manufacturing techniques alters, procurement can benefit by remaining flexible and innovative in the face of 3D printing and other technologies. (Pellengahr et al., 2016)

Beyond explicitly PSM related new skills, technical infrastructure all over the networks needs to enhance regarding speed, security, availability of service and real-time efficiency. Curricula in practice as well academia should focus on increasing ICT skills in general. How these tasks can be embedded in the company depends on existing structures, process and know-how and will turn out clearer in the future. Procurement is one of the functions investigating intelligence solutions. Technological progress and related skills development can be pushed and controlled top down from management and strategic relevance. (Zhang, 2006)

4 Conclusion

One of the additional findings during the literature review is that the topic of additive manufacturing in connection with procurement processes or procurement organization was not often the main topic of the related papers. Most of them focused on technological or production issues. The changes in the area of procurement were only mentioned as a marginal note in the majority of the papers. Still, the systematic literature review has shown that a lot of advantages based on the production technology additive manufacturing can only be exploited by incorporating the specific features of additive manufacturing very early in the product lifecycle, right at the point of designing products in the supply chains and purchasing.

In addition to that, some of the main challenges procurement faces can be affected by progressing utilization of additive manufacturing in a positive way.

Certain risks can be minimized, e.g. by increasing own production of parts instead of relying on suppliers and by decentralization of production to locations close to the customer. Being a technology leader enhancing innovation potential can have positive influence on the company's reputation. The environmental impact decreases through the reduction of waste, transportation and inventory, and additive manufacturing can contribute to sustainability challenges and goals. It is



Figure 3: Challenges of procurement (Adapted from: Crawshaw, 2017)

important to demonstrate a partnership approach and develop a close network with the suppliers of new technology and data as well as with external and internal customers. These details are very important to exploit all advantages of additive manufacturing along the supply chain. Mentioned advantages have to be implemented already at the supplier stage to functionally integrate product parts at an early stage, so that the reduction of product complexity and a reduction of temporary storage can be spread as a monetary benefit alongside the complete supply chain. Centers of excellence can help establish the required know-how of new technologies and spread it throughout the organization as a support function. Procurement needs greater connections with stakeholders in the future to stay competitive, which is especially important for additive manufacturing net-works.

To point out the further research demands, based on the previous literature review, it can be summarized that past research concerning additive manufacturing and procurement is not broadly positioned in the research landscape. Most of the research focuses on additive manufacturing as a new technological production method and on the new products that can be realized with it. The changes in procurement as an early step in companies' processes have not been developed so far.

Additional to this finding, the competences of procurement employees as important decision-makers who affect the whole supply chain (e.g. if a part should be bought or produced additively) are not yet explored scientifically in a strategic way. Nevertheless, it is obvious that especially at this point in the supply chain process a decision concerning additive manufactured parts is indispensable. That indicates that especially interdisciplinary research teams containing the disciplines of engineering and educational science should work together for the development of new education programs containing additive manufacturing topics.

References

- Arnolds, H. (2013). Materialwirtschaft und Einkauf: Grundlagen Spezialthemen Übungen. 12., aktualisierte und überarbeitete Auflage 2013. Wiesbaden: Springer.
- Baldinger, M. (n.d.). "Supply chain management für additive manufacturing: Konzepte, Werkzeuge und Prozesse für die Zusammenarbeit mit Dienstleistern zur Reduktion der Risiken beim Einstieg in additve Manufacturing". In: ().
- Bogers, M., R. Hadar, and A. Bilberg (2016). "Additive manufacturing for consumer-centric business models: Implications for supply chains in consumer goods manufacturing". In: *Technological Forecasting and Social Change* 102, pp. 225–239.
- Campbell, T., C. Williams, O. Ivanova, and B. Garrett (2018). Could 3D Printing Change the World? Technologies, Potential, and Implications of Additive Manufacturing.
- Conner, B. P., G. P. Manogharan, A. N. Martof, L. M. Rodomsky, C. M. Rodomsky, D. C. Jordan, and J. W. Limperos (2014). "Making sense of 3-D printing: Creating a map of additive manufacturing products and services". In: Additive Manufacturing 1-4, pp. 64–76.
- Crawshaw, G. (2017). The top six challenges facing procurement.
- Drake, M. (2011). Global Supply Chain Management. Business Expert Press.
- European Environment Agency (2014). Annual European Union greenhouse gas inventory 1990-2012 and inventory re-port 2014.
- Flint, D. J., E. Larsson, B. Gammelgaard, and J. T. Mentzer (2005). "LOGISTICS INNOVATION: A CUSTOMER VALUE-ORIENTED SOCIAL PROCESS". In: *Journal of Business Logistics* 26.1, pp. 113– 147.
- Gress, D. R. and R. V. Kalafsky (2015). "Geographies of production in 3D: Theoretical and research implications stemming from additive manufacturing". In: *Geoforum* 60, pp. 43–52.
- Huang, S. H., P. Liu, A. Mokasdar, and L. Hou (2013). "Additive manufacturing and its societal impact: A literature review". In: *The International Journal of Advanced Manufacturing Technology* 67.5-8, pp. 1191–1203.
- Jiang, R., R. Kleer, and F. T. Piller (2017). "Predicting the future of additive manufacturing: A Delphi study on economic and societal implications of 3D printing for 2030". In: *Technological Forecasting and Social Change* 117, pp. 84–97.

Johannknecht, F. and R. B. Lippert (n.d.). "Nachhaltigkeit und Business-Cases". In: pp. 31–44. Kushner, M. (2018). Procurement in 2025: Five Megatrends & Their Implications. GEP, Clark.

Liu, P., S. H. Huang, A. Mokasdar, H. Zhou, and L. Hou (2014). "The impact of additive manufacturing in the aircraft spare parts supply chain: Supply chain operation reference (scor) model based analysis". In: Production Planning & Control 25.13-14, pp. 1169–1181. Mellor, S., L. Hao, and D. Zhang (2014). "Additive manufacturing: A framework for implementation". In: International Journal of Production Economics 149, pp. 194–201.

Pellengahr, K., A. Schulte, J. Richard, and M. Berg (2018). Pilot Study Procurement 4.0. The Digitalisation of Procurement. Fraunhofer In-stitute for Material Flow and Logistics.

Plowman, E. G. (1964). Lectures on elements of business logistics (Stanford transportation series). Stanford University.

PricewaterhouseCoopers (2014). Einkauf - Die neue Macht in den Unternehmen.

- Scharlach, A., C. Schuh, and M. Strohmer (2014). Procurement 2020+. 10 Mega-Trends, die den Einkauf verändern werden.
- Schuh, G., S. Aghassi, D. Bremer, and M. Graw (n.d.). "Einkaufsstrukturen". In: pp. 25-74.
- Stucker, B. (2011). Additive Manufacturing Technologies. Frontiers of Engineering 2011: Reports on Leading-Edge Engineering from the 2011 Symposium.
- Tranfield, D., D. Denyer, and P. Smart (2003). "Towards a Methodology for Developing Evidence-Informed Management Knowledge by Means of Systematic Review". In: *British Journal of Management* 14.3, pp. 207–222.
- Weele, A. J. van (2010). Purchasing & supply chain management: Analysis, strategy, planning and practice. 5. ed. Andover: Cengage Learning.
- Weller, C., R. Kleer, and F. T. Piller (2015). "Economic implications of 3D printing: Market structure models in light of additive manufacturing revisited". In: *International Journal of Production Economics* 164, pp. 43–56.
- Zeyn, H. (2017). Industrialisierung der Additiven Fertigung: Digitalisierte Prozesskette von der Entwicklung bis zum einsetzbaren Artikel Industrie 4.0. Beuth Innovation. Berlin: Beuth Verlag.
- Zhang, Q., M. A. Vonderembse, and M. Cao (2006). "Achieving flexible manufacturing competence". In: International Journal of Operations & Production Management 26.6, pp. 580–599.

Part II

Innovation and Technology Management
Efficiency Analysis for Digitalised Working Systems of Truck Drivers

Dominic Loske¹, Matthias Klumpp¹

1 - FOM University of Applied Sciences

The current research discussion postulates a digital transformation, characterized by new and in its extent unknown development and application potentials. At this stage scientists and practitioners know little about how these changes will transform specific work tasks. And the question is still even more opaque, whether there are real efficiency gains to be won by digitalization steps. Therefore, this paper develops an efficiency analysis to evaluate the effect of changing levels of digitalization within the working systems of professional truck drivers. Data Envelopment Analysis (DEA) is used in order to quantify truck driver performance in the logistics processes of a German food retailer. As inputs we use loading time and cost. Outputs are load factor of units, invoice charged to shops and value of damage. The findings indicate that a change in the level of digitalization entails a loss of efficiency in the first instance which can partly be compensated later. By using correlation analysis we prove a low statistical linear relationship of age and efficiency plus a strong statistical linear correlation of employer size and efficiency as well as period of employment and efficiency, always regarding the changing levels of digitalization in the working system of professional truck drivers. We derived the importance of employee retention programs for HR management along with a positive working environment provided for truck drivers to reduce fluctuation effects. Furthermore, we advise to design software for truck drivers as commonplace as possible and in the style of widespread smart phone software user interfaces.

Keywords: Digitalisation; Data Envelopment Analysis; Efficiency analysis; Truck driver

First recieved: 15.May.2018 Revised: 28.Jun.2018 Accepted: 03.Jul.2018

1 Introduction

According to the results of a cross-sectoral survey of more than 500 German companies by the digital association Bitkom, 84% of companies with logistics processes rely on digital technologies (Bitkom, 2017, online). Referring to this digital transformation, the current research discussion postulates almost unanimously a development boost that is characterized by a new and in its extent unknown development with non-foreseeable application potentials (Avent, 2014, p. 2). The trend is referred by different views as "Second Machine Age" (Brynjolfsson and McAfee, 2014, p. 7), "Third Industrial Revolution" (Rifkin, 2011, p. 2; Markillie, 2012, p. 1), "Distributed Capitalism" (Zuboff, 2010, p. 1) or "Industry 4.0" (Kagermann, et al., 2011, p. 1). The social and economic consequences can be described as disruptive and affect logistics as a science discipline, activity and industry (Avent, 2014, p. 2; Koch, et al., 2014, p. 4; Wee, et al., 2015, p. 7). Yet scientists and practitioners know little about what exactly is behind these buzz words and how the emerging changes of working systems will transform work tasks or working relationships - and especially the question of efficiency improvements is without widespread empirical answers yet.

Therefore, this paper aims to develop and apply an efficiency analysis to evaluate the effect of changing levels of digitalization within the working systems of professional truck drivers. We aspire to answer the following research questions: What are the components of an efficiency analysis aimed to evaluate the effect of changing levels of digitalization within the working systems of professional truck drivers? (RQ1) Which level of efficiency exists before the change in the level of digitization, how does it develop during and after the changeover? (RQ2) What effect has age, size of the employer and period of employment - hence work experience - on the level of efficiency? (RQ3).

After this introduction (section 1) we present a short literature review on impacts of digitalization on the human workforce in logistics and identify existing research gaps (section 2). In section 3, the efficiency analysis for digitalised working systems of truck drivers is elaborated by explaining basics of Data Envelopment Analysis, selecting applicable inputs and outputs for the efficiency analysis and explaining the setup and data collection. Based on the findings of the case study which are presented in section 4, section 5 of the paper discusses these results and derives implications for logistics management. The paper concludes with a summary and outlook in section 6.

2 Digitalization and human workforce in logistics

The logistics sector plays a key role in implementing Industry 4.0 solutions (BVL, 2017, p. 5), which raises the questions of consequences for employees and employers. From a sociological point of view, the human workforce can be theoretically structured in job clusters with mainly skilled and unskilled work tasks. The first group can be described as jobs that do not require profound vocational training, work that is carried out after short training on the job or a brief induction phase. On the other hand, qualified skilled workers play an important role: Typical oc cupational profiles include e.g. specialists in warehouse logistics, dispatchers, skilled workers in harbors and professional truck drivers (Eisenmann and Ittermann, 2017, p. 4). When investigating the human workforce, recent scientific literature also differentiates between blue-collar and white-collar workers.

The first describes a class of workers with primary physical working activities and is inspired by the blue working clothes that employees in the production sector use to wear. Their counterparts are white-collar workers who perform in professional, managerial, or administrative work whereby the expression is derived from the idea of a typical manager wearing white shirts. In this paper we examine qualified skilled workers in the working class of blue-collar workers.

2.1 Findings of literature review and resulting research gap

To identify an existing research gap, the development boost mentioned in section 1 was analyzed by a systematic bibliometric analysis based on Elsevier's abstract and citation database Scopus. Therefore, we concentrated on publications dealing with the impact of digitalization between 2008 and 2018 within the document types of textbooks, conference papers and journal articles. Furthermore, publications related to the sections of social sciences or business/management/accounting were investigated. We used TITLE-ABS-KEY ("impact" AND "digitalization") AND PUBYEAR > 2008 AND (LIMIT-TO (SUBJAREA, "SOCI") OR LIMIT-TO (SUBJAREA, "BUSI")) as search term. To illustrate the 154 resulting publications we applied VOSviewer, a software tool for constructing and visualizing bibliometric networks (van Eck and Waltman, 2010, p. 523). The bigger a circle, the more frequently the keyword appears in the publication set from Scopus. The keywords are assigned to a cluster, based on a computer algorithm. Each cluster has its own color related to the year of publication.



Efficiency Analysis for Digitalised Working Systems of Truck Drivers

Figure 1: Keywords map of investigated publications

The results show that the current research discussion concentrates on the investigation of digitalization in the industrial sector (keyword e.g. industry 4.0, manufacture) and moreover on the impact on human workforce and existing working systems (keyword e.g. human, working environment, education). Therefore, the blue-collar workers in the logistics sector are hardly addressed. The research approaches of the publications are either conceptual frameworks (Dombrowski, et al., 2014; Hirsch-Kreinsen, 2015; Huchler, 2016; Eisenmann and Ittermann, 2017; Klumpp, 2018; Hirsch-Kreinsen, et al., 2018) or empirical studies based on qualitative research methods (Stowasser and Jeske, 2015; Windelband and Dworschak, 2015; Bauer and Schlund, 2015; Ahrens, 2016; Will-Zocholl, 2016; Wróbel-Lachowska, et al., 2018). Consequently, we identified a research gap for investigating the impacts of digitalization in logistics based on a quantitative methodology.

2.2 Impacts of digitalization on human workforce

Scientific contributions investigating the changes in organizations are often related to change management. It addresses the preparation, realization and analysis for developments or far-reaching changes aiming to improve the current situation of a company and the efficiency of company's activities (Vahs and Weiand, 2010, p. 7). There are several theoretical-conceptual approaches to describe organizational changes on a macro-level, e.g. Kotter's 8-Step Process for Leading Change (1995) or micro-level, e.g. Lewin's 3-Step Model (1947, 1963). Lewin was one of the first researchers who investigated the procedure and effects of changes, especially the reactions of groups that have to face planned changes. According to him, the process of change can be divided in 3 phases called "unfreezing", "moving" and "refreezing". Furthermore, the model states individual levels of efficiency depending on the current change phase (Lewin, 1947, p. 34). The following figure illustrates this development.



Figure 2: Theory of level of efficiency in change processes, following Lewin, 1963, p. 2017 in Mohr, 1997, p. 73.

In order to explain where the change takes place exactly, the model of a working system created by Hardenacke et al. is used (Hardenacke, et al., 1985). The focus of the model is the work task which is derived by a superior mission of the organization. To accomplish the working task, the human resource impinges on the work object by using work equipment (Luczak, 1997, p. 13). An interaction between humans and the work equipment is necessary and requires a certain level of gualification for successful interaction. Besides the inputs into the working system (e.g. energy, material and information) and the output factors (e.g. work results), there are exogenous factors influencing the working system and subsystems like the physical and social environment (Schultetus, 2006, p. 15). The truck drivers examined in this paper have to face changes in digital innovations related to track and trace systems. The work equipment is a mobile device used to display work tasks for the truck driver e.g. to load a certain amount of containers for a grocery store and deliver it within a given time window. In order to fulfill the task, the truck driver has to scan all relevant 1D barcodes which are attached to the containers, load them into his truck and record differences between the data provided by the mobile device and the determined condition of transported goods.

3 Efficiency analysis for digitalised working systems of truck drivers

There is a wide variety of definitions for the term efficiency in scientific literature. Forsund and Hjalmarsson already described in 1974: "Efficiency is a word easy to use, but very difficult to give a precise operational meaning (Forsund and Hjalmarsson, 1974, p. 152)." Current research approaches postulate almost unanimously that there is no uniformly widespread definition for the concept of efficiency in business literature (Tzika, et al., 2017, p. 530). Efficiency in this research is understood as the relationship between the results achieved (outputs) and the resources used (inputs) for this purpose (DIN EN ISO 9000, 2000, p. 22). To measure efficiency quantitatively, performance measurement differs between key figures, parametric (e.g. Ordinary Least Squares, Stochastic Frontier Analysis) and non-parametric efficiency analysis (e.g. (Stochastic) Data Envelopment Analysis). The main difference is that non-parametric methods do not require a functional relationship between input and output factors whereas parametric methods do (Burger, 2008, p. 48).

3.1 Data Envelopment Analysis

The data envelopment analysis (DEA) method is a non-parametric optimization method of mathematical programming for measuring the relative efficiency of decision making units (DMUs) that have multiple inputs and outputs. A basic model was introduced by Charnes et al. (Charnes, et al., 1978, pp. 429–444) and is based on Pareto's definition of economic efficiency (Pareto, 1897), Koopmans activity analysis concept (Koopmans, 1951) together with the publications of Debreu and Farrell dealing with radial efficiency measurement (Farrell, 1957; Debreu, 1951). The increasing use in scientific research is related to the extensive scope of application where DMUs can be organizations or organizational units such as banks (Chen, et al., 2018; Huang, et al., 2018), universities (Lee and Worthington, 2016; Yang, et al., 2018), airlines (Omrani and Soltanzadeh, 2016; Chen, et al., 2017), employees (Khodamoradi, et al., 2016; Dugelova and Strenitzerova, 2015; Azadeh and Mousavi, 2014; Du, 2013; Zbranek, 2013; Shirouyehzad, et al., 2012) or nations (Malhotra and Malhotra, 2009). Further advantages beyond multiple inputs and outputs included are the facts that DEA is solely based on empirical data without the need of a priori existing production function and the fact that there is no need to weight factors, as this is done endogenously by the mathematics optimization model. The production process or throughput is seen as a black box. The basic mathematical notation is as follows (Wilken, 2007, p. 35):

$$\mathsf{eff}\,\mathsf{DMU}_0 = \frac{virtual\,ouputs\,of\,DMU_0}{virtual\,inputs\,of\,DMU_0} = \frac{\sum_{j=1}^s u_{0,j}y_{0,j}}{\sum_{i=1}^m v_{0,i}x_{0,i}} \tag{1}$$

eff	Abbreviation	for efficiency

- DMU₀ DMU with index 0 as exemplary decision unit
- s: Number of outputs to each DMU
- m: Number of inputs to each DMU
- $u_{0,j}$: The weight assigned to the output
- $v_{0.i}$: The weight assigned to the input
- $y_{0,j}$: Amount of the j output produced by DMU 0
- $x_{0.i}$: Amount of the i input consumed by DMU 0

The basic idea is to calculate an efficiency frontier that is used as a best practice input-output-combination for the underlying production scenario. DMUs that are on the efficiency frontier are considered as 100% efficient, whereas the relative inefficiency of a DMU can be determined by measuring the distance between individual DMU performance and the efficiency frontier. As the equation above requires n calculations for all n DMUs, the optimization problem is solved by a linear programming formulation (Charnes, et al., 1978, pp. 435–437). The optimization method can furthermore be based on constant returns on scale (CRS) in the CCR-model (Charnes, et al., 1978) or variable returns on scale (VRS) in the BCC-model (Banker, et al., 1984) and in each case with an input or output orientation.

3.2 Selection of applicable inputs and outputs for DEA

In this paper we analyze the efficiency of truck drivers working in the sector of distribution logistics for a large German food retailing company. Their transportation unit is responsible for delivering food and non-food items from the central logistics center to all grocery shops of the relevant delivery area complete and on time as well as for returning recyclable materials plus empty load carriers from the grocery shops back to the logistics center. To get a clear idea of the underlying scenario, the description is divided in financial flows and physical material flows related to the retailers supply chain.

From a financial point of view, the sales unit of the retailing company is charged by the logistics unit with individual service costs based on cost rates determined at the beginning of a financial year. It can be seen as a revenue stream for the logistics unit. To deliver all orders, the food retailing company uses solely carriers of different sizes whereby the tours are planned by the dispatchers of the retailing company. The carriers receive a payment for the physical distribution which is based on transported units. On the other hand the carriers have to pay for goods damaged in the delivery process.

Focusing the daily business of professional truck drivers and the physical material flows, the work process can be divided in the following steps: (1) Register at the responsible dispatcher in the logistics center, (2) Receive data for delivery tour through mobile device, (3) Load the truck by scanning barcodes on load carriers through mobile device, (4) Receive freight documents from dispatcher, (5) Drive to n grocery shops, (6) Unload cargo at n grocery shops, (7) Return recyclable



Figure 3: Illustration of work tasks of professional truck driver

material and empty load carriers back to logistics center. Hence, the question arises where truck drivers get in touch with a change in the level of digitalization and where they feel a change in their daily business. The following Figure 3 illustrates the work tasks of a professional truck driver in the given scenario.

Therefore, we choose to investigate the loading process (2) and (3). Analyzing the whole chain is unsuitable due to the fact that too many confounding factors e.g. waiting time at logistics center or grocery shops, traffic jams and vehicle breakdowns can distort the factor time as an important element in an efficiency analysis. The sub-processes (1), (4) and (5) are unsuitable because a change in the level of digitalization does not affect them. Furthermore, (6) and (7) are suitable for the scenario but require less interaction with the mobile device than (2) and (3).

For the DEA we use the loading time needed to load the truck and occurring transportation costs as inputs. The second factor is quantified by the payment to

carriers for transported units. Choosing the whole payment is unsuitable as this does not take into account that only a sub-process, the process of loading the truck, is analyzed. Therefore, we use a percentage of the whole payment based on the ratio of needed loading time and total time of a tour, whereby latter is the planned total time of a tour based on the calculation of the route planning software to avoid confounding factors like traffic jams or waiting time.

$$Relevant transportation costs = \frac{Actual loading time}{Planned total time of tour} \times$$
(2)
Total payment

As outputs we use 3 factors. The first one is the invoice of the transportation unit of the retailing company to the sales unit for all rendered logistics services where we do not use the whole amount but the equal percentage as for the input transportation costs. It can be seen as a revenue for the logistics sector which is often used as an output factor for performance evaluation of logistics firms based on a DEA model (Hong and Xu, 2015; Momeni, et al., 2015; Tang, et al., 2015; Ye, 2015). The second output factor is the monetary value of all goods that are damaged or destroyed while loading the truck. In this case we do not use the sales prices of goods that can be found in the grocery shops but the purchase price of the food retailing company which presents the product value at the time of damage. As high output values increase the efficiency of DMUs we use the reciprocal calculated by the following formula.

> Damage value of DMU = Max.observed damage damage caused by DMU(3)

The third output factor is related to the transport units that are loaded into the truck. We could simply use the amount of units and argue that when loading a roller container (square measure 72cm x 81,5cm and max. load 500kg) or a bigger pallet (square measure: 120cm x 80cm and max. load 1500kg) one barcode scan is required and therefore, the effort is equal. But with regard to the driver loading his truck, the different weights of the transport units cause a fundamental difference in the effort of handling. We take this unequal effort into account by rating loading equipment based on their geometrical square measure. Therefore,



3 Efficiency analysis for digitalised working systems of truck drivers



Figure 4: Components of the efficiency analysis

1.0 transport unit (TU) shall correspond to the dimensions of a roller container and all others are converted based on this standard, e.g. 1.64 TU for a pallet [(120cmx80cm)/(72cmx81,5cm)]. With the justification of every single input and output factors we can answer RQ1 "What are the components of an efficiency analysis aimed to evaluate the effect of changing levels of digitalization within the working systems of professional truck drivers? (RQ1)" and summarize the results of the elaboration in the following Figure 4.

3.3 Setup and data collection

To analyze the efficiency levels for digitalised working systems of professional truck drivers, the daily loading activities for 30 of them were examined during 4.5 weeks by evaluating 1,344 delivery tours. The digital change in the underlying scenario was the replacement of mobile devices based on windows mobile software with complex operation using a keyboard, by new mobile devices based on Android software with a user friendly full touch display. Another major modification was the integration of more and new processes into the existing workflow that is handled by the mobile device and have not been included before, e.g. special application for high value products like cigarettes, elimination and digitalization of accompanying documents along with a clear menu navigation.

To be able to prove that the developments in efficiency levels are not based on external factors, the results of a n = 15 group with n = 15 of professional truck drivers with digitalization (DG) are contrasted with a n = 15 control group (CG, n = 15) of professional truck drivers without any changes in the level of digitalization. The change was done at the beginning of calendar week 16 (t_0). To get an own empirical curve progression based on Lewin's 3-Step Model we analyzed the 3 phases "unfreezing" (t - 4, t - 3, t - 2, t - 1), "moving" (t 0) and "refreezing" (t + 1, t + 2, t + 3, t + 4). The sociodemographic structure of the digitalized group and the control group has been chosen identically in order to

guarantee a maximum level of construct validity. The following figure shows the sociodemographic structure.

Furthermore, we analyzed the period of employment of relevant professional truck drivers. As the retailing company only uses carriers, information about the period of employment of truck drivers is not relevant, when investigating the impact of digitalization. Therefore, we enquired the truck drivers about their time of occupation at the transport unit of the food retailing company. Figure 5 shows the results for the digitalization group and the control group.



Figure 5: Sociodemographic structure of digitalized and control group

4 Findings of the case study

In the case study we use an input oriented model as the company aims to minimize loading time of professional truck drivers and cost for transportation. The

Period	Digit	alized gro	oup D	Cor	ntrol Grou	ір С	Diff. of mean values
	Min	Mean	Мах	Min	Mean	Мах	
t-4	0.80	0.93	1.00	0.64	0.94	1.00	+0.01
t-3	0.78	0.93	1.00	0.68	0.94	1.00	+0.01
t-2	0.79	0.94	1.00	0.86	0.94	1.00	±0.00
t-1	0.72	0.92	1.00	0.85	0.93	1.00	+0.01
t0	0.63	0.80	1.00	0.61	0.86	1.00	+0.06
t+1	0.74	0.89	1.00	0.76	0.91	1.00	+0.02
t+2	0.77	0.93	1.00	0.76	0.92	1.00	-0.01
t+3	0.67	0.89	1.00	0.66	0.91	1.00	+0.02
t + 4	0.78	0.92	1.00	0.69	0.88	1.00	-0.04

Table 1: Statistical measures for results of CCR-model

investigation was done during 4.5 weeks what we divided into 9 periods with 3 days (MO-WE and TH-SA) each. We chose t - 4, t - 3, t - 2, t - 1 as periods before the change, t 0 as the time frame of the change and t + 1, t + 2, t + 3, t + 4 as periods after the change. All truck drivers of the digitalized group received their new mobile devices at the beginning of t 0 and before starting their daily work. The following table shows relevant statistical measures for the results of the DEA CCR model calculation.

We chose a CCR-model with constant returns on scale which is most commonly used as a basic model in scientific literature. This can be explained by the underlying assumption regarding the optimal operational size (Cantner, et al., 2007, p. 11). A DMU under evaluation that operates with CRS performs with a most productive scale size (MPSS). Therefore CRS assume that all DMUs exhibit an optimal operational size. The concept of MPSS has been defined by Banker, et al. where DMUs do not naturally operate at MPSS but would achieve it by scaling its operations up and down via VRS (Banker, 1984, p. 39). Hence BCC-model is only applied when it is specifically required to check for increasing or decreasing returns to scale. As we assume that the MPSS, in the case of professional truck drivers performance defined as the individual performance capability, is equal for each DMU the CCR-model is applied. The following Figure 6 illustrates the curve progression of DEA efficiency measures for t - 4 to t + 4.

The graphs in Figure 6 show that the efficiency of the digitalized group decreases significantly in t 0 where the change in digital technology took place. The data



Figure 6: Curve progression of DEA efficiency measures for t - 4 to t + 4

cannot be interpreted solely by the total efficiency values, but by the difference between the mean values of the 2 groups. Before the digitalization process the curves were closely related with a deviation of max. 1% but in *t* 0 the differences in efficiency values are 6% in the BCC-model. This is enough evidence to answer RQ2 "Which level of efficiency exists before the change in the level of digitization, how does it develop during and after the changeover?": Before the change in the level of digitalization we monitored a significant loss of efficiency that increased to the former level in *t* + 2. After the changing level of digitalization the level of efficiency averages 91% and is therefore higher than during the change but lower than before. We assume that the effect of an increasing level of efficiency through digitalization will occur in long-term investigations which outline further research needs.

To be able to answer RQ3 "What effect has age, size of the employer and period of employment on the level of efficiency?" we use a correlation analysis to investigate

the statistical relationship between the level of efficiency and sociodemographic attributes. We chose the efficiency values of t 0 (CCR-model) for the group with digitalization as it is the period where the change in the level of digitalization takes place. A high efficiency in this period proves that the driver is affected by the change but still efficient in the work he does.

The correlations coefficient r for the relationship of "age in years" (x) and "level efficiency" (y) is r = 0.1496. Therefore, there is a low statistical linear relationship between these 2 factors. The following figure shows the correlation graphs for the relationship of "total trucks of company" for the carrier's size (x) and "level of efficiency" (y) as well as the relationship of "period of employment" at the food retailing company and not as a truck driver (x) and "level of efficiency" (y).

Due to the correlation coefficients there is a strong positive linear statistical relationship between a company's size and the level of efficiency during a digital change. Furthermore, we can determine a strong positive linear statistical relationship between the period of employment in a certain occupation and the level of efficiency during a digital change. These findings answer RQ3.



Figure 7: Correlation graphs for company size and period of employment

5 Discussion and implications for management

The curve progression of the analysis indicates a loss of efficiency during the first days of the change which has been explained in the previous chapter. In addition, the investigation shows that the initial level is not constantly reached in the periods after. Due to the limited periods included into the analysis we cannot answer the question if the increasing level of digitalization can increase the level of efficiency for this case.

The results of the correlation analysis indicate that age is less relevant when discussing the changing levels of digitalization for professional truck drivers. One reason for this is that mobile devices like mobile phones became a part of our every day's life, independent of our age or social status. More than 10 years ago the use of internet and mobile devices was interpreted as a border that had to be crossed and where users had to intentionally go to. This can be understood by reenacting the motto of the internet conference re:publica in 2007 called "living in the internet". Digital devices and the internet are no parallel worlds to go into any more, it rather became a medium that permanently and unconsciously expands the reality we are living in. All of the professional truck drivers investigated in this case possess a private smart phone and are therefore used to the full touch display introduced with the mobile devices. A recommendation for IT business units and software engineers is to design virtual dialogues on mobile devices as commonplace as possible. For HR management our analysis gives indications that the age of employers will less be a problem in regards to changing levels of digitalization. The results of the correlation analysis investigating the size of the company can be traced back to the fact that bigger carriers use supervisors to coordinate truck drivers and their service at the logistics unit of the food retailing company. A direct contact for questions has a positive impact on the efficiency level when changing the levels of digitalization. This can be a hint for logistics process management to invest in experts helping all truck drivers by placing them on the area of outgoing goods in the logistics center to weaken the negative effects on efficiency levels. The last correlation analysis with the period of employment at the food retailing company indicated that being proficient in the company's processes can be essential for a successful analog-to-digital conversion. An employment of more than 5 years and the knowledge gained in this time seems to be a valuable asset. This means that HR management of carriers should concentrate on developing and promoting employee retention programs to lower fluctuation effects of professional truck drivers. For the contracting authority, in our case the

food retailing company, it also means to provide a positive working environment with short waiting times, clear objectives and permanent support.

6 Conclusion and outlook

The aim of this paper was to develop and apply an efficiency analysis to evaluate the effect of changing levels of digitalization within the working systems of professional truck drivers. RQ1 "What are the components of an efficiency analysis aimed to evaluate the effect of changing levels of digitalization within the working systems of professional truck drivers?" was answered by the development of a DEA analysis using loading time and percentage of transportation costs as input factors plus load factor of TU, percentage of charge to sales unit and value of damage (reciprocal) as output factors. We applied the analysis in order to evaluate the truck driver's (DMU) performance in the logistics sector of a food retailing company. On the basis of this case study we answered RQ2 "Which level of efficiency exists before the change in the level of digitization, how does it develop during and after the changeover?" by deriving the curve progression of relative efficiency during 4.5 divided in 9 periods with 3 days each. Before the change there was an efficiency level of an average of 95% (BCC-model) and during the change in the level of digitalization we monitored a significant loss of efficiency that increased to the former level in t + 2. Finally RQ3 "What effect has age, size of the employer and period of employment on the level of efficiency?" was answered by correlation analysis comparing the level of efficiency during the change in the level of digitalization (t 0) with the age of the truck drivers, the size of their employers and their period of employment at the German food retailer. We showed a low statistical linear relationship of age and efficiency plus a strong statistical linear relationship of employer size and efficiency as well as period of employment and efficiency, always regarding the changing levels of digitalization in the working system of professional truck drivers.

The contributions and implications presented in this article were taken from a quantitative (DEA analysis) and explanative (related to Lewin's 3-Stage Model of Change) approach, which is why the results have limitations and an enlargement of the scope would enable new insights in the future: We did not include further qualitative input and output factors like the satisfaction or motivation of truck drivers. Moreover, we examined a change in the level of digitalization without any training methods applied to prepare the professional truck drivers.

Therefore, research is necessary and interesting along the lines of the following possible research questions: (i) How can qualitative soft factors like employee's satisfaction and motivation be integrated in an efficiency analysis for professional truck drivers? Current research approaches regarding human resource management include soft factors into DEA for efficiency measurement of employees (Shirouyehzad, et al., 2012; Zbranek, 2013; Du, 2013; Azadeh and Mousavi, 2014; Dugelova and Strenitzerova, 2015; Khodamoradi, et al., 2016). Furthermore, (ii) How and to which extend can training methods like seminar based training or self-studies with digital mock ups reduce negative effects in the level of efficiency when changing the level of digitalization for professional truck drivers? These questions would constitute interesting research pathways following the first quantitative results presented here.

Altogether, the presented analysis and management implications have shown the importance and value of a quantitative-empirical approach regarding the digital transition in specified logistics processes and human work contexts. This will be a long-term research and business practice question for the future of logistics.

References

- Ahrens, D. (2016). "Neue Anforderungen im Zuge der Automatisierung von Produktionsprozessen: Expertenwissen und operative Zuverlässigkeit". In: Arbeits- und Industriesoziologische Studien 9.1, pp. 43–56.
- Avent, R. (2014). "The third great wave". In: The Economist Special Report, pp. 1-14.
- Azadeh, A. and A. Mousavi (2014). "The Impact of Job Security, Satisfaction and Stress on Performance Assessment and Optimization of Generation Companies". In: *Journal of Loss Prevention in the Process Industries* 32, pp. 343–348.
- Banker, R. D. (1984). "Estimating most productive scale size using data envelopment analysis". In: European Journal of Operational Research 17.1, pp. 35–44.
- Banker, R. D., A. Charnes, and W. W. Cooper (1984). "Some Models for Estimating Technical and Scale Inefficiencies in Data Envelopment Analysis". In: *Management Science* 30.9, pp. 1078– 1092.
- Bauer, W. and S. Schlund (2015). "Wandel der Arbeit in indirekten Bereichen: Planung und Engeneering". In: Digitalisierung industrieller Arbeit. Ed. by H. Hirsch-Kreinsen, P. Ittermann, and J. Niehaus. Baden-Baden: Nomos Verlagsgesellschaft, pp. 54–69.

Bitkom (2017). Digitalisierung der Logistik.

- Brynjolfsson, E. and A. McAfee (2014). The second machine age: Work, progress, and prosperity in a time of brilliant technologies. New York: W.W. Norton & Company.
- Burger, A. (2008). "Produktivität und Effizienz in Banken: Terminologie, Methoden und Status quo". In: Frankfurt School – Working Paper Series No. 92.

- BVL (2017). "Logistik als Wissenschaft: Zentrale Forschungsfragen in Zeiten der vierten industriellen Revolution". In: Positionspapier Bundesvereinigung Logistik (BVL) e. V.
- Cantner, U., H. Hanusch, and J. Krüger (2007). Produktivitäts- und Effizienzanalyse: Der nichtparametrische Ansatz. Berlin: Springer.
- Charnes, A., W. W. Cooper, and E. Rhodes (1978). "Measuring the efficiency of decision making units". In: European Journal of Operational Research 2.6, pp. 429–444.
- Chen, Z., R. Matousek, and P. Wanke (2018). "Chinese bank efficiency during the global financial crisis: A combined approach using satisficing DEA and Support Vector Machines?" In: *The North American Journal of Economics and Finance* 43, pp. 71–86.
- Chen, Z., P. Wanke, J. J. M. Antunes, and N. Zhang (2017). "Chinese airline efficiency under CO2 emissions and flight delays: A stochastic network DEA model". In: *Energy Economics* 68, pp. 89– 108.
- Debreu, G. (1951). "The Coefficient of Resource Utilization". In: Econometrica 19.3, pp. 273–292.
- DIN EN ISO 9000 (2000). Qualitätsmanagementsysteme: Grundlagen und Begriffe. Berlin: Beuth.
- Dombrowski, U., C. Riechel, and M. Evers (2014). "Industrie 4.0: Die Rolle des Menschen in der vierten industriellen Revolution". In: *Industrie 4.0*. Ed. by W. Kersten. Schriftenreihe der Hochschulgruppe für Arbeits- und Betriebsorganisation e.V. (HAB). Berlin: Gito, pp. 129–153.
- Du, J.-Y. (2013). "Staff performance appraisal based on data envelopment analysis (DEA)". In: Journal of Chemical and Pharmaceutical Research 5.11, pp. 102–105.
- Dugelova, M. and M. Strenitzerova (2015). "The Using of Data Envelopment Analysis in Human Resource Controlling". In: *Procedia Economics and Finance* 26, pp. 468–475.
- Eisenmann, M. and P. Ittermann (2017). "Hybride Dienstleistungen und Wandel der Arbeit: Herausforderungen und Perspektiven in der Logistik". In: Soziologisches Arbeitspapier 50, pp. 1– 46.
- Farrell, M. J. (1957). "The measurement of productive efficiency". In: Journal of Royal Statistical Society 120.3, pp. 253–290.
- Forsund, F. R. and L. Hjalmarsson (1974). "On the Measurement of Productive Efficiency". In: The Swedish Journal of Economics 76.2, pp. 141–154.
- Hardenacke, H., W. Peetz, and G. Wichardt (1985). Arbeitswissenschaft. Studienbücher der Wirtschaft. München: Hanser.
- Hirsch-Kreinsen, H. (2015). "Gestaltungsperspektiven von Produktionsarbeit bei Industrie 4.0". In: Arbeit in der digitalisierten Welt. Ed. by C. Schlick. Frankfurt am Main: Campus.
- Hirsch-Kreinsen, H., P. Ittermann, and J. Niehaus, eds. (2015). Digitalisierung industrieller Arbeit: Die Version Industrie 4.0 und ihre sozialen Herausforderungen. 1. Auflage. Baden-Baden: Nomos Verlagsgesellschaft.
- Hirsch-Kreinsen, H., M. ten Hompel, P. Ittermann, J. Dregger, J. Niehaus, T. Kirks, and B. Mättig (2018). ""Social Manufacturing and Logistics": Arbeit in der digitalisierten Produktion". In: Zukunft der Arbeit. Ed. by S. Wischmann and E. Hartmann. Berlin: Springer Vieweg.
- Hong, H. and D. Xu (2015). "Performance evaluation of logistics firms based on DEA model". In: International Journal of Services, Technology and Management 21.4-6, pp. 199–213.
- Huang, T.-H., K.-C. Chen, and C.-I. Lin (2018). "An extension from network DEA to copula-based network SFA: Evidence from the U.S. commercial banks in 2009". In: *The Quarterly Review of Economics and Finance* 67, pp. 51–62.
- Huchler, N. (2016). "Die "Rolle des Menschen" in der Industrie 4.0: Technikzentrierter vs. humanzentrierter Ansatz". In: Arbeits- und Industriesoziologische Studien 9.1, pp. 57–79.

- Kagermann, H., W.-D. Lukas, and W. Wahlster (2011). "Industrie 4.0: Mit dem Internet der Dinge auf dem Weg zur 4. industriellen Revolution". In: VDI Nachrichten 13.
- Kersten, W., ed. (2014). Industrie 4.0: Wie intelligente Vernetzung und kognitive Systeme unsere Arbeit verändern. Schriftenreihe der Hochschulgruppe für Arbeits- und Betriebsorganisation e.V. (HAB). Berlin: Gito.
- Khodamoradi, M., M.-H. Behzadi, and M. Rostamy-MaleKhalifeh (2016). "Evaluating of staff's performance with job satisfaction and Organizational Commitment approach by DEA: Case study: Khuzestan Oxin Steel Company". In: *International Journal of Data Envelopment Analysis* 4.2, pp. 945–949.
- Klumpp, M. (2018). "Automation and artificial intelligence in business logistics systems: Human reactions and collaboration requirements". In: International Journal of Logistics Research and Applications 21.3, pp. 224–242.
- Koch, V., R. Geissbauer, S. Kuge, and S. Schrauf (2014). Chancen und Herausforderungen der vierten industriellen Revolution. Frankfurt and München: Pwc und Strategy&.
- Koopmans, T. C., ed. (1951a). Activity Analysis of Production and Allocation: Proceeding of a Conference. London: Wiley.
- Koopmans, T. C. (1951b). "An analysis of production as an efficient combination of activities". In: Activity Analysis of Production and Allocation. Ed. by T. C. Koopmans. London: Wiley, pp. 33–97.
- Lee, B. L. and A. C. Worthington (2016). "A network DEA quantity and quality-orientated production model: An application to Australian university research services". In: *Omega* 60, pp. 26–33.
- Lewin, K. (1947). "Frontiers in Group Dynamics". In: Human Relations 1.1, pp. 5–41.
- Luczak, H. (1997). "Kerndefinition und Systematiken der Arbeitswissenschaft". In: Handbuch Arbeitswissenschaft. Ed. by H. Luczak, W. Volpert, and T. Müller. Stuttgart: Schäffer-Poeschel, pp. 11–19.
- Luczak, H., W. Volpert, and T. Müller, eds. (1997). Handbuch Arbeitswissenschaft. Stuttgart: Schäffer-Poeschel.
- Malhotra, R. and D. K. Malhotra (2009). "Evaluating the efficiency of European Union integration". In: International Journal of Commerce and Management 19.3, pp. 233–252.
- Markillie, P. (2012). "A third industrial revolution". In: The Economist Special Report, pp. 1–10.

Mohr, N. (1997). Kommunikation und organisatorischer Wandel. Heidelberg: Gabler Verlag.

Momeni, E., M. Azadi, and R. F. Saen (2015). "Measuring the efficiency of third party reverse logistics provider in supply chain by multi objective additive network DEA model". In: *International Journal of Shipping and Transport Logistics* 7.1, pp. 21–41.

Omrani, H. and E. Soltanzadeh (2016). "Dynamic DEA models with network structure: An application for Iranian airlines". In: *Journal of Air Transport Management* 57, pp. 52–61.

- Pareto, V. (1897). Cours d'économie politique. Paris.
- Rifkin, J. (2011). The third industrial revolution: How lateral power is transforming energy, the economy, and the world. New York: Palgrave Macmillan.
- Schlick, C., ed. (2015). Arbeit in der digitalisierten Welt: Beiträge der Fachtagung des BMBF 2015. 1. Aufl. Frankfurt am Main: Campus.
- Schultetus, W. (2006). Arbeitswissenschaft von der Theorie zur Praxis: Arbeitswissenschaftliche Erkenntnisse und ihr wirtschaftlicher Nutzen. Köln: Wirtschaftsverl. Bachem.
- Shirouyehzad, H., F. Hosseinzadeh Lotfi, M. B. Aryanezhad, and R. Dabestani (2012). "A Data Envelopment Analysis Approach for Measuring Efficiency of Employees: A case study". In: *The South African Journal of Industrial Engineering* 23.1, pp. 191–201.

- Stowasser, S. and T. Jeske (2015). "Arbeitswelt der Zukunft". In: Arbeit in der digitalisierten Welt. Ed. by C. Schlick. Frankfurt am Main: Campus.
- Tang, L., X. Huang, Y. Peng, and Z. Xiao (2015). "Analysis and evaluation of relative efficiency of warehousing and distribution operations based on mixed DEA model". In: *Chemical Engineering Transactions* 46, pp. 583–588.
- Tzika, A., S. Zelewski, and M. Klumpp (2017). "Effizienzentwicklung von Hochschulen in Deutschland". In: Hochschuleffizienz. Ed. by S. Zelewski, M. Klumpp, and N. Akca. Berlin: Logos Berlin, pp. 517–600.
- Vahs, D. and A. Weiand (2010). Workbook Change-Management: Methoden und Techniken. Stuttgart: Schäffer-Poeschel.
- van Eck, N. J. and L. Waltman (2010). "Software survey: VOSviewer, a computer program for bibliometric mapping". In: *Scientometrics* 84.2, pp. 523–538.
- Wee, D., R. Kelly, J. Cattel, and M. Breunig (2015). Industry 4.0: How to navigate digitization of the manufacturing sector. München et al.: McKinsey & Company.
- Wilken, R. (2007). Dynamisches Benchmarking: Ein Verfahren auf Basis der Data Envelopment Analysis. Wiesbaden: Deutscher Universitäts-Verlag.
- Will-Zocholl, M. (2016). "Die Verlockung des Virtuellen: Reorganisation von Arbeit unter Bedingungen der Informatisierung, Digitalisierung und Virtualisierung". In: Arbeits- und Industriesoziologische Studien 9.1, pp. 25–42.
- Windelband, L. and B. Dworschak (2015). "Arbeit und Kompetenzen in der Industrie 4.0: Anwendungsszenarien Instandhaltung und Leichtbaurobotik". In: *Digitalisierung industrieller Arbeit*. Ed. by H. Hirsch-Kreinsen, P. Ittermann, and J. Niehaus. Baden-Baden: Nomos Verlagsgesellschaft, pp. 71–86.
- Wischmann, S. and E. Hartmann, eds. (2018). Zukunft der Arbeit: Eine praxisnahe Betrachtung. Berlin: Springer Vieweg.
- Wróbel-Lachowska, M., Z. Wisniewski, and A. Polak-Sopińska (2018). "The role of the lifelong learning in Logistics 4.0". In: International Conference on Applied Human Factors and Ergonomics 1, pp. 402–409.
- Yang, G.-l., H. Fukuyama, and Y.-y. Song (2018). "Measuring the inefficiency of Chinese research universities based on a two-stage network DEA model". In: *Journal of Informetrics* 12.1, pp. 10– 30.
- Ye, Y. (2015). "Performance evaluation of logistics enterprise based on DEA model". In: Proceedings of the International Conference on Management, Information and Educational Engineering, MIEE 2014 1.
- Zbranek, P. (2013). "Data envelopment analysis as a tool for evaluation of employees' performance". In: Acta Oeconomica et Informatica 16.1, pp. 12–21.
- Zelewski, S., M. Klumpp, and N. Akca, eds. (2017). Hochschuleffizienz: Konzeptionelle Herausforderungen und Lösungsansätze aus Sicht der betriebswirtschaftlichen Forschung. Berlin: Logos Berlin.
- Zuboff, S. (2010). Creating value in the age of distributed capitalism. McKinsey Quarterly.

Alternative Development Paths for Supply Chains in 2030

Denis Daus¹, Ana Cristina Barros², Dimitra Kalaitzi³, Victoria Muerza⁴, Irene Marchiori $^{\rm 5}$

- 1 Fraunhofer Institute for Material Flow and Logistics (IML)
- 2 INESC TEC
- 3 Aston University
- 4 Zaragoza Logistics Center (ZLC)
- 5 Institute of Industrial Technologies and Automation National Research Council (ITIA-CNR)

Many different megatrends such as digitalization, are posing specific challenges and opportunities for supply chains creating the urgent need to adapt and rethink the way they are organized. This paper aims to define development paths (DPs), which constitute projections into the future. The DPs are based on the analysis of megatrends, which might have an impact on the design of supply chains until the year 2030. The results are 51 possible DPs, which are derived from the identified megatrends and clustered under 22 descriptors, which are in turn assigned to the PESTLE (Political, Economic, Social, Technological, Legal, and Environmental) dimensions. These DPs consider the underlying criteria reasonability and conceivability. They describe how the future might look like in 2030 and can be used to address developments, challenges and opportunities that may arise in supply chains. Thus, this paper creates the starting basis for further research that deals with the creation of holistic industrial scenarios affecting future supply chains.

Keywords: Supply Chain; Megatrends; Development Paths; Scenarios

First recieved: 28.May.2018 Revised: 22.Jun.2018 Accepted: 11.Jul.2018

1 Introduction

The biggest trader and exporter is the political and economic European Union (EU) of 28 member states. It is expected that in the next 10-15 years 90% of the global economic growth will come from outside the EU (ALICE, 2014). Therefore, on one hand, the EU companies have to be able to adapt and gain a competitive advantage by accessing these new markets. On the other hand, companies must configure accordingly their supply chains to respond efficiently to incoming challenges, such as customization and scarcity of resources. Digitalization accompanied by acceleration of technological developments affect different industries, such as process, discrete manufacturing and logistics industry that are significant sectors for the EU's economy.

The process industries are industries where the production processes are either continuous, or the batches of materials are indistinguishable (Institute of Industrial and Systems Engineers, 2018). The product is created by using a formula to refine the raw materials leaving no way to break down the final product into its basic components (Cole and O'Donnell, 2017).

Different divisions of the process industry sector such as chemicals, minerals nonferrous metals, are united under SPIRE - Sustainable Process Industry through Resource and Energy Efficiency (Tello and Weerdmeester, 2012). These divisions comprise more than 450,000 companies, employ 6.8 million people and have a yearly sales volume of more than EUR 1.6 billion. The common feature between these types of companies is the high dependence on resources (e.g. energy). The process industry is important for Europe since it covers 20% of the overall European industry in terms of employment and sales volume (Tello and Weerdmeester, 2012).

The discrete manufacturing sector has 29.7 million employees in over 2 million companies generating a turnover of EUR6.98 billion and a value added nearly to EUR 1.63 billion; the sector represents 17% of the EU GDP (EFFRA, 2013). Companies in this sector produce distinct items that can be easily counted and the products are measurable in distinct units.

The top five branches for value added and employment in the discrete manufacturing sector are machinery and equipment, food products, metal products, motor vehicles and electric equipment (EFFRA, 2013).

As reported by Eurostat and the alliance for European Logistics, the logistics sector constitutes the single biggest industry in Europe: it creates 7 million employees

and the revenue per year is more than EUR 900 billion, representing around 7% of total European GDP (Savills Investment Management, 2016).

In order to address these challenges and benefit from the opportunities provided by megatrends, companies use mechanisms such as collaboration and reconfiguration of supply chains. Berger (2015) identified the top ten trends impacting the supply chain: (i) rising customer requirements regarding reliability, (ii) rising customer requirements regarding flexibility, (iii) increasing demand volatility, (iv) rising customer requirements regarding lead times. (v) rising customer requirements regarding cost, (vi) increasing uncertainty, (vii) increasing complexity, (viii) increasing globalization, (ix) increasing heterogeneous customer requirements, and (x) increasing speed of change. Cost and reliability are revealed as the most important targets in supply chains (Berger, 2015). In the future, lead time and flexibility will also have a high importance. More recently, Kersten et al. (2017) analyzed the trends and opportunities in supply chain management that lead to digital transformation opportunities from a twofold perspective: endogenous and exogenous. Endogenous trends include: digitalization of business processes, business analytics, transparency in the supply chain, automation, networking/ collaboration, and decentralization. Exogenous trends comprise: cost pressure, demand fluctuations, government regulations/ compliance, individualization, staff shortages, risks/ interruptions, complexity, sustainability, and changed consumer behavior.

The analysis of challenges and opportunities in the supply chain has generally been addressed in the literature by focusing on one or several dimensions. For example, Zhong, et al. (2016) focus on Big Data technology and identify current challenges, opportunities and future perspectives from six aspects: data collection, data transmission, data storage, processing, decision-making, and applications. Similarly, Bechtsis, et al. (2017) study the impact of Automated Guided Vehicles for a sustainable supply chain. Barbosa-Póvoa, da Silva and Carvalho (2018) include three dimensions (economic, environmental and social) in their analysis to identify opportunities and challenges from an operational perspective. However, few studies have considered other dimensions (e.g. legal, political).

Challenges increment the complexity of supply chains in three main aspects: operational, logical and administrative (Yami, 2018). According to Yami (2018), a study carried out in 2006 performed a ranking of complexity drivers in the supply chain leading to the following result: product/ services, customers, direct materials, ship-to locations, manufacturing locations, suppliers, and distribution

centers. The actions focused on the control of the complexity of the supply chain that will have a positive impact in the competitiveness of the firm.

In this regard, handling change is an integral part of managing supply chains. However, the speed of change is so fast that it is difficult to identify where to focus the company's management resources. Under this uncertain environment a framework for the identification of future scenarios in the supply chain has been developed. The framework comprises a PESTLE analysis to identify megatrends that interact to create the challenges and opportunities future supply chains might face. This paper uses the process, discrete manufacturing and logistics industries as input in order to be able to analyze megatrends from different points of view. The results of this paper are final DPs that consider these three industries like a single system. This approach provides the opportunity to create generic DPs, which are required to build industrial scenarios.

This paper is divided into five sections. Section 2 includes a literature review related to scenario development in the field of supply chain management (SCM). Section 3 describes the methodology developed for the definition of DPs. The defined DPs are summarized in Section 4 and elaborated in detail for the technological dimension in Section 5. Section 6 entails the conclusion and an outlook on further research.

2 Literature Review

Today, supply chains face several challenges and opportunities, such as globalization and emerging technologies that change and shape the future of industries. Thus, it is crucial for companies to develop robust strategies and prepare for the future (Singh, 2004; Melnyk, et al., 2009). To this end, scenario planning is the most appropriate approach for a long-term planning to support decision making in uncertain situations (Schoemaker, 1993).

Scenario planning has been utilized by several studies in the field of SCM (e.g. Mazzarino, 2012; PwC, 2009; von der Gracht and Darkow, 2010). As for the planning horizon, most of the papers in the field of SCM focus on a range of 8–10 years (e.g. von der Gracht and Darkow, 2013) and a few take into account a planning horizon of more than 10 years (Jiang, Kleer and Piller, 2017). Regarding the context of the scenarios, the focus of those papers is mainly on the logistics industry (e.g. DHL, 2012; von der Gracht and Darkow, 2016). There are a few papers

that develop scenarios for the discrete manufacturing industry (e.g. Arora and Putcha, 2013; Jiang, Kleer and Piller, 2017) and process industry (e.g. Willigens and von der Gracht, 2013). To the best of our knowledge, there is not any study on industry scenarios that support decision-making in all of these three industries so that companies could take a holistic perspective into consideration. This requires a more thorough approach to manage long-term planning in the process, discrete manufacturing and logistics industry. Our research closes this research gap by developing alternative DPs that consider all the three aforementioned industries.

Scenario publications mainly use the Delphi technique to develop and present only the final scenarios (e.g. von der Gracht and Darkow, 2010; Jiang, Kleer and Piller, 2017). In the contrary, our study follows the Gausemeier and Plass (2014) approach, which is described in the methodology section. This paper also presents potential future DPs that enable companies to recognize different developments based on megatrends. From the DPs final scenarios, which reflect the overall system relations between the chosen dimensions, can be developed in further research.

The central idea of developing alternative paths for long-term planning is to lead decision makers to specific directions within the PESTLE dimensions and provide support in uncertain times (Powell, 1992). However, most of the studies give emphasis mainly on the environmental dimension (e.g. Arora and Putcha, 2013; von der Gracht and Darkow, 2013; von der Gracht and Darkow, 2013; von der Gracht and Darkow, 2016; PwC, 2009). Only a few studies (e.g. DHL, 2012; Mazzarino, 2012) have addressed all of them. Our focus lies within all of the PESTLE dimensions. Based on the gaps discussed above, this study tries to answer the following research questions:

- 1. What development paths might emerge from megatrends in the industrial sector by 2030?
- 2. How will the technological dimension particularly shape the DPs for the industrial sector by 2030?

3 Methodology

This paper follows the scenario technique developed by Gausemeier (Gausemeier, et al, 1998; Gausemeier and Plass, 2014) in order to create projections, which constitute developments into the future, specifically until the year 2030.

The Gausemeier approach has five stages: (i) preparation, (ii) scenario field analysis, (iii) scenario prognosis, (iv) scenario development, (v) scenario transfer. This paper presents the third stage, in which a range of possible futures (typically 2-4) is defined for each descriptor (i.e. megatrends such as globalization). These projections represent alternative and dissimilar developments of the descriptor.

Thus, this study shows all the potential future developments, which enable companies to evaluate different paths for different megatrends, instead of presenting the final scenarios that result from different combinations of a variety of DPs within the underlying dimensions. This basis allows building final consistent scenarios in further research.

These DPs are derived from the analysis of identified megatrends in previous research (Kalaitzi, et al., 2018), which in turn have been organized according to the PESTLE dimensions. Some megatrends have been rephrased if necessary in order to derive the descriptors. The decision to rephrase a certain megatrend is subject to its tendency. If the megatrend has a positive or negative inclination, it is rephrased to represent a neutral position; e.g. the megatrend "protectionism" has been renamed to the descriptor "trade policy". A descriptor can have a positive, negative and neutral DP. However, deviating settings are also feasible. Consequently, the number of DPs per descriptor is not fixed and varies by descriptor. Each descriptor is characterized by diverging DPs that express possible future states of the descriptor and describe circumstances companies and societies might face.

The analysis of the megatrends considered statistics, forecasts and descriptions of the respective megatrends found in literature (Kalaitzi, et al., 2018). Hence, quantitative and qualitative data is combined in order to derive the DPs. Additionally, several iterative workshops with three to 17 experts from different departments, sectors and with different backgrounds were conducted in order evaluate the DPs derived from literature as well as elaborate on further possible DPs. This approach provides a comprehensive picture throughout all PESTLE dimensions as well as the three industries under consideration.

Each DP is assigned to one of the PESTLE dimensions. Nevertheless, certain DPs also can have impacts on other dimensions since it is rarely that developments affect only one of the PESTLE dimensions. This is due to the comprehensive nature of the study and a holistic analysis of opportunities and challenges of megatrends.

The main criteria for the creation of DPs are its reasonability and conceivability (Gausemeier and Plass, 2014). Hence, we propose that every DP needs to fulfil certain quality criteria, namely: (i) plausibility - a DP needs to be plausible to the complete scenario team, (ii) dissimilarity - all DPs have to be distinct to each other, (iii) completeness - a set of projections within a descriptor has to provide a comprehensive set of possible developments, (iv) relevance - each DP requires a check regarding its future relevance, and (v) information content - each DP needs to add further value to the set of DPs within a descriptor. A DP can be futuristic but needs to rely on valid arguments or requires justification by statistical developments.

Suitable DPs must be distinct, so that the consistency check, needed in subsequent research for the scenario building, does not lead to many different evaluations and, hence, to inconsistent scenarios. Rather reasonable combinations of DPs are necessary for the creation of consistent scenarios.

Section 4 presents the results of this research by listing the descriptors with a definition and the assigned DPs organized by the six PESTLE dimensions.

4 Overview of the Development Paths

Table 1 contains the descriptors per PESTLE dimension along with a definition for each descriptor, and provides an overview of the identified DPs per descriptor.

		t in Eu- in Eu-	edo	S
ısions	lopment Path	tant developmen rnment collapse	cal concord in Eur ctionism rade	anted union able confederation nentation
E dimer	Deve	Cons rope Gove rope	Politi Prote Free 1	Conte Unsta Fragr
ary of development paths in the PESTLE	Definition	Political setting describes the polit- ical activity in a society, the satis- faction level of the population and general risks that might affect a country (Campos and Gassebner, 2009).	"Trade policies are policies aimed at influencing the international commercial relations and the flow of goods and services across bor- ders" (Jarman, 2017) affecting the availability, and therefore price and use goods and services.	"That form of association between states in which the general govern- ment is dependent upon the re- gional governments has often been described as a confederation" (Ben- nett, 1964).
Table 1: Summa	Descriptor	Political Setting	Trade Policies	Confederation
	Dimension		Political	

	Global Shift	Trade	Global Trade Shift describes the changes in location of economic activities between industrialized countries and emerging economies (Dicken, 2015; UNCTAD, 2012).	The pendulum shifts Steady titans US & Europe
	Global Co Structures	s	Global Corporate Structures "de- fine and clarify responsibilities for operational, control, and reporting processes" (Baret, et al., 2013, p.2) on a global governance level.	Think global, act local Rise of born-global firms
Economic	Digital Ecc	ymono	Digital Economy describes the changing environment in which digitalization is becoming more and more integrated into the busi- ness environment enabling new business concepts (Brynjolfsson and Kahin, 2000).	Traditional economy persists Digital potential Digital impediment
	Financial tions	Innova-	Financial innovation is "something new that reduces costs, reduces risks, or provides an improved pro- duct/service/instrument that bet- ter satisfies participants' demands" (Frame and White, 2002, p.3) within a financial system.	Bank and Fintech collabora- tion A world without banks Big five are the banks of the fu- ture

129

Ageing population and acceler- ation of disparities Awareness of inequalities and wealth redistribution	Smart regions Smart cities	Much and cheap Consumption awareness DIY society Individualized consumption
Demographic change describes the changes and tendencies of the pop- ulation regarding age, gender, birth and death rate and migration. It also comprises longer and health- ier life expectancy (Cambridge Dic- tionary, 2017).	"The process of the economic development which leads to a significant concentration of hu- man resources, economic activi- ties, and resource consumption in cities (modern environment or re- furbished buildings, studios and lofts" (Madlener and Sunak, 2011).	Consumption patterns describe the buying behavior and the handling of the purchased goods or services (Fletcher and Emmanuel-Stephen, 2016).
Demographic Change	Urban Living	Consumption Patterns
	Social	

Customer Orien- tation	"Customer orientation refers to the importance an employee places on meeting customers' needs and ex- pectations" (Nguyen, et al., 2014, p.1097).	Individualism – focus on vari- ety Collectivism – focus on the crowd
Knowledge- based economy	Knowledge-based economy describes "trends in advanced economies towards greater depen- dence on knowledge, information and high skill levels, and the increasing need for ready access to all of these by the business and public sectors" (Organisation for Economic Cooperation and Development, 2007, pp.434).	Investments equalize the labor market Rapid changes cause unem- ployment
Digital Transfor- mation	Digital transformation describes the changes related to the applica- tion of technologies and their inte- gration into all aspects of human life and society, e.g. to improve per- formance of enterprises or conve- nience of social life or to change the way business is done (Probst, et al., 2017, p.10).	Rapid advancement of digitiza- tion and digitalization Obstacles restraining digital transformation Digital stagnation

	Autonomous Sys- tems	Autonomous systems describe ob- jects or devices that can act and make situation-dependent deci- sions independently without inter- ference by humans or other outside forces and have the ability for self- governance in the performance of control functions (European Group on Ethics in Science and New Tech- nologies, 2018; Antsaklis, et al., 1991, p.5).	Dynamic development of au- tonomous technologies Innate reluctance to accept au- tonomous technologies
Technological	Alternative en- ergy generation, storage usage	Alternative energy generation, stor- age and usage illustrates how en- ergy can be obtained and stored in sustainable ways. Moreover, it de- scribes the usage of alternative en- ergy for industrial, transportation and mobility purposes (Kleiner, et al, 2017; Hydrogen Council, 2017, p.10).	Established electrification technologies and green sys- tems Ongoing electrification and alternative energy endeavors
	Decentralized connection of information and physical devices	Decentralized connection of infor- mation and physical devices de- scribes how companies cope with the masses of data and take advan- tage of the smart decentralized link- age (Dickson, 2016).	Dominance of global players Start-ups and SMEs take up business

	Disruptive Production Technologies	Disruptive technologies describe developments that bring revolu- tionary changes to social life and to the way companies understand and do business. In the context of production technologies, the con- cept of Industry 4.0 reflects this emerging pattern, encompassing the integration of different tech- nologies into an autonomous pro- duction system that can regulate itself based on knowledge and sen- sors (Lasi, et al., 2014; Hofmann and Rüsch, 2017).	Continuous exploitation of dis- ruptive technologies Coexistence of conventional and disruptive technologies
Legal	Consumer Protection Laws Intellectual Prop- erty Laws	Consumer protection law is de- fined as "all legal principles and rules governing relationships and problems between various parties or each other in relation to goods and / or services in the aspects of life" (Azis, 2018, p.56). Intellectual property (IP) laws deal with the laws that apply for "cre- ations of the mind: inventions; liter- ary and artistic works; and symbols, names and images used in com- merce" (World Intellectual Prop- erty Organization, 2018, p.2).	Promotion of laws and full product transparency Legislation is lagging behind dynamic market development Full security for inventors and data providers Low confidentiality for data and market participants

Comprehensive regulatory framework Heterogeneous regulations	Our planet is recovering Our planet on the brink	Countering resource depletion Rise in depletion of natural re- sources
Social and Environmental regula- tions describe regulations that urge companies and societies to act and use environmental functions in an environmentally friendly and ethical way (Organisation for Eco- nomic Cooperation and Develop- ment, 2007, pp.253, 725; Shift2Rail, 2017).	It is a "a change in global or regional climate patterns, started from the mid to late 20th century onwards and attributed to the increased lev- els of atmospheric carbon dioxide produced by the use of fossil fuels" (Oxford's dictionary, 2017).	It describes the management of hu- man interaction with the environ- ment and aims to assure that the state of an environmental resource affected by humans is maintained for future generations, and for ecosystem integrity through con- sidering ethical, economic, and sci- entific (ecological) variables (Pahl- Wostl, 2007).
Social and Envi- ronmental Regu- lations	Climate Change	Environmental Environmen- tal Resource Management
Due to numerous DPs in each of the six PESTLE dimension, it is not feasible to illustrate all DPs in detail in this paper. Hence, the focus in Section 5 is on the technological dimension. More specifically, "Autonomous Systems" are chosen as example since these technologies are expected to progress on a high pace with vast impacts on all industries and the related supply chains.

5 Development Paths for Autonomous Systems in the Technological Dimension

The two DPs under the descriptor "Autonomous Systems" are exemplarily outlined for the technological dimension.

Autonomous systems describe objects or devices that can act and make situationdependent decisions independently without interference by humans or other outside forces (European Group on Ethics in Science and New Technologies, 2018). These technologies have an immense impact on ways of working, particularly how people will collaborate with other people, machines and virtual formats in entirely new ways (Wisskirchen, et al., 2017).

Based on the previous research (Kalaitzi, et al., 2018), autonomous systems encompass the trends robots, drones, automated vehicles/ automated guided vehicles and cyber-physical systems. The DPs take up on these trends and are described in the following Subsections 5.1 and 5.2.

This section focuses on the DPs. DPs describe how the future in 2030 might look like. Since this study is part of a comprehensive scenario generation approach, it is subject to further research to analyze which DPs are feasible in the context of a holistic scenario. The scenarios are composed of combinations of several DPs from various descriptors and different PESTLE dimensions. Hence, at this stage of research it is yet not possible to derive which implications arise, how companies need to prepare for the future, how supply chains will change concerning their configuration or how the developments will affect specific industries (e.g. process, discrete manufacturing or logistics).

5.1 Dynamic Development of Autonomous Technologies

This DP describes that companies are taking advantage of digital technologies to find new markets, business models and revenue streams and that human workforce can benefit from this development.

Technology advancements, especially with regard to robots, drones and autonomous vehicles are progressing at a rapid pace. Cyber-physical systems play a key role for autonomous systems in the industrial environment. They constitute enablers for efficient communication and control by transferring and exchanging data over the internet in real time. An increased exploitation of these technologies leads to a highly automated and autonomous environment which permits to improve the productivity rate (for example robots can perform 24/7) reducing quality problems, errors and down times (Wisskirchen, et al., 2017). Coupled with the ability to share and act upon the associated data and derived insights, new service and production related business models emerge both within and across organizations, removing traditional silos as well as simplifying trust and contractual agreements. Automation (both physical and virtual) replaces an increasing range of human tasks (Bingley, et al., 2016).

Implications on supply chains

The described technological advancements allow the administration of supply chains to be simplified and to operate with less cost and better customer satisfaction. In particular, the first step in the advancement of autonomous systems is at the process level in order to automate non-value added activities. In this regard, rapid changes could lead to high rates of unemployment. In this case, companies need to define approaches for reallocation of staff along the supply chain and consider the possibility to improve their capabilities since these technologies require advanced IT skills. By tracking the whole supply chain, transparency between the supply chain actors can be increased, which improves the capability to react efficiently and quick to external influences. If new business models are implemented to handle the increasing complexity of autonomous systems, the supply chain can achieve high performance in terms of agility, reliability and transparency.

5.2 Innate Reluctance to Accept Autonomous Technologies

This DP states that the use of advanced technologies is reshaping the work landscape intensifying competition on the labor market and causing worker displacements.

The technological development enables a high degree of automation and autonomiztion. Suppliers provide modular and standardized components so that technology solutions become affordable, but still often lack profitability (Gausemeier and Plass, 2014). Since an autonomous technology requires suites of expensive sensors, the average cost of this technology is high and this could slow down the application of this technology. A particular technological roadmap to reduce those costs is yet to be established (Omohundro, 2014; Anderson, 2016). Although autonomous technologies often lack profitability (Gausemeier and Plass, 2014) or regulations prevent the full exploitation of their potentials, automated and partly autonomous factories are progressively becoming the standard in Europe. The result is a shift from assistance of human activities to a more machine-centered environment. This development radically reshapes the work landscape and creates new business models (Bingley et al., 2016). Users struggle with operating the highly complex machines (Gausemeier and Plass, 2014). Hence, employees fear for their jobs, get frustrated and demotivated and eventually adopt a negative attitude towards emerging advanced technologies.

Implications on supply chains

High costs, privacy and cyber security issues, low IT penetration into processes as well as a lag of technology standards make the adoption of new technologies slower and restrain digital transformation. Supply chains need to revise processes and move from traditional supply chains towards a connected, smart, and highly efficient supply chain ecosystem in order to achieve comprehensive agility and transparency.

6 Conclusion and Outlook

This paper identifies 51 DPs that might emerge from megatrends in the industrial sector by 2030. DPs constitute projections of each megatrend into the future.

They are grouped under 22 so-called descriptors and assigned to the PESTLE dimensions. Each DP creates different conditions that will have an impact on supply chains and their configuration.

In order to address our first research question, we have extended previous scientific publications and grey literature by taking a holistic approach and considering three different industries (i.e. process, discrete manufacturing, and logistics) as well as six PESTLE dimensions. We assessed megatrends and their associated challenges and opportunities in order to define DPs for the future. The systematic collection of information regarding megatrends and their analysis permits a complete and well-structured illustration of a set of DPs.

To answer our second research question, we exemplarily elaborated on two DPs for autonomous system by providing a comprehensive description of the two divergent DPs. Furthermore, possible implications of these DPs on supply chains and companies are described.

The construction of DPs is part of a comprehensive scenario technique. Future research will show the impacts of each DP on each specific industry and the related supply chains. Then, by using the Gausemeier and Plass (2014) methodology, it will be necessary to create the most plausible and diversified industrial scenarios. These scenarios help to define the best strategies to engage the different challenges of the future.

Our research is limited in the way that process, discrete manufacturing and logistics industries are considered as an input only. The resulting DPs are not classified by industries. Thus, this study follows a holistic approach and can be used to create scenarios for different industries in future research and to derive specific DPs for the aforementioned industries. Additionally, the DPs were validated with a small group of experts. More experts could provide other insights. Some DPs are concentrated in a European context since the focus was on existing European roadmaps. Practically, this study enables supply chain professionals to understand possible DPs and serves as basis to develop future scenarios in subsequent work.

7 Acknowledgements

This research has received funding from the European Union's Horizon 2020 Research and Innovation Program under the Grant Agreement No. 768884. The authors thank all contributors of the Next-Net project team and all experts from industry participating in the workshops.

References

ALICE (2014). ALICE Recommendations to H2020 Work Programs 2016-2017.

- Anderson, J. e. a. (2016). Autonomous Vehicle Technology: A Guide for Policymakers. Ed. by RAND Corporation.
- Antsaklis, P. J., K. M. Passino, and S. J. Wang (1991). "An introduction to autonomous control systems". In: *IEEE Control Systems* 11.4, pp. 5–13.
- Arora, V. and S. Putcha (2013). "Future Scenarios for Green Chemical Supply Chains". PhD thesis. Massachusetts Institute of Technology: Massachusetts Institute of Technology.
- Azis, D. E. P. (2018). "Juridical Review The Implementation of Oral Agreement is associated with the Law of Treaties and Law Number 8 Year 1999 concerning Consumer Protection". In: Substantive Justice International Journal of Law. Vol. 1, pp. 56–64.
- Barbosa-Póvoa, A. P., C. da Silva, and A. Carvalho (2018). "Opportunities and challenges in sustainable supply chain: An operations research perspective". In: *European Journal of Operational Research* 268.2, pp. 399–431.
- Baret, S., N. Sandford, E. Hid, J. Vazirani, and S. Hatfeld (2013). Developing an effective governance operating model: A guide for financial services boards and management teams. Ed. by Deloitte Development LLC.
- Bechtsis, D., N. Tsolakis, D. Vlachos, and E. Iakovou (2017). "Sustainable supply chain management in the digitalisation era: The impact of Automated Guided Vehicles". In: *Journal of Cleaner Production* 142, pp. 3970–3984.

Bennett, W. H. (1964). American Theories of Federalism. Tuscaloosa: University of Alabama Press. Berger, R. (2015). The Supply Chain Excellence Study 2015.

Bingley, M., J. F. Esteban Lauzán, J. Hall, G. Jagdale, M. Pfeil, M. Smith, F. Pinheiro, and H. Tardieu (2016). Journey 2020: Digital Shockwaves in Business. Ed. by Atos SE.

Brynjolfsson, E. and B. Kahin (2000). Understanding the Digital Economy: Data, Tools, and Research. Cambridge Dictionary (2017). Demographics.

Campos, N. F. and M. Gassebner (2009). "International terrorism, political instability and the escalation effect". In: IZA discussion papers 4061.

Cole, B. and J. O'Donnell (2017). Discrete vs. process manufacturing.

DHL (2012). "Delivering tomorrow Logistics 2050 A Scenario Study". In:

Dicken, P. (2015). Global shift: Mapping the changing contours of the world economy. 7. ed. Los Angeles, Calif.: Sage Publ.

Dickson, B. (2016). Decentralizing IoT networks through blockchain.

EFFRA (2013). Factories of the future: Multi-annual roadmap for the contractual PPP under Horizon 2020. Luxembourg: Publications Office of the European Union.

European Battery, Hybrid and Fuel Cell Electric Vehicle Congress (2017). Geneva, Switzerland.

European Group on Ethics in Science and New Technologies (2018). Artificial Intelligence, Robotics and 'Autonomous' Systems. Ed. by European Commission. Luxembourg.

- European Union (2015). *EU transport in figures: Statistical Pocketbook 2015*. EU transport in figures. Luxembourg: Office for Official Publications of the European Communities.
- Fletcher, K.-A. P. and C. M. Emmanuel-Stephen (2016). "Social Media Engagement: Reshaping the Consumption Pattern of Generation Y Caribbean and Latin American Consumers". In: Consumerism and Buying Behavior in Developing Nations. Ed. by IGI Global, pp. 218–238.
- Frame, W. S. and L. J. White (2002). Empirical Studies of Financial Innovation: Lots of Talk, Little Action? Ed. by Federal Reserve Bank of Atlanta. Atlanta, GA, USA.
- Gausemeier, J., A. Fink, and O. Schlake (1998). "Scenario Management". In: Technological Forecasting and Social Change 59.2, pp. 111–130.
- Gausemeier and Plass (2014). Zukunftsorientierte Unternehmensgestaltung Strategien, Geschäftsprozesse und IT-Systeme für die Produktion von morgen. 2nd ed. München: Carl Hanser Fachbuchverlag.
- Hydrogen Council (2017). "Hydrogen scaling up: A sustainable pathway for the global energy transition". In:
- IGI Global, ed. (2016). Consumerism and Buying Behavior in Developing Nations.
- Institute of Industrial and Systems Engineers (2018). Process Industries Division.
- Jarman, H. (2017). "Trade policy governance: What health policymakers and advocates need to know". In: *Health policy (Amsterdam, Netherlands)* 121.11, pp. 1105–1112.
- Jiang, R., R. Kleer, and F. T. Piller (2017). "Predicting the future of additive manufacturing: A Delphi study on economic and societal implications of 3D printing for 2030". In: *Technological Forecasting & Social Change* 117, pp. 84–97.
- Kalaitzi, D., A. Matopoulos, R. Fornasiero, A. Zangiacomi, I. Marchiori, A. C. Barros, K. Pires, P. Campos, V. Amorim, V. Muerza, S. Sardesai, D. Daus, M. Witthaut, L. Aerts, and S. Balech (2018). D2.1: Report on trends and key factors.
- Kersten, W., M. Seiter, B. von See, N. Hackius, and T. Maurer (2017). Trends And Strategies In Logistics And Supply Chain Management: Digital Transformation Opportunities. Ed. by BVL International -The Supply Chain Network.
- Kleiner, F., M. Beermann, B. Çatay, E. Beers, H. Davies, and O. T. Lim (2017). "Current status of the electrification of transport logistic vehicles - Early niche markets and commercialization opportunities". In: European Battery, Hybrid and Fuel Cell Electric Vehicle Congress. Geneva, Switzerland.
- Lehmacher, W., F. Betti, P. Beecher, C. Grotemeier, and M. Lorenzen (2017). Impact of the Fourth Industrial Revolution on Supply Chains. Ed. by World Economic Forum. Cologny/Geneva, Switzerland.
- Madlener, R. and Y. Sunak (2011). "Impacts of urbanization on urban structures and energy demand: What can we learn for urban energy planning and urbanization management?" In: *Sustainable Cities and Society* 1.1, pp. 45–53.
- Mazzarino, M. (2012). "Strategic scenarios of global logistics: what lies ahead for Europe?" In: European Transport Research Review 4.1, pp. 1–18.
- Melnyk, S. A., R. R. Lummus, R. J. Vokurka, L. J. Burns, and J. Sandor (2009). "Mapping the future of supply chain management: a Delphi study". In: *International Journal of Production Research* 47.16, pp. 4629–4653.
- Nguyen, H., M. Groth, G. Walsh, and T. Hennig-Thurau (2014). "The Impact of Service Scripts on Customer Citizenship Behavior and the Moderating Role of Employee Customer Orientation". In: Psychology & Marketing 31.12, pp. 1096–1109.

Omohundro, S. (2014). "Autonomous Technology and the Greater Human Good". In: Journal of Experimental & Theoretical Artificial Intelligence 26.3, pp. 303–315.

Organisation for Economic Co-operation and Development (2007). *Glossary of Statistical Terms*. Oxford's dictionary (2017). *climate change*.

- Pahl-Wostl, C. (2007). "The implications of complexity for integrated resources management". In: Environmental Modelling & Software 22.5, pp. 561–569.
- Powell, T. C. (1992). "Strategic Planning as Competitive Advantage". In: Strategic Management Journal 13.7, pp. 551–558.
- Probst, L., B. Pedersen, O.-K. Lonkeu, C. Martinez-Diaz, L. Novelle Araujo, PwC, Klitou, Demetrius, J. Conrads, M. Rasmussen, and CARSA (2017). *Digital Transformation Scoreboard 2017: Evidence* of positive outcomes and current opportunities for EU businesses.
- PwC (2009). "How will supply chains evolve in an energy-constrained, low-carbon world?" In: Transportation & logistics 2030 series 1.
- Savills Investment Management (2016). European Logistics: warehousing the future. Ed. by Savills Investment Management LLP. London.
- Schoemaker, P. J. H. (1993). "Multiple scenario development: Its conceptual and behavioral foundation". In: Strategic Management Journal 14.3, pp. 193–213.
- Shift2Rail (2017). Shift2Rail Joint Undertaking (S2R JU): Annual Work Plan 2018 & Budget.
- Singh, M. (2004). "A Review of the Leading Opinions on the Future of Supply Chains: Supply Chain 2020 Project Working Paper". PhD thesis. Boston, MA: MIT.
- Speranza, M. (2018). "Trends in transportation and logistics". In: European Journal of Operational Research 264, pp. 830–836.
- Substantive Justice International Journal of Law (2018). Vol. 1.
- Tello, P. and R. Weerdmeester (2012). SPIRE Roadmap.
- UNCTAD (2012). Global trade trends. Ed. by UNITED NATIONS PUBLICATION.
- von der Gracht, Heiko A. and I.-L. Darkow (2010). "Scenarios for the logistics services industry: A Delphi-based analysis for 2025". In: *International Journal of Production Economics* 127.1, pp. 46–59.
- von der Gracht, Heiko A. and I.-L. Darkow (2013). "The future role of logistics for global wealth scenarios and discontinuities until 2025". In: Foresight 15.5, pp. 405–419.
- von der Gracht, Heiko A. and I.-L. Darkow (2016). ""Energy-constrained and low-carbon scenarios for the transportation and logistics industry"". In: *The International Journal of Logistics Management* 27.1, pp. 142–166.
- Willigens, S. E. and von der Gracht, Heiko A. (2013). The Future of the Supply Chain 2020+: Scenarios and Requirements for the Process Industry. Ed. by msg systems ag. Ismaning/Munich.
- Wisskirchen, G., Biacabe, B., Bormann, U., Muntz, A., Niehaus, G., Soler, G. and von Brauchitsch, B. (2017). Artificial Intelligence and Robotics and Their Impact on the Workplace.

World Intellectual Property Organization (2004). What is Intellectual Property? Geneva, Switzerland. Yami, Z. (2018). Supply Chain Complexity – Dealing with a Dynamic System.

Zhong, R. Y., S. T. Newman, G. Q. Huang, and S. Lan (2016). "Big Data for supply chain management in the service and manufacturing sectors: Challenges, opportunities, and future perspectives". In: Computers & Industrial Engineering 101, pp. 572–591.

About the Future Role of Software in the Product

Henning Skirde¹, Robert Steinert¹

1 - ID-Consult GmbH

Product development is shifting towards realizing an extended scope of product functions in software. This leads to new challenges in terms of a methodological integration and synchronization of different development disciplines. This paper provides insights on how to systematically manage the harmonization of traditional development disciplines with agile software development based on an integrated data model. To achieve this, the traditional product data model is extended to a seamless system data model that covers – based on a common function structure – the product itself but also services induced by digitalization and infrastructure backends.

Keywords: Product Data Model METUS; Software Development; Agile and SCRUM; Harmonization Digitalization

First recieved: 25.May.2018 Revised: 15.Jun.2018 Accepted: 04.Jul.2018

1 Software as a Function Owner in Digitalized Products

"Bending sheet metal is not our value creation's core anymore", this is – slightly exaggerated – how manufacturing companies put a major change into a nutshell, that has become essential in product development. The functions of technical products are to an increasing extent determined by Software and digital connectivity, "bent sheet metal" does not ensure market success anymore.

Today, software is an indispensable ingredient of technical products. An example is embedded software like in an automobile's control unit or as a control system of a machine or plant. Nevertheless, traditional development processes based on a two-stage freezing of requirements often lack the consideration of software and do not recognize it as an essential element of the development process according to its role in the completed product.

Mechanics and electronics may feature different development methods, but their common ground and their origin descend from a physical world. Where these disciplines think based on product structures, software development choses a structure based on data models. In addition, software development as a discipline has come up with its own methodology: Agile approaches such as SCRUM (Schwaber & Suther-land 2017) are entirely different from traditional product development processes.

Up to now, a common solution has been to encapsulate the discipline of software development right into the traditional product development process. By this, software development could retain its own approaches and data models. An integration with other development disciplines' statuses has been conducted based on completed releases, milestones and snapshots. Against the background of major differences regarding methodology and data models of other development disciplines, such an approach seemed to be sufficient. Today on the contrary, such a methodological segregation fails to be a promising way to go (Eigner et al. 2018).

Why is this the case? Software as an ingredient of technical products in the previous sense contains a system related machine control. It might be embedded in a control unit's firmware with the aim to interpret signals from a machine's or plant's sensors and pilots corresponding actuators based on algorithms.

In an easy example, the software of a car's engine control unit uses the values from sensors at the crankshaft, throttle valve and intake air temperature to calculate



Figure 1: In future products, the software's significance will be enhanced (Source: own representation)

an optimal timing and duration for the fuel injection and subsequently provides the resulting values to the corresponding actors.

This previous sense of software that is despite all complexity well manageable has already been outdated (Figure 1): To this day, software has been modified from an integrated system element to a self-contained function owner. Mainly mechanically determined products have been altered into today's mechatronic products. In the future, due to a higher share of software in products as well as of digital connectivity, this will result in "cyberphysical systems" (Eigner et al. 2016).

2 Software's Promise: More Value from Less Effort

A product's competitive edges can be realized in software to an increasing degree, without changing the hardware in the ideal case. An example from the automotive industry would be an entertainment module that comes with an identical hardware (screen, control unit, operating controls) to minimize the number of physical variants. Particular functions, e.g. a hands-free-speakerphone, can be activated by software. To achieve this, the software covers the maximum functionality. By this, variants can be derived by locking and unlocking individual functions. In this context, software offers the attractive promise of an increased customer value without bearing the costs of a higher physical variance (Khaitan & McCalley 2015).

The opportunities of software as a function owner in a product are numerous and attractive:

- 1. Variance in software can be realized in a more flexible and more costeffective way compared to hardware.
- 2. Additional prospects for differentiation can be derived by additional product functions and services provided by software.
- Existing products can be updated with more recent software to run new functions. This enhances customer satisfaction and extends the hardware's competitiveness in the life cycle.
- 4. Elongated operating times regarding hardware, that can be updated with software, reduce the TCO and therefore might be a competitive advantage.

However, these opportunities face several risks:

- 1. Traditional competitive edges regarding products more and more lose their significance.
- 2. The high innovation frequency in agile software development can only be transferred to other engineering disciplines with several limitations.
- 3. Customers have a limited willingness to pay for software due to their experience regarding apps that are free of charge or at least cheap.

Software is not only an essential ingredient of a product – be it as a machine control or as a hands-free-speakerphone's software – but also to an increasing degree a "virtual" system element. This refers to software that an end user will experience as a part of the product functionality, but it is not part of the particular product. To provide an example, an app on the end user's own device like a smartphone will serve as a control for his digitally connected coffee machine.

From this point of view, products increasingly lose their physical boundaries, become more and more digitally connected elements of cyberphysical systems (Eigner et al. 2016). Thereby, new value propositions and business models can be identified and subsequently implemented. But before implemented, both the product development process and also product structures have to be adapted to the new requirements.

For this, the following three areas inhibit new challenges:

- Methodological integration: The development methods and displaying of requirements in software engineering differ considerably from those in the development disciplines for mechanics and electronics.
- 2. Integration of "virtual" product components: Components that are a product's ingredient from a functional perspective, but are not in the physical product structure's scope (e.g. a smartphone app for machine control) are not considered in conventional development processes (Conti et al. 2012).
- Synchronizing the diverse product and innovation cycles among different development disciplines: Agile principles in software development often contain iterations that are not sufficiently synchronized with development cycles of the corresponding hardware.

The following section provides insights on how to systematically manage these three areas, in order to utilize a digital product's advantages to a full extent.

3 Requirements: Traditional, Agile or Both?

Product functionality and innovation are increasingly shifted from mechanics to software. In several areas, mechanical designs have achieved high degrees of maturity. Consequently, disruptive innovations become rather unlikely or require disproportionally high development efforts. Therefore, software can serve as an adequate leverage for new and innovative functionalities.

In the utmost case, the hardware's role is downgraded to a software's operating environment. This has the consequence that requirements to hardware are determined by the dedicated software functionalities. If software becomes a functional driver, it will provide the basis for hardware development. Accordingly, hardware in terms of an operating environment for a software has to be designed with the aim to be able to "carry" an ex ante defined number of software cycles with more and more comprehensive functionality.

This requires enhancing the harmonization of different development disciplines based on a common functional level. This means after initially determining a common function structure, the development team has to decide not only how, but also in which discipline a function shall be realized. In addition, realizing functions in software also calls for a comprehensive management of requirements that incorporates all disciplines.

Agile software development allows for a realization of a product's market-ready versions in short iteration loops that feature additional functions that are successively implemented in further releases. Compared to this, traditional development disciplines have a higher effort to realize a new product release because an incremental approach comparable to software engineering is not feasible (Vogel & Lasch 2016). The reason for this is the extent to which changes in product's hardware impact the operations – e.g. starting with molding design and subsequently logistics and production.

Agile and traditional methods for requirements management and engineering will thus need to coexist even though a convergence can already be noticed. Against this background, the argument can be derived that traditional and agile requirements processes need to be further harmonized and interrelated to display an overall view on the product system. A seamless product data model plays an important role not only in the course of the entire product development process, but also regarding a common understanding in different development disciplines.

This requires a methodological approach that consolidates a market and product view on the one hand as well as the phases of product definition, design and implementation on the other hand into a common data model that subsequently allows for flexible and numerous visualizations. The METUS methodology and the corresponding software have proven to be appropriate for this (METUS 2018).

4 From Product Architecture to System Architecture

Traditional product architecture models start an entire product's description of with requirements and functions, followed by the product structure consisting of individual function owners that can be both, mechanical and electronical components and modules (Göpfert & Tretow 2013). Without adaptions, such an architecture may not be able to incorporate the external or "virtual" scopes of a product into its structure.

Following the goal of a purposeful enhancement of a product's customer value driven by software functionality (or shifting the realization of functions from hardware (HW) to software (SW)), the traditional product architecture model does not suffice anymore. The model has to be enlarged in order to also manage





new components and services as an ingredient of the extended architecture. This requires raising the product architecture to a system architecture (figure 2). If this raise is successful, the product's system architecture also covers new "digital" scopes that feature similar characteristics in the function structure like mechanically or electrically realized functions. In addition, this step may also contain changes of the infrastructure, e.g. the provision of a backend or a cloud for value-added services that are required by digitalization (Crawley et al. 2015).

5 Synchronizing Tact Rates of Several Development Disciplines

In a similar way, like software will become a major driver of new product functionalities, the pulsing of software development will increasingly determine the pulsing of the entire development process.

In an age of mainly mechanical products, the product lifecycles corresponded to the innovation cycles (figure 3). Modularizing product architectures has enabled a decoupling of these two different cycles (Skirde 2015; Bahtijarevic et al. 2014; Lau

et al. 2011; Mikkola 2006). Technology and innovation leaps can be realized only in particular modules that replace outdated modules in the original product. The increase of mechatronic products has come with diverging the pulsing of product lifecycle and innovation cycles: New software releases, for example the already explained automobile engine's control unit, are deployed without changing the operating environment provided by the hardware.

To realize an extended scope of product functionalities with software functions, the hardware design has to be adequately dimensioned in advance: it has to be able to operate a software's future releases right from the start. For this, product management has to define up to which future software releases the hardware is supposed to remain unmodified. The ability for updates, known by every end user of a smartphone or tablet computer, will become a feature of nearly every technical product.

The conceptual design of updateable products requires a function roadmap containing software updates planned by the product management. This roadmap has to be transparent for all involved parties and agreed as an ingredient of the entire system architecture represented by a product embedded into this architecture. If this succeeds, new perspectives can be raised: particular scopes of the product variance can be shifted from hardware to software. In total, an extension of the performance spectrum towards a comprehensive product-service-system is likely (figure 4).





6 Pricing for Software-Based Product Functions

The increasing usage of software as a function owner determines that a higher share of the product requirements has to be fulfilled by software. Consequently, the share of the product's value proposition that has to be covered by software is increased accordingly. From this change, the question can be raised how to achieve an appropriate pricing for software as a function owner.

Today, software development is often accounted into the company's indirect costs. In comparison to the production of physical products, software comes with high non-recurring costs (NRC), but causes only very little recurring costs (RC). In terms of transparency, assigning the costs of individual software development to the corresponding scope of the function structure seems to be more beneficial from an economical perspective than accounting these into the indirect costs (Skirde et al. 2016). An analogous argumentation can be derived for assigning software development costs to requirements as well as the value proposition.

The major success factor that can be derived from a function-based pricing is again the overall view on the product, which is the basis that allows for a cost-oriented analysis of functions:

By connecting the value proposition with requirements (traditional and agile), function structure and finally the product structure with its mechanical, electronical and now also digital scopes, comprehensive data continuity from the market to the product view can be established.

If this procedure in the sense of a function analysis is then conducted the other way around – from the software scope in the direction of the function structure, the software's contribution towards fulfilling a requirement and realizing the value proposition can directly be derived.

The chance to increase a product's customer value by enhancing the software ingredients and thus to realize a greater extent of variance in software on a low level of costs, simultaneously shortening innovation cycles, seems to be exceedingly attractive.

Completely exploiting the chances outlined in this paper will probably only succeed for those companies, that already today turn their attention to a seamless product development process that interconnects a market and a product perspective on a reliable methodological basement.

7 Summary and Conclusion

Product development is shifting towards realizing an extended scope of product functions in software, challenging the past development paradigms. In this paper, the new challenges in terms of a methodological integration and synchronization of different development disciplines have been pointed out. To conclude, our recommendation to companies is to systematically manage the harmonization of traditional and software related development disciplines based on an integrated product data model. A common functional structure representing the entire product can support this harmonization. In addition, it allows for deriving data for the product data model can enable a company to consider new services induced by digitalization as well as accordingly required infrastructure scopes in their value proposition.

References

- Bahtijarevic, J. and M. De Murcia E Paes (2014). "Applying Modular Function Deployment (MFD) to software architecture". In: KTH Production Engineering and Management.
- Conti, M., S. Das, C. Bisdikian, M. Kumar, L. Ni, A. Passarella, G. Roussos, G. Tröster, G. Tsudik, and F. Zambonelli (2012). "Looking ahead in pervasive computing: Challenges and opportunities in the era of cyber-physical convergence". In: *Pervasive Mobile Computing*, Vol. 8 (1), pp. 2–21.
- Crawley, E., B. Cameron, and D. Selva (2015). "System Architecture: Strategy and Product Development for Complex Sys-tems". In: Prentice Hall Press, Upper Saddle River, NJ, USA.
- Eigner, M., H. Apostolov, and P. Schäfer (2018). ", Nachhaltigkeit 4.0" Nachhaltige Produktentwicklung im Zeitalter der vier-ten industriellen (R)evolution". In: *Fort-schritte in der Nachhaltigkeitsforschung*, pp. 253–272.
- Eigner, M., U. August, and M. Schmich (2016). "Smarte Produkte erfordern ein Umdenken bei Produktstrukturen und Pro-zessen". In: White Paper, Siemens PLM Software.
- Göpfert, J. and G. Tretow (2013). "Produktarchitektur". In: *In: Feldhusen, J. and & Grote, K.-H. [Eds.].* Pahl/Beitz Konstruktionslehre, 8. Edi-tion, Springer, Berlin/Heidelberg, p. 252.
- Khaitan, S. and J. McCalley (2014). "Design Techniques and Applications of Cyberphysical Systems: A Survey". In: In: IEEE Systems Journal, Vol. 9 (2), pp. 350–365.
- Lau, A., R. Yam, and E. Tang (2011). "The Impact of Product Modularity on New Product Performance: Mediation by Product Innovativeness". In: In: Journal of Product Innovation Management, Vol. 28 (2), pp. 270–274.
- METUS (2018). "Die METUS Software ist die Schaltzentrale f
 ür Produktentscheidungen". In: [online]: https://www.id-consult.com/loesungen/metus-software.html, called on May 15, 2018.
- Mikkola, J. (2006). "Capturing the Degree of Modularity Embedded in Product Architectures". In: Journal of Product Innovation Management, Vol. 23, pp. 128–146.

- Schwaber, K. and J. Sutherland (2017). "The SCRUM Guide". In: [online]: http://www.scrumguides.org/ docs/scrumguide/v2017/2017-Scrum-Guide-US.pdf, called January 22nd, 2018.
- Skirde, H. (2015). "Kostenorientierte Bewertung modularer Produktarchitekturen". In: Eul Verlag, Lohmar-Köln.
- Skirde, H., W. Kersten, and M. Schröder (2016). "Measuring the Cost Effects of Modular Product Architectures – A Conceptual Approach". In: International Journal of Innovation and Technology Management, 13(4).
- Skirde, H. and R. Steinert (2018). "The Future Role of Software in the Product Realizing Variance in Software". In: *Whitepaper, DOI: 10.13140/RG.2.2.25525.17126.*
- Vogel, W. and R. Lasch (2016). "Complexity drivers in manufacturing companies: a literature review". In: Logistics Research, 9: 25. doi:10.1007/s12159-016-0152-9.

Environmental Innovation of Transportation Sector in OECD Countries

Duygu Şahan¹, Okan Tuna¹

1 - Dokuz Eylül University Maritime Faculty

Climate change is a global concern and transport sector contributes to it significantly. This study aims to identify the factors which contribute to the development of environmental innovation in transport sector and to examine their effects. The analysis is carried out via a panel regression model with a dataset for 23 OECD countries for the period between 1997-2012. Environmental patent data in transportation is used as a proxy for the innovation capacity. The independent variables consist of value added, environmental stringency, CO2 emissions and GDP growth. Empirical exercises suggest that innovation in transport has a positive relationship with CO2 emissions and a negative relationship with environmental stringency. The negative impact of environmental regulation on innovation in transport sector is an important insight. This can be associated with excessive adjustment costs of regulation with respect to benefits of improved efficiency by innovation. Furthermore, innovation may be realized in response to rising fuel prices rather than in response to environmental mitigation policies. The positive effect of CO2 emissions may imply that as the CO2 emission caused by transport sector rises, innovation capacity increases through the search for more energy-efficient vehicles. This study contributes to the literature by analyzing the utilization of technology for environment specifically in the field of transport. The analysis can be conducted in a more comprehensive manner including manufacturing sector. The results might provide some important insights for policy makers as well as executives in transportation sector.

Keywords: environmental innovation; environmental stringency; technology ; transport

First recieved: 28.May.2018 Revised: 25.Jun.2018 Accepted: 10.Jul.2018

1 Introduction

Concerns on climate change have caused the adoption of United Nations Framework Convention on Climate Change (UNFCC) in 1992 to avoid damage caused by human activity on the climate system (UNFCC, 1992). With Paris Agreement, countries have become dependent on long term climate change goals of keeping the rise in global temperature under 2 °C above pre-industrial levels and further restricting global warming under 1.5 °C above pre-industrial levels to the end of the century (United Nations, 2016).

Transport sector plays an important role in the economies as it serves as an enabler for international trade by transportation of goods and as a global connector by passenger transportation. In terms of climate change, transport is one of the most important sectors as a source of greenhouse gas emissions. In a report by IEA (2009, p.3), it is stated that 25% of energy-related CO2 emissions are produced by transport sector which holds crucial importance in climate change mitigation. Moreover, technological transition is designated as a requirement for sustainability in companion with policies adopted to utilize these technologies in the report. In this context, transport sector experience diligent attention to achieve such goals due to its fuel dependency and CO2 emission reduction potential (Rogelj et al. 2015). Technological development and related applications is regarded as an area to mitigate climate change on a large scale (Sims et.al.,2014, p.613). The prominence of transport sector for stringent climate change mitigation is also stressed by Zhang et al. (2018), indicating that technological transformation in the sector offers the most remarkable capacity to lower CO2 emissions.

Climate change mitigation in transportation is an area which attract great attention by policymakers, international organizations, governments and researchers. Jolley (2004) asserts that transportation demand rises at a faster pace than income growth in developing countries. Moreover, it is pointed out that technology holds significant potential to compensate the mismatch between exponential increase in transportation demand and relatively fixed environmental capacity in the long run.

This study aims to detect the determinants of climate change mitigation technology development in transport sector and to scrutinize their impacts. The rest of the paper proceeds as follows. In part 2, literature on the relationship between innovation and environmental regulation is discussed and then studies investigating the role of transport sector in climate change mitigation are reported. In part 3, the determinants of innovation in transport sector to mitigate climate change are identified in line with the existing studies and empirical analysis is conducted. Afterwards, the results are discussed in section 4. The last part includes concluding remarks on the contribution of the study and recommendations for upcoming research.

2 Literature Review

Research on the determinants of environmental innovation in specific to transport sector lacks in the literature to the best of our knowledge. So, the literature review starts with presentation of empirical studies which explore the relationship between environmental regulation and technology development. Subsequently, studies on climate change mitigation in the transport sector are outlined.

The role of technology in environmental protection is underlined by Porter and Van Der Linde (1995). They argue that environmentalism and industrial competitiveness are not necessarily opposites, asserting that it can enhance competitiveness with higher productivity. Jaffe and Palmer (1997) find that environmental compliance costs have a positive effect on R&D expenditures, however they find no significant evidence for the relationship between environmental costs and patents for a panel of US manufacturing industries by using Pollution Abatement Control Expenditures (PACE) data as a proxy for stringency of environmental policy. Lanjouw and Mody (1996) show the correlation between environmental regulation and innovation by analyzing patent data and PACE data for the US, Germany and Japan for the period between 1972-1986. They employ all R&D and patent data without eliminating groups which are irrelevant to environment. In order to deduct better insights, they suggest to study chosen industries in focus with disaggregated data. Moreover, Brunnermeier and Cohen (2003) utilize panel data for 146 US manufacturing industries for the span between 1983-1992 to identify determinants of environmental innovation. They report that PACE is positively associated with environmental innovation. Popp (2006) analyzes air pollution control patent data for US, Japan and Germany between 1970-2000. Analysis reveals that environmental regulation has a direct impact on domestic emission control innovation. From a different perspective, Jaffe et al. (2003) draw attention to link between technological development and environmental policy within the context of environmental economics.

Another line of research gathers on connection between transport, urbanization and climate change mitigation. Economic growth and transport are intertwined

and mobilization of products in large distances along with decentralization because of rapid urbanization has increased transport demand further (Kejun, 2010). Dulal et al. (2011) advocate high-density settlement and high-density employment to mitigate climate change. Emission reduction alternatives are examined via scenario analysis in terms of cost and scope for urban transportation in developing countries (Wright and Fulton, 2005). A set of measures with an emphasis on modal shift is proposed as the possible minimum cost policy after comparing fuel technology and policies for modal shift.

Technological innovation is not the solution to reduce emissions from transport merely, but behavioral change to promote modal shift to more environmentally friendly modes and policies to control demand for mobility are other areas to tackle as discussed in IEA (2015) and EEA (2012). In terms of policy development, Colvile et al. (2001) suggest that as technology and transport system gradually become capable of stable decrease of emissions from road transport, policies on air quality will become more rigid. This argument implies a long run cointegration between technology and climate change mitigation. In the same manner, Howey et al. (2010) assess climate change mitigation stringent goals of UK and conclude that dramatic changes which dictate innovative technologies coupled with long run coherent policies are required for UK in fulfilling the CO2 reduction commitment. Likewise, an important mitigation tool for climate change caused by transport sector is stated as advances in vehicle technology in Shaheen and Lipman (2007). These improvements consist of increased utilization of electric vehicles and utilization of alternative energy sources accompanied with the necessary technology to use them. By the same token, Chapman (2007) asserts that improvements in transport technology is essential to address climate change issue in the long run. On the other hand, it is also claimed that policy to promote behavioral change holds critical role in the short run to benefit from the technological developments in a solid way. Jolley (2004) supports this view by stating that emissions caused by transportation can be reduced with a sustainable transportation strategy integrated with developments in transport technologies. Van Der Zwaan et al. (2013) conduct a scenario analysis for the period until 2100 to examine technology diffusions to fulfill a predetermined climate change policy. It is predicated that the dominating vehicle technology along with alternatives can be identified by R&D practices of private sector. As implied by this finding, transformation pathway can be designated by innovation. In a recent empirical work by Beltrán-Esteve and Picazo-Tadeo (2015), environmental performance in the transport sector is studied for 38 countries between 1995-2009. The results show that environmental performance improvement is primarily driven by eco-innovations.

3 Data and Methodology

In the scope of this study, factors which determine environmentally related innovation in transport sector are identified to analyze their relationship with innovative output across time and countries. The analysis is conducted with a panel data covering the period between 1997-2012 with annual frequency for 23 OECD countries. The countries are Australia, Austria, Belgium, Czech Republic, Denmark, Finland, France, Germany, Greece, Hungary, Ireland, Italy, Japan, Korea, Netherlands, Norway, Portugal, Slovak Republic, Spain, Sweden, Switzerland, United Kingdom, United States. Innovative capacity of transport sector in relation to environment has been measured by patents as a dependent variable following Costantini et al. (2017) and Dechezleprêtre et al. (2013). Besides, to capture differences across countries, patent data based on inventor's residence is considered. The number of granted patents at the European Patent Office (EPO) for climate change mitigation technologies related to transportation are retrieved from OECD (2018a).

Explanatory variables set consists of a diverse group of parameters. Firstly, following Jaffe and Palmer (2007), value added is taken as an explanatory variable. In order to take the size of the sector into account, data for share of value added by the transport sector is obtained from OECD (2018b). Environmental regulation is also considered as a determinant in environmental innovation in transport. For this variable, Environmental Policy Stringency Index is taken from OECD (2018c). The index measures the degree of penalty for actions causing pollution or environmental damage within a range between 0 and 6, the former implying zero stringency and the latter referring to highest level. Bearing in mind that transport is one of the primary contributors to emissions as discussed earlier, CO2 emissions from transport sector in relation to GDP data acquired from OECD (2018d) is also included as an independent variable. From a macroeconomic point of view, the integration of transport sector and economic growth suggests the inclusion of a variable about demand. Thus, GDP growth series are taken from OECD (2018e) as it is referred to as one of the determinants of transport demand in Zhang et al. (2018). Descriptive statistics for the panel dataset is presented in Table 1.

Variables	Obs.	Mean	Std. Dev.	Min.	Max.
Patents	368	50	102	0	499
Value Added	368	5	1	3	9
Environmental Stringency	368	2	1	0	4
CO2 Emissions	368	85	42	24	250
GDP Growth	368	2	3	-9	11

Table 1: Descriptive Statistics of Variables

The modeling procedure is to be discussed after a brief description of panel data and explanation on panel regression. To begin with, panel data can be defined as observations for the same subjects (firms, countries etc.) at multiple points in time. Panel data is a rich source to analyze as it has two dimensions: crosssectional units and time. It enables to explain heterogeneity across subjects and dynamic effects that are not obvious in cross sections (Greene, 2010).

In panel estimation, two methods can be utilized: fixed effect and random effect. The fixed effect model controls for unobserved data or omitted variables due to unavailability of data. On the other hand, random effect model assumes all relevant variables are included.

By the structure of our dataset for 23 countries and 16 time periods, a panel regression model of the following form is to be estimated;

$$PAT_{it} = \beta_0 + \beta_1 VALUE_{it} + \beta_2 STR_{it} + \beta_3 CO2 + \beta_4 GROWTH_{it} + u_{it}$$
(1)

where

$$u_{it} = \mu_i + v_{it}$$

In (1), PAT is patents, β_0 is intercept term, VALUE is value added, STR is environmental stringency index, GROWTH stands for GDP growth and CO2 refers to CO2 emissions. In these kind of models, year dummies can be used to control for year specific effects in the data. In that case, the model to be estimated is as follows;

$$PAT_{it} = \beta_0 + \beta_1 VALUE_{it} + \beta_2 STR_{it} + \beta_3 CO2 + \beta_4 GROWTH_{it} + \beta_5 YD_{it} + u_{it}$$
(2)

where

$$u_{it} = \mu_i + v_{it}$$

In (2), YD stands for year dummies. In these models, μ_i are assumed to absorb individual specific effects (Baltagi, 2008, p.14). These two model types will be compared based on Akaike Information Criterion (AIC). AIC developed by Akaike (1973) is a technique to compare different model specifications. In statistical modeling, two common considerations about the predictive capacity are overfitting and under-fitting. An over-fitted model includes unnecessary variables which inflate the variation. An under-fitted model lacks relevant information by omitting related variables, in which case the model fails to capture the true relationship. AIC is based on the view that a model should seize the real relationship between variables without including irrelevant parameters. AIC is calculated by the following equation for each model i:

 $AIC_i = -2MLL_i + 2k_i$ where MLL stands for the maximum log likelihood value and k is the number of estimated parameters.

4 Empirical Findings

Panel modeling is not a straightforward process as the properties of cross-sectional data impose some restrictions. So, we adopt a stepwise approach. Initially, the evaluation of random effect or fixed effect model specification is carried out by Hausman test (Greene, 2010, p.420). This analysis has led to fixed effect modeling. Afterwards, assessment of model specification (1) and (2) reveals that year dummy variables improve AIC in the estimation. The estimation results are presented in Table 2. According to the results, all variables except for value added are significant determinants of innovation capacity building in transport sector. Besides coefficient of determination also known as R2 of 19.28 % indicate that the model can explain almost 20% of the deviation in innovation of transport sector. Moreover, the F-statistic of 4.1 with p-value of 0.00 reveals that the model is significant.

Diagnostic tests for the base model including cross-sectional dependence test developed by Pesaran (2004), modified Wald test for groupwise heteroscedasticity in (Greene, 2010) and Wooldridge autocorrelation test (Wooldridge, 2002) are shown in Table 3. All explanatory variables along with time dummies are included for preliminary analysis with only exception for autocorrelation test as test specification does not allow for time dummies. According to diagnostic test statistics, there is cross-sectional dependence, heteroscedasticity and autocorrelation problems in the model. In this case, main assumptions of Ordinary Least Squares (OLS) estimators are violated and model is not reliable.

Variables	Coefficient	P-value	Year Dummies	Coefficient	P-value
VALUE STR -8.68** CO2 0.19** GROWTH 0.78** INTERCEPT	3.28	0.30	1998	4.20	0.54
		0.02	1999	11.23*	0.10
		0.04	2000	13.40*	0.05
		0.02	2001	15.74**	0.02
	0.94	0.96	2002	19.82**	0.01
			2003	30.13***	0.00
			2004	32.79***	0.00
			2005	41.54***	0.00
			2006	49.63***	0.00
			2007	55.41***	0.00
			2008	49.79***	0.00
			2009	54.80***	0.00
			2010	54.30***	0.00
			2011	53.08***	0.00
			2012	37.91***	0.00

Table 2: Base Model Estimation Results

Notes: *, ** and *** indicate 10%, 5% and 1% levels of significance respectively.

Tuble 5. Blughostie resis for the Buse Model			
Test	Statistics	P-value	
Wooldridge Test	50.98***	0.00	_
Modified Wald Test	5,,375.97***	0.00	
Pesaran's CD Test	13.80***	0.00	

Table 3: Diagnostic Tests for the Base Model

Notes: *, ** and *** indicate 10%, 5% and 1% levels of significance respectively.

In order to handle the aforementioned problems, Driscoll and Kraay regression is employed as it gives heteroscedasticity consistent standard errors and provides robustness to general forms of cross-sectional and temporal dependence (Hoechle, 2007). The results of estimations are displayed in Table 4.

Table 4	: Regression	Results with I	Driscoll-Kraay St	andard Error	S
Model 1			Model 2		
Variables	Coef.	P- value	Variables	Coef.	P- value
VALUE	3.28	0.31	VALUE	-	-
STR	-8.68 *	0.05	STR	-8.94 **	0.03
CO2	0.19 *	0.06	CO2	0.22 *	0.06
GROWTH	0.78	0.45	GROWTH	-	-
INTERCEPT	0.94	0.97	INTERCEPT	19.16	0.16
YD			YD		
1998	0.62 ***	0.00	1998	3.83 ***	0.00
1999	0.63 ***	0.00	1999	11.22 ***	0.00
2000	1.39 ***	0.00	2000	13.42 ***	0.00
2001	1.60 ***	0.00	2001	14.46 ***	0.00
2002	1.99 ***	0.00	2002	18.20 ***	0.00
2003	3.18 ***	0.00	2003	28.92 ***	0.00
2004	3.73 ***	0.00	2004	32.94 ***	0.00
2005	5.24 ***	0.00	2005	41.38 ***	0.00
2006	6.10 ***	0.00	2006	49.98 ***	0.00
2007	5.95 ***	0.00	2007	55.98 ***	0.00
2008	7.81 ***	0.00	2008	48.10 ***	0.00
2009	10.99 ***	0.00	2009	49.24 ***	0.00
2010	8.21 ***	0.00	2010	53.72 ***	0.00
2011	9.05 ***	0.00	2011	51.69 ***	0.00
2012	9.41 ***	0.00	2012	35.33 ***	0.00
R2	19%		R2	19%	
F-test (19,15)	23.92 ***	0.00	F-test (17,15)	23.46 ***	0.00

Notes: *, ** and *** indicate 10%, 5% and 1% levels of significance respectively.

In Table 4, Model 1 is estimated with all independent variables along with time dummies and Model 2 is estimated by eliminating insignificant variables to obtain a parsimonious model. The F-test results of both models indicate overall significance of the variables in explaining innovation in transport. As shown in Table 4, all the year dummies are significant. This result is reasonable as the tech-

nology development along with environmental regulations entail a long period of time. Besides, there is a lag between a technological change and its adaptation. Environmental stringency is a significant factor and has a negative effect on environmental innovation in transportation. This inverse relationship can be explained by an implied and indirect negative impact of environmental policies on firms in reference to the lack of consensus on the directional impact of environmental regulation as stated by Leiter et al. (2011). Companies operating in transport sector may anticipate that cost of adjustment to environmental regulations outweighs the efficiency gains of technology development. Moreover, as innovation process does not give results in short term, firms may be in expectation of looser environmental regulation in the future thereby they don't undertake capital and organizational responsibility by investing in research and development. In addition, innovation might be driven by efforts to remedy dependence of transport sector on petroleum, likely in response to volatility in energy prices which is another implication for focus of transport companies on costs. The positive sign of CO2 emissions might indicate that search for energy-efficient vehicles results in increased technology development when emissions of the sector rise. This plausible insight might indicate that transport sector embraces environmental mitigation after it is realized that the damage is continuous.

5 Conclusion

The role of technology development draws close attention to mitigate environmental damage. As an important source of air pollution (Colvile et al., 2001), innovation of transportation sector is important to control climate change. The objective of this study is to identify the determinants of such innovative capacity and analyze their effects. To do this, a panel regression for environmental patents in transport sector is performed on input variables of environmental stringency, CO2 emissions by transportation, GDP growth and share of value added by the sector for the period between 1997-2012 for 23 OECD countries. The empirical findings suggest that environmental stringency and sectoral CO2 emissions are significant factors determining innovative capacity in transportation. The stringency of environmental regulation has negative correlation with innovation in transportation. This might be due to costs being firms' primary focus rather than environmental mitigation and energy prices acting as a primary catalyst in technology development for transportation. The positive effect of CO2 emissions on innovation suggest that efforts for higher energy efficiency in transport sector end up with innovation as CO2 emissions rise. This study is limited to the analysis of transport sector only, however manufacturing sector can be also considered due to the interaction between one another. Besides the availability of data limits the time span in the analysis, thus time horizon can be extended by utilizing different parameters to proxy environmental stringency.

References

- (EEA), E. E. A. (2010). Econometric analysis. 7th ed. Essex: Prentice Hall.
- (EEA), E. E. A. (2012). The contribution of transport to air quality TERM 2012: Transport indicators tracking progress towards environmental targets in Europe. (Report No: 10/2012). Copenhagen: Rosendahls-Schultz Grafisk.
- (IEA), I. E. A. (2015). "Energy and climate change. World energy outlook special report, 1-200". In:
- Akaike, H. (1973). "Information theory and an extension of the maximum likelihood principle". In: In Second international symposium on information theory [Petrov, B. N. and Csaki, B. F. (eds.)] Pp. 267–281.

Baltagi, B. (2008). Econometric analysis of panel data. John Wiley & Sons.

- Beltrán-Esteve, M. and A. J. Picazo-Tadeo (2015). "Assessing environmental performance trends in the transport industry: eco-innovation or catching-up?" In: *Energy Economics* 51, pp. 570–580.
- Brunnermeier, S. B. and M. A. Cohen (2003). "Determinants of environmental innovation in US manufacturing industries". In: *Journal of environmental economics and management* 45.2, pp. 278–293.
- Chapman, L. (2007). "Transport and climate change: a review". In: *Journal of transport geography* 15.5, pp. 354–367.
- Climate Change, I. P. on (2015). Climate change 2014: mitigation of climate change. Vol. 3. Cambridge University Press.
- Colvile, R., E. Hutchinson, J. Mindell, and R. Warren (2001). "The transport sector as a source of air pollution". In: Atmospheric environment 35.9, pp. 1537–1565.
- Costantini, V., F. Crespi, G. Marin, and E. Paglialunga (2017). "Eco-innovation, sustainable supply chains and environmental performance in European industries". In: *Journal of cleaner* production 155, pp. 141–154.
- Dechezleprêtre, A., M. Glachant, and Y. Méniére (2013). "What drives the international transfer of climate change mitigation technologies? Empirical evidence from patent data". In: Environmental and Resource Economics 54.2, pp. 161–178.
- Dulal, H. B., G. Brodnig, and C. G. Onoriose (2011). "Climate change mitigation in the transport sector through urban planning: A review". In: *Habitat International* 35.3, pp. 494–500.
- FCCC, U. (1992). "United Nations Framework Convention on Climate Change. 1992". In: http: //www.unfccc.de.
- Hoechle, D. et al. (2007). "Robust standard errors for panel regressions with cross-sectional dependence". In: Stata Journal 7.3, p. 281.

- Howey, D. D., D. R. North, and D. R. Martinez-Botas (2010). "Road transport technology and climate change mitigation. Grantham Institute for Climate Change(Briefing Paper No. 2)." In: http:// workspace.imperial.ac.uk/climatechange/Public/pdfs/press/GranthamTransportBriefingPaper. pdf.
- Jaffe, A. B., R. G. Newell, and R. N. Stavins (2003). "Technological change and the environment". In: Handbook of environmental economics. Vol. 1. Elsevier, pp. 461–516.
- Jaffe, A. B. and K. Palmer (1997). "Environmental regulation and innovation: a panel data study". In: *Review of economics and statistics* 79.4, pp. 610–619.
- Jolley, A. (2004). "New Technologies, Industry Developments and Emission Trends in Key Sectors: The Land Transportation Sector Draft Working Paper No. 1". In:
- Kejun, J. (2010). "Mapping Climate Mitigation Technologies and Associated Goods within the Transport Sector". In: ICTSD Programme on Trade and Environment, http://ictsd.org/downloads/ 2010/11/mappingclimate-change-mitigation-technologies-and-associated-goods-within-thetransport-sector-jm.Pdf.
- Lanjouw, J. O. and A. Mody (1996). "Innovation and the international diffusion of environmentally responsive technology". In: *Research Policy* 25.4, pp. 549–571.
- Leiter, A. M., A. Parolini, and H. Winner (2011). "Environmental regulation and investment: Evidence from European industry data". In: *Ecological Economics* 70.4, pp. 759–770.
- Nations, U. (2016). "Paris agreement". In: https://unfccc.int/sites/default/files/english_paris_ agreement.pdf.
- OECD (2018a). "Climate change mitigation technologies related to transportation. Science, Technology and Patents, Patents, Patents in Environmentally Related Technologies". In: http://stats. oecd.org/.
- OECD (2018b). "CO2 emissions from transport in tonnes per one million units of current US GDP Transport, Performance Indicators, Energy and Environment". In: http://stats.oecd.org/.
- OECD (2018c). "Environmental policy stringency. Environment, Environmental Policy". In: http: //stats.oecd.org/.
- OECD (2018d). "GDP growth rate. National Accounts, Annual National Accounts, Main Aggregates, Gross Domestic Product". In: http://stats.oecd.org/.
- OECD (2018e). "Share of value added by the transport sector. Transport, Performance indicators.Economic and Social]". In: http://stats.oecd.org/.
- Pesaran, M. H. (2004). "General diagnostic tests for cross section dependence in panels". In: Working paper (No. 1229), University of Cambridge, Cambridge, United Kingdom.
- Popp, D. (2006). "International innovation and diffusion of air pollution control technologies: the effects of NOX and SO2 regulation in the US, Japan, and Germany". In: *Journal of Environmental Economics and Management* 51.1, pp. 46–71.
- Porter, M. E. and C. Van der Linde (1995). "Toward a new conception of the environment competitiveness relationship". In: *Journal of economic perspectives* 9.4, pp. 97–118.
- Rogelj, J., G. Luderer, R. C. Pietzcker, E. Kriegler, M. Schaeffer, V. Krey, and K. Riahi (2015). "Energy system transformations for limiting end-of-century warming to below 1.5 C". In: *Nature Climate Change* 5.6, p. 519.
- Shaheen, S. A. and T. E. Lipman (2007). "Reducing greenhouse emissions and fuel consumption: Sustainable approaches for surface transportation". In: *IATSS research* 31.1, pp. 6–20.
- Wooldridge, J. M. (2010). Econometric analysis of cross section and panel data. MIT press.
- Wright, L. and L. Fulton (2005). "Climate change mitigation and transport in developing nations". In: *Transport Reviews* 25.6, pp. 691–717.

Environmental Innovation of Transportation Sector in OECD Countries

- Zhang, R., S. Fujimori, and T. Hanaoka (2018). "The contribution of transport policies to the mitigation potential and cost of 2 C and 1.5 C goals". In: *Environmental Research Letters* 13.5, p. 054008.
- Zwaan, B. C. van der, H. Rosler, T. Kober, T. Aboumahboub, K. V. Calvin, D. Gernaat, G. Marangoni, and D. McCollum (2013). "A cross-model comparison of global long-term technology diffusion under a 2 C climate change control target". In: *Climate Change Economics* 4.04, p. 1340013.
Design of an Added Plan with Social Responsibility

Kathya Alexandra Suesca Rozo¹, Andrés Felipe Santos Hernández¹

1 - Escuela Colombiana de Ingeniería "Julio Garavito"

For decades and worldwide, steel has been one of the most important commodity in the development of the industry due to its multiple uses. At present, the offer of steel in Colombia is too low compared to the sector's demand. This has caused market speculation and a low added value regarding all products made with steel. In such a competitive market, the organizations involved have been developing strategies to improve their incomes, some of these are developed in the operations area, where the planning is the most important part regarding the cost reduction, minimize the rate of rotation of the personal in the operations area and increment in the effectivity of the supplies chain. Having this need and tendency created, the investigation group designed an added planning model, based on the case of an enterprise that transforms steel in cloves. Using this model in the enterprise will guarantee decrease the shortage in posterior periods and as consequence lower the costs and wastes caused by shortfalls, the costs of keeping an inventory and costs due to high lead time to market. Furthermore, the model will minimize the rate of rotation of the personal in the operations area, this will result encouraging social responsibility which contribute to the organization's productivity.

Keywords: Steel transformation; Added planning; Optimization; Social responsibility

First recieved: 30.May.2018 Revised: 09.Jul.2018 Accepted: 16.Jul.2018

1 Introduction

Since the start of the 50's, steel in Colombia has become a resource quite relevant regarding development of the industry and the country's infrastructure. 500.000 tons of steel are expected to be the demand in the projects of infrastructure, such as the routes 4G (El Espectador, 2017). Nowadays, according to Camacero (Cámara Colombiana del Acero), Colombia produces 1.2 million tons of steel and consumes 3.6 million tons, this shows a shortfall in the primary supply chain of the sector. Therefore, the producers and marketers must import around 2.4 million tons per year, so the demand can be satisfied (Portafolio, 2017).

Production of this sector represents 7% of the national industrial GDP and even though it creates more than 40.000 jobs, it still shows problems in reducing the production and logistics costs. This is a consequence of the lack of knowledge regarding the practices in the supply chain and exterior facts, such as the increment of imports in lower prices (Dinero, 2017).

Keeping in mind the late technological and industrial tendencies in the world, China has shown to be the asiatic country with the highest production of steel, with around 800 tons per year, provides about 50% of the global demand (El nuevo siglo, 2016). Their companies are owned by the state, so they are highly subsidized and are not governed by market rules (Alacero, 2017). For this reason, their prices are lower than the market, which is known as dumping (El español, 2016).

This has generated an oversupply of this product and a drop in international prices (Portafolio, 2017). Not to mention the technical stoppages and closures of steel plants that the overproduction of this asiatic giant has generated (Alacero, 2017).

Consequently, the Chinese government has hardworking to close those companies that produce low quality steel made of scrap. Same, with the objective to reduce its production capacity by more than 50 million tons, increment their prices and minimize environmental impacts (Infoacero, 2018). In addition, these decisions are part of the government's goal of changing its economic model, with less dependence of the industry and more focalized on new technologies (Angencia EFE, 2018).

Regarding this inconvenient, the sector must know that it will continue to fight with the price against this rival, since the asiatic country will continue to produce steel at very competitive prices. For this reason, the national companies of sector should implement best practices throughout the supply chain, specifically in the primary chain and the planning of its production, to guarantee the supply of this product at a competitive cost.

Given the fact that this product is a commodity, the enterprises involved with this supply chains have had to deal with the prices, and therefore the organizations are concerned and have worked slowly in reducing the operational costs with the purpose of improving their opportunities and therefore their oncomes.

Even though detecting the wasting and overspends along the process is relevant (Nave, 2002), this practice does not reduce the costs significantly in comparison to an adequate planning in its supply chain. As Chopra (2008) says, an adequate plan will guarantee the focalization of the resources along the chain so that an emerging market can be handled, even more if there are high lead time to market, uncontrolled product portfolios, over-stock, low service quality, lost sales, and unsatisfied customers.

However, the arrival of enterprises such as Gerdau, Votorantim and Techint has improved the sectors know how and its technology level. This, marked by the demand of competitiveness, requires that the customer service and the attention to the market demand an expansion in the investments of the information systems and adequate planning in the supply chain (Munar, 2008).

By identifying this need in the sector, the research group has built a model of added planning, through the application of linear programming, for an enterprise located in Bogotá (Colombia) that works in the transformation of steel in construction products, such as rod, mesh, cloves, picks, and other products. The model's aim is to lower the costs of the planning in the supply chain, the costs are the inversions made with the expectations of getting benefits in the future, but also in the present. How we try to impact in the enterprise is to generate benefits along the whole chain so that we can accomplish a balance between the demand and the offer of cloves.

To meet this end, a model has been developed so that the effectivity and the productivity of the resources is improved. A model like this is a representation of reality using linear mathematic expressions, in which we can know with certainty the parameters and by giving it a solution, the model proposes which should be the decisions to make in one or more time periods (Bohórquez Quiroga, Sarmiento Lepesqueur, & Jaimes Suárez, 2013).

2 Added Planning Models

There are many methods of added planning which can be divided in three groups according to Boiteux, these are: comparative alternatives, use of decision rules and based in models of linear programming (Boiteux, 2007). The linear programming models are the most used nowadays.

There are some authors that have developed applications, in this sector, there is Cañas (2013), who planned the production of an industry that belong at the metallurgical industry so that he could maximize utilities by introducing multiphase, multi period, multiproduct models of linear programming and by the development of a constructive heuristic based on the principles of the restrictions theory. Chavez (2014), designed a model of linear programing with the purpose of resolving a problem of added planning in production for Huachipato metallurgical company, considering the restrictions concerning inventory, demand, and capacity. Zurito (2010), made a model of linear programming to minimize the lacks in the inventory and therefore the associated costs. Gholamian & Tavakkoli & Nezam (2015), developed an entire mixed model of linear programming with service quality, minimize fluctuations in the rates of change of the work force and maximize sales.

3 Case Study

3.1 Company and Process Background

The enterprise to be analyzed is part of the labor union Camacero (Cámara colombiana del acero), which was created in 2014 with the objective of gathering more than 40 national enterprises dedicated to the production, transformation, and commercializing of steel. This with the purpose of generating solutions in the value chain (El Espectador, 2014). Nowadays, the labor union counts with more than 400 producers registered.

In Latin-American, Colombia is stablished as the fourth steel producer after Brazil, México, and Argentina (Alacero, 2017). Nevertheless, Colombia must work to increase its production and improve its competitiveness. According to the ANDI (Asociación nacional de empresarios de Colombia) between 2007 and 2016 the sector grew 13% (Andi, 2018), despite the hard situation that it has been facing.

Given this need and considering the shortage of raw material, the strong competence with foreign enterprises, a model of added planning is proposed so that it determines the ideal production, capacity inventory, stock-out and out-sourcing levels required to satisfy the cloves demand, the product that was chosen to this case of study (Sunil Chopra, 2008). Nevertheless, if the organizations in addition to making a correct planning would add practices of social responsibility, they would assure not only the decrease of costs, but also will assure an improve in their competitiveness.

It is also necessary to identify the stage in which the workers are about the learning curve, this means to know if they are beginners, medium or advanced. If enterprises reduce the rotation of their advanced personnel, they could assure not only quality but also the flux in production. We know this because the workers with more experience have a more efficient handling of the resources and guarantee the improvement of the production.

3.2 The Process

The fabrication process of the 3"x 24 -kilogram galvanized clove has 5 steps. The raw material for this product is 5-milligram SAE 1023¹ steel which comes in approximately 2-metric ton rolls. The supplying of this material is made with forklifts, leaving 2 rolls per round in two reelings. These feed two drawing machines, the koch 1 and koch 2. These machines are automatic, the operator handling is needed during the time of charge and discharge. The drawing is the plastic conformation of a material through tensile continuous stresses, through rows, it is possible it's thinning (Boca, 2017). In this case, a diameter of 5.5-milligram. This wire is deposited in colls of up to 900-kilogram, which are transported by the forklift to the area of hot galvanizing. On each route, the vehicle can carry up to four coils. In the hot galvanizing line, usually 2 lines are programmed for this type of product.

¹It is a quality of low carbon steel that can be used in cold. It has high tenacity and low mechanical resistance.



The entire line has 50 threads, but only two lines are programmed for this type of clove; these coils are unrolled at a slow pace, going through a process of hot galvanizing, which is a surface treatment in a zinc vat at high temperatures that allows the wire to be protected from corrosion, so that later, the galvanized wire is winded at the end of the line, also, with up to 900-kilogram (Berto, 2017). This material is taken by the forklift to the cloves ship, where 4 wafios equipment are programmed. These machines turn the wire into cloves, temporarily storing them in baskets of up to 300-kilogram. Using a bridge crane, in the same ship, the baskets are taken to the polishing area, where 4 drums that turn with sawdust. clean, and polish the clove. These polished cloves are deposited again in the same baskets and loaded by the bridge crane to the packing section. Generally, 4 packers are scheduled for the packing of this material. Each packer manually weighs about 600-grams with a tolerance of 10-grams. In each pack, and each one of these, the packer deposits the clove in a cardboard box of 24-kilogram. As they finish each carton they stack it in a stowage, until completing 54 cartons. The operator that is completing the stacking, packs the stowage with stretch film².

²Vinipel, translucid plastic film with a high resistence.

Operation	Equipment	Production speed (ton/hour)	Efficiency (%)	Real production (ton/hour)
Drawing	Koch 1	2.1	65	1.4
	Koch 2	2.5	57	1.4
Galvanizing	2 lines	3.5	63	2.2
Dicing	Wafios 1	1.7	72	1.2
-	Wafios 2	1.3	67	0.9
	Wafios 3	1.5	70	1
	Wafios 4	1.2	55	0.7
Polishing	drum 1	0.7	85	0.6
	drum 2	0.8	92	0.7
	drum 3	0.7	80	0.6
	drum 4	0.6	89	0.5
Packing	packer 1	0.024	78	0.008
	packer 2	0.024	82	0.008
	packer 3	0.024	72	0.007
	packer 4	0.024	86	0.009

Table 1: Statistics of the process of the 3"x 24-kilogram galvanized clove. May 2017.

4 Methodology

4.1 Data Gathering

The product that was chosen to carry out this research was the galvanized 3"x 24-kilogram clove, which was selected through an ABC analysis, which is a tool used to classify products into, high, medium and low turnover categories, also known as pareto's 80:20 law, which indicates that 80% of the money from the sales of a company is generated by 20% of the existing products in its portfolio, so that this product has an A classification (Bravo, 1996).

4.2 Model Formulation

In order to develop the added planning model, the planning horizon of six months and the demand to be supplied are identified. Based on Table 2, the demand of the 3"x 24-kilogram clove is shown based on a historical data given by the organization. Furthermore, Table 3 shows the costs associated with the aggregated plan.

In order to calculate the kilograms per standard hour produced by each of the drawing, galvanizing, dicing and polishing machines, the speed of production and efficiency for each machine are considered.

Then, to calculate the kilograms per standard hour produced by each of the drawing, galvanizing, dicing and polishing machines, the speed of production and efficiency for each machine are considered.

Demand unit (kilogram)
16500
13800
13100
18450
18465
22350

Table 2: 3"	x 24-	kilogram	cloves	demand
10010 2.0	~ ~ .	into gi ann	000000	actitutie

ltem	Price	Unit
Material	11	€/kilogram
Inventory maintenance cost	1.4	€ /kilogram/month
Stock-out cost	1.8	€/kilogram/month
Hiring cost	225	€/worker
Dismissal cost	315	€ /worker
Packing MOD cost	8.3	€/hour
Extra hours cost	14.3	€/hour
Drawing cost	19.5	€ /hour
Galvanizing cost	75	€/hour
Dicing cost	33	€ /hour
Polish cost	12.6	€/hour
Outsourcing cost	60	€ /kilogram

Table 3: Costs associated with the added plan. May 2017.

Drawing operation: It has two drawing equipment, the koch1 and the koch 2.

$$Drawing machine = \left(2100 \frac{kilogram}{hour} * 65\%\right) + \left(2500 \frac{kilogram}{hour} * 57\%\right) = 2790 \frac{kilogram}{hour}$$
(1)

Galvanizing operation: It has one machine

$$Galvanizing machine = \left(3500 \frac{kilogram}{hour} * 63\%\right) = 2205 \frac{kilogram}{hour}$$
(2)

180

Dicing operation: It has four machines (Wafios).

$$\begin{aligned} Dicing machine &= \left(1700 \frac{kilogram}{hour} * 72\%\right) + \left(1300 \frac{kilogram}{hour} * 67\%\right) + \\ &\left(1500 \frac{kilogram}{hour} * 70\%\right) + \left(1200 \frac{kilogram}{hour} * 55\%\right) \\ &= 3805 \frac{kilogram}{hour} \end{aligned}$$

Polish operation: It has four machines

$$Polish \ operation = \left(700 \frac{kilogram}{hour} * 85\%\right) + \left(800 \frac{kilogram}{hour} * 92\%\right) + \left(700 \frac{kilogram}{hour} * 80\%\right) + \left(600 \frac{kilogram}{hour} * 89\%\right) = 2425 \frac{kilogram}{hour}$$

Based on the kilogram per standard hour of each machine, the cost per hour is calculated. Table 4 shows the costs associated with the drawing, galvanizing, dicing and polishing operations. The kilogram per hour produced by the packing operation depends on the efficiency of each worker, because this operation, unlike the drawing, galvanizing, dicing and polishing operations, uses labor for its operation. The efficiency of each packer is determined by the seniority. The classification is shown in Table 5.

In order to calculate the labor of packing, the following assumptions are considered: 1 shift has 7.5 hours a day and 1 month has 25 days.

Beginner packer labor per hour

$$Beginner = \frac{\left(54\frac{boxes}{turn} * \frac{24}{1}\frac{kilogram}{box} * \frac{1}{7.5}\frac{turn}{hours} * 55\%\right)}{4} = 24\frac{kilogram}{hour}$$
(5)

Operation cost	Price	Unit	
Drawing cost	0.01	€/kilogram	
Galvanizing cost	0.03	€/kilogram	
Dicing cost	0.01	€ /kilogram	
Polish cost	0.01	€/kilogram	

Table 4: \$ per-kilogram costs. Source: the authors

Table 5: Personal classification in the Packing operation

Туре	Efficiency	Seniority
Beginner Medium Advanced	55% 65%	Less than 3 months Between 3 and 4 More than 4 months
Advanced	85%	More than 4 months

Medium packer labor per hour

$$Medium = \frac{\left(54\frac{boxes}{turn} * \frac{24}{1}\frac{kilogram}{box} * \frac{1}{7.5}\frac{turn}{hours} * 65\%\right)}{4} = 28\frac{kilogram}{hour}$$
(6)

Advanced packer labor per hour

$$Advanced = \frac{\left(54\frac{boxes}{turn} * \frac{24}{1}\frac{kilogram}{box} * \frac{1}{7.5}\frac{turn}{hours} * 85\%\right)}{4} = 37\frac{kilogram}{hour}$$
(7)

Furthermore, based on the amount of days we can have an inventory of the product in the process of drawing, dicing, and polishing. The used inventory policies are shown in Table 6.

The first step to build a model of added planning is to identify the decision that the enterprise can make (Sunil Chopra, 2008).

Drawing (Kilogram)	Galvanizing (Kilogram)	Dicing (Kilogram)	Polish (Kilogram)	Final Product Inventory (Kilogram)	Stockout (Kilogram)
10 days	4 days	6 days	8 days	15 days	Policy
5500	2200	3300	4400	8250	250
4600	1840	2760	3680	6900	250
4367	1747	2620	3493	6550	250
6150	2460	3690	4920	9225	250
6155	2462	3693	4924	9233	250
6548	2619	3929	5239	9823	250
6856	2742	4114	5485	10284	250

Table 6: Inventory policy

4.2.1 Decision Variables

Be t= 1, ... ,6 the horizon time to analyze, Where 1=June, 2= July, ..., 6 = November.

The next step in the construction of the model is establishing the objective functions.

Variable	Description
B _t	Number of employee's beginner for packing at the beginning of month t.
Md_t	Number of employee's medium for packing at the beginning of month t.
A_t	Number of employee's advanced for packing at the beginning of month t.
D_t	Number of employees dismissed for packing at the beginning of month t.
C_t	Number of employees hired for packing at the beginning of month t.
F_t	Amount of kilogram produced in drawing in month t.
P_t	Amount of kilogram produced in dicing in month t.
G_t	Amount of kilogram produced in galvanizing in month t.
U_t	Amount of kilogram produced in polishing in month t.
M_t	Amount of kilogram of finished product in packing in month t.
EB_t	Number of extra hours of beginner hired for packing at the beginning of month t.
EMd_t	Number of extra hours of medium hired for packing at the beginning of month t.
EA_t	Number of extra hours of advanced hired for packing at the beginning of month t.
IF_t	Inventory of drawing at the end of month t.
IG_t	Inventory of galvanizing at the end of month t.
IP_t	Inventory of dicing at the end of month t.
IU_t	Inventory of polishing at the end of month t.
IM_t	Inventory of finish product at the end of month t.
S_t	Number of pending units in month t.
SC+	Number of subcontracted units in month t.

Table 7: Decisions can be taken

4.3 Objective function:

Minimize the total cost of the organization during the time of the planning, in this case its 6 months.

$$\begin{split} \operatorname{Min} \mathsf{Z} &= \sum_{t=1}^{6} 225 * C_t + \sum_{t=1}^{6} 315 * D_t + \sum_{t=1}^{6} 1532 * B_t + \sum_{t=1}^{6} 1532 * M_t + \\ &\sum_{t=1}^{6} 1532 * A_t + \sum_{t=1}^{6} 14.3 * (EB_t + EM_t + EA_t) + \\ &\sum_{t=1}^{6} 0.01 * F_t + \sum_{t=1}^{6} 0.03 * G_t + \sum_{t=1}^{6} 0.01 * P_t + \sum_{t=1}^{6} 0.01 * U_t + \\ &\sum_{t=1}^{6} 1.4 * IM_t + \sum_{t=1}^{6} 1.8 * S_t + \sum_{t=1}^{6} 60 * SC_t + \sum_{t=1}^{6} 11 * F_t \end{split}$$
(8)

4.4 Subject a:

The size of the work force: the size of the work force, which is the total of the hired workers in the month t, considering the actual workers (beginner, medium and advanced), but not the dismissed ones.

$$B_t + Md_t + A_t = B_{t-1} + Md_{t-1} + A_{t-1} + C_t - D_t$$
(9)

Extra hour: a worker (beginner, medium and advanced) can work until 10 extra hours in the month t (Gerencie, 2017).

$$EB_t \le 10 * B_t \tag{10}$$

$$EMd_t \le 10 * Md_t \tag{11}$$

185

$$EA_t \le 10 * A_t \tag{12}$$

Packing capacity: the amount of kilogram cloves produced should not be above the available capacity of the packer (beginner, medium and advanced).

$$\left(1 turn * \frac{7.5 hours}{1 turn} * \frac{25 days}{1 month} * \frac{24 - kilogram}{hour}\right) * B_t + \left(\frac{24 - kilogram}{hour}\right) * EB_t \ge M_t$$
(13)

$$\left(1 turn * \frac{7.5 hours}{1 turn} * \frac{25 days}{1 month} * \frac{28 - kilogram}{hour}\right) * Md_t + \left(\frac{28 - kilogram}{hour}\right) * EMd_t \ge M_t$$

$$(14)$$

$$\left(1 turn * \frac{7.5 hours}{1 turn} * \frac{25 days}{1 month} * \frac{37 - kilogram}{hour}\right) * A_t + \left(\frac{37 - kilogram}{hour}\right) * EA_t \ge M_t$$
(15)

Machine capacity: the production capacity of the machines must be greater than the production required in the drawing, galvanizing, dicing and polishing operations.

$$\left(1 turn * \frac{7.5 hours}{1 turn} * \frac{25 days}{1 month} * \frac{2790 - kilogram}{hour}\right) \ge F_t \qquad (16)$$

$$\left(1 turn * \frac{7.5 hours}{1 turn} * \frac{25 days}{1 month} * \frac{2205 - kilogram}{hour}\right) \ge G_t \qquad (17)$$

$$\left(1 turn * \frac{7.5 hours}{1 turn} * \frac{25 days}{1 month} * \frac{3805 - kilogram}{hour}\right) \ge P_t \qquad (18)$$

186

$$\left(1 turn * \frac{7.5 hours}{1 turn} * \frac{25 days}{1 month} * \frac{2425 - kilogram}{hour}\right) \ge U_t \qquad (19)$$

Inventory: the inventory in each operation must be equal to the inventory in the immediately previous period, plus the production carried out in that period, minus the production required in the next season, minus the inventory that should remain in the operation; to guarantee the flow of production.

$$IM_{t} = IM_{t-1} + M_{t} + SC_{t} - S_{t+1} + S_{t} - demand(t)$$
⁽²⁰⁾

$$IF_t = IF_{t-1} + F_t - G_t \tag{21}$$

$$IG_t = IG_{t-1} + G_t - P_t \tag{22}$$

$$IP_t = IP_{t-1} + P_t - U_t (23)$$

$$IU_t = IU_{t-1} + U_t - M_t$$
(24)

Constrain positive: ensure that the variables take positive values.

$$B_{t}, Md_{t}, A_{t}, C_{t}, D_{t}, EB_{t}, EMd_{t}, EA_{t}, F_{t}, G_{t}, P_{t}, U_{t}, M_{t}, IF_{t}, IG_{t}, IP_{t}, IU_{t}, IM_{t}, S_{t}, SC_{t} \ge 0$$
(25)

On the other hand, it must be guaranteed that the out-sourced units do not exceed 250 units and that the inventory policies established are equal to the inventory of each operation in each month.

5 Results

The total cost with the proposed model is 1.51 M €. The Tables 8, 9, 10, 11, 12, and 13 shown the results. This Table shows the production that is required in each operation, drawing, galvanizing, dicing, polishing and final product to satisfy the demand of claves.

These tables show the inventory that each operation should have according to the corresponding inventory policies. As can be observed, the only month with shortage in units is November.

Month	Production drawing (kilogram)	Production galvanizing (kilogram)	Production dicing (kilogram)	Production polishing (kilogram)	Final product (kilogram)
June	38600	32000	33100	29800	25400
July	10290	11370	11190	11730	12450
August	12190	12470	12423	12563	12750
September	25405	23265	23622	22552	21125
October	18485	18479	18480	18477	18473
November	20929	20457	20536	20300	19985

Table 8: Production during the planning horizon (2017)

Table 9: Inventory WIP (2017)

Month	Drawing inventory (kilogram)	Galvanizing inventory (kilogram)	Dicing inventory (kilogram)	Polishing inventory (kilogram)	Final product inventory (kilogram)
June	5500	2200	3300	4400	8250
July	4600	1840	2760	3680	6900
August	4367	1747	2620	3493	6550
September	6150	2460	3690	4920	9225
October	6155	2462	3693	4924	9233
November	6548	2619	3929	5239	9823

Table 10: Policies o	Stock-out and	Outsourcing
----------------------	---------------	-------------

Month	Stock-out (kilogram)
November	250

Table 11: Work force required for each period

Month	Worker Force	
June	15	
July	8	
August	8	
September	12	
October	11	
November	12	

The number of workers after the month of August is regular. The distribution according to the time horizon is shown in Table 12.

Month	Hiring cost	Dismissal cost	Beginner	Medium	Advanced
December					4
June	11		6	5	4
July		7	3	3	2
August			3	3	2
September	4		5	4	3
October		1	4	4	3
November	1		5	4	3

Table 12: Distribution of work force

Table 13 shows distribution of costs according to the type and time period such as the total costs of hiring, dismissal and extra hours.

Month	June	July	August	September	October	November
Hiring cost	2475			006		225
Dismissal cost		2205			315	
Beginner	9281	4641	4641	7734	6187	7734
Medium	7734	4641	4641	6187	6187	6187
Advanced	6187	3094	3094	4641	4641	4641
Extra hours cost				215	391	
Drawing cost	270	72	85	178	129	146
Galvanizing cost	1088	387	424	791	629	969
Dicing cost	287	97	108	205	160	178
Polish cost	155	61	65	117	96	105
Inventory cost	11137	9315	8842	12453	12464	13260
Stock-out cost						450
Outsourcing cost						
Material cost	405300	108045	127995	266753	194087	219755

costs
Total
13:
Table

The kilograms of the final product that the organization should produce to satisfy the demand are 110183 kilograms with the raw material unit cost estimated as 11 €, which gives a total cost of 1.2 M€ and a total cost of planning of 1.5 M€. The cost of buying raw material represents 77 % of this cost. This shows a shortage in the supply chain of this sector.

6 Improvements

Aggregate planning plays a fundamental role in the supply chain, because it allows any organization to determine the quantity of production, inventory, outsourced units, minimum of inventory shortage and capacity that it must have in a given period, to satisfy the demand. In turn, it will guarantee the reduction of costs and improve the level of service.

On the other hand, performing an aggregate planning model, considering the level of learning of workers, i.e., if the worker is beginner, medium or advanced, allows the organization to classify workers according to their efficiency in each period. In this way, a fair wage could be guaranteed for them.

Whit the proposed model, the organization may:

- Determine the optimal level of inventory that this should have, to ensure a good level of service and incur a low cost to maintain inventory.
- Identify the cost in which the organization incurs to keep the three kinds of workers. (beginners, medium and advanced)
- Verify if it complies whit the established inventory policies of the organization.
- Include social responsibility issues to ensure a fair wage for their employees which will depends on their efficiency.
- Reduce the inventory shortage and their associated costs.
- Minimize the operations personnel turnover.

- Manage the variability of delivery time and demand.

7 Relevance in the Sector

The combination of aggregate planning with social responsibility is essential if organizations seek to increase their productivity and consequently their competitiveness. Each one fulfills a fundamental role, in the first case it seeks to make the best decisions at the aggregate level, and in the second case, to ensure a focus on the organization's human capital, so that a fair payment is made.

One of the decisions that any organization should take when planning is to decide how many workers to hire and how many to dismiss, if the organization considers the efficiency of its workers and plans based on this, it will be able to reduce the rotation of the Operations personnel, because their workforce will be enough to supply all the demand. Consequently, their workforce would be motivated, and the organization would not have to incur costs for fired.

On the other hand, being steel is one of the scarcest raw materials at the national level and with greater use, its cost of acquisition is high compared to international prices, for this reason the organizations immersed in the sector must contend at price level.

8 Conclusions

The aggregate planning being more than a mathematical model, is a tactic tool that enables the determination of the inventory, production, stock-out levels as well as the workers that the organization needs to satisfy the demand in the corresponding period of time. In addition, it guarantees the optimization of cost, time and resources.

The planning guarantees an organization a high level of service, reduced costs and controls the variability of demand. However, the planning should consider the social responsibility in order to ensure fair wages and level of workforce in all working periods.

Finally, the organization should considers its operations within the whole supply chain. This includes the added value from the side of raw material to the final

product in order to decrease the lead time to market along with the associated costs and waste caused by shortages and inventory.

References

Alacero (2017a). Latin America in figures 2017.

- Alacero (2017b). Steel subsidized from China puts at risk the employment of thousands of Latin Americans. We call on our governments to act urgently and ensure fair competition.
- Alacero (2017c). Uses of Steel.

Andi (2018). Colombian Committee of steel producers.

Berto, R. and M. Ayatollai (2017). "Mechanical behavior of hot-dip galvanized welded steel under".

Bohorquez Quiroga C., S. L. A. J. S. (2013). Mathematical model.

Boiteux, O. D. (n.d.). "State of the art on aggregate production planning". PhD thesis.

Bravo, J. (1996). Purchases and inventories.

Camacero (2014). "The guild of steel is born, Camacero". In:

Camacero (2017). "Economic information". In:

Cañas Catañeda, J. S. (2013). "Planning of the production applying linear programming models and theory of restrictions for an industry in the metalworking sector".

Chase, R., R. Jacobs, and N. Aquilano (2009). Production operations management and supply chain. Chopra, S. (2008). Management of the supply chain Strategy, Planning and Operation.

- Clifford, F. A. and H. Reinoso (2014). "Formulation of a programming problem for the aggregate planning of steel company Huachipato".
- Dinero (2017a). Despite dumping and low demand, the steel industry does not stop investing.
- Dinero (2017b). Improve prospects for the steel industry How is it and where is it going?

Donald J. Bowersox, David J. Closs, and M. Bixby Cooper (2007). Administracion y logistica en la cadena de suministros, p. 426.

El español (2016). Why China is a growing threat to European steel.

El Espectador (2017). An industry that stomps.

El Nuevo Siglo (2016). China threatens steel sector in Colombia.

F, A. (2018). China will continue to cut capacity for steel and coal production in 2018.

Gerencie (2017). It is not legal to work more than 10 hours a day.

Gestión (2017). China, the world leader in steel and aluminum accused of "dumping" by its partners.

Herrera, M. (2018). Another year of high prices.

Infoacero (2017a). How is the panorama of the steel industry going?

Infoacero (2017b). The Academy and its Importance to Promote the Use of Steel.

Mahdavi-Amiri, N. G. I. M. R., M. Tavakkoli, and M.-A. Nezam (n.d.). "A Comprehensive fuzzy multiobjective multi-product multi-site aggregate production planning decisions in a supply chain under uncertainty".

Munar, X. V. (2008). Strategic direction for the steel industry in Colombia for the year 2020.

pais, E. (2016). China, the great steel super producer.

Portafolio (2008). Why does China sell so cheap?

Portafolio (2014). In Colombia, production is much lower than demand.

Portafolio (2017). Colombian steel, between China and local demand.

Quality, A. S. for (2002). "How To Compare Six Sigma, Lean and the Theory of Constraints". In: American Society for Quality.

República, L. (2017). Contraband and low prices in China, the main problems of the steel industry.

Sas-Boca, I. M., M. Tintelecan, M. Pop, D.-A. Iluţiu-Varvara, and A. M. Mihu (2017). "The wire drawing process simulation and the optimization of geometry dies". In: *Procedia Engineering* 181, pp. 187–192.

Vargas, L. D. (2016). Steel and metalworking: a sector of importance for the Colombian industry. Zurito Olea, C. A. (n.d.). "Production planning at the Gerdau Aza steel company. S.A".

Part III

Supply Chain Analytics

Multi-Method Decision Support Framework for Supply Network Design

Giuseppe Timperio¹, Robert de Souza¹, Boy Panjaitan Bernado², Sumit Sakhuja¹, Yoseph Sunardhi ²

- 1 National University of Singapore
- 2 PT Pos Indonesia

In a business context characterized by increased competition due to digital transformation and technological breakthroughs, massive penetration of online purchasing, greater customers' expectations and switch of manufacturing paradigms, the role of logistics has become today more critical to firms than ever before. One critical initiative that logistics providers can undertake in order to gain a truly competitive edge is to optimally seize and utilize key resources such as storage and transportation assets. This paper is an effort in that direction and proposes a decision support framework to design a cost effective supply network in the last leg of deliveries. A case study based approach about a postal service provider in Southeast Asia is presented to showcase the applicability of the proposed framework. By leveraging on the integration of data analytics, network optimization and simulation, this work highlights the advantages of adopting a holistic approach to decision making for the used case. Results show that number of storage facilities, and their locations, affect speed and cost-effectiveness of last mile distribution. For the case at hand, 18% of savings in transportation and warehousing cost with no impact on service level can be achieved by reducing the number of facilities in the network from 9 to 4. By reading the present paper, decision makers will gain insights on how to address challenges related with supply network design, transportation costs reduction and optimization of overstretched transportation routes.

Keywords: Network Design; Network Optimization; Last Mile Logistics; Transportation Optimization

First recieved: 28.May.2018 Revised: 01.Jun.2018 Accepted: 22.Jun.2018

1 Introduction

Over the past few years, the attention given by academics and practitioners to supply chain management and logistics issues is increased tremendously (Yazdani et al., 2017). Logistics encompasses several business activities, which mostly relate with planning, organization, and control of the material flow (direct and reverse) from raw material at manufacturers to final points of consumption (Jonsson & Mattsson, 2005). With the surge in online purchasing, the role of logistics has become today more critical than ever before, with the sector registering a spike in the level of competition accompanied by a severe increase in usage of Third Party Logistics Service Providers (LSP). Between 2010 and 2016, the LSP industry has registered a steady yearly growth in revenue, reaching the global amount of 802.2 billion U.S. dollars in 2016 (+64% compared to 2010) (Statista, 2018 a). Worldwide, the market size for LSP in Asia Pacific is the largest. In 2016, the regional market was seized at approximately 305 billion U.S. dollars 2016 (38% of total), followed by North America (25% of total), and Europe (21% of total) (Statista, 2018 b). Constant regional economic growth and rapid surge of middle-class, demographic profile of consumers and businesses, advantageous trading conditions, and recent crystallization of the ASEAN Economic Community (AEC) make in fact Asia Pacific region particularly appealing to global freight forwarders and related firms (Spire Research and Consulting, 2016). Moreover, elements such as low cost for outsourcing, supply chain inefficiencies, and increased demand for LSPs, creates even more favorable conditions for the further increase in the presence of logistics players in near future (Spire Research and Consulting, 2016). Additionally, the gradual liberalization of regional logistics services is slowly increasing competitiveness and foreign participation, thereby introducing more logistics players into the market (Logistics, Insights Asia, 2016). With competition shifting from firm-to-firm to supply chain against supply chain (Christopher & Towil, 2001) there is clear need for LSP to rethink about supply chain and logistics management strategies so as to reduce operational costs and increase competitiveness (Tartavulea & Radu, 2013).

Thus, for a local LSP that wants to operate profitably in such a competitive context, there is a burning need of more streamlined logistical processes so as to gain concurrent cost effectiveness and superior service level. One such important initiative that LSP can undertake in order to gain competitiveness is to optimally allocate and utilize their logistics assets such as storage and transportation. Particularly, practical experience and academic research suggest that number of storage facilities, and their locations, greatly affect both speed and cost-effectiveness of logistics operations (Zhang & Xu, 2014). Therefore, a seamless and effective navigation of storage assets through an optimum design of supply network (SN) would be highly beneficial for an LSP that aims at gaining a truly competitive edge in both offline and online marketplaces, particularly in the final leg of deliveries. Last mile logistics is considered in fact the least efficient, costly, and most polluting sections of the entire logistics chain [(Gevaers, et al., 2014), (Wang, et al., 2016)]. Therefore, by taking these "last mile problems" into consideration, the present work is motivated by the real need to develop location decision models for interested LSPs that aim to streamline their logistics processes through an optimum design of their last mile distribution network and by the potential gain that such models can bring into practice. By focusing on the last mile distribution network of the National postal service provider (PSP), one of the major LSP in Indonesia, we will demonstrate how the proposed models can lead to significant gains in terms of costs, with no impact on service levels, using a real-life case in the Asia-Pacific region. The scope of work is on the PSP' supply network structure which, in the context of one of the most populated urban areas in Indonesia, comprises of 1 large Distribution Centre (DC), 9 intermediate DCs, and several demand points. This paper will assist researchers and practitioners in understanding how the following research questions are to be addressed

- RQ1: How to determine the suitable number of intermediate DCs (nodes) to serve a highly populated urban area in one of the fastest developing economy of the Pacific region (Indonesia), and what is the suitable number of DCs for the particular case at hand?
- RQ2: How to decide on the locations for the intermediate DCs of an optimized supply network, and what is the optimum network configuration for the particular case at hand?

The remainder of the paper is structured in the following sections. Section 2 presents a review of relevant literature. Section 3 describes the real world application by presenting the logistics landscape of Indonesia, and by setting the background of the case study. Section 4 describes the methodological approach. Section 5 discusses the model implementation for the selected case and results to date. Finally, section 6, highlights managerial implications, limitations, and next steps to reinforce findings to date.

2 Literature Review

Decisions on logistics system design are typically about number of storage facilities, their locations, allocation of products to facilities and to market (Korpela & Tuominen, 1996). Logistics system design studies have been largely applied to commercial sector, and a fair number of research papers are available in the domain of network design. Some research groups looked at strategic and tactical decisions such as flows of commodities across supply networks [(Keskin & Uster, 2007), (Qin & Ji, 2010)], or transportation policies (Tancrez, et al., 2012). Some others focused upon operational issues (Chow, et al., 2012) such as service level improvement (Rodziah, 2017). Substantial contributions have been made in addressing cross-level decisions such as inventory control (Ho & Emrouznejad, 2009), transportation flows (Lee, et al., 2008), location decisions [(Sun, 2002), (Balcik & Beamon, 2008)] or warehouse capacity planning (Francas & Minner, 2009). In literature, network design problems are typically tackled using either mathematical modeling/optimization, Multi-Criteria Decision Making (MCDM), and simulation models. Seldom, two methods of the above are integrated.

2.1 Supply network optimization

Network optimization refers to a series of methods which, by defining objective functions and constraints, are capable of determining the optimal supply chain network design with the lowest total cost structure [(Beamon, 1998), (Supply Chain Acuity, Management Consulting, 2013)]. A recent work by Kovács, G. et al. (2017) highlighted the need of establishing efficient supply chain operations to withstand the increasing global competition, and suggested to focus on minimum total cost and lead time, and improved customer service level, as objectives of the supply chain optimization. Benyoucef, Xie and Tanonkou (2013) considered a two period SC design model addressing facility location/supplier selection problem. The authors used a Monte Carlo approach in combination with Lagrangian relaxation, so as to minimize fixed DCs location costs, inventory and safety stock costs at the DCs as well as ordering costs and transportation costs across the network. Zokaee, S. et al. (2017) proposed the use of robust optimization to determine the strategic locations and tactical allocation for a four-tier supply chain. Their model was then extended to incorporate uncertainty in key input parameters such as demand, supply capacity and major cost data including transportation and shortage cost parameters. Ghaffari-Nasab, N. et al. (2015) proposed the use of mixed

integer nonlinear programming model and linearization using a step-by-step approach to decide on the number, location and operation of hub facilities so as to minimize the total logistics costs for the network for a LSP. Authors compared the performances of two alternative distribution strategies namely hub-and-spoke and direct shipment, with the first showing significant advantages in terms of cost.

2.2 Multi-Criteria Decision Making (MCDM)

MCDM is concerned with structuring and solving decision problems on the basis of multiple decision criteria. A common application of MCDM is to tackle those decision problems for which a unique optimal solution does not exist and it is necessary to use decision maker's qualitative inputs to differentiate between solutions [(Majumder, 2015), (Roh, et al., 2015)] such as facility location problem as per Timperio, et al. (2017). Ho & Emrouznejad (2009), explored the use of a combined approach analytic hierarchy process (AHP) and goal programming (GP) to design a logistics distribution network in consideration of qualitative inputs from domain experts (e.g. on delivery and service level elements), with limited resource availability. Galvez, D. et al. (2015) integrated Mixed Integer Linear Programming (MLP) optimization and Analytical Hierarchical Process (AHP) for the implementation of a logistics network in the domain of sustainable energy production processes.

2.3 Supply chain simulation

Is used for supply network design decisions or design of supply chain policies. The main advantage of simulation is that it allows to assess supply chain behavior in a virtual environment, hence reducing the risk of making costly mistake in implementation (Thierry, et al., 2008). Munasinghe, I. U. et al. (2017) proposed a simulation-based approach to assess the impact of the facility location and product differentiation on supply chain network design. Six different scenarios were tested, with the goal of minimizing total distribution network cost. Salem R. W. & Haouari M. (2017) proposed the use of a simulation-optimization based approach to address a three-echelon stochastic supply chain network design problem. Outcomes of their work included selection of suppliers, determination

of warehouses locations and sizing, as well as the material flow, with the objective of minimizing total expected cost.

2.4 Research Gaps and Contribution of the Current Research

Despite the comprehensive literature available in the domain of network design, to our knowledge, most research groups have handled the facility location problem by selecting a single method, and only in few cases two methods were integrated (mostly optimization and simulation). Although previous works provide fair solutions to this category of problems, it is also evident the lack of an adequate holistic support (strategic, tactical, and operational) to decision making. Thus, it appears that none of the research groups have considered an integration of methodologies, and this paper fills this gap. By reading the present paper, decision makers will be able to gain comprehensive answers to the following questions:

How to determine the suitable number of DCs for a cost-effective and time responsive last mile urban delivery?

How to decide on the locations of DCs for a cost-effective and time responsive last mile urban delivery?

As such, the work presented in this manuscript will apply data analytics, network optimization, and dynamic simulation to demonstrate the theoretical and practical relevance of the proposed solution approach. In particular, the study seeks to bring the following contributions:

- Generate deeper insights (static and dynamic) in the domain of network design by uncovering strategic, tactical and operational elements;
- Determine the optimal network configuration for the case study at hand;
- Bridge theoretical knowledge with practice in the nice of network design;

3 Indonesia Logistics Landscape

Located in the South East Asia region, Indonesia is the 14th largest nation by size in the world which covers 1,811,569 square kilometers of land and 5,800,000 square kilometers of water. The country spans three time zones and counts over 260 million people spread over more than seventeen thousand islands.

According to the World Bank's Logistics Performance Index of 2016, Indonesia ranks as 63rd on a global scale in regards to key logistics elements such as customs procedures, infrastructure, international shipments, logistics competence, tracking & tracing, and timeliness (The World Bank, 2016). This translates in logistics costs accounting for the 26% of national GDP (US\$ 861 billion), worse than its neighboring countries like Singapore (8%) and Malaysia (14%). Poor logistics performance affect a) Country's economic competitiveness and b) Incountry disparities in terms of accessibility and pricing of primary commodities (Indonesia-Investments, 2013). In such a challenging context, having in place an optimized supply network would be highly beneficial for increasing profits and optimizing overstretched routes

3.1 The Case at Hand

This case study focuses on the distribution network of the PSP in Indonesia that operates a countrywide network serving various type of businesses and consumers. However, the focus of this work is on the PSP' supply network dedicated to serve a highly populated urban area in Indonesia comprising of 1 large Distribution Centre (DC), 9 intermediate DCs, and several demand points (Figure 1). After a series of field visits, interactions with operations team, and basic data analytics, it has been identified that the existing network structure is leading to high operational and distribution costs, and therefore in need of a restructuring.





Figure 1: Schematic Representation of PSP's Supply Network (for Illustration Purpose only)

4 Solution Approach

To tackle the research problem of above, a solution approach integrating data analytics, network optimization and simulation has been conceived and tested. The proposed approach encompasses three Phases:

Phase 1. Identification of optimum number of DCs to be included in the supply network;

Phase 2. Structuring and Optimizing the Supply Network;

Phase 3. Stress-testing the supply network and measuring performances based on pre-identified parameters.

Figure 2 shows the proposed integrated decision support framework.

The proposed solution approach is used in this particular case to restructure an existing supply network. However, in those instances whereby a new network is to be established, the same framework can be also applied.



Figure 2: Proposed Integrated Decision Support Framework

In order to restructure an existing supply network the necessary steps include the development of the "AS-IS" Model, followed by the "TO-BE" (Ideal) model, and finally the "TO-BE" (real) in consideration of real-life constraints (Figure 3):

(1) "AS-IS" model. This model is required to understand the existing network configuration, operational requirements, and identify bottlenecks and areas for improvement. Performances of "AS-IS" network will be used to set the benchmark for any proposed changes.





Figure 3: Stages to Supply Network Restructuring

- Ideal "TO-BE" model. This intermediate model is needed to identify the "unconstrained solution" (ideal) in terms of number of DCs and their locations;
- (3) Real "TO-BE" model. This final model will lead to the identification of real-life solution, that is an adjustment of "TO-BE" ideal based on real-life constraints set by the PSP;
- 4.1 Phase 1: Identification of number of suitable locations for the nodes

Network design begins with the identification of number of suitable locations for the nodes. These decisions are traditionally based on total cost (warehousing, transportation, lost sales). Two different strategies, centralized and decentralized supply networks, can be adopted based on logistics requirements. Centralized warehousing is a system where a single (or few) DC is used to serve a particular area whereas a decentralized approach encompasses the use of several facilities


Figure 4: Determining the Optimum Number of DCs in a Conventional Supply Chain

spread out to cover a particular area (Kokemuller, 2014). At changes of number of network nodes (DCs), corresponds a trend of costs as shown in Figure 4.

Source: Barloworld Logistics Africa (PTY) LTD

- Warehouse & Inventory Cost. Since fewer resources are needed to run one warehouse as opposed to several, lower number of DCs has a positive effect on costs related to warehousing activities. Additionally, variable costs of warehousing such as labor, warehouse management, equipment, and training of personnel can be also kept at minimum in case of centralized system.
- Primary Transport Cost. The transportation cost of the first tier (e.g. from main DC to intermediate DCs) are lower in case of centralized supply network. Consolidation of shipments can be in fact be implemented, with a positive impact on costing structure.
- 3. Local Delivery Cost. The transportation cost of the second tier (e.g. from intermediate DCs to final consumers) are higher in case of highly cen-

tralized system. In case of decentralized network in fact, the distance in between intermediate DCs and customers is lower.

4. Cost of Lost Sales. A decentralized system helps in achieving shorter lead times and higher percentages of on-time delivery, resulting in higher service level, and thus lower cost of lost sales.

In the case study undertaken in this research, data analytics and green field analysis were used to determine the cost items. Green field analysis is a GIS/center of gravity based approach (Russel and Taylor, 2010), which seeks to determine the geographic coordinates for a potential new facility. Computations are based on minimum transportation cost (calculated as "distance" * "Product Amount") (AnyLogistix, 2018) in consideration of aggregated demand for each customer and product, customer locations (direct distance between customers and DCs/Warehouses), and service distance (or number of facilities to locate) (AnyLogistix, 2018). To perform green field analysis, 2 years of operational information on historical demand (by location, amount, and time distribution), product flows, and costs were required. A template in MS Excel was shared with the company to facilitate data provision. The company provided the historical data referring to the biennium 2016 and 2017.

4.2 Phase 2: Structuring and Optimizing the Supply Network

Once Phase 1 is completed and the number of suitable locations for the network nodes is determined, Phase 2 will be used to identify the optimum network configuration in terms of both nodes (location of DCs) and arcs (connections in between DCs) using Network Optimization (NO). NO is an optimization technique which seeks to find the best configuration of a supply chain network structure as well as the flows based upon an objective function, which typically maximizes profits (Sample in Table 1),

Component	Amount [\$]			
Revenue	XXX, XXX, XXX, XXX			
Supply Cost	(XXX, XXX, XXX)			
Production Cost	(XXX, XXX, XXX)			
Transportation Cost	(XXX, XXX, XXX)			
Inbound Processing Cost	(XXX, XXX, XXX)			
Storage Cost	(XX, XXX, XXX)			
Outbound Processing Cost	(XXX, XXX, XXX)			
Fixed Cost	(XX, XXX, XXX)			
Opening Cost	(XX, XXX, XXX)			
Closing Cost	(XX, XXX, XXX)			
Profit	XXX,XXX,XXX			

Table 1: NO Cost Components

The mathematical model can be formulated as such:

Let:

 x_{ijk} = quantity of product k shipped from DC i to demand point j

 y_i = indicator (binary, 1 = Yes and 0 = No) variable for the selection of DC i

 δ_{ij} = indicator (binary, 1 = Yes and 0 = No) variable for the selection of shipping from DC i to demand point j

 f_i = operating cost for each DC i

 c_{ij} = unit cost of moving a unit weight, w_k of product k from DC i to demand point j

 p_{ik} = unit cost of processing a unit weight of product k in DC i

m = total number of DC to consider

n = total number of demand points

 d_j = demand from demand point j

M = maximum number of DC in the network

 α = maximum number of DC to fulfil a demand point j

The formulation is as follows:

$$\min\left[\begin{pmatrix}facility\\operating\ cost\end{pmatrix} + \begin{pmatrix}transportation\\cost\end{pmatrix} + \begin{pmatrix}facility\\processing\ cost\end{pmatrix}\right]$$
$$\min\sum_{i}^{m} f_{i}y_{i} + \sum_{i}^{m} \sum_{j}^{n} \sum_{k}^{K} c_{ij}x_{ijk}w_{k} + \sum_{i}^{m} \sum_{j}^{n} \sum_{k}^{K} p_{ik}x_{ijk}w_{k}$$
(1)

Subject to Maximum number of DC. As an additional planning consideration, the maximum number of DCs to include in the network can be limited due to constraints such as budget.

$$\sum_{i}^{m} y_i \le M \tag{2}$$

Demand-supply balance. In the case of deliveries of items, the network must (hard constraint) be able to fully fulfil the demand.

$$\sum_{i}^{m}\sum_{j}^{n}\sum_{k}^{K}x_{ijk} = \sum_{i}^{m}\sum_{j}^{n}\sum_{k}^{K}d_{jk}\delta_{ij}$$
(3)

Flow constraint. The connection of a demand point to a DC can only exist if and only if the DC is open.

$$\delta_{ij} \le y_i \forall i, j \tag{4}$$

Fulfilment constraint. Demand fulfilment can be either from multiple sources (multiple DCs) or constrained to a single source (single DC).

$$\sum_{i}^{m} \delta_{ij} y_i \le \alpha \forall j \tag{5}$$

In this case, the NO model was developed using AnyLogistix (ALX) software.

210

4.3 Phase 3: Stress-testing the supply network and measuring performances based on preidentified parameters

Once the ideal network configuration is defined (Phase 2), Phase 3 will stress-test the supply network based on performance measures (e.g. service level, profits). This final phase will assess network robustness and resilience, as well as measure operational performances. This Phase can be undertaken by using various computer simulation techniques such as discrete-event simulation in AnyLogistix (ALX) software.

5 Implementing the Solution Approach

The implementation of the integrated solution approach described above for the PSP case was initiated by collecting structural (existing network structure, transportation assets, product flows) and operational data (demand patterns, costs). An initial analysis of data led to the identification of bottlenecks, criticalities, as well as areas for improvements. A sample of district-level demand patterns is in Figure 5.



Figure 5: District-level Demand Patterns

5.1 Phase 1: Identification of number of suitable locations for the nodes

Given the unique nature of the business whereby the PSP in consideration operates in, the only cost item driving the decision about optimum number of facilities are transportation cost. In fact:

Warehouse and inventory cost are not relevant due to 1) PSP does not carry any inventory, and 2) Manpower is fixed, regardless the number of DCs; Cost of lost sales does not apply since there is no other competitor offering the same service; Transportation cost, which includes primary transport cost and local delivery cost, have been calculated using green field analysis in AnyLogistix (ALX) (sample in Figure 6). Transportation cost and marginal cost reduction with changed number of DCs is in Figure 7.

Results show that by increasing the number of intermediate DCs, the overall transportation costs to serve final customers would reduce.

5.2 Phase 2: Structuring and Optimizing the Supply Network

Once Phase 1 is completed, next step consists of optimizing the baseline network configuration. This phase will select the optimal network nodes out of the subset of suitable candidates from Phase 1, based on costs. The optimization problem of calibrating the baseline network configuration consists of finding the optimum number and location of the distribution centers as to effectively distribute goods, while satisfying multiple constraints.

In this work, Transportation Cost, Inbound Processing Cost, Outbound Processing Cost and Fixed Cost were considered for formulating the NO model. Other cost items listed in Table 1 were not considered due to a) Data unavailability (opening and closing costs) and b) Limited relevance based on the nature of the PSP business (Supply, production, storage).

Results show that the ideal "TO-BE" (ideal) network configuration should be inclusive of 4 intermediate DCs (Figure 8 and Figure 9), which will minimize overall supply chain cost, with no changes on service level as compared to "AS-IS".

In order to define the "TO-BE" (REAL) network configuration, the constraint on maximum distance travelled per delivery man (100 km) in a day was included. Results highlighted that "TO-BE" Ideal and "TO-BE" real overlap.

A comparative analysis of "TO-BE" network versus "AS-IS" network shows that while terms of service level the two network perform similarly, in terms of cost effectiveness the "TO-BE" network surpasses the "AS-IS" by nearly the 1% of total costs. Although a lower number of DC determines higher transportation cost (+7.74%), the saving coming from a lower fixed costs of facilities (-8.76%) is more substantial. On these two cost items in fact, an overall improvement of over 4% in costs can be appreciated. Although not yet assessed, further savings could be achieved via economies of scale e.g. through consolidation of manpower and/or transportation assets.



Figure 6: Illustration of Green Field Analysis



Figure 7: Transportation Cost and Marginal Cost Reduction with Changed Number of DCs



Figure 8: "TO-BE" (Ideal) Solution



Figure 9: "TO-BE" Ideal, Material Flow

6 Conclusions & Managerial Implications

This paper has proposed a decision support framework for network design and has demonstrated the applicability to real life case on a PSP operating in Indonesia. Starting from an initial identification of suitable locations for the network nodes, the proposed framework depicts procedures and tools to leverage on across the various phases of the problem's solution seeking. The full implementation of the proposed approach would lead to the definition of the optimum network configuration, assessment of its robustness and measurement of operational performances. However we applied the framework partially, as to define a preliminary solution to the problem at hand.

As anticipated in the introduction, this paper can assist researchers and practitioners in understanding how the following research questions are to be addressed, and key findings include:

RQ1: How to determine the suitable number of intermediate DCs (nodes) to serve a highly populated urban area in one of the fastest developing economy of the Pacific region (Indonesia), and what is the suitable number of DCs for the particular case at hand?

The suitable number of intermediate DCs (nodes) to serve a highly populated urban area in one of the fastest developing economy of the Pacific region (Indonesia) can be determined by combining data analytics with green field analysis, and network optimization. For the geography of reference (greater Surabaya), and with the provided datasets, the number of suitable intermediate DCs should be equal to 4.

RQ2: How to decide on the locations for the intermediate DCs of an optimized supply network, and what is the optimum network configuration for the particular case at hand?

Locations for the intermediate DCs of an optimized supply network can be selected using a network optimization approach. The optimum network configuration for PI in Greater Surabaya should include the four nodes as per Figure 8. The identified set of facilities guarantees enhanced cost effectiveness (-18% of transportation and warehousing cost) at comparable service level;

Managerial Implications. This study is able to support decision makers in a wide range decisions in the context of network design. GFA can help decision makers

with the determination of transportation cost at increased number of DCs, as well as identification of potentially suitable locations. The NO can help logistics managers to make strategic decisions about DCs' locations.

Limitations. This work has few limitations. First, the dataset on demand is limited to the biennium 2016-2017. An extension of the dataset with inclusion of more data points would provide a more accurate solution to the problem at hand. Secondly, inclusion of cost items such as cost for opening or closing a DC and manpower allocation would help to fine tune the proposed solution.

Next steps. In order to reinforce the findings to date, a dynamic simulation model can be developed (Phase 3). This would allow to:

- Determine transportation (fleet size) and storage requirements;
- Perform what-if analysis with comparison of alternative network configurations;

References

Anylogistix (2018). "ALX Documentation, Experiments, GFA". In:

Atan, R. binti (2016). "Enhancing service quality through Service Level Agreement (SLA) full implementation". In: Science in Information Technology (ICSITech), 2016 2nd International Conference on. IEEE, pp. 1–1.

Balcik, B. and B. M. Beamon (2008). "Facility location in humanitarian relief". In: 11.2, pp. 101–121. Bank, T. W. (2016). "Global Rankings 2016". In:

Beamon, B. M. (1998). "Supply chain design and analysis:: Models and methods". In: 55.3, pp. 281–294.

Benyoucef, L., X. Xie, and G. A. Tanonkou (2013). "Supply chain network design with unreliable suppliers: a Lagrangian relaxation-based approach". In: 51.21, pp. 6435–6454.

Chow, P.-S., T.-M. Choi, and T. Cheng (2012). "Impacts of minimum order quantity on a quick response supply chain". In: 42.4, pp. 868–879.

Francas, D. and S. Minner (2009). "Manufacturing network configuration in supply chains with product recovery". In: 37.4, pp. 757–769.

Galvez, D., A. Rakotondranaivo, L. Morel, M. Camargo, and M. Fick (2015). "Reverse logistics network design for a biogas plant: An approach based on MILP optimization and Analytical Hierarchical Process (AHP)". In: 37, pp. 616–623.

- Gattorna, J., W. Selen, and R. Ogulin (2004). "Characteristics, strategies and trends for 3PL/4PL in Australia". In: pp. 11–16.
- Gevaers, R., E. Van de Voorde, and T. Vanelslander (2014). "Cost modelling and simulation of last-mile characteristics in an innovative B2C supply chain environment with implications on urban areas and cities". In: 125, pp. 398–411.

Ghaffari-Nasab, N., M. Ghazanfari, and E. Teimoury (2016). "Hub-and-spoke logistics network design for third party logistics service providers". In: 11.1, pp. 49–61.

Ho, W. and A. Emrouznejad (2009). "Multi-criteria logistics distribution network design using SAS/OR". In: 36.3, pp. 7288–7298.

"Indonesia-Investments" (2013). In:

Jonsson, P. and S.-A. Mattsson (2016). Logistik: Läran om effektiva materialflöden. Studentlitteratur. Keskin, B. B. and H. Üster (2007). "Meta-heuristic approaches with memory and evolution for a multi-product production/distribution system design problem". In: 182.2, pp. 663–682.

- Korpela, J. and M. Tuominen (1996). "A decision aid in warehouse site selection". In: 45.1-3, pp. 169– 180.
- Kovács, G., Z. Varga, and M. Mihalik (2017). "Optimal formation of logistics networks". In: Vehicle and Automotive Engineering. Springer, pp. 391–403.
- Lee, B. K., K. H. Kang, and Y. H. Lee (2008). "Decomposition heuristic to minimize total cost in a multi-level supply chain network". In: 54.4, pp. 945–959.
- Majumder, M. (2015). "Multi criteria decision making". In: Impact of urbanization on water shortage in face of climatic aberrations. Springer, pp. 35–47.
- Munasinghe, I., T. Rupasinghe, and R. Wickramarachchi (2017). "A simulation-based modeling approach to assess the multi-echelon supply chain network design". In: Industrial Engineering and Engineering Management (IEEM), 2017 IEEE International Conference on. IEEE, pp. 1372– 1376.
- Nagurney, A. and L. S. Nagurney (2010). "Sustainable supply chain network design: A multicriteria perspective". In: 3.3, pp. 189–197.
- Qin, Z. and X. Ji (2010). "Logistics network design for product recovery in fuzzy environment". In: 202.2, pp. 479–490.
- Roh, S., S. Pettit, I. Harris, and A. Beresford (2015). "The pre-positioning of warehouses at regional and local levels for a humanitarian relief organisation". In: 170, pp. 616–628.
- Russell, R. S. and B. W. Taylor-Iii (2008). Operations management along the supply chain. John Wiley & Sons.
- Salem, R. W. and M. Haouari (2017). "A simulation-optimisation approach for supply chain network design under supply and demand uncertainties". In: 55.7, pp. 1845–1861.
- Sun, M. (2002). "The transportation problem with exclusionary side constraints and two branchand-bound algorithms". In: 140.3, pp. 629–647.
- Tancrez, J.-S., J.-C. Lange, and P. Semal (2012). "A location-inventory model for large three-level supply chains". In: 48.2, pp. 485–502.
- Tartavulea, R. I. and R. I. Petrariu (2013). "Logistics Market Statistics and Opinions about the Supply Chain Management in Romania". In: 15.33, p. 180.
- Thierry, C., G. Bel, and A. Thomas (2008). Supply chain management simulation: an overview.
- Timperio, G., G. B. Panchal, A. Samvedi, M. Goh, and R. De Souza (2017). "Decision support framework for location selection and disaster relief network design". In: 7.3, pp. 222–245.
- Wang, Y., D. Zhang, Q. Liu, F. Shen, and L. H. Lee (2016). "Towards enhancing the last-mile delivery: An effective crowd-tasking model with scalable solutions". In: 93, pp. 279–293.
- Yazdani, M., P. Chatterjee, E. K. Zavadskas, and S. H. Zolfani (2017). "Integrated QFD-MCDM framework for green supplier selection". In: 142, pp. 3728–3740.
- Zhang, W. and D. Xu (2014). "Integrating the logistics network design with order quantity determination under uncertain customer demands". In: 41.1, pp. 168–175.

Zokaee, S., A. Jabbarzadeh, B. Fahimnia, and S. J. Sadjadi (2017). "Robust supply chain network design: an optimization model with real world application". In: 257.1-2, pp. 15–44.

Business Model of Aircraft Fleet Planning using ANN

Partha Kumar Pandit¹, M. Ahsan Akhtar Hasin²

1 - Bangladesh Airlines Training Center BATC

2 - Bangladesh University of Engineering and Technology

The purpose of Aircraft Fleet Planning is to provide the number and type of aircraft for acquisition, its time of acquisition, and trade-in or phase out of fleet. Airlines select a particular type of aircraft from the manufacturer on the basis of optimum cost, considering a number of constraints. Because of existence of several conflicting criteria, the solution approach becomes an NP-hard problem. As such, a traditional linear programming approach cannot optimize the system in a reasonable time frame. This paper aims at developing a model for selecting aircraft using Artificial Neural Networks. Key inputs have been obtained from the major areas of aircraft design characteristics, aircraft physical performance, maintenance needs, operating economics, acquisition cost, operating cost and customer satisfaction. The input values are fuzzy in nature. However, several methods for combined use of fuzzy logic systems and neural networks have been suggested. Experience, which is conventionally used for selecting a particular type of aircraft, has been used for training of the proposed network. Single-layer ANN model provided a good solution with optimality in cost, without sacrificing time constraint and algorithmic complexity. The airline business will be immensely benefited from the solution procedure.

Keywords: Aircraft Fleet Planning; ANN; Fuzzy

First recieved: 24.May.2018 Revised: 04.Jun.2018 Accepted: 22.Jun.2018

1 Introduction

Aircraft Fleet planning process determines type of aircraft the airline should buy and their numbers, in order to achieve organizational goals. Corporate planning deals with the need of the airlines, current resources, corporate objective, projected industry environment and marketing strategy. Specific requirement of the airlines is then identified through marketing research and analysis on the basis of the set company objectives.

Given a flight schedule, which is a set of flight segment with specified departure and arrival times and a set of aircraft, the fleet assignment problem is to determine which aircraft type should fly each flight segment (Clark et al., 1996). The objective is to maximize revenue minus operating cost. In the basic fleet assignment problem considered by Clark et al.(1996) a daily, domestic fleet assignment problem is modeled and solved with up to eleven fleets and 2500 flight legs.

Satisfied customers usually return back and buy more, they tell other people about their experiences. Customer Satisfaction is one of the most important processes in airline industry and is recognized as the key to the success of business competition. Like any other industry, airlines first measure the expectation of its passenger, analyzes the consumer behaviour, prepares a customer satisfaction program to provide a set of standard customer services, identifies the service gap and tries for continual improvement.

This research considered the issues of aircraft cost and customer satisfaction in selecting aircraft and developing models integrating these two separate issues – cost estimation and customer satisfaction. Solution methodologies of Fuzzy Logic and Artificial Neural Network have been adopted.

1.1 Aircraft Selection Process

Air transportation offers both passenger and freight services that are essential for economic growth and development. In a highly competitive environment, airline companies have to control their operating costs by managing their flights, aircraft, and crews effectively.

Fleet assignment involves the optimal allocation of a limited number of fleet types to flight legs in the schedule subject to various operational constraints (Jacobs and Johnson, 2008). Airlines typically manage their annual business cycle by

subdividing the year into a sequence of scheduling periods that span about a month each (Ahmad, 2000).

Aircraft selection process is a dynamic function, coordinated by a number of group functions including corporate planning. The life cycle starts with the corporate objective of acquisition of new aircraft in the fleet. The planning life cycle is shown in Figure 1.

Data acquisition is the next step. Information related to any kind of lease or purchase of aircraft and making comparative statements, relevant information is required. Then evaluation of data is the logical sequence.

The next step is the purchase negotiation and finally acquisition of the aircraft. Many sections of the airlines will be involved when aircraft is going for its operation. Continuous updating/modifying the aircraft will be going on until the aircraft is disposed off.



Figure 1: Fleet planning life cycle

1.2 Cost Consideration of Airline

In order to maximize anticipated profit, cost is a vital area for all business. Costing is also important for pricing of its product. The price of average revenue per passenger mile must be sufficient to cover average cost per passenger mile flown. Airline operating cost is categorized in two areas – direct operating cost or variable costs and indirect operating cost or fixed-overhead costs.

A flight sequence or route is built for each individual aircraft so as to cover each flight exactly once at a minimum cost while satisfying maintenance requirements. Finally, in the crew scheduling or pairing optimization problem, a minimum cost set of crew rotations or pairings is constructed such that every flight is assigned a qualified crew and that work rules and collective agreements are satisfied. In practice, most airline companies solve these problems in a sequential manner to plan their operations, although recently, an increasing effort is being made to develop novel approaches for integrating some of the airline operations planning problems while retaining tractability.

The cost categories (Swan and Adler, 2006) include pilot, cabin crew, fuel, Landing fee, airframe maintenance, engine maintenance and the ownership costs of the aircraft. The flight crew costs include all costs (Ssamula and Mistro, 2004) associated with the flight and cabin crew including allowances, pensions, salaries, etc.





The costs of an airline incurs in producing available seat miles (ASM) depend upon the type of adjustments it is able to make in the amounts of the various resources it employs. The quantities of many resources used – labour, fuel and so forthcan be varied relatively quickly in the short run. But amounts of other resources demand more time of adjustment.

1.3 Customer Satisfaction - a primary Paramater

Introduction of wide body service in the early 1970s marked the climax of production –sales orientation in the air transport industry. Excess capacity and shortage of customer changed the marketing concept to a customer –oriented approach.

The industry entered into its matured stage since 1960s. Weaker competitors left the industry. The remaining competitors have become well entrenched, and the marketing policy and images are well known. Customer loyalties and market shares became stabilized.

Evaluation of customer satisfaction differs from one study to another. Studies have been differing with their focus and coverage. Most of the studies focused on evaluating factors influencing customer satisfaction or associated customer satisfaction and quality of services (Josephat and Ismail, 2012). Logistic regression was used to develop customer satisfaction model for Precision Air.

Customers' satisfaction degree is customers' comparative evaluation between expectation and perceived value for the products and the services after making relevant payment for their proceeds. Satisfaction or dissatisfaction after the purchase, depends not only on the difference between performance that customers actually feel and expected, but also upon their costs that they pay in order to get the profits, that is to make customers obtain the maximum discount. Customers' delivered value means the difference between the total value of customers and the total cost of customers.

Service encounters that do not meet expectations should have a bigger impact on customers than service encounters that exceed expectations. Furthermore, if customers are concerned about the possibility of service encounters that do not meet expectations, future sales and profitability may be negatively influenced (Shaun et al., 2005).

2 Theoretical Background

The flying performance of different aircraft model is different, for example, range of aircraft, flying altitude ceiling, maximum takeoff weight, maximum zero fuel weight, available cargo space and many more. So a particular route is not suitable for all models of the aircraft to perform. In addition, different models have different seating capacity, layout and their operating costs are not same (Li, and Tan, 2013). Profit maximization is the primary goal of an airline whereas customer satisfaction is considered to be the secondary goal through which strategic advantage can be achieved.

Conventionally, a computer operates through sequential linear processing technologies. They apply formulas, decision rules, and algorithms instructed by users to produce outputs from the inputs. Conventional computers are good at numerical computation. But ANNs improve their own rules; the more decisions they make, the better the decisions may become.

ANN can determine the optimal solution to an NP-complete (nondeterministic polynomial) problem, such as the travelling salesperson problem. This is equivalent to linear and integer programming in management sciences, where the objective functions are optimized using a heuristic search procedure. there is no doubt that ANNs are feasible for business applications. Many phenomena that are difficult to describe can be modeled by ANNs, if carefully designed.

Artificial Neural Networks have emerged as a very popular area of research, both from the design and the usage points of view. There is considerable research emphasis on designing better and more efficient neural networks, more powerful "learning algorithms", better transfer functions etc. On the other hand there is a great amount of academic interest in the applications of neural networks. Disciplines as diverse as the biological sciences and finance have applications that use neural networks. There is a significant volume of research on neural networks in the engineering and science literature. There also exists a reasonable body of neural network research as related to business.

2.1 Profit Maximization Model

Considering the features of a particular route and flight plan, under constraints of pre-announced flight schedules, not considering the flight stopovers and having

enough airport capacity, the model (Pandit et al., 2010) which considers matched flying area, as well as the traffic match conditions can be proposed as follows:

$$\operatorname{Max}\sum_{i=0}^{m}\sum_{j=1}^{n}P_{ij}x_{ij} \tag{1}$$

$$\sum_{i=1}^{m} \sum_{j=1}^{n} x_{ij} = m$$
 (2)

$$\sum_{j=1}^{n} x_{ij} = 1 \tag{3}$$

$$x_{ij}d_j \ge D_j$$
 $i = 1, 2, 3..., m; \ j = 1, 2, 3..., n$ (4)

$$x_{ij}s_j \ge S_j$$
 $i = 1, 2, 3..., m; \ j = 1, 2, 3..., n$ (5)

$$x_{ij}r_j \ge R_j$$
 $i = 1, 2, 3..., m; \ j = 1, 2, 3..., n$ (6)

Where,

$$\begin{split} i &= 1, 2, 3 \dots, m \, m & \text{ is overall flight number} \\ j &= 1, 2, 3 \dots, n \, n & \text{ is the number of aircraft model} \\ P_{ij} & \text{ the revenue of aircraft model } j \text{ to perform the flight } i \end{split}$$

$$P_{ij} = E_{ij} - F_{ij} - V_{ij}$$

$$x_{ij} \begin{cases} 0 \\ 1 \end{cases}$$

 d_j is the range of the aircraft model j

 D_i is the distance of the flight i

else otherwise

 s_j is the seating configuration of the aircraft model j

 E_{ij} Revenue earned from model j to flight i F_{ij} Fixed operating cost of model j to flight i

flight *i* is performed by the aircraft model *j*

 V_{ij} is the variable operating cost of model j to fly flight i

- S_j is the required traffic of the flight i
- r_j is the flying requirement (ETOPS, TCAS,...) of the model j
- R_j is the flying requirement (ETOPS, TCAS,..) of the route
- g_j is the cargo space available in model j
- G_j is the cargo space requirement in flight i

The objective function (1) means that the total income of all flights is largest, after the aircraft types are assigned to all flights considering the bulk of the flights of global optimization. Constraint (2) is to ensure that an equal number of models are selected for flights. Constraint (3) is to ensure that only one model is assigned to each flight. Constraint (4) is to ensure that the model assigned to the flight meets the flight distance requirements. Constraint (5) is to ensure that the model assigned to the flight meets the traffic requirement is fulfilled by the seating capacity. Constraint (6) is to ensure that the selected model meets the technical requirements of the route.

2.2 Integrated Fleet Planning Model

As a decision making process integrated fleet planning depends on Design characteristics, Physical Performance, Maintenance Needs, Acquisition cost, operating economics (Pandit and Hasin, 2010).

In this research, the fleet planning problem (FPP) has been defined as a multiobjective optimization function with cost minimization and satisfaction maximization as follows:

$$\mathsf{FPP} = \mathsf{Selection} \left[\mathsf{Min} \sum C_{ij}, \, \mathsf{Max} \sum S_{ik} \right] \tag{7}$$

Where

i = alternative aircraft

j = cost parameters

k = satisfaction parameters

The constraints are the required Design characteristics, Physical Performance, Maintenance Needs, Operating Economics and availability of Acquisition Cost.

Selection of aircraft on the basis of optimal solution is the unique goal of this research. The input values are considered as fuzzy in nature. However, in contrast to neural networks, fuzzy decision logic systems are not capable of automatic learning. To solve this problem, this research utilized combined use of fuzzy logic systems and neural networks.

In this research, fuzzy neural network approach is used, where neural network part is primarily used for its learning and classification capabilities and for pattern association and retrieval. For neural network analysis, a two-layer hidden function is used which proves sufficient for learning. The data set for learning is derived from the available business results, which are optimized using conventional linear or integer programming approach, as used typically by some airlines research organizations.

The quantitative features, considered here, are costs and expenses of acquisition for purchase of aircraft, its spares, ground support equipment, maintenance training, flight training, financing, pre-payment, maintenance, fuel, general administrative activities, sales promotion activities, passenger service and handling, cockpit and cabin crew, dispatcher, insurance and rental charges, landing and navigation fees.

The fleet planning data used in this research are – available aircraft and specific requirement of fleet planning in terms of destination, type of load to be carried, time of aircraft acquisition.

Once these features were extracted, their probable fuzzy relations were determined. It is necessary to note that fuzzy relation provides an analyzer with a rich framework within which many problems emerging from practical applications of the theory are formulated.

This research also utilized this strength of fuzzy relations. However, the great utility of fuzzy relation equations was somewhat hampered by the rather high computational demand of conventional linear programming optimization models, which is supported by the existing literatures as well (Oliver, 1997).

A sensitivity analysis was also performed for minimization of deviations from desired outputs. The desired response of the network to which each input pattern and its comparison with the actual output of the network were used in the learning algorithm for appropriate adjustments of the weights. These adjustments whose purpose was to minimize the difference between the desired and actual outputs were made incrementally.

That is small adjustments in the weights are made in the desired direction for each training pair. This is essential for facilitating a convergence to a solution (specific values of the weight) in which patterns in the training set are recognized with high fidelity.



Output

Figure 3: Detailed Model

3 Outline of Methodology

In order to carry out this research work, steps that have been adopted are mentioned below:-

3.1 Fuzzification

Uncertainties regarding all satisfaction data have been handled with the help of fuzzy membership function and finally Customer Satisfaction fuzzy ranking is prepared for different aircraft.

Passenger satisfaction data was collected from the reviews made by the passengers for different flights with different aircraft. In those reviews, they had options to give comment and make ranking on a number of issues like-seat comfort, satisfaction for Catering, satisfaction for entertainment, value for money and staff comfort.

3.2 Neural Network Design

The work flow for the general neural network design process has seven primary steps:

- 1. Collection of data
- 2. Creating the network
- 3. Configuring the network
- 4. Initializing the weights and biases
- 5. Training of the network
- 6. Validating the network (post-training analysis)
- 7. Using the network

Aircraft specification and data sheets have been collected from individual manufacturer and other sources. Aircraft parameters and cost related data have been collected from different airlines. Customer satisfaction fuzzy ranking is also being used as input of the ANN.

Database has been prepared and updated for all aircraft. Database has also been developed for customer satisfaction, taking into account factors like seat comfort, catering service, entertainment and overall remarks on different past flights.

Route data has been collected from flight history of aircraft. Route data have been collected for five routes- Dhaka-Bangkok, Dhaka-Dubai, Dhaka-Kathmandu, Dhaka-Kolkata and Dhaka-Kuala Lumpur. For selecting aircraft for a particular route, flight histories for five routes have been taken. The routes and the aerial distances are mentioned in Table 1.

Flight history has been taken from all these routes. Airline operating in these routes have different types of aircraft having differences in design data, performance data, seat capacity, cost data, satisfaction data. It is worthy to mention here that there are some common aircraft also being operated by different airlines.

Route No.	From	То	Distance KM
1	Dhaka (DAC)	Bangkok (BKK)	1533
2	Dhaka (DAC)	Dubai (DXB)	3555
3	Dhaka (DAC)	Kathmandu (KTM)	676
4	Dhaka (DAC)	Kolkata (CCU)	245
5	Dhaka (DAC)	Kuala Lumpur (KUL)	2579

Table 1: Routes used for flight history

Feed Forward network of Artificial Neural Network has been implemented for training, validating and finally testing. After establishment of network it was tested by different data sets to check its response.

3.3 Defuzzification

Defuzzification is the process of converting a fuzzified output into a single crisp value with respect to a fuzzy set. A common and useful defuzzification technique is center of gravity. The centroid of this shape, called the fuzzy centroid, is calculated. The x coordinate of the centroid is the defuzzified value.

4 Result

Satisfaction data is full of uncertainties. Application of fuzzy logic with a number of rules could provide a crisp result on ranking. This ranking was used as a special input for Artificial Neural Network.

4.1 Customer Satisfaction Fuzzy Ranking

The remarks given by the passengers were collected for preparing customer satisfaction fuzzy ranking. These remarks were made flight wise for a particular airline. Passengers put their recommendations on seat comfort, satisfaction for Catering, satisfaction for entertainment, value for money and staff comfort in a ranking scale of 10. Airlines wise such remarks were collected and categorized according to aircraft.

Two columns namely value for money and staff comfort have no direct correlation with the type of aircraft the airline used. So for the sake of this research, these two columns were omitted. For Seat comfort, catering and entertainment the linguistic variables are

Very Low	VL
Low	LO
Medium	MI
High	HI
Very High	VH

For final recommendations, following were the linguistic variables

Not recommended	NR
Recommended but needs improvement	RC
Highly Recommended	HR

The mode (with highest frequency) is considered. If the remarks are transferred into linguistic comments, the table becomes as follows.

ICAO Code	Seat Comfort	Catering Comfort	Entertai- nment	Satisfaction Fuzzy	Coding
A310	LO	МІ	VL	1	NR
A319	MI	MI	VL	1	NR
A320	HI	HI	MI	2	RC
A333	LO	HI	MI	2	RC
A336	LO	HI	MI	2	RC
B735	HI	HI	MI	2	RC
B736	MI	LO	VL	1	NR
B737	HI	HI	MI	2	RC
B738	HI	HI	MI	2	RC
B739	HI	HI	HI	3	HR
B772	HI	HI	MI	2	RC
B77W	VH	VH	VH	3	HR
B762	LO	MI	VL	1	NR
DH8C	MI	MI	VL	1	NR
DH8D	MI	LO	VL	1	NR
E145	HI	MI	MI	2	RC
AT73	LO	н	VL	1	NR
MD11	HI	MI	VL	1	NR

Table 2: Linguistic expression of Satisfaction

Fuzzy Inference System has been used in this situation. The system has three inputs (Seating, Catering and Entertainment) and one output (Satisfaction). 8 rules are used to setup the Inference system. The summary of the GUI system is placed here. Triangular membership function were used with the following parameters for the input variables

Input Variable	MF	MF Definition	MF Type
Seating	VL	[0.037 0.602 1.24]	trimf
	LO	[0.7882 1.448 2.189]	trimf
	MI	[1.78 2.36 3.221]	trimf
	HI	[2.73 3.34 4.24]	trimf
	VH	[3.76 4.23 5.02]	trimf
Catering	VL	[0.0114 0.575 1.157]	trimf
	LO	[0.853 1.554 2.27]	trimf
	MI	[1.82 2.36 3.142]	trimf
	HI	[2.743 3.353 4.253]	trimf
	VH	[3.81 4.25 5.033]	trimf
Entertainment	VL	[-0.0331 0.5754 1.21]	trimf
	LO	[0.708 1.396 2.25]	trimf
	MI	[1.753 2.277 3.127]	trimf
	HI	[2.7 3.21 4.22]	trimf
	VH	[3.75 4.18 5.033]	trimf

Table 3: Fuzzy	Input Variable
----------------	----------------

Та	b	le 4: Fuzzy	Output Var	iab	le (Triangu	lar Mem	bersl	nip)
----	---	-------------	------------	-----	------	---------	---------	-------	-----	---

Output Variable	MF	MF Definition	MF Type
Satisfaction	NR	[0.0104 0.853 2.03]	trimf
	LO	[1.52 2.44 3.485]	trimf
	MI	[3 3.909 5.01]	trimf



Figure 4: Membership Plot of input variable Seating



Figure 5: Membership Plot of output variable Satisfaction

The rules used in this FIS are

Rule 1. If (Seating is VL) or (Catering is VL) or (Entertainment is VL) then (Satisfaction is NR)

- Rule 2. If (Seating is HI) and (Catering is not VL) and (Entertainment is not VL) then (Satisfaction is RC)
- Rule 3. If (Seating is not VL) and (Catering is HI) and (Entertainment is not VL) then (Satisfaction is RC)
- Rule 4. If (Seating is not VL) and (Catering is not VL) and (Entertainment is HI) then (Satisfaction is RC)
- Rule 5. If (Seating is HI) and (Catering is HI) and (Entertainment is HI) then (Satisfaction is HR)
- Rule 6. If (Seating is VH) and (Catering is not VL) and (Entertainment is not VL) then (Satisfaction is HR)
- Rule 7. If (Seating is not VL) and (Catering is VH) and (Entertainment is not VL) then (Satisfaction is HR)
- Rule 8. If (Seating is not VL) and (Catering is not VL) and (Entertainment is VH) then (Satisfaction is HR)

After fuzzification the output value is now ready to be used for Artificial Neural Network as described in the detailed model in figure 3.

4.2 Artificial Neural Network

The nftool of MATLAB is used for the Artificial Neural Network. nftool leads through solving a data fitting problem, solving it with a two-layer feed-forward network using Levenberg-Marquardt Algorithm and trained with trainlm algorithm.

4.2.1 Neural Network Summary

Input Layer :	26 Neurons
Hidden Layer:	10 Neurons
Output Layer :	1 neuron
Algorithm used:	Levenberg-Marquardt Algorithm

ICAO Code	Seating	Catering	Ent.	Sat.	Sat. Rank
A310	1.67	2.3	0.3	0.981	3
A319	1.9	3.1	0.45	1.37	6
A320	2.85	3.9	2.88	3.04	17
A333	1.1	2.9	2.8	1.8	9
A336	0.89	3.56	3	1.67	8
B735	3.12	2.9	2.87	2.99	16
B736	2.25	1.45	0.6	0.965	1
B737	2.85	3.4	2.33	2.48	11
B738	2.85	2.87	2.29	2.5	13
B739	2.88	3.4	3.4	2.93	15
B772	3.5	3.83	2.26	2.57	14
B77W	4.1	3.9	3.85	3.39	18
B762	1.2	1.9	0.45	0.969	2
DH8C	2.4	1.9	1.1	1.01	5
DH8D	1.8	1.7	1	0.995	4
E145	2.9	2.8	2.28	2.49	12
AT73	1.4	2.99	0.45	1.37	7
MD11	3.23	1.93	1.1	2.13	10

Table 5: Satisfaction Ranking



Figure 6: Neural Network Layers

After establishing the network it needs training. For that purpose developed data sets are provided to the network as data_in from a data file. Also the target was provided through another data file called data_out. 70% data (772 Samples) are used for training, 15% (166 Samples) for validation and another 15% (166 Samples) are used for testing purpose.

Results			
	🖏 Samples	🔄 MSE	🖉 R
🕡 Training:	772	2.00225e-1	9.84167e-1
Validation:	166	3.18077e-2	9.97122e-1
🕡 Testing:	166	3.49275e-2	9.97542e-1
	Plot Fit Plo	t Error Histogram	

Figure 7: Training, Validation and Testing data samples

4.2.2 Training Summary

The process to train a neural network involves tuning the values of the weights and biases of the network to optimize network performance, as defined by the network performance function. The training of neural network depends on many factors - complexity of the problem, the number of data points in the training set, the number of weights and biases in the network, the error goal and whether the network is being used for function approximation (regression).

In Levenberg-Marquardt Algorithm of feed forward network, trainlm algorithm is used for training. Iteration is one step taken in the gradient descent algorithm towards minimizing the loss function using a mini-batch. An epoch is the full pass of the training algorithm over the entire training set. The training state is presented in Figure 8.



Figure 8: Training State of the network

The neural network randomly selects data from the given data set for training, testing and validating. In each set of data the relationship between target and actual output exists which is performed by a correlation analysis to establish a linear relationship.

Regression analysis describes a relationship between target and actual output of the developed network. Regression analysis in Figure 9, presents least-squares fit for four distinct situation of Training, Validation, Testing and All samples.

^{4.2.3} Result of Regression Analysis



Figure 9: Regression Analysis

4.2.4 Performance of Network

When the training in is complete, the network performance is checked and determined if any changes need to be made to the training process, the network architecture, or the data sets. The performance of the network is shown in Figure 10.


Figure 10: Performance of the network

4.2.5 Simulink

With the help of Simulink, the system is simulated interactively and viewed the results on scopes and graphical displays. The integration of Simulink and MATLAB enables to run unattended batch simulations of Simulink models using MATLAB commands.



Figure 11: Simulink Diagram

5 Findings

Main objectives of this research were to develop a fleet planning model for selecting aircraft in a particular route where cost is minimized and customer satisfaction is maximum. By using fuzzy logic for customer satisfaction ranking and Artificial Neural Network for the established databases.

The Artificial Neural Network has been established with 26 neurons in the input layer, 10 neurons in the middle hidden layer and 1 output neuron. The network has been trained, validated and tested for being used by the user.

Feed Forward network using Levenberg-Marquardt Algorithm do not show always the same accuracy as the human brain. Generally it is a very good algorithm to reach its global minima when the error learning is concerned. Summary of the

Sample Type	Number of Sample
Total	1104
Training -70%	772
Validation -15%	166
Testing -15%	166

Table	6:	List	of	Samp	ole	used
-------	----	------	----	------	-----	------

Epoch		MSE			Regressio	n	
	Train	Valid	Test	Train	Valid	Test	All
13	0.0736	0.0346	0.0422	0.988	0.9635	0.9976	0.985
20	0.0402	0.0278	0.0624	0.984	0.9968	0.9968	0.988
18	0.0976	0.0284	0.0536	0.992	0.9975	0.9543	0.987
14	0.2002	0.0318	0.0342	0.984	0.9971	0.9975	0.987

Table 7: Network Accuracy

Mean squared Errors (MSE) and network accuracy (Regression-R) values are as follows:

The network has been tested for known and unknown inputs for recognizing the output.

Sample Input

[5; 202; 2579; 2;56000; 897; 5170; 80142; 142000; 41000; 280; 1204; 5.82; 23591.176; 940.65; 1457.43; 320; 26309.256; 1827.92; 28137.176; 280; 100.4899143; 0.981; 3; 25.7; 10]

From the Simulink output window, the output is 1 and when it is decoded for finding the real value of 1, it is found that the selected aircraft is A310.

Aircraft specification, design characteristics, aircraft performance data, power plant data, customer satisfaction data needs regular updating for correct decision making by the neural network.

Business Model of Aircraft Fleet Planning using ANN



Figure 12: Simulink Output window

6 Conclusion

Airline industry is suffering from lack of integrated fleet planning mechanism, considering multiple input factors, along with uncertainties in different forms and at various stages. This research developed a mathematical model with dual objectives of minimizing operating costs and maximizing customer satisfaction. In this perspective the model developed by the researcher and its implementation through fuzzy logic and artificial neural network has given good results with respect to both objectives. Computer simulation has provided useful sensitivity with respect to varied input data sets. With slight modifications and updating of data on a regular basis, the model can serve the purposes of other regions of the world.

For specific optimization problem, other optimization algorithms may find better solutions than Fuzzy Neural Network combination (given the same amount of

computation time). In this particular fleet planning problem, future research potential lies in the area of using other algorithms like evolution strategies, evolutionary programming, simulated annealing, Gaussian adaptation, hill climbing and swarm intelligence and methods based on integer linear programming.

References

- Clarke, L. W., C. A. Hane, E. L. Johnson, and G. L. Nemhauser (1996). "Maintenance and crew considerations in fleet assignment". In: *Transportation Science* 30.3, pp. 249–260.
- Jacobs, T. L., B. C. Smith, and E. L. Johnson (2008). "Incorporating network flow effects into the airline fleet assignment process". In: *Transportation Science* 42.4, pp. 514–529.
- Jarrah, A. I., J. Goodstein, and R. Narasimhan (2000). "An efficient airline re-fleeting model for the incremental modification of planned fleet assignments". In: *Transportation Science* 34.4, pp. 349–363.
- Josephat, P. and A. Ismail (2012). "A logistic regression model of customer satisfaction of airline". In: International Journal of Human Resource Studies 2.4, p. 197.
- Li, Y. and N. Tan (2013). "Study on fleet assignment problem model and algorithm". In: *Mathematical* Problems in Engineering 2013.
- Mcquitty, S., M. Hyman, R. Oliver, P. Sautter, and A. Stratemeyer (2005). "Service variability and its effect on consumer perceptions and intentions". In: pp. 264–294.
- Oliver, R. (1997). Satisfaction: a behavioral perspective on the consumer.
- Pandit, P. K., M. A. A. Hasin, and M. A. Sharif (2010). "Aircraft Fleet Planning Under Varying Demand using Aggregate Resource Planning Concepts". In: International Journal of Logistics and Transport.
- "Pioneering Solutions in Supply Chain Management: A Comprehensive Insight Into Current Management Approaches" (2010). In: Erich Schmidt Verlag, Germany. Chap. Integrated Aircraft Fleet Planning using Fuzzy Logic and Artificial Neural Network Approach.
- Ssamula, B. and R. F. Del Mistro (2004). *Developing a cost model for running an airline service*. Citeseer.
- Swan, W. M. and N. Adler (2006). "Aircraft trip cost parameters: A function of stage length and seat capacity". In: Transportation Research Part E: Logistics and Transportation Review 42.2, pp. 105–115.

Part IV

Risk and Security Management

Adequate Flexibility Potential to handle Supply Chain Uncertainties

Immanuel Zitzmann¹, David Karl¹

1 - Otto-Friedrich-Universität Bamberg

Risks and uncertainties (e.g. IT-system failure or demand fluctuation) threaten the performance of supply chains. This paper gives insight into the planning of flexibility potential as a crucial tool for managing the consequences of operational and disruptive uncertainties. A simulation study of lot-sizing decisions in a two-stage decentralised supply chain is used. The modelled supply chain faces operational as well as disruptive uncertainties. It is analysed how capacity and/or stock flexibility on each stage cope with unexpected events. The location of flexibility within the supply chain is key for its ability to handle uncertainties. The paper shows that stock flexibility can substitute capacity flexibility to a certain degree. However, disruptive uncertainties cannot be handled by stock flexibility alone. Therefore, trade-offs in flexibility potential have to be considered. In contrast to other studies, this simulation models operational and disruptive uncertainties in three areas: internal processes, supply side and demand side. Also flexibility management in a decentralised decision making process is analysed. Contrary to the lean thinking approach it is shown that inventory plays an important role in managing uncertainties. Therefore, management should use the right amount of inventory to create flexibility depending on the individual risk situation.

Keywords: Uncertainty; Supply Chain; Flexibility; Simulation Study First recieved: 24.May.2018 Accepted: 07.Jun.2018

1 Introduction

Globalisation, digitalisation and new technologies change the processes of value creation. As a result, global and complex supply chains can be observed (Meixell and Gargeya, 2005). The scale of these value creating networks makes them vulnerable (Peck, 2005; Peck, 2006). They are not only affected by unexpected events geographically near to the final consumer but also by incidents that occur on the other side of the world (Simangunsong et al., 2012). Examples for such events are the eruption of the volcano Eyjafjallajökull on Iceland in 2010 or an earthquake in Taiwan (Papadakis and Ziemba, 2001). In addition to those low-probability/high-impact risks operational uncertainties jeopardize supply chains (Tang and Tomlin, 2008; Sodhi and Tang, 2012). The focus on efficiency has led to supply chains which are not able to compensate even small disorders like delayed delivery or fluctuation in production rates and consumer demand (Craighead et al., 2007).

The literature on supply chain risks and uncertainties mentions different approaches to handle unexpected situations. Thus, a supply chain should be robust, resilient and agile (Naylor et al., 1999; Christopher and Peck, 2004; Klibi et al., 2010). The importance of each aspect depends on the individual circumstances of a supply chain (Cabral et al., 2012). However, all three concepts need flexibility to be utilised (Zitzmann, 2014; Zitzmann, 2016). It is the key ability to handle risks and uncertainties in supply chains and to use opportunities which emerge from unexpected situations.

Regarding flexibility management and flexibility in supply chains many research questions can be explored (Stevenson and Spring, 2007). The focus of this paper is to look into planning flexibility potential. In particular, we aim to give answers on the following questions: Where in the supply chain is flexibility needed? How can capacity and stock flexibility substitute each other?

This article is structured as follows. First, we provide background information on risk and uncertainties in supply chains as well as on flexibility and its creation. Subsequently, in the methodology section a simulation study is introduced. Section 4 presents the results of the simulation study and discuss its implications. The final section summarises the findings of the paper.

2 Review of Relevant Literature

This paper builds on existing literature regarding risks and uncertainties as well as flexibility in the context of supply chains. First, the terms risk and uncertainty will be distinguished in section 2.1 as they are often used interchangeably. Afterwards, section 2.2 will consider the complex concept of flexibility, especially flexibility in supply chains. How flexibility can be created will be the subject of section 2.3

2.1 Risk and Uncertainties in Supply Chains

Making decisions under uncertainty is part of every management process and therefore in supply chain management as well. Uncertainty can be described as inability to predict something (Milliken, 1987). The term risk is often used interchangeably with uncertainty, but they differ. Figure 1 shows two opinions how the terms can be distinguished. In both cases, uncertainty encompasses more than risk. According to decision theory, risk and uncertainty can be differentiated according to their predictability. If it is possible to quantify a probability of occurrence, it is called risk; if not it is called uncertainty (Knight, 1971). The second approach, which will be followed in this paper, considers the consequences of uncertainty. If they are positive, they are called chances; if they may be negative, then they are risks (Simangunsong et al., 2012).

The literature review of Simangunsong et al. (2012) identifies 14 sources of uncertainties in supply chains which can be classified into three groups: within an institution of a supply chain, within the regarded supply chain, or external to the value creating network. Chopra and Sodhi (2004), Jüttner (2005), Tang and Tomlin (2008), Sodhi and Tang (2012) as well as Tiwari et al. (2015) also identify these sources of uncertainties for supply chains.

uncer	tainty		uncer	tainty
without probabaility	with probability = risk	relationship between risk and uncertainty	positive impact = chance	negative impact = risk

Figure 1: Relationship between risk and uncertainty

The focus of this paper is flexibility in supply chains and its ability to handle uncertainties, and not about reducing uncertainties, as it is done in (supply chain) risk management. Therefore, more important than the sources of uncertainties are their impact on the supply chain. According to their magnitude it can be distinguished between operational and disruptive uncertainties (Chopra and Sodhi, 2004; Tang, 2006; Tang and Tomlin, 2008; Sodhi and Tang, 2012):

Operational uncertainties are inherent fluctuations in the processes of a supply chain, e.g. uncertain customer demand, variations in production rate or delays in transportation. Uncertainties are operational when fluctuations are limited to a certain range around an average value.

Disruptive uncertainties refer to events with major impact on the supply chain. These are natural or man-made disruptions that happen rarely like earthquakes, floods, terrorist attacks or machine breakdowns. Their consequences for the supply chain are massive and cannot be predicted to a certain degree.

Operational as well as disruptive uncertainties can result from the sources mentioned previously and can affect the flow of goods, information and finance within a supply chain. This paper considers the flow of goods and how flexibility can handle the impact of emerging uncertainties. Usually, no organisation exists that manages the whole supply chain. Therefore, decisions about flexibility management are made on the institutional level (Lummus et al., 2003). Taking this viewpoint, supply chain uncertainties have impact on three different areas: the supply side, the process of value creation within the regarded institution itself and the demand side (Chopra and Sodhi, 2004; Tang and Tomlin, 2009). Even a single source of uncertainty can lead to consequences in more than one area, e.g. the breakdown of a machine is an uncertainty located within the manufacturing process. When it happens in the regarded institution, it affects the process of value creation. The supply process is affected when the breakdown occurs at the side of a vendor. Natural disasters which are external uncertainties often affect all three areas of uncertainty impact. An earthquake may not only destroy factories as well as streets and therefore, endanger the supply and manufacturing process, it also affects the demand side.

2.2 Flexibility in Supply Chains

According to Garavelli (2003), "[...] flexibility reflects the ability of a system to properly and rapidly respond to changes, coming from inside as well as outside the

system [...]⁴. As supply chains are the considered system in the context of supply chain flexibility, this definition shows that flexibility is a suitable instrument to handle uncertainties which may be internal or external to a supply chain (Tang, 2006). Flexibility is topic in economic and organizational literature as well as in the context of manufacturing (Sethi and Sethi, 1990; Jain et al., 2013). So far, a holistic flexibility theory does not exist (Yu et al., 2015). Sánchez and Pérez (2005), as well as Garavelli (2003), summarize the aspects of flexibility according to six dimensions. These include functional, hierarchical, measurement and strategic aspects as well as the time horizon and the object of change. Different approaches exist on how flexibility may be achieved. The models of Vickery et al. (1999) and Duclos et al. (2003) are the most common concepts (Singer, 2012). Due to a more process orientated view, the latter is more suited when looking at supply chains.

The dimensions of Duclos et al. (2003) as well as the attributes of other approaches (Pujawan, 2004; Kumar et al., 2006; Stevenson and Spring, 2007; Manders et al., 2017), describe features of a flexible supply chain. They also explain what these dimensions are used for, yet no guideline is given how planning flexibility works and how the right scale of flexibility is established. Also, these categorisations and frameworks do not solve some of the major issues in managing flexibility. There is still the problem that no independent measure for flexibility exists (Stevenson and Spring, 2007). Due to multiple dimensions, indicator systems have to be applied to measure flexibility. These indicators cannot be used universally but rather have to be developed for each regarded system individually. Therefore, they are subjective and situational (Gerwin, 1993; Koste et al., 2004). Additionally, it is not possible to determine the benefit of flexibility in advance (Jain et al., 2013). Only in retrospect the contribution of flexible components to the supply chain performance can be evaluated.

flexibility	potential
strategic	operational
flexibility	flexibility
=	=
redundant	redundant
capacity	stock

Figure 2: Flexibility potential

Planning flexibility on a strategic level means creating potential through additional capacity in the regarded system. Thus, we describe it as capacity flexibility. It may be established by an additional production line, spare transportation vehicles or a multiple instead of a single sourcing concept. Alternative or additional to strategic flexibility capabilities flexibility potential can also be created on the operational level. This operational flexibility is established through stock flexibility (Vickery et al., 1999; Wang, 2008). Thereby, a certain amount of inventory is hold on purpose to react to uncertainties. Safety stocks are an example of flexibility created by inventory. Thus, flexibility potential can be created by capacity or stock potential as shown in Figure 2. What kind of flexibility is more useful in a supply chain to handle uncertainties and where it should be established is considered in the simulation study in section 3 and 4 of this paper.

3 Simulation

As we look at situations where operational as well as disruptive uncertainties may occur at multiple places within a supply chain an analytical analysis is not possible. To investigate if capacity and/or stock flexibility are adequate approaches to handle uncertainties in supply chains we therefore conducted a simulation study. The supply chain considered in the study is introduced in section 3.1, whereas, the modelling of strategic and operational flexibility in the supply chain is explained in section 3.2.

3.1 Two-stage Supply Chain as Object of Analysis

A two-stage supply chain is the object of the simulation. Its structure is based on the supply chain introduced by Banerjee (1986). The author analysed joint lotsizing in a producer-retailer relationship. This case will be modified to the purpose of the simulation study. Figure 3 summarises the structure of the modelled supply chain. Additional to Banerjee's case a transportation process is added. Hence, the modelled supply chain includes three processes: The first one is the production of goods at the place of the producer that is followed by the transportation process. Thereby, the finished goods are carried from the warehouse of the producer to the warehouse of the retailer. The third modelled process of the supply chain is the customer demand which is satisfied at the place of the retailer. In this supply chain the producer as well as the retailer determine their individual lot size which will then be produced and transported. The planning as well as the execution process is rolling and consists of six periods. In each period with the length of a month decisions about production and order sizes have to be made. In all three processes of the supply chain operational and possibly disruptive uncertainties exist. The study analyses the effect of capacity as well as safety stock to handle such uncertainties.

Operational uncertainties are modelled with the help of probability distributions. In all three processes – production, transport and demand – the production and transport performance or request for goods can differ from the average rates. In production as well as in transport triangular distributions are used. Table 1 presents the basic parameters with operational uncertainties used for the simulation model.

Adequate Flexibility Potential to handle Supply Chain Uncertainties



Figure 3: Supply chain of the simulation study

Table 1: Model parameters including operational uncertainties

Parameter	Value (Explanation)
Production time	Triangular distributed, mode 36 minutes/piece, lower bound 32.4 minutes/piece (-10 %), upper bound 54 min- utes/piece (+50 %)
Warehouse capacity (producer or retailer)	unlimited
Transportation time	Triangular distributed (-10 %/+50 %): mode 120 hours, lower bound 108 hours, upper bound 180 hours/piece (e.g. 1 month for carrying goods from Asia to Europe by ship)
Demand	Normal distributed: mean 100 pieces/month, S.D. 20 pieces/month, lower bound 0 pieces/month

Additional to operational uncertainties disruptive events are possible in some simulation models. They are also modelled in all three processes. Within production and transport disruptive uncertainties will lead to an interruption of the respective process. The nature of unexpected events is that their occurrence cannot be predicted. Nevertheless, they happen. Table 2 explains possible disruptive events for the different supply chain processes.

Event	Probability	Impact
Production interruption	0.1 %/hour	Interruption for a uniformly distributed length of at least one day (6 working hours) up to 10 days (60 working hours)
Transportation interruption	0.1 %/hour	Interruption for a uniformly distributed length of at least one day (6 working hours) up to 10 days (60 working hours)
Unexpected demand peak	0.05 %/hour	Uniformly distributed demand from 120 % up to 200 % of the actual demand of the period
Unexpected demand drop	0.05 %/hour	Uniformly distributed demand from 0 % up to 80 % of the actual demand of the period

Table 2: Model parameters regarding disruptive events

We assume minimum costs for the retailer at an order size of 80 pieces/shipment. Depending on the disposable inventory in its warehouse and the expected demand in two months the retailer will therefore order nothing or a multiple number of the lot size every month. The order must be placed two months in advance because of the production and transport time. The producer uses the order quantities of the retailer and its own disposable inventory to decide if production is necessary or not. When production is started the production lot size is 200 pieces.

3.2 Modelling Flexibility Potential in the Supply Chain

Flexibility potential can be created at four points in the observed supply chain. In two cases it is capacity flexibility and the others are stock flexibility, which are implemented in the simulation model according to Table 3.

The first point for capacity flexibility is the place of the producer. If uncertainties occur the flexibility potential can be used to compensate them. The transportation process is the second point where capacity flexibility can be integrated in the process. Stock flexibility can be created in both, the producer's and the retailer's

warehouse. In this case a certain level of additional inventory is defined. It is used as safety stock to handle unexpected events and uncertainties. The producer as well as the retailer have to consider the defined level of safety stock within their planning process. That is done by reducing the disposable inventory by the amount of additional inventory.

Event	Probability	Impact
Production	Reduction of production time by 33 % = average production capacity increases to 300 pieces/month (additional production lines and work force)	Safety stock: 20 % of the average de- mand (= 20 pieces)
Transportation	Reduction of transportation time by 33 % (faster ship or use of an alternative means of transportation)	-
Demand	-	Safety stock: 20 % of the average de- mand (= 20 pieces)

Table 3: Flexibility potential in the simulation model

Producer and retailer can use capacity and stock flexibility. Nevertheless, it is also possible to use just one kind of flexibility potential or none at all. Therefore, 16 alternative combinations of creating flexibility potential are possible in the observed supply chain. For each of them a simulation model was created. Additionally, we distinguish between situations where only operational uncertainties exist and such where operational as well as disruptive uncertainties occur. This differentiation leads to additional simulation models. The models -1- to -16- are regarding operational uncertainties alone, whereas the models -17- to -32- consider operational as well as disruptive uncertainties. Each model will run for 10,000 iterations. Supplementary to the 32 supply chain configurations we model a supply chain with no flexibility potential and no uncertainties. This deterministic model (-0-) is used to verify the correct modelling.

4 Results and Findings

To analyse what kind of flexibility is best to handle uncertainties in the regarded supply chain and at which process it should be created a performance index is needed. As the purpose of a supply chain is to match supply with customer demand we will use hours with out-of-stock situations at the retailer as performance indicator for the whole supply chain. The simulation calculates the performance within six periods which are 720 working hours (6 hours per day). We use this time as basis to calculate the service level. As the focus of the study is on the benefit and location of flexibility, we do not consider the costs of implementing flexibility (Tang and Tomlin, 2008).

Since there are no uncertainties in the deterministic model -0- the demand can always be satisfied (service level = 100 %). Such a situation with no uncertainties is not possible in a real world supply chain.

Section 4.1 presents the performance of the supply chain with uncertainties. The implications of the results are explained in section 4.2.

4.1 Simulation Results

Figure 4 shows the performance of the supply chain with operational uncertainties (flexibility configurations -1- to -16-) The amount of hours with out-of-stock situations given in Figure 4 is the mean of 10,000 iterations.

With a mean of 70 hours of out-of-stock situations, the performance of a supply chain with operational uncertainties and no flexibility at all (-13-) is the lowest. Thus, the service level is 90 %. The highest service level of 97 % can be achieved in the models -7-, -8- as well as -11- and -12-. These configurations have in common that capacity flexibility in the transportation process and stock flexibility at the retailer exists. This is the first indication that the other forms of flexibility are not as important. Especially stock flexibility of 20 % at the producer has low benefit to handle uncertainties in the supply chain. The service level of model -14- which simulates a situation where only this kind of flexibility exists is just 1 % above model -13- with no flexibility at all.

Comparing simulations with and without stock flexibility at the place of the producer but otherwise similar configurations (e.g. -1- with -2-, -3- with -4- or -15 with -16-) confirms the small benefit of stock flexibility at the producer.

Simulation model	Capacity flexibility- producer	Capacity flexibility- transport	Stock flexibility – producer	Stock flexibility – retailer	Out-of-stock situations	Service level
-1-	Х				65	91 %
-2-	imes		imes		63	91%
-3-	imes			\boxtimes	27	96 %
-4-	imes		imes	\boxtimes	26	96 %
-5-		imes			45	94 %
-6-		imes	imes		43	94 %
-7-		imes		\boxtimes	23	97 %
-8-		imes	imes	\boxtimes	21	97 %
-9-	imes	imes			42	94 %
-10-	imes	imes	imes		45	94 %
-11-	imes	imes		\ltimes	21	97 %
-12-	imes	imes	imes	\boxtimes	23	97 %
-13-					70	90 %
-14-			imes		67	91 %
-15-				imes	31	96 %
-16-			Х	\mathbb{X}	30	96 %

Adequate Flexibility Potential to handle Supply Chain Uncertainties

Figure 4: Supply chain performance with operational uncertainties (mean out of 10,000 simulation runs)

The same can be said about capacity flexibility of the producer. Comparing the models -2- with -14-, -4- with -16- or -6- with -10- the service levels are identical. These configurations only differ in the regard of capacity flexibility at the producer. In models that have this flexibility the amount of out-of-stock hours is lower. Therefore, it can be noted that capacity flexibility helps to handle uncertainties. However, the benefit is equally small like the one of stock flexibility at the producer. In both cases the reduction of out-of-stock hours does not affect the service level. It only changes compared to a supply chain with no flexibility potential.

The kind of flexibility that increases supply chain performance the most is stock flexibility at the retailer. In this simulation model (-15-) a service level of 96 % can be achieved. That is just 1 % below the highest level of models that consider operational uncertainties alone.

The results of the supply chain with operational as well as disruptive uncertainties simulated in the models -17- till -32- can be seen in table 4.2. Again, with one exception the service level does not change when stock flexibility at the producer exists or not (e.g. comparing -17- with -18; -19- with -20- or -31- with -32-). This statement also applies to the comparison of a situation with no flexibility potential and stock flexibility at the producer alone. Overall, the analyses of situations with operational and disruptive uncertainties together confirm the findings regarding stock flexibility at the producer that were made by analyzing service levels when only operational uncertainties exists. The results of table 4.2 also confirm that the capacity flexibility in transport together with stock flexibility at the retailer lead to the highest service levels (95 % or 94 %). Regarding capacity flexibility at the producer the results of table 4.2 differ from them in table 4.1. The performance of models without this flexibility potential (e.g. -29- till -32-) is nearly as good as in models with capacity flexibility at the producer (-17- till -20-). Therefore, it can again be noted that capacity flexibility at the producer does not play an imported role by handling uncertainties.

Simulation model	Capacity flexibility- producer	Capacity flexibility- transport	Stock flexibility – producer	Stock flexibility – retailer	Out-of-stock situations	Service level
-17-	Х				97	87 %
-18-	imes		\boxtimes		94	87 %
-19-	imes			imes	52	93 %
-20-	imes		\boxtimes	\boxtimes	51	93 %
-21-		\boxtimes			74	90 %
-22-		\boxtimes	\boxtimes		69	90 %
-23-		\boxtimes		\boxtimes	46	94 %
-24-		\boxtimes	\boxtimes	\boxtimes	41	94 %
-25-	\boxtimes	\boxtimes			63	91%
-26-	\boxtimes	\boxtimes	\boxtimes		63	91%
-27-	\boxtimes	\boxtimes		\boxtimes	40	94 %
-28-	imes	imes	\bowtie	\ge	39	95 %
-29-					104	86 %
-30-			\ge		98	86 %
-31-				imes	58	92 %
-32-			\bowtie	\bowtie	57	92 %

Figure 5: Supply chain performance with operational and disruptive uncertainties (mean out of 10,000 simulation runs)



Figure 6: Achieved service level in 10,000 simulation runs in a supply chain with operational uncertainties and no flexibility



Figure 7: Achieved service level in 10,000 simulation runs in a supply chain with operational and disruptive uncertainties and no flexibility

The simulation models -1- to -16- in Figure 4 consider only operational uncertainties. Therefore, it is expected that the amount of out-of-stock hours is lower than in the simulations models -17- to -32- of Figure 5 which consider operational as well as disruptive uncertainties. This can be confirmed. The service level of models with operational as well as disruptive uncertainties is about two till five percentage points below the one that can be achieved when only operational uncertainties. This is also illustrated by comparing Figure 6 and Figure 7 which show the achieved service levels of the supply chain with no flexibility within 10,000 simulation runs. When only operational uncertainties exist it is possible to achieve a service level of 95 % in 6,035 simulation runs (Figure 6). In one run the lowest service level of 44 % occurs. When operational as well as disruptive uncertainties exist a service level of 95 % can only be achieved in 2,983 runs (Figure 7). The lowest performance is a service level of 11 %. This shows that it is a huge difference if a mean service level of 90 % like in model -13- or of 86 % (model -29-) can be achieved.

4.2 Implications of Results

The simulation study shows how important it is to consider uncertainties in all stages of supply chains. The modelled uncertainties are independent from each other. However, at the final stage of a supply chain all effects of uncertainties may add up and interrupt customer satisfaction. The study shows that flexibility potential at the retailer is most effective. It is able to compensate not only uncertainties in demand but also in processes upstream. If sufficient flexibility potential is available here it can compensate nearly all uncertainties. In the simulation study, the stock flexibility at the retailer is not dimensioned to handle all uncertainties. But operational flexibility potential in the amount of 20 % from the average demand can already reduce out-of-stock hours tremendously. As stock flexibility as operational flexibility potential is the only way to handle short-term uncertainties at the end of a supply chain, it is highly important.

Capacity as well as stock flexibility at previous stages of the supply chain can handle uncertainties as well. But their effect is limited to uncertainties that occur at the place of the flexibility potential or upstream from it. Uncertainties downstream of flexibility potential cannot be handled by this capability. But it is still needed. It protects the final stage from situations where uncertainties of the whole supply chain add up. Such situations could only be managed by huge inventory levels which are undesirable considering cost efficiency and other supply chain objectives. However, medium levels are not able to handle situations where effects of different sources of uncertainties add up.

It can be noted that every kind of flexibility helps in handling uncertainties and will increase supply chain performance (Sánchez and Pérez, 2005). That is confirmed by the study. A small degree of flexibility can increase performance enormous when it is located at the right place. As Garavelli (2003) as well as Graves and Tomlin (2003) already pointed out it is not necessary to build total flexibility into a system. Crucial in handling uncertainties is stock flexibility (Tsay and Lovejoy, 1999). It is available immediately and has its biggest impact when located at the end of a supply chain. The respective models of the simulation study have the highest service levels.

5 Conclusions

This paper presented insights into handling uncertainties in supply chains with the help of flexibility. It regarded the effects of existing uncertainties. To overcome them a supply chain needs flexibility potential which has to be created proactive. As the study shows, the location of flexibility within the supply chain is key for its ability to handle uncertainties. The further down in the supply chain the potential is located the more effective it is in compensating uncertainties that occur somewhere in the value creation process. Nevertheless, flexibility only in the last stage of a supply chain is not sufficient to achieve the best performance regarding the matching of supply and customer demand. As uncertainties cumulate during the processes of value creation they may be too much to be handled at just one point in the supply chain. Therefore, flexibility potential should be established at the end of the supply chain but also at other critical processes.

Our study has a number of limitations. While the results of the simulation fit into existing literature on flexibility in supply chains the findings are limited to the observed two-stage supply chain. Analyses of different supply chain structures could validate and generalise our findings. Furthermore, strategic as well as operational flexibility is modelled by a certain amount of redundant capacity and additional stock. Different levels of flexibility were not considered. How variations in the level of flexibility potential influences the ability of handling uncertainties therefore still have to be studied. As investments in additional capacity are probably more expensive then into stock flexibility cost considerations may be another point of further research.

If management faces the challenge of handling uncertainties it therefore should first of all define levels of safety stock near the end customer. To a certain degree they have to abandon the lean thinking approach. After defining a base level of stock flexibility the supply chain has to be analysed to identify critical processes. At these processes, additional flexibility potential should be created. This can be capacity and/or stock flexibility. These steps have to be integrated not only in the management process of the individual institutions within the supply chain but also in the process of coordination and collaboration between institutions. By integrating considerations about flexibility potential can only be utilised.

References

- Banerjee, A. (1986). "A Joint Economic-Lot-Size Model for Purchaser and Vendor". In: Decision Science 17.3, pp. 292–311.
- Cabral, I., A. Grilo, and V. Cruz-Machado (2012). "A decision-making model for Lean, Agile, Resilient and Green supply chain management". In: *International Journal of Production Research* 50.17, pp. 4830–4845.
- Chopra S.; Sodhi, M. S. (2004). "Managing Risk to Avoid Supply-Chain Breakdown". In: MIT Sloan Management Review 46.1, pp. 53–61.
- Christopher M.; Peck, H. (2004). "Building the Resilient Supply Chain". In: The International Journal of Logistics Management 15.2, pp. 1–14.
- Craighead, C. W., J. Blackhurst, M. J. Rungtusanatham, and R. B. Handfield (2007). "The Severity of Supply Chain Disruptions: Design Characteristics and Mitigation Capabilities". In: *Decision Sciences* 38.1, pp. 131–156.
- Duclos, L. K., R. J. Vokurka, and R. R. Lummus (2003). "A conceptual model of supply chain flexibility". In: Industrial Management & Data Systems 103.6, pp. 446–456.
- Garavelli, A. C. (2003). "Flexibility configurations for the supply chain management". In: International Journal of Production Economics 85.2, pp. 141–153.
- Gerwin, D. (1993). "Manufacturing flexibility: a strategic perspective". In: *Management Science* 39.4, pp. 395–410.
- Graves, S. C. and B. T. Tomlin (2003). "Process Flexibility in Supply Chains". In: *Management Science* 49.7, pp. 907–919.
- Gupty, Y. P. and S. Goyal (1989). "Flexibility of manufacturing systems: Concepts and measurements". In: European Journal of Operational Research 43.2, pp. 119–135.
- Jain, A., P. K. Jain, T. S. Chan, and S. Singh (2013). "A review on manufacturing flexibility". In: International Journal of Production Research 51.19, pp. 5946–5970.
- Jüttner, U. (2005). "Supply chain risk management understanding the business requirements from a practitioner perspective". In: *The International Journal of Logistics Management* 16.1, pp. 120–141.
- Klibi, W., A. Martel, and A. Guitouni (2010). "The design of robust value-creating supply chain networks: A critical review". In: European Journal of Operational Research 203.2, pp. 283–293.

Knight, F. H. (1971). Risk, Uncertainty and Profit. London: The University of Chicago Press.

- Koste, L. L., M. J. Malhotra, and S. Sharma (2004). "Measuring dimensions of manufacturing flexibility". In: Journal of Operations Management 22.2, pp. 171–196.
- Kumar, V., K. A. Fantazy, U. Kumar, and T. A. Boyle (2006). "Implementation and management framework for supply chain flexibility". In: *Journal of Enterprise Information Management* 19.3, pp. 303–319.
- Lummus, R. R., L. K. Duclos, and R. J. Vokurka (2003). "Supply Chain Flexibility: Building a New Model". In: Global Journal of Flexible Systems Management 4.4, pp. 1–13.
- Manders, J. H. M., M. C. J. Caniels, and P. Ghijsen (2017). "Supply chain flexibility: A systematic literature review and identification of directions for future research". In: *The International Journal of Logistics Management* 28.4, pp. 964–1026.
- Meixell, J. M. and V. B. Gargeya (2005). "Global supply chain design: A literature review and critique". In: *Transportation Research Part E: Logistics and Transportation Review* 41.6, pp. 531–550.
- Merschmann, U. and U. W. Thonemann (2011). "Supply chain flexibility, uncertainty and firm performance: An empirical analysis of German manufacturing firms". In: *International Journal* of Production Economics 130.1, pp. 43–53.
- Milliken, F. J. (1987). "Three Types of Perceived Uncertainty about the Environment: State, Effect and Response Uncertainty". In: Academy of Management Review 12.1, pp. 133–143.
- Naylor, J. B., M. M. Naim, and D. Berry (1999). "Leagility: Integrating the lean and agile manufacturing paradigms in the total supply chain". In: *International Journal of Production Economics* 62.1, pp. 107–118.
- Papadakis, I. S. and W. T. Ziemba (2001). "Derivative Effects of the 1999 Earthquake to US Personal Computer Manufactures in Mitigation and Financing of Seismic Risks". In: *Mitigation and Financing of Seismic Risks: Turkish and International Perspectives*. Ed. by P. R. Kleindorfer and M. R. Sertel. Dordrecht: Springer, pp. 261–276.
- Peck, H. (2005). "Drivers of supply chain vulnerability: an integrated framework". In: International Journal of Physical Distribution and Logistics Management 35.4, pp. 210–232.
- Peck, H. (2006). "Reconciling supply chain vulnerability, risk and supply chain management". In: International Journal of Logistics: Research and Application 9.2, pp. 127–142.
- Prater, E., M. Biehl, and M. A. Smith (2001). "International supply chain agility Tradeoffs between flexibility and uncertainty". In: International Journal of Operations & Production Management 21.5/6, pp. 823–839.
- Pujawan, I. N. (2004). "Assessing supply chain flexibility: a conceptual framework and case study". In: International Journal of Integrated Supply Management 1.1, pp. 79–97.
- Sánchez, A. M. and M. P. Pérez (2005). "Supply chain flexibility and firm performance". In: International Journal of Operations & Production Management 25.7, pp. 681–700.
- Sethi, A. K. and S. P. Sethi (1990). "Flexibility in manufacturing: A Survey". In: The International Journal of Flexible Manufacturing Systems 2.4, pp. 289–328.
- Simangunsong, E., L. C. Hendry, and M. Stevenson (2012). "Supply-chain uncertainty: a review and theoretical foundation for future research". In: *International Journal of Production Research* 50.16, pp. 4493–4523.
- Singer, C. (2012). Flexibilitätsmanagement zur Bewältigung von Unsicherheiten in der Supply Chain. Köln: Eulverlag.
- Slack, N. (1983). "Flexibility as a manufacturing objective". In: International Journal of Production Management 3.3, pp. 4–13.
- Sodhi, M. S. and C. S. Tang (2012). Managing Supply Chain Risk. New York: Springer.

- Stevenson, M. and M. Spring (2007). "Flexibility form a supply chain perspective: definition and review". In: International Journal of Operations & Production Management 27.7, pp. 685–713.
- Tang, C. S. (2006). "Robust strategies for mitigation supply chain disruptions". In: International Journal of Logistics: Research and Applications 9.1, pp. 33–45.
- Tang, C. and B. Tomlin (2008). "The power of flexibility for mitigating supply chain risks". In: International Journal of Production Economics 116.1, pp. 12–27.
- Tang, C. and B. Tomlin (2009). "How Much Flexibility Does It Take to Mitigate Supply Chain Risks?" In: Supply Chain Risk – A Handbook of Assessment, Management and Performance. Ed. by G. A. Zsidisin and B. Ritchie. New York: Springer, pp. 155–174.
- Tiwari, A. K., A. Tiwari, and C. Samuel (2015). "Supply chain flexibility: a comprehensive review". In: Management Research Review 38.7, pp. 767–792.
- Tsay, A. A. and W. S. Lovejoy (1999). "Quantity Flexibility Contracts and Supply Chain Performance". In: Manufacturing & Service Operations Management 1.2, pp. 89–111.
- Upton, D. M. (1994). "The Management of Manufacturing Flexibility". In: California Management Review 36.2, pp. 72–89.
- Vickery, S., R. Calantone, and C. Dröge (1999). "Supply Chain Flexibility: An Empirical Study". In: The Journal of Supply Chain Management 35.2, pp. 16–24.
- Wang, Y.-C. (2008). "Evaluating flexibility on order quantity and delivery lead time for a supply chain system". In: International Journal of Systems Science 39.12, pp. 1193–1202.
- Yu, K., J. Cadeaux, and B. N. Luo (2015). "Operational flexibility: Review and meta-analysis". In: International Journal of Production Economics 169, pp. 190–202.
- Zitzmann, I. (2014). "How to Cope with Uncertainty in Supply Chains? Conceptual Framework for Agility, Robustness, Resilience, Continuity and Anti-Fragility in Supply Chains". In: Next Generation Supply Chains. Ed. by T. W. Kersten Blecker and C. M. Ringle. Berlin: Springer, pp. 361– 378.
- Zitzmann, I. (2016). "RoRA-SCM Integrating flexibility in the planning process to handle uncertainties". In: Logistics Management – Contributions of the Section Logistics of the German Academic Association for Business Research, 2015, Braunschweig, Germany. Ed. by T. B. J. D. Mattfeld Spengler and M. Grunewald. Heidelberg: Springer, pp. 51–62.

Remote Sensing in Humanitarian Logistics: An Integrative Approach

Christian Hein¹, Henning Hünemohr², Rainer Lasch¹

1 - TU Dresden

2 – Lilienthal Digitaler-Weinbau GmbH

Remote sensing is becoming increasingly important for the acquisition of information in humanitarian logistics. While different platforms (e.g. satellites, drones) with their own characteristics and advantages exist, the literature often refers to just one of them. This paper aims at integrating platforms, making full use of their specific advantages. The work is conceptual, comparing different remote sensing platforms according to their advantages and limitations as well as evaluating the potentials of their combination drawing on both academic and practitioner literature. Moreover, drone test flights (hexacopter) demonstrate the practical implementation of data acquisition, processing, as well as their integration. The results show that the combination of data gathered via different remote sensing platforms is highly beneficial for the management of humanitarian logistics in terms of quality, time, flexibility, and cost. Furthermore, the feasibility has been demonstrated by the drone experiment and the respective data processing and integration. To the best of our knowledge, this is the first work systematically examining the integration of data acquired via different remote sensing platforms considering their specific advantages and the area of application in the context of humanitarian logistics as well as demonstrating the feasibility through the use of an experimental drone flight. The results discussed represent the first step in the use of combined remote sensing data in humanitarian logistics; specific applications should be examined more in detail. Furthermore, challenges in the use of such data have to be overcome, such as their integration into the information management processes of humanitarian actors.

Keywords: Humanitarian Logistics; Humanitarian Operations; Remote Sensing; Drones

First recieved: 28.May.2018 Revised: 21.Jun.2018 Accepted: 08.Jul.2018

1 Introduction

Natural and human-made disasters have unfortunately become almost commonplace. Even though some of these tragedies might be preventable, the occurrence of others cannot be influenced. Extensive multifaceted responses are coordinated by the international community, represented by numerous of humanitarian organizations or agencies, in order to alleviate the suffering of those affected. In this context, logistics plays a crucial role (Kovács and Spens, 2011). Information is increasingly recognized as a key factor of successful humanitarian operations (Altay & Pal, 2014; Kim et al., 2012), whereas, at the same time, information management is recognized as underresearched (Gupta et al., 2016). Concerning the rapid acquisition of profound data, remote sensing technologies are of rising importance. Different platforms are applicable, such as satellites, aircrafts, and drones, to gather remote sensing data. Previous work consider only one of these remote sensing platforms (Delmonteil and Rancourt, 2017; Tatham et al., 2017). As each platform inheres specific advantages and limitations, this paper aims to combine them. Therefore, the different remote sensing platforms will be examined and compared comprehensively in order to answer the following research questions: Which are the specific advantages and limitations of different remote sensing platforms in the context humanitarian logistics and are there potentials of combining data from these platforms? And how can such an integrative approach be realized practically by utilizing drones? The remainder of this work is organized as follows: After introducing humanitarian logistics and remote sensing in the second section, a systematic overview of the potential humanitarian uses of remote sensing technology is given by not distinguishing between the different platforms. In section three, the different platforms are compared according to their advantages and limitations and the potentials of integration are examined. Following, the experimental drone flight is described concerning the methodology, the technical realization, as well as the results in section four. Finally, the conclusion and an outlook are provided.

1.1 Humanitarian Logistics

Although logistics has always been involved in relief operations, its appearance in practice and academic literature was very limited before the beginning of the 21st century (Kovács and Spens, 2011). This recognition seems overdue since logistics plays a particularly important role in humanitarian operations, as it influences

their effectiveness significantly and determines the speed, the coverage, and the costs of humanitarian organizations (Baumgarten, 2011). Although commercial and humanitarian logistics are closely related, there are significant differences which characterize the humanitarian relief work environment and pose specific challenges. Nevertheless, humanitarian logistics definitions resemble those of logistics in general. Thomas and Kopczak (2005) specify humanitarian logistics as 'the process of planning, implementing and controlling the efficient, cost-effective flow and storage of goods and materials, as well as related information, from the point of origin to the point of consumption for the purpose of alleviating the suffering of vulnerable people'. In contrast to commercial logistics, attention here is placed on the reduction of people's suffering. Humanitarian logistics includes a wide range of activities and processes such as preparedness, procurement, transport, warehousing, and training (Schwarz, 2012). In order to visualize relief chain management and humanitarian logistics tasks and to contribute a standardization of terms, definitions, and activities, Blecken (2009) developed a reference task model. He describes the two dimensions of hierarchical and structural decomposition (Figure 1). Within the dimension of the structural decomposition it is distinguished between assessment, procurement, warehousing and transport activities. Assessment is required to identify the affected people or the community needs; procurement tasks ensure the availability of all required material resources. Warehousing tasks are related to the storage and the transshipment of supplies: transportation includes all distribution activities. Finally, vertical and horizontal cross-section functions accompany these tasks. In disaster management, there is often a distinction between different phases. While diverse approaches exist, the cyclic classification into the phases mitigation, preparedness, response, and recovery has become commonplace (OCHA, 2013). The first two phases relate to the pre-disaster and the others to the post-disaster phase. The activities cannot be modeled strictly in sequence, as they might vary, repeat, or overlap according to the disaster scenario.

	Assessment	Procurement	Warehousing	Transport	Reporting
Strategic level	Create mission statement Plan emergency preparedness Plan program strategy	Negotiate framework agr. Plan emergency supply strat. Plan kits Plan standard item catalogue	Plan warehouse capacities Plan warehouse network	Han transport capacities Han transport network Han transport strategy	Donor report Inventory report Damage/loss rep. Needs assess. rep.
Tactical level	Plan demand Plan emergency team Plan project activities Plan standard item list	Plan program item list Plan purchasing methods Plan sourcing methods Plan tender procedures	Plan emergency stock pos. Plan quality assurance Plan warehouselayout	Pan consolidation policy Pan transport goods Pan transport modes Pan transport routes	
Operational level	Assessiocal capacities Assessiocal sources of supply Forecast demand Identify number of benefic.	Mobilitze supplies (ad-hoc) Qualify suppliers Record order/shipment info. Select supplier	Mark and label goods Monitor stock Pick and pack goods Receive/store goods	Load goods Schedule transport/ deliveries Select transport mode/route Track and traceshipment	
	Operations support Im	nplement basic infrastructure; m	obilize equipment; mobilize per	sonnel; operate support system	g

Figure 1: Humanitarian logistics framework with task samples (Blecken, 2009)

Remote Sensing in Humanitarian Logistics: An Integrative Approach

1.2 Remote Sensing

Remote sensing involves the contactless and areal extensive exploration of the earth 's surface (Heipke, 2017). Chen et al. (2016) define this as "the science and technology of capturing, processing and analyzing imagery [...] of the earth [...], from sensors in space, in the air and on the ground". Further terms are commonly utilized: Photogrammetry, which describes the part of remote sensing focusing mainly on analysis (Toth and Jutzi, 2017), and earth observation, often used interchangeably with remote sensing, but in a more comprehensive way (McInerney and Kempeneers, 2015).

A key component of remote sensing systems are the sensors. They are determined by their spatial, temporal, and spectral resolution. Passive sensors record the visible environment - such as photo cameras - and can represent different spectral bandwidths. Active sensors - mainly LiDAR (Light Detection and Ranging) and Radar (Radio Detection and Ranging) - are less determined by visual contact, as they send electromagnetic waves, which they receive in turn. Hence, they are applicable in cloudy environments or even at night (Toth and Jutzi, 2017). Remote sensing sensors are installed on so called platforms, which can be divided into satellite-based, airborne, and drones. Furthermore, there also exist terrestrial (mobile and static) platforms (Toth and Jutzi, 2017). Satellites are very common in remote sensing. Today, approximately 50 countries operate earth observation satellites, which can differ in their constellation, sensor type, resolution, strip width, their altitude, and their orbit. Airborne platforms have been deployed for quite some time, exclusively in remote sensing. Most common are aircrafts and helicopters. Drones represent a relatively new form of platform. Here, there are a number of terms which can be identified: Often used are also the terms unmanned aircraft systems (UAS), remotely piloted aircraft system (RPAS) and unmanned aerial vehicle (UAV) (Heipke, 2017). The range of drones is very diverse, including micro very light weight (under 1 kg) drones, having only a short range and altitude, up to drones with the size and capabilities of conventional aircraft. Hence, there is no commonly accepted classification as of yet. Griffin (2014) distinguishes four types of drones: Fixed wing, rotary wing, multi-rotary wing, and airships. In summary, a wide range of technological solutions exists which can be applied in humanitarian context.

2 Humanitarian Uses of Remote Sensing

The area of application for remote sensing in humanitarian operations is manifold and will be outlined systematically in this section. As stated above, there exists a variety of different platforms for remote sensing. It should be mentioned that all these platforms are also characterized by multiple uses, as they can be utilized in areas other than remote sensing as well. Satellites, for example, enable the functionality of positioning technologies, such as portable GPS devices or vehicle tracking systems. Further, telecommunication satellites often provide the only reliable form of communication in certain disaster situations (Delmonteil and Rancourt, 2017). Aircrafts and helicopters are mainly used for tasks other than remote sensing, primarily for transportation and evacuation services in disasters scenes. The application of drones is relatively new in humanitarian operations. In addition to their application in remote sensing, they can be used for communication support as well as (small) package delivery (Soesilo and Sandvik, 2016). All these other applications are of course of high importance for humanitarian operations and particularly in humanitarian logistics. However, as this article concentrates on remote sensing, in the following possible humanitarian uses of this specific technology will be examined.

A uniform systematization of remote sensing applications in humanitarian operations does not vet exist. Terms such as assessment, mapping, or monitoring are used inconsistently and often without any definition. For the following description of humanitarian uses, the relevant terminology will be outlined briefly. One of the most common applications of remote sensing can be seen in the field of assessments. Aerial images and the resulting maps are used to evaluate the (potential) extent of disasters and to determine the scope of the required help. Assessments are regularly performed in the context of humanitarian operations. Risk and vulnerability assessments pertain to the pre-disaster phase and aim to detect the exposure to hazards and to determine the vulnerability of people regarding the identified risks (Bündnis Entwicklung Hilft, 2017). Needs assessments are mainly performed after the onset of disasters and have an enormous importance in international relief operations. They should make clear whether an intervention is needed and how to best being performed. Further, needs assessments can be useful for the determination of demand regarding relief commodities. However, the main aim of assessments is to understand the situation and to identify the problems, their sources, as well as their consequences (ICRC and IFRC, 2008). Via data acquired and processed by remote sensing, adjustments can be made to these assessment processes. In the context of this work, the term assessment

will be understood as an umbrella term for all activities related to these tasks. In order to specify assessments more in detail, the term mapping will be used if the concerning activities relate to remotely acquired and accordingly processed map material. The detection of changes by comparing data over a period of time will correspond to the term monitoring. Finally, the use of remote sensing material for management and illustration purposes will relate to the term visualization. As remote sensing techniques can be applied across all disaster management phases, a respective differentiation is helpful.

2.1 Assessment

Concerning the published works, assessments are the largest and also most diverse area of application for remote sensing in humanitarian operations. In their simplest form, remotely acquired images or videos can be used to get an overview of disaster scenes and to estimate the extent of a disaster (Tatham et al., 2017). Immediately after the onset of a disaster, remote sensing can be applied particularly in search and rescue activities. Special sensors are often required for these operations. However, for deeper analysis, further processing steps are necessary, such as creating maps or time series material.

2.1.1 Mapping

Cartographic representation and the related processing of this material can be utilized in manifold ways across the different phases of disaster management. In the pre-disaster phase, the main objective is to detect hazards, determine risks, and finally to develop disaster response plans. Hence, remote sensing supports both the mitigation and preparedness phase. Flood management can be supported by remotely acquired map material in order to detect potential breaches in dike systems or to identify flooding areas (Sharma et al., 2017). Sambah and Miura (2014) propose a tsunami vulnerability assessment by utilizing satellite images and processed digital elevation models (DEM). Further, for forest fire susceptibility and risk mapping, remote sensing has been proposed (Pradhan et al., 2007).

In post-disaster scenarios, remote sensing is mostly applied in the response phase immediately after the disaster's onset. Rapid post-disaster assessments are applicable in the management of the aftermaths of various disaster types, such as earthquakes, floods, storms, and fire as well as in the case of man-made disasters

(Griffin, 2014). Automated building damage assessments using remotely acquired data are proposed by different authors, e.g. Dinesh el al. (2013) for a tsunami case or Kakooei and Baleghi (2017) for earthquake and hurricane cases. Concerning logistics activities, the assessment of infrastructure can be beneficial, e.g. for route planning. Wang, J. et al. (2015) propose a road damage assessment by using high resolution remote sensing images. Also the identification of best locations for facilities can be supported by mapping activities. A further application is the estimation of displaced populations (Wang, S. et al., 2015). In the recovery phase, post-disaster restoration and reconstruction assessments are suggested (Adams and Friedland, 2011).

2.1.2 Monitoring

While the above mentioned mapping applications of remote sensing relate to the situation analysis at one specific time, monitoring activities are performed continuously over a longer period, comparing different stages. In the pre-disaster phase, a typical application can be found in food security management, monitoring agriculture assets and warning in the case of critical developments (Enenkel et al., 2015). Generally, warning is a highly beneficial field of application for remote sensing (Strunz et al., 2017); it might be particularly suitable for some specific disaster types with a certain lead time or specific indicators, such as tsunamis, hurricanes, or floods. In the post-disaster phase, dynamic disaster events are critical for monitoring activities. Multitemporal satellite images can be used for the watershed and flood monitoring (Rau et al., 2007). In relation to a longer time horizon, Guo et al. (2010) propose a method for monitoring the damage reduction and reconstruction after the Wenchuan earthquake via satellite imagery.

2.2 Visualization

In contrast to the above stated areas of applications, remote sensing material can also be used in order to support general management processes. On this occasion, the visual presentation of cartographic material is used for the development of emergency plans or for communication aspects (Voigt et al., 2007). Providing near time information after the onset of a disaster supports the collaboration efforts of the different humanitarian organizations (Strunz et al., 2017). These activities can be performed in both pre-disaster and post-disaster phase.
3 Comparison of Platforms

Following the systematic description of the manifold uses of remote sensing in humanitarian operations and logistics, this section aims to compare the different platforms and describe advantages and disadvantages in order to subsequently discuss the potentials of their integration.

3.1 Advantages and Limitations

The comparison of the different remote sensing platforms is to be executed keeping various aspects in mind. A complete list of these criteria can be found in Table 1. Even though the spatial resolution of satellite images has increased in recent years, their maximum is with 30 cm (for some specific satellite sensors) still lower than those of airborne and drone platforms, whose resolution can be up to 1 cm; also the spatial accuracy of geo-positions in the acquired data is higher for the latter (Toth and Jutzi, 2017). Furthermore, airborne and drone platforms provide the opportunity to acquire oblique images for more specific processing and analysis methods. Moreover, drones can be applied for ad hoc, quick response assessments with live broadcasts in a 360° angle and their susceptibility to cloudy weather situations is lower (Kakooei and Baleghi, 2017; Toth and Jutzi, 2017). The higher spatial resolution and accuracy of airborne and drone platforms result from the fact that their flying altitude is significantly lower, particularly those of small drones. As one would expect, the observed space and the resulting surface coverage is substantially lower as well (Toth and Jutzi, 2017). Here, satellites make use of their inherent advantages. Contrarily, satellites are restricted by their predefined orbit and speed, which consequently limits their repetition rate and their preparation time compared to other platforms. The periodic flight path of satellites might be a disadvantage regarding their flexibility; however, it enables the availability of pre-disaster images, which is beneficial for comparison reasons and necessary for specific map analysis methods. Advantages of drones arise from their high flexibility; they are rapidly deployable and their application can be adjusted to the operations' requirements easily (Kakooei and Baleghi, 2017). Boccardo et al. (2015) note that drones are only rapidly deployed if they are already on site. Otherwise, the transportation to the disaster scene can restrict their short preparation time. Furthermore, the complexity of use as well as the operational risk is lower for drones than for the other platforms, whereas skilled personnel is

generally required for remote sensing applications. However, whereas the deployment of satellite technology and airborne platforms has been established in the humanitarian context, the use of drones implies several legal and ethical issues. Most countries do not have any legal framework for the use of drones and also questions regarding privacy issues have not yet been answered. Moreover, the utilization of drones particularly in conflict-based disasters is seen as problematic (Tatham et al., 2017). Finally, the costs of the different remote sensing platforms vary significantly. Regarding the cost aspect, drones are the most favorable platform (Toth and Jutzi, 2017). Considering all these advantages and limitations, the potential of integrating several platforms is discussed briefly.

	Satellite	Airborne	Drone
Spatial resolution	0.3-300 m	0.5-0.25 m	0.01-0.05 m
Spatial accuracy	1-3 m	0.05-0.1 m	0.01-0.25 m
Angle of view	Vertical	Vertical/oblique	360°
Weather susceptibility	High	Medium	Medium
Space of observation	Global	Regional	Local
Surface coverage	High (>10 km)	Medium (1 km)	Low (0.1 km)
Repetition rate	Day(s)	Hours	Minutes
Preparation time	1-2 days	Several hours	Within an hour
Pre-disaster image	Available	Probable	No
Maneuverability	No/limited	Medium	High
Usability	Difficult	Complex	Easy
Operational risk	Medium	High	Low
Operat. conditions	Established	Well established	Immature
Costs	High	Medium	Low

Table 1: Platform comparison (Kakooei and Baleghi, 2017; Toth and Jutzi, 2017)

3.2 Potentials of Integration

Concerning the differences between the platforms for remote sensing, it has been shown that the disadvantages of the one platform are often the advantages of the other one. Hence, it seems plausible to consider the integration or combination of remote sensing data acquired via different platforms. In academic literature, a systematic discussion about such an integration for the humanitarian context does not yet exist. To the best of our knowledge, only Kakooei and Baleghi (2017) demonstrated the integration for the specific application of building damage assessments. However, the opportunities of an integrated approach are greater.

The benefits of integration can be systemized according to quality, time, flexibility, and cost aspects. In terms of quality the application of drones might be more reasonable due to their higher resolution. However, according to the performed task, it might be more important to get an overview of a specific region. For example, the condition of the road network and the related infrastructure could be assessed via drone, whereas the rest of the environment is gathered via satellite. In addition, oblique images from airborne and drone platforms can result in enhanced damage scale estimations but in turn they are limited in its spatial extension. On this occasion, the time aspect is critical as well. Considering the above stated uses of remote sensing in humanitarian operations, satellites could be mainly deployed for monitoring and to generate baseline material in the predisaster phase which can be used for comparison. The post-disaster data might be gathered via airborne or drone platforms as they inhere a higher repetition rate and a lower preparation time. On the one hand, this might be applicable for the assessment of critical logistical infrastructure, such as airports, ports, or warehouses. On the other hand, enhanced assessments can be useful for rapid demand determination. This flexibility is also a benefit of an integrated consideration of multiple platforms. Drones and airborne platforms are rapidly deployable after the disaster's onset and easier in use than satellites. They can therefore support the immediate response phase whereas the latter one are favorable in the longer term. Finally, cost aspects can be considered as well. Using the most reasonable platform for each task, can reduce the cost of remote sensing activities significantly.

4 Experimental Remote Sensing by Drone

In order to demonstrate the practical application of the integration of remote sensing data, experimental drone flights have been performed and the acquired data processed. The aim is to shown how the data acquired by drone can be processed. Therefore, a 3D model will be computed in order to generate orthoimages as well as elevation models, which in turn can be integrated into a Geographic Information System (GIS). These systems are capable of combining data from different remote sensing platforms. The technical realization of these tasks includes the three main working steps of data acquisition, processing, as well as their integration and will be demonstrated more in detail in the following. Finally, results are presented and compared and a judgment in terms of the applicability of the proposed integration will be given.

4.1 Technical Realization

The technical realization includes the following three main steps: First of all, the data has to be acquired, followed by the steps of data processing and the integration into a GIS.

4.1.1 Data Acquisition

For data acquisition, a commercially available hexacoper (Airborne Robotics XR6, diameter : 950 mm, max. payload: 1.2 kg, max. speed: 15 m/s, max. flight time: 15 min) was used. The Sony NEX-7 Alpha camera was utilized with a fixed focallength of 19 mm, providing pictures with a resolution of 24 megapixels. With its 25-area contrast autofocus and the quick shutter lag, this camera is particularly suitable for remote sensing. For flight planning, the software MissionPlanner (1.3.49) was applied. By compiling waypoints, this plan can be transmitted to the flight-control device of the drone, specifying its precise flight route. For the experimental flights, a test area was chosen, which meets specific requirements. At a minimum, this area should include one street with one intersection, as well as accompanying infrastructure (e.g. power cables) and vegetation within a small range of about two hectares (this limitation is set by the technical range of the drone). Optimally, the area further includes different forms of vegetation and different types of streets as well as buildings. Such an area, meeting almost all



Figure 2: Test flight area and drone (own figure)

these requirements, was found in the district of Meißen (Germany) north of the municipality of Sörnewitz. Figure 2 illustrates the location of the flight area and the flight plan as well as the used drone. The drone flights were performed on a sunny day in October 2017. Two flights at different altitudes were conducted: The first flight at an altitude of 40 m (total flight duration: 12:00 min) and the second flight at an altitude of 80 m (total flight duration: 7:10 min).

4.1.2 Processing

A first evaluation of the acquired data already took place on site in order to decide whether enough images were produced and if their quality was sufficient. During the flights an adequate number images for the given tasks were taken; only a small number had to be removed due to fuzziness. First of all, the taken pictures had to be geo-referenced, which was done by utilizing the GPS data of the drone. Therefore, a camera position tool was used. This step can be omitted if the camera

Remote Sensing in Humanitarian Logistics: An Integrative Approach

Figure 3: Thick point cloud generated by Agisoft Photoscan (screenshot)

already contains a GPS function (which was not the case for the utilized camera). For the 3D model generation, the software Agisoft PhotoScan (1.3.4) was applied. Agisoft is able to use the taken 2D images in order to generate a geo-referenced point cloud (Figure 3). This step is essential for generating orthoimages and DEMs, which in turn are needed to embed the data into a GIS. Orthoimages are straight, distortion-free and scaled illustrations of the earth's surface, whereas DEMs contain the respective elevation information. For their generation, the 2D images have to be aligned to create a thin point cloud in a first step, followed by the creation of the thick point cloud and the polygon mesh.

4.1.3 Integration

For the final step, the integration into a GIS, the software QGIS (2.18.13) was used. It is one of the most common open source GISs and enables the recording, manipulation, management, and presentation of spatial data. The system's capabilities go far beyond the demonstrated aspects here. However, in our case the GIS was used for the integration of the previously generated orthoimage, the refining of the respective geo-references, and the representation of the DEM. A GIS enables the combination of different remote sensing data, such as satellite or airborne



Figure 4: QGIS with Google layer and orthoimage, 0.1 x 0.1m (screenshot)

images, as well as geographical maps. In this case, a referenced, freely available Google Maps layer (airborne images) and the orthoimage were imported. The result of the integration within QGIS with our own layer in the foreground can be seen in Figure 4. As the geo-references of the drone's positioning system are not sufficient in all cases, a refining can be performed. Therefore, the two layers can be aligned according to distinctive points.

4.2 Results

As a result of the two drone test flights and the subsequent processing of the acquired remote sensing data, orthoimages and DEMs with a resolution of up 1 cm2/Px could be generated. The quality of geo-referencing can also be stated as high-grade, as their deviation is less than 20 cm (absolute) and the total computing time was between 107 and 165 min. All data could be processed by the GIS without any difficulties and obvious errors. The high level of detail is shown in Figure 5 which presents the resulting orthoimage. The resolution of the overview orthoimage corresponds approximately with the quality of the airborne image of the freely available Google Maps layer. The enlarged image details make clear one of the main advantages of drones compared to other remote sensing platforms. It is demonstrated that even cracks in the asphalt or mud holes as well as power cables can be detected.



Figure 5: Orthoimage (40 m, 1 cm²/Px) (own figure)

5 Conclusion

Remote sensing platforms inhere their specific advantages and limitations as shown in section 3. Hence, it can be assumed that an integrated approach in terms of combining data from different platforms will be beneficial for remote sensing activities in humanitarian context. The practical applicability of such an integration has been demonstrated in section 4. In the following, possible application of an integrated view in humanitarian logistics will be discussed. Moreover, a summary and an outlook for further research will be given.

5.1 Discussion

The above stated integrated approach in remote sensing can be supportive for different humanitarian logistics tasks. First of all, procurement and distribution processes are highly affected by sound information beneficiaries' needs. Knowing which and where these needs exist is crucial for logistical activities. Here, improved procedures comparing pre-disaster (generated by satellite) and postdisaster data (airborne or drone) could be applied for damage and needs assessments. Furthermore, the utilization of remote sensing for mapping task should be highlighted. In humanitarian operations, the integrated view could be beneficial in the distribution context. Infrastructure assessments are of high importance and they should be balanced regarding high surface coverage and high level of detail. Hence, satellite data could be used for general overview assessments, extended by drone data for detailed infrastructure assessment. According to different types of disasters, the possible applications are manifold, for example concerning the accessibility of certain areas by roads blocked by landslides, water or fire. The strength of satellite images in getting an overview and drones in providing detailed data can be mutual supportive. Further application areas can be found in warehousing concerning the facility location. Similar aspects as mentioned above can be beneficial. The initial assessment of best locations for transit points or warehouses could be done by satellite, whereas the detailed analysis will be supported by drone images. Finally, the visualization aspect can be mentioned. Decision makers in humanitarian logistics need to be provided with detailed information from the site. As drones are not capable to provide images from all areas, particularly in disasters with a high coverage, the supplement with according satellite images is appropriate. This approach should not be limited solely to the different remote sensing platforms, but should also consider data from different humanitarian actors.

5.2 Summary and Outlook

This work shows that the combination of different remote sensing platforms is highly beneficial in the context of humanitarian operations and logistics. Therefore, the possible areas for application have been categorized and outlined regarding certain assessment tasks, including mapping and monitoring activities as well as visualization tasks. Following, the advantages and limitations of the platforms were highlighted. It was shown that the platforms are mutually supportive. Hence, an integration of remote sensing data from different sources can be highly beneficial as support for humanitarian logistics activities such as procurement, distribution and warehousing in terms of quality, time, flexibility, and cost. In the last section, the practical implementation of such an integrative approach was demonstrated. Therefore, drone test flights were conducted and the gathered material processed and integrated.

The area of application for remote sensing in humanitarian operations and particularly for logistics activities is manifold. Further research is necessary to systematically identify the potentials of remote sensing, regardless of their platform, and according to the disaster management phases, on the one hand, and the different disaster types, on the other. Moreover, the requirements regarding these areas of applications have to be identified. This relates mainly to the necessary resolution and the time horizon concerning the gathered data. Finally, the integration into the information management processes of humanitarian organizations and the sharing of information amongst the various actors should be evaluated. Investigating these questions could lead to more efficient and expedient humanitarian operations, not just in terms of logistical aspects.

References

- Adams, S. and C. Friedland (2011). "A Survey of Unmanned Aerial Vehicle (UAV) Usage for Imagery Collection in Disaster Research and Management". In: *Proceedings of the 9th International Workshop on Remote Sensing for Disaster Response*. 15-16 September 2011. Stanford, CA, USA.
- Altay, N. and R. Pal (2014). "Information Diffusion among Agents: Implications for Humanitarian Operations". In: *Production and Operations Management* 23.6, pp. 1015–1027.

Baumgarten, H. (2011). "Humanitäre Logistik: Herausforderungen und Potenziale der Logistik in der humanitären Hilfe". In: ed. by H. Baumgarten, J. Schwarz, and M. Keßler. Hamburg: DVV Media Group. Chap. Humanitäre Logistik, pp. 3–15.

Blecken, A. (2009). "A Reference Task Model for Supply Chain Processes of Humanitarian Organisations". PhD thesis. Universität Paderborn. Bündnis Entwicklung Hilft (2017). WorldRiskReport: Analysis and Prospects 2017. online.

- Chen, J., I. Dowman, S. Li, Z. Li, M. Madden, J. Mills, N. Paparoditis, F. Rottensteiner, M. Sester, C. Toth, T. J., and C. Heipke (2016). "Information from Imagery: ISPRS Scientific Vision and Research Agenda". In: *ISPRS Journal of Photogrammetry and Remote Sensing* 115, pp. 3–21.
- Delmonteil, F.-X. and M.-È. Rancourt (2017). "The Role of Satellite Technologies in Relief Logistics". In: Journal of Humanitarian Logistics and Supply Chain Management 7.1, pp. 57–78.
- Dinesh, C., A. Bari, R. Dissanayake, and M. Tamura (2013). "Detecting and Assessment of Tsunami Building Damage Using High-Resolution Satellite Images with GIS Data". In: International Journal of Disaster Resilience in the Built Environment 4.2, pp. 132–144.
- Enenkel, M., L. See, M. Karner, M. Álvarez, E. Rogenhofer, C. Baraldès-Vallverdú, C. Lanusse, N. Salse, and H. El-Shemy (2015). "Food Security Monitoring via Mobile Data Collection and Remote Sensing: Results from the Central African Republic". In: *PLOS ONE* 10.11, e0142030.
- Griffin, G. (2014). "The Use of Unmanned Aerial Vehicles for Disaster Management". In: *Geomatica* 68.4, pp. 265–281.
- Guo, H., L. Liu, L. Lei, Y. Wu, L. Li, B. Zhang, Z. Zuo, and Z. Li (2010). "Dynamic Analysis of the Wenchuan Earthquake Disaster and Reconstruction with 3-Year Remote Sensing Data". In: International Journal of Digital Earth 3.4, pp. 355–364.
- Gupta, S., M. Starr, R. Farahani, and N. Matinrad (2016). "Disaster Management from a POM Perspective: Mapping a New Domain". In: *Production and Operations Management* 25.10, pp. 1611– 1637.
- Heipke, C. (2017). "Photogrammetrie und Fernerkundung: Handbuch der Geodäsie". In: ed. by C. Heipke. Berlin: Springer. Chap. Photogrammetrie und Fernerkundung: Eine Einführung, pp. 1–27.
- ICRC and IFRC (2008). Guidelines for Assessment in Emergencies. online.
- Kakooei, M. and Y. Baleghi (2017). "Fusion of Satellite, Aircraft, and UAV Data for Automatic Disaster Damage Assessment". In: International Journal of Remote Sensing 38.8-10, pp. 2511–2534.
- Kim, M., R. Sharman, C. Cook-Cottone, H. Rao, and S. Upadhyaya (2012). "Assessing Roles of People, Technology and Structure in Emergency Man-agement Systems: A Public Sector Perspective". In: Behaviour & Information Technology 31.12, pp. 1147–1160.
- Kovács, G. and K. Spens (2011). "Trends and Developments in Humanitarian Logistics: A Gap Analysis". In: International Journal of Physical Distribution & Logistics Management 39.6, pp. 506– 528.
- McInerney, D. and P. Kempeneers (2015). Open Source Geospatial Tools: Applications in Earth Observation. Cham: Springer International Publishing.
- OCHA (2013). Disaster Response in Asia and the Pacific. online.
- Pradhan, B., M. Dini Hairi Bin Suliman, and M. Arshad Bin Awang (2007). "Forest Fire Susceptibility and Risk Mapping Using Remote Sensing and Geo-graphical Information Systems (GIS)". In: Disaster Prevention and Management: An International Journal 16.3, pp. 344–352.
- Rau, J.-Y., L.-C. Chen, J.-K. Liu, and T.-H. Wu (2007). "Dynamics Monitoring and Disaster Assessment for Watershed Management Using Time-Series Satellite Images". In: IEEE Transactions on Geoscience and Remote Sensing 45.6, pp. 1641–1649.
- Sambah, A. and F. Miura (2014). "Remote Sensing and Spatial Multi-Criteria Analysis for Tsunami Vulnerabil-ity Assessment". In: Disaster Prevention and Management: An International Journal 23.3, pp. 271–295.
- Schwarz, J. (2012). "Humanitäre Logistik für die Versorgungsproblematik in Subsahara-Afrika: Aufbau von Logistikkapazität durch Wissenstransfer". PhD thesis. Technische Universität Berlin.

- Sharma, V., N. Mishra, A. Shukla, A. Yadav, G. Rao, and V. Bhanumurthy (2017). "Satellite Data Planning for Flood Mapping Activities Based on High Rainfall Events Generated using TRMM, GEFS and Disaster News". In: Annals of GIS 23.2, pp. 131–140.
- Soesilo, D. and K. Sandvik (2016). Drones in Humanitarian Action: A Survey on Perceptions and Applications. Geneva: Swiss Foundation for Mine Action (FSD).
- Strunz, G., S. Martinis, and E. Schöpfer (2017). "Photogrammetrie und Fernerkundung: Handbuch der Geodäsie". In: ed. by C. Heipke. Berlin: Springer. Chap. Beiträge der Fernerkundung zur Unterstützung des Katastrophenmanage-ments und der humanitären Hilfe, pp. 805–839.
- Tatham, P., C. Ball, Y. Wu, and P. Diplas (2017). "Long-Endurance Remotely Piloted Aircraft Systems (LE-RPAS) Support for Humanitarian Logistic Operations". In: *Journal of Humanitarian Logistics* and Supply Chain Management 7.1, pp. 2–25.
- Thomas, A. and L. Kopczak (2005). From Logistics to Supply Chain Management: The Path in Humanitarian Sector. online.
- Toth, C. and B. Jutzi (2017). "Photogrammetrie und Fernerkundung: Handbuch der Geodäsie". In: ed. by C. Heipke. Berlin: Springer. Chap. Plattformen und Sensoren für die Fernerkundung und deren Geopositionierung, pp. 29–64.
- Voigt, S., T. Kemper, T. Riedlinger, R. Kiefl, K. Scholte, and H. Mehl (2007). "Satellite Image Analysis for Disaster and Crisis-Management Support". In: *IEEE Transactions on Geoscience and Remote Sensing* 45.6, pp. 1520–1528.
- Wang, J., Q. Qin, J. Zhao, X. Ye, X. Feng, X. Qin, and X. Yang (2015). "Knowledge-Based Detection and Assessment of Damaged Roads Using Post-Disaster High-Resolution Remote Sensing Image". In: *Remote Sensing* 7.12, pp. 4948–4967.
- Wang, S., E. So, and P. Smith (2015). "Detecting Tents to Estimate the Displaced Populations for Post-Disaster Re-lief Using High Resolution Satellite Imagery". In: International Journal of Applied Earth Observation and Geoinformation 36, pp. 87–93.

Impact of Managerial Risk-taking and IRM on Innovation

Fatemeh Seidiaghilabadi¹, Ebrahim Abbassi¹, Zahra Seidiaghilabadi¹

1 - Islamic Azad University

Innovation in any business comes with risk. This study aimed to explore the role of managerial risk-taking and integrated risk management (IRM) on innovation. To verify the hypotheses, a questionnaire was designed. Data were collected from 109 Insurance Managers from Iran Insurance Companies. PLS structural equation modeling was employed to test both measurement and structural model. The results demonstrate that integrated risk management and managers' risktaking have a positive impact on innovation. This is the first study to explicitly and separately consider the effects of managerial risk-taking and integrated risk management (IRM) on innovation in the Insurance sector. Due to the lack of scales to analyze IRM, questionnaires which adopted from previous studies may not be the best scales to measure variables. This study focuses on Iran insurance industry, which limits its generalizability. Our findings highlight the need for managers in high risk-taking behavior to encourage employees to be more creative and develop organizational innovation. The authors discovered that firms should perform an effective IRM system that oversees many systematic risks through organization innovation process.

Keywords: Innovation; Managerial risk-taking; IRM; PLS-SEM

First recieved: 28.May.2018 Revised: 02.Jul.2018 Accepted: 12.Jul.2018

1 Introduction

Innovation is a popular concept that has attracted the interest of both academic and business communities (Santos et al., 2014). In addition, Innovation is critically important and necessary for firms to respond to rapid changes (Kwak, Seo, and Mason, 2018), and dynamic competitive business environment in order to grow and survive (Tushman and O'Reilly III, 1996; Dess and Picken, 2000; Bowers and Khorakian, 2014; Wu and Wu, 2014; Kim, Choi and Skilton, 2015). Innovation is a high-risk process and is prone to uncertainty in business.

On the other hand, risk management is emerging as an essential contributor to most fields of decision making and firm's control in any business at the complexity and uncertainty business environment (Giannakis, Croom, and Slack, 2004; Ritchie and Brindley, 2007; Eckles, Hoyt, and Miller, 2014). This complexity shows that risks arise from various sources. Scholars believe that the extensive use of explicit risk managing might reduce expenditure on innovation failure and if it is well implemented, firms will reach their innovative plans more accelerator (Bowers and Khorakian, 2014; Stosic et al., 2017). Clearly, appropriate risk management would encourage innovation in the organization.

March (1996), argues that risk-taking is necessary for innovation. Confirming March's argument, previous research has shown that there exists a close relationship between risk-taking and innovation. Researchers believe that managers play the role of strategic decision-makers, capable of identifying opportunities and making the right decisions about innovation within the organization (García-Granero et al., 2015). In fact, managerial risk-taking requires significant investment that hopes to bring out success or profit, with a high possibility of failure or heavy debt(Latham and Braun, 2008).

Additionally, not only managers and CEO's risk-taking stimulate innovation, but also firms need to manage all sorts of risks surrounded an innovation project from planning to implementation (Wu and Wu, 2014). Integrating all risks through the different phases of innovation can (Wu and Wu 2014) help managers make the critical decision to abandon poor innovation plan (Bowers and Khorakian,2014). This integrated approach provides for real risk management that is coordinated among all parties involved in the control of innovation's risk. It also provides management with the capacity to monitor risk mitigation performance. Besides, the potency of the risk management within an integrated approach is determined by its clear classification of any levels of business (Bilgin and Danis, 2016). The integrated risk-taking management encompasses operational risks. According to Meulbroek(2002a), operational risks, generated from individuals, processes, and physical properties, can exert influence on innovation performance. Thus, such risks need to be carefully managed.

Although the effect of risk-taking on innovation has been well documented in various studies, research has failed to examine the impact of integrated risk-taking management on innovation. Moreover, the effect of these two types of approach on innovation has remained underexplored.

Accordingly, since innovation in the insurance industry has received little attention, the chief aim of this article is to fill the gap and to examine the effect of managerial risk-taking and IRM on innovation. The study, therefore, attempts to provide answers to the following research questions:

1. What is the effect of managers' risk-taking on insurance companies' innovation?

2. What is the effect of integrated risk-taking management on insurance companies' innovation?

The article is structured as follows. First, reviewed the literature . Next, a hypothetical model will be presented. The SEM model will be tested using data from 109 insurance managers working in Iran Insurance Companies. The article concludes with discussion, conclusion, managerial implications, and limitations of the study.

2 Literature Review

This chapter, briefly explain innovation within the context of the organization. Then it provides an introduction to the relationship between managers' risktaking and integrated risk management which aim at promoting organizational innovation.

2.1 Innovation

Within the service industry, the primary way to gain a competitive advantage is innovation. Birkinshaw et al.,(2011) showed that product/service innovations are

just the tip of the innovation iceberg (Medrano and Olarte-Pascual, 2016). Scholars have thoroughly mentioned that the quest for innovation can be an important strategy which organizations with hyper-competitive characteristics use to remain competitive in a progressively dynamic, speedily changing and complicated market (Liu et al., 2017). In the innovation literature, various types of innovation have been considered. It can be product, process, administrative, technological, marketing, radical or incremental, (García-Granero et al., 2015; Liu et al., 2017). Service or product innovation is the first-time commercial usage of a product or service, which is new to the marketplace, whereas process innovation is the execution of methods that are new to the company, but not necessarily new on the market (Jeschke et al., 2017). Hence, this process requires clearness of thought and the capability to get things done. Evan(1966), pointed out that administrative innovation can be a concept for new policy, the allocation of resources, the structuring of responsibilities, of authority, of rewards (Pauget and Dammak, 2018). Administrative innovation increases productivity by guaranteeing efficiency in internal processes and individuals and business. Specifically, administrative innovation refers to changes in the characteristics of organizational or institutional elements. Then, adoption of any types of innovation is determined by an attitude towards the organization innovation partially.

2.2 Managing Risk-taking

Risk-taking is considered as one of the most important activities of managers to encourage innovation throughout the organization. Risk-Taking behavior is an individual's behavior in risky conditions, which is characterized by using the degree of risks involved in decision-making (Nkundabanyanga et al., 2015).

Research has shown that managers' inclination toward risk-taking has considerable influence on a company's capacity for innovation. As noted earlier, managerial risk-taking "involves investing significant resources in activities with a high possibility of failure, which includes incurring heavy debt or making large resource commitments in the hope of reaping potentially high benefits" (García-Granero et al., 2015). Hence, if expected values of taking risks for two strategies are comparable, but one is a considerable uncertain, managers will choose the one with a more specific result.

2.3 Integrated Risk Management (IRM)

Typically, RM has been about each administrator dealing with specific risks which might affect some firms' goals. In a very meanwhile, regarding risk literature, RM methods are changing from controlling risk independently and departments and in a Silo-based approach to an integrated risk approach (Tommerberg, 2010; Meulbroek, 2001, 2002b). An IRM allows firms to control a wide array of risks holistically (Togok, Isa, and Zainuddin, 2016) and consists of many different facets of an organization's activities (Wu and Wu, 2014). Such an integrated risk model combines all sorts of risks with an integrative focus (Andersen, 2008) and helps managers to identify, control, evaluate, and monitor all risks on the specific categories to make a great decision. Thus, IRM offers a steady picture of risk within the entire organization.

Accordingly, since the integrated approach to managerial risk-taking includes differing facets of the organization, companies adopting this approach, are capable of dealing with the skills and expertise necessary to face potential risks.

Although integrated risk management differs from Enterprise risk management (ERM), the other concept of holistic risk procedure, both add a wide variety of strategic, functional, and financial decisions for handling risk (Oxelheim, Wihlborg and Thorsheim, 2011; (Meulbroek, 2002b). Table 1 provides some recent definitions of IRM. Some scholars maintained that the success of IRM is greatly dependent on how proficiently it is applied in an organization(Nocco and Stulz, 2006). Meulbroek, (2002a, 2002b), observed that there are three ways of employing risk management aims: modifying the firm's procedure, altering its capital structure and utilizing targeted financial instruments. IRM identifies a theory that managers for choosing the perfect solutions (Tommerberg, 2010). To put it simply, IRM implementation elements depict the critical path that a comprehensive risk containment program should take to ensure proactive well as reactive measures to lessen systematic risk exposures.

Comprehensive risk containment program should take to ensure proactive well as reactive measures to lessen systematic risk exposures.

Table 1: IRM definitions and descriptions

Scholars	Definitions
Meulbroek (2002a); Bromiley et al., (2015)	IRM is the identification and evaluation of the common risks that affect the significant value and enforcing a company-wide approach to managing those risks.
Miller and Waller, (2003)	IRM is the consideration of a full variety of uncertain contingencies that affect enterprise overall performance.
(Andersen, 2008)	Activities into one unified framework and enables identification of such interdependencies as a consequence.
(Wu and Wu, 2014)	IRM entails identity of unique activities or situations applicable to product innovation's risks and possibilities typically, assessing and measuring them, integrating the risks, and formulating plans to restrict them. The process additionally consists of executing those plans and tracking development.

3 Developing Hypotheses

According to the above-stated discussion, a hypothetical model is presented in Figure 1 illustrates the effect of managers' inclination towards risk-taking and integrated risk-taking management on innovation.

3.1 Risk-appetite and Innovation

Risk-taking comes with an essential influence on organizations' long-term development and innovative activities (Li and Tang, 2010). Consequently, risk-taking is manifested in the managers' willingness to utilize opportunities, aiding their firms in getting competitive advantages through innovation activities (Faccio, Marchica and Mura, 2011). Researchers on innovation literature agree that innovations and creative behaviors are closely associated with risk-taking. In the primary, however, risk-taking should be considered as a relevant attribute of managers to accomplish innovation. In addition, managers have the authority to improve and shape the organization by their high-risk decisions. Therefore, we put forward the following hypothesis:

Hypothesis 1: Managerial risk-taking has a positive impact on innovation.

3.2 IRM and Innovation

Innovation aligns with actions that possess an excessive degree of risk, and to be successful; managers ought to manipulate this procedure and eliminate the risks into account. Tidd, Bessant, and Pavitt (2005) emphasized that even though innovation has different types, scales, and sectors, it is a process, which needs to be managed. Johnson(2010), believed that not only does risk management circulate innovation in advance, but it additionally increases its speed (Bowers and Khorakian, 2014). IRM is set the recognition and assessment of precise events or conditions related to products' risks and possibilities for innovation, integrating risks and plans to reduce risks. Implementation of those packages and their progress is likewise part of this procedure. As a result, even though each risk control hobby reduces dangers one after the other (e.g., Technical risks), integrated hazard control eliminates a few risks and preserves others by growing risk



Figure 1: Conceptual model

graphs (Meulbroek, 2002b). Because of different risks in the innovation procedure, the IRM approach has emerged as one of the very most essential additives of innovation. Therefore, we formulated the second hypothesis as follows:

Hypothesis 2: integrated risk management has a positive impact on innovation

4 Methodology

The authors verified the research hypotheses based on the data obtained from insurance companies in Iran. Information in the present research was gathered from human-resource, risk, research and development managers, and managers who somehow handle innovation of insurance, and assessment of its related risks in Iran's insurance companies.

Focusing on one industry, closely examined its specific characteristics associated with innovation. The reason was that, according to Santarelli and Piergiovanni, (1996), researching a small unit make differences get reduced in data.

4.1 Data Collection

To increase response rates, and to ensure the participation of the managers, emails were sent to some managers of insurance companies. The aim of the study was explained in those emails. After receiving the managers' agreement, a questionnaire was sent to them. In addition to sending emails to some directors, some phone calls were made to directors and deputies, and they were asked to participate in the study. Moreover, some meetings were arranged with respondents to complete questionnaires. This method ensured sufficient access to the right audience, proper use, and understanding of words, and increased the response rate. Finally, 109 valid responses were obtained, equivalent to a response rate of 94.78% (out of 115 distributed questionnaires).

4.2 Questionnaire Development

There were three primary variables in this study, including the manager's risktaking, integrated risk management (IRM) and Innovation. A seven-point Likertscale ranging from "1" (strongly disagree) to "7" (strongly agree) was used to measure the existing variables. Three constructs for innovation, were adapted from Jimenez-Jimenez, Sanz Valle and Hernandez-Espallardo, (2008). The managerial risk-taking and IRM were measured by three and one items, respectively, which were adapted from previous research studies by Covin and Slevin (1986) and García-Granero et al. (2015), whereas IRM was adopted from the study of (Wu and Wu, 2014). We only control firm age (number of years since the company's creation); Firm age was measured in two separate categories:" lower than 20 years= 1 and more than 20 years activity= 2".

4.3 Data Analysis and Results

Verifying the adequacy of the data check is essential before performing factor analysis. The KMO index and Bartlett's test were both used to look for the appropriateness of sample adequacy. Thus, KMO in a factorial analysis was 0.896, and thus, the sampling adequacy for this approach is significantly above the threshold value of 0.7(Bilgin and Danis, 2016). The significance level of Bartlett's test (x2 = 873.474; df= 78, p= 0.000) is smaller than 5 percent strengthened its adequacy. Finally, the results verified that the executed dataset in this paper was ideal for exploratory factor evaluation procedures.

4.4 Measurement Model

The authors used Smart PLS software V.2.3.7 (Ringle, Wende and Becker, 2015) to analyze the measurement and structural models (Hair et al., 2017, 2016). We examined the measurement model, through Convergent and discriminant validity. Assessment of the convergent validity includes factor loadings> 0.7, composite reliability (CR)> 0.7 and average variance extracted (AVE)> 0.5 and Cronbach alpha > 0.7 (Hair et al., 2013).

רוו אר-טו מפו כטואנו מכוא	Second-order constructs	ltem	Loadings	t-value	AVE	CR	a
Managerial Risk-taking		MR1	0/7065	7/32***	0/64	0/84	0/71
(MR)		MR2	0/9078	49/45			
		MR3	0/7803	10/42			
IRM		IRM	1/0000		1/0	1/0	1/0
Product innovation		product_1	0/9044	38/40	0/76	06/0	0/84
		product_2	0/8970	51/48			
	Innovation	product_3	0/8236	20/04			
Process innovation		process_1	0/7397	12/14	0//0	0/87	0/79
		process_2	0/8905	41/63			
		process_3	0/8872	36/41			
Administrative innovation		Admin_1	0/8714	23/60	0/80	0/92	0/87
		Admin_2	0/9312	63/76			
		Admin_3	0/8854	32/04			

Table 2: Measurement model

4 Methodology

Construct	IRM	MR	Administrative	Process	Product
IRM	1/0000				
MR	0/4597	0/8025			
Administrative	0/6369	0/3419	0/8964		
Process	0/5707	0/5207	0/6464	0/8421	
Product	0/5982	0/5107	0/6163	0/6921	0/8758

Table 3: Discriminant validity – Fornell- Larcker criterion

Note: The bold values in the above matrix are the squared correlations between the latent constructs, and the diagonal values are AVEs

Table 2 shows that the results of the measurement model exceeded the recommended values, thus indicating sufficient convergent validity. In addition to Table 3, Table 4, and Table 5 shows the Discriminant validity. In this regard, the square root of the AVE in table 3, is higher than its highest correlation with any other construct (Fornell and Larcker, 1981; Hair et al., 2017, 2016).

The comparison of cross-loadings in Table 4 indicates that an indicator's loadings are higher than other loadings for its construction in the same column and same row. The Heterotrait- the Monotrait ratio of correlations (HTMT), is another Discriminant validity.

HTMT values shown in Table 5 imply that all values are below the threshold of 0.9 (Hair et al., 2017). Thus, the discriminant validity is established between the latent constructs and, overall, the reflective constructs are reliable and valid.

4.5 Structural Equation Modeling

As mentioned in the previous section SmartPLS 3.2.7 was used to test the structural model and hypotheses (Ringle, Wende and Becker, 2015). The primary criterion for spiritual model evolution is R2, which represents the amount of explained variance of each endogenous latent variable (Hair et al., 2016). Table 6 shows that the R2 for the entire model is 0/5276, which presents a reasonable explanation of the model. In addition, the effects of control variable (CV) on innovation was tested by adding Firm age as a CV to the model. By adding firm age to the model, the R2 of innovation has change from0/526 to 0/539. Using effect

	Managerial risk-taking	IRM	Product	Process	Administrative
Mr1	0/7065	0/2769	0/3747	0/3351	0/1975
Mr2	0/9078	0/4104	0/4527	0/4906	0/3599
Mr3	0/7803	0/4085	0/4002	0/4123	0/2437
IRM1	0/4597	1/0000	0/5982	0/5707	0/6369
product_1	0/4931	0/5259	0/9044	0/6298	0/5531
product_2	0/4565	0/5399	0/8970	0/6745	0/6149
product_3	0/3858	0/5065	0/8236	0/4989	0/4357
process_1	0/3844	0/2794	0/3904	0/7397	0/4570
process_2	0/4012	0/6154	0/6901	0/8905	0/6071
process_3	0/5285	0/5015	0/6284	0/8872	0/5562
Admin_1	0/2167	0/5057	0/4914	0/5308	0/8714
Admin_2	0/2863	0/5647	0/5732	0/6037	0/9312
Admin_3	0/4094	0/6376	0/5879	0/6000	0/8854

Table 4: Cross-Loadings

Notes: Italic values are loadings for each item that is above the recommended value of 0.5; an item's loadings on its variable are higher than all of its cross-loadings with other variable

Table 5: Discriminant validity– Heterotrait- the Monotrait ratio of correlations (HTMT)

Construct	IRM	MR	Administrative	Process
MR	0/5384			
Administrative	0/6783	0/4170		
Process	0/6212	0/6821	0/7681	
Product	0/6500	0/6527	0/7059	0/8183

Note: The criterion for HTMT is 0.90 (Hair et al., 2017)

Impact of Managerial Risk-taking and IRM on Innovation

Endogenous latent variables	R Square	Q2-value
Innovation	0/539	0/284

Table 6: Results of R2 and Q2 value*

Notes: *Q2 value = effect size: 0.02 = small; 0.15 = medium; 0.35 = large

size suggested by Chin, Marcolin, and Newsted, (2003):f2=R2included – R2 excluded/1-R2included, The effect size 0/026 was obtained, which shows small effect. Thus, the result illustrates that firm age does not have a significant effect.

When blindfolding is run for all endogenous latent constructs in the model, they all have Q2 values considerably above zero. Table 6 shows that all Q2 values are providing support for predictive relevance (Hair et al., 2017). Table 6 shows the results of R2 and Q2 values. The result of R2 value based on Table 6, indicates that 53 percent of the variance, adjusted R2, in Innovation is explained By IRM and risk-taking with firm age as CV.



Figure 2: Structural Model Test result

Hypothesis	Path direction	Path coeff Without CV	icient With CV	T-value Without CV	With CV	Decision
H1	MR->Innovation	0/2582	0/266	3/52	3/701	Supported
H2	IRM->Innovation	0/5705	0/589	6/97	7/27	Supported

Table 7: Structural results and hypothesis testing

Table 7 and Figure 2 show the structural model analysis. The results showed that the relationships between Managerial risk-taking and innovation (Path coefficient = 0/266, t-value = 3/701 with a p-value<0.01). Thus, H1 is supported and predicted that managerial risk-taking positively affects innovation in the insurance industry. Furthermore, as Table 7 shows, the relationship between IRM and innovation is significant (Path coefficient = 0/589, t-value = 7/27 and with a p-value<0.01). Thus, the H2 is supported as well.

5 Conclusion

The main aim of our study has been the study of risk management, both integrated risk management (IRM) and managerial risk-taking (MR). To the best of our knowledge, it is a new line of study, which is still fragmented and with little empirical evidence. Given this gap, our study presents, after a theoretical review of the variables, an empirical model which was tested on a sample of 109 managers from Iran's' insurance companies in different types of positions.

The results of this paper suggest that both IRM and Managerial risk-taking (MRT) enable the Company's development through innovation in any types. Results show that the risk arises from a variety of sources and most importantly, innovation has a high probability of failure at several stages of organizational development; that is, all types of innovation involve risk, and all risk includes the possibility of failure.

This study offers two important implications. First, it confirms previous research findings and shows that how insurance company as a business enterprise can, to a large extent, bring success in innovation through integrated risk management.

Hence, a firm which adopts an IRM approach may gain more and useful insight about the affection of RM on the firm value and developing innovation activities because these companies have this opportunity to identify all innovation barriers through a systematic IRM framework.

On the other hand, the study predicts that managers who take risks are more inclined to develop innovative products, services, and processes.

If a manager has a low-tolerance for innovative and creative ideas, then it is unlikely to bring new product/process on a regular basis. Thus, managers with the risk appetite ability not only have a higher level of tolerance to innovation uncertainty but also have more assurance in completing innovation projects (Simon and Houghton, 2003).

The second implication of the study is that, even though the company's age did not have a significant effect on innovation; it seems that the company's age can strengthen the effect of managers' risk-taking and integrated risk on innovation. In other words, as the years of activity of insurance companies, more opportunities to innovate and use Risk Management Experience. Therefore, the nature of risk management can be associated with the company's age.

Additionally, We found that IRM was positively related to innovation. In addition, both managerial risk-taking and IRM have a significant indirect effect on product/process and administrative innovation. Thus, these risk management procedures help the companies to improve their organizational performance. This paper contributes to the literature on innovation in the insurance industry through the studying the role of IRM and Managerial risk-taking.

This study offers tremendous managerial implications. First, the hypothesized relationships are supported through the data. This contemporary model has validated to significantly explain both managerial risk-taking and IRM to acquire innovation within the insurance industry. Innovation risks in the insurance industry are among the issues that demand considerable attention. Risk management in insurance has an undeniable role in the improvement and effectiveness of insurance services and economic domains and consequently develops insurance industry. Since Iranian insurance companies, as a business enterprise, are subject to large fluctuations including operational, national, and political fields, dealing with all risks associated with these areas in the innovation process requires the use of national leadership. This indicates the high status of IRM in an insurance company in Iran.

Consequently, the current study could assist both insurance managers and CEO's to alter better guidelines and strategies to promote the outstanding risk framework throughout the entire company.

Second, our findings provide new insights concerning the role of IRM and managers' risk-taking in facilitating innovation consequences. The results of our study challenge the existing studies that propose the complementary effects of IRM and managerial risk-appetite for innovation. Third, insurance managers ought to consider the efforts and rewards cautiously, when adopting the IRM framework. More particularly, insurance companies will reap an excessive level of revenue and overall performance and will gain more perception about every risk across innovation procedure. Therefore, insurance companies' should construct an effective IRM system that oversees many systematic and strategic risks through organizational innovation. Fourth, insurance managers, need to be conscious that managers' risk-taking will inspire employees to be creative and could proportion their new thoughts freely in their organizational development.

This study has certain limitations regarding the study's sample and setting. First, data were collected in the insurance sector as one of the leading financial instructions, in Iran; therefore, the generalization of the study' results to other financial institutions or other industries and countries with different risk cultures might be rather. Thus, further studies in other areas and sectors to test the model are highly recommended. Second, there is not a specific scale to measure IRM. Therefore, the IRM scale in this study may not be the best measurement scale.

Third, we examined the only effect of managerial risk-taking and IRM on innovation. Future research is needed to examine the effect of other variables on innovation such as innovation culture or adding some control variables such as organizational risk climate or company's size.

References

Andersen, T. J. (2008). "The performance relationship of effective risk management: Exploring the firm-specific investment rationale". In: *Long range planning* 41.2, pp. 155–176.

Bilgin, M. H. and H. Danis (2016). Entrepreneurship, Business and Economics-Vol. 2: Proceedings of the 15th Eurasia Business and Economics Society Conference. Springer.

Bilgin, N., C. Kuzey, G. Torlak, and A. Uyar (2015). "An investigation of antecedents of organizational citizenship behavior in the Turkish hospitality industry: a structural equation approach". In: *International Journal of Culture, Tourism and Hospitality Research* 9.2, pp. 200–222.

- Birkinshaw, J., C. Bouquet, and J.-L. Barsoux (2011). "The 5 myths of innovation". In: *MIT Sloan Management Review* 52.2, p. 43.
- Bowers, J. and A. Khorakian (2014). "Integrating risk management in the innovation project". In: *European Journal of innovation management* 17.1, pp. 25–40.
- Bromiley, P., M. McShane, A. Nair, and E. Rustambekov (2015). "Enterprise risk management: Review, critique, and research directions". In: Long range planning 48.4, pp. 265–276.
- Chin, W. W., B. L. Marcolin, and P. R. Newsted (2003). "A partial least squares latent variable modeling approach for measuring interaction effects: Results from a Monte Carlo simulation study and an electronic-mail emotion/adoption study". In: *Information systems research* 14.2, pp. 189–217.
- Covin, J. G. and D. P. Slevin (1986). "The development and testing of an organizational-level entrepreneurship scale". In: Frontiers of entrepreneurship research 1.3, pp. 628–639.
- Dess, G. G. and G. T. Lumpkin (2005). "The role of entrepreneurial orientation in stimulating effective corporate entrepreneurship". In: *The Academy of Management Executive* 19.1, pp. 147– 156.
- Dess, G. G. and J. C. Picken (2000). "Changing roles: Leadership in the 21st century". In: Organizational dynamics 28.3, pp. 18–34.
- Eckles, D. L., R. E. Hoyt, and S. M. Miller (2014). "Reprint of: The impact of enterprise risk management on the marginal cost of reducing risk: Evidence from the insurance industry". In: *Journal* of Banking & Finance 49, pp. 409–423.
- Evan, W. M. (1966). "Organizational lag". In: Human organization 25.1, pp. 51–53.
- Faccio, M., M.-T. Marchica, and R. Mura (2011). "Large shareholder diversification and corporate risk-taking". In: *The Review of Financial Studies* 24.11, pp. 3601–3641.
- Fornell, C. and D. F. Larcker (1981). "Evaluating Structural Equation Models with Unobservable Variables and Measurement Error". In: Journal of Marketing Research 18.1, pp. 39–50.
- García-Granero, A., Ó. Llopis, A. Fernández-Mesa, and J. Alegre (2015). "Unraveling the link between managerial risk-taking and innovation: The mediating role of a risk-taking climate". In: Journal of Business Research.
- Giannakis, M., S. Croom, and N. Slack (2004). "Supply chain paradigms". In: Understanding supply chains, pp. 1–22.
- Hair, J. F., G. T. M. Hult, C. M. Ringle, M. Sarstedt, and K. O. Thiele (2017). "Mirror, mirror on the wall: a comparative evaluation of composite-based structural equation modeling methods". In: *Journal of the Academy of Marketing Science* 45.5, pp. 616–632.
- Hair, J., G. T. Hult, C. Ringle, and M. Sarstedt (2016). A primer on partial least squares structural equation modeling (PLS-SEM). Sage Publications.
- Jimenez-Jimenez, D., R. Sanz Valle, and M. Hernandez-Espallardo (2008). "Fostering innovation: the role of market orientation and organizational learning". In: *European Journal of innovation* management 11.3, pp. 389–412.
- Johnson ?, M. W. (2010). Risk management and innovation? Vol. ?3, p.41PM.
- Kim, Y., T. Y. Choi, and P. F. Skilton (2015). "Buyer-supplier embeddedness and patterns of innovation". In: International Journal of Operations and Production Management 35.3, pp. 318– 345.
- Kwak, D.-W., Y.-J. Seo, and R. Mason (2018). "Investigating the relationship between supply chain innovation, risk management capabilities and competitive advantage in global supply chains". In: International Journal of Operations & Production Management 38.1, pp. 2–21.

- Latham, S. F. and M. Braun (Feb. 2008). "Managerial Risk, Innovation, and Organizational Decline". In: Journal of Management 35.2, pp. 258–281.
- Li, J. and Y. I. Tang (2010). "CEO hubris and firm risk taking in China: The moderating role of managerial discretion". In: Academy of Management Journal 53.1, pp. 45–68.

Liu, G., W. W. J. Ko, I. Ngugi, and S. Takeda (2017). "Proactive entrepreneurial behaviour, market orientation, and innovation outcomes: A study of small-and medium-sized manufacturing firms in the UK". In: *European Journal of Marketing* 51.11/12, pp. 1980–2001.

March, J. G. (1996). "Learning to be risk averse." In: Psychological review 103.2, p. 309.

- Medrano, N. and C. Olarte-Pascual (2016). "The effects of the crisis on marketing innovation: an application for Spain". In: *Journal of Business & Industrial Marketing* 31.3, pp. 404–417.
- Meulbroek, L. (2001). "A better way to manage risk". In: Harvard business review 79.2, p. 22.
- Meulbroek, L. (2002a). "Integrated risk management for the firm: a senior manager's guide". In:
- Meulbroek, L. (2002b). "The promise and challenge of integrated risk management". In: Risk Management and Insurance Review 5.1, p. 55.
- Miller, K. D. and H. G. Waller (2003). "Scenarios, real options and integrated risk management". In: Long range planning 36.1, pp. 93–107.
- Nkundabanyanga, S. K., J. Opiso, W. Balunywa, and I. N. Nkote (2015). "Financial service outreach correlates: Managerial competence and risk-taking behaviour". In: *International Journal of Social Economics* 42.4, pp. 404–420.

Nocco, B. W. and R. M. Stulz (2006). "Enterprise risk management: Theory and practice". In: Journal of applied corporate finance 18.4, pp. 8–20.

Oxelheim, L., C. Wihlborg, and M. Thorsheim (2011). "The CFO's Information Challenge in Managing Macroeconomic Risk". In: *The Strategic CFO*. Springer, pp. 189–208.

- Pauget, B. and A. Dammak (2018). "Towards a relational innovation". In: Society and Business Review 13.1, pp. 5–14.
- Ringle, C. M., S. Wende, and J.-M. Becker (2015). SmartPLS 3. Boenningstedt: SmartPLS GmbH.
- Ritchie, B. and C. Brindley (2007). "Supply chain risk management and performance: A guiding framework for future development". In: International Journal of Operations & Production Management 27.3, pp. 303–322.
- Salinah, T. (2016). "Factors influencing the effectiveness of enterprise risk management (ERM) in public listed companies/Salinah Togok". PhD thesis.
- Santarelli, E. and R. Piergiovanni (1996). "Analyzing literature-based innovation output indicators: the Italian experience". In: *Research Policy* 25.5, pp. 689–711.
- Santos, D. F. L., L. F. C. Basso, H. Kimura, and E. K. Kayo (2014). "Innovation efforts and performances of Brazilian firms". In: *Journal of Business Research* 67.4, pp. 527–535.
- Simon, M. and S. M. Houghton (2003). "The relationship between overconfidence and the introduction of risky products: Evidence from a field study". In: Academy of Management Journal 46.2, pp. 139–149.
- Stosic, B., M. Mihic, R. Milutinovic, and S. Isljamovic (2017). "Risk identification in product innovation projects: new perspectives and lessons learned". In: *Technology Analysis & Strategic Management* 29.2, pp. 133–148.
- Tidd, J., J. Bessant, and K. Pavitt (2005). Managing innovation integrating technological, market and organizational change. John Wiley and Sons Ltd.
- Togok, S. H., C. R. Isa, and S. Zainuddin (2016). "Enterprise risk management adoption in Malaysia: A disclosure approach". In: Asian Journal of Business and Accounting 9.1, pp. 83–104.

Tommerberg, P. (2010). "Risk management models in two Norwegian institutions". PhD thesis.

Tushman, M. L. and C. A. O'Reilly III (1996). "Ambidextrous organizations: Managing evolutionary and revolutionary change". In: California management review 38.4, pp. 8–29.

Wu, J. and Z. Wu (2014). "Integrated risk management and product innovation in China: The moderating role of board of directors". In: *Technovation* 34.8, pp. 466–476.

Enhanced FMEA for Supply Chain Risk Identification

Lu Lu¹, Zhou Rong¹, Robert de Souza¹

1 - National University of Singapore

Supply chain risk identification is fundamental for supply chain risk management. Its main purpose is to find critical risk factors for further attention. The failure mode effect analysis (FMEA) is well adopted in supply chain risk identification for its simplicity. It relies on domain experts' opinions in giving rankings to risk factors regarding three decision factors, e.g. occurrence frequency, detectability, and severity equally. However, it may suffer from subjective bias of domain experts and inaccuracy caused by treating three decision factors as equal. In this study, we propose a methodology to improve the traditional FMEA using fuzzy theory and grey system theory. Through fuzzy theory, we design semantic items, which can cover a range of numerical ranking scores assessed by experts. Thus, different scores may actually represent the same semantic item in different degrees determined by membership functions. In this way, the bias of expert judgement can be reduced. Furthermore, in order to build an appropriate membership function, experts are required to think thoroughly to provide three parameters. As the results, they are enabled to give more reliable judgement. Finally, we improve the ranking accuracy by differentiating the relative importance of decision factors. Grey system theory is proposed to find the appropriate weights for those decision factors through identifying the internal relationship among them represented by grey correlation coefficients. The results of the case study show the improved FMEA does produce different rankings from the traditional FMEA. This is meaningful for identifying really critical risk factors for further management.

Keywords: supply chain risk identification; FMEA; grey system theory; fuzzy set theory

First recieved: 01.Jul.2018 Revised: 10.Jul.2018 Accepted: 15.Jul.2018

1 Introduction

Risk identification involving both risk classification and risk ranking can be seen as a fundamental work for risks assessment. It identifies critical risks that need further assessment and treatment (Berman and Putu, 2012). Generally, researchers categorize risks into several groups for systematically risk identification. In our study, risks are classified into three levels—the macro level, the company level and the industry level. Risks on the macro level may influence the whole supply chain's operation; risks on the company level are from operation activities of a company; risks on the industry level are from the development of industry (Zhou et al., 2012).

Subsequently, we still need to identify most relevant risk factors so that only those relevant and important ones are studied further. As a large number of risk factors that may be involved, the easiest way is through risk ranking so that a company can effectively mitigate them (Chopra and Sodhi, 2004). A comparison of nine risk ranking techniques is shown in Table 1, where techniques are compared in five attributes—complexity of application, risk consequence analysis, risk probability analysis, quantified output, and objectivity.

		-	-		
Techniques	Complexity of applica-	Risk con- sequence	Risk prob- ability	Quantitative output	Objectivity
		didiysis	didiysis		
Structured "What- f"(SWIFT)	Low	A	A	No	Low
Fault tree analysis	Medium	NA	A	Yes	High
Cause and conse-	High	A	A	Yes	High
quence analysis					
Cause and effect anal-	Medium	A	NA	No	Medium
vsis					
Decision tree	High	A	A	Yes	Medium
FMEA	Medium	A	A	Yes	Medium
Hazard analysis and	Medium	A	NA	No	Medium
critical control points					
Analysis hierarchy	Medium	NA	NA	Yes	High
orocess					
Bayesian statistics	High	A	NA	Yes	High
and Bayes nets					

Table 1: Comparison of different techniques

According to the above table, risk ranking techniques can be classified into four big categories. The first category is a supporting method, which can only give a general analysis about the risks. For example, SWIFT uses a series of "What if" questions to identify the deviations from normal conditions with the help of a predefined checklist. The second category uses scenario analysis, which is good at analyzing the causes of risks. Fault tree analysis, cause and consequence analysis, cause and effect analysis, and decision tree belong to this category (Dakas et al., 2009, Hauptmanns, 2010, Hichem and Pepijn, 2007). The third category is function analysis method including FMEA, hazard analysis, and critical control points. They focus on analyzing the effects of risks. The final category is the statistical method, which applies the statistical knowledge into the analysis process. AHP and BBN belong to this category.

FMEA has been adopted widely as it can produce quantitative output, which is desirable for risk ranking. However, it may be biased as the opinions of domain experts can be subjective. The target of the current study is to improve FMEA for its objectivity using fuzzy set theory and grey system theory.

The structure of the paper is as follows. The next section describes the traditional FMEA; section 3 presents the enhanced FMEA. A case study is provided in section 4. Finally, conclusions are made in section 5.

2 The Traditional FMEA

FMEA identifies failure modes and mechanisms as well as their effects. There are several types of FMEA, e.g. design FMEA, system FMEA, process FMEA, service FMEA, software FMEA, etc. The current study adopts the FMEA methodology. In the study, the system means the supply chain while the failure mode refers to the potential supply chain risk. For each risk, experts give three scores between one and ten regarding the risk's occurrence frequency (OF), detectability and severity. Then the risk priority number (RPN) can be calculated through multiplying these three score and represents the risk impact of the risk factor. The higher is the RPN, the more critical is the risk.
Risk	Experts	OF	Detectability	Severity
Risk 1	B1	4	5	9
	B2	5	1	5
	B3	2	2	8
	Average	3.7	2.7	7.3
Risk 2	B1	9	3	4
	B2	6	2	2
	B3	8	3	5
	Average	7.7	2.7	3.7

Table 2: Scores marked by experts

2.1 An Example

We assume that there are three domain experts—B1, B2 and B3, who give ranks regarding the risk impacts of two risk factors: risk 1 and risk 2 in a scale of 1 to 10. The greatest rank, 10, refers to the greatest risk impact. The summary of scores marked by experts is shown in Table 2.

Then, using the average numbers of three experts' rankings, RPNs for risks 1 and 2 can be calculated as RPN₁=3.7*2.7*7.3=7.3 and RPN₂=7.7*2.7*3.7=7.7, respectively. Since RPN₂ is greater than RPN₁, risk 2 is more risky than risk 1 according to those three experts.

2.2 Limitations of Traditional FMEA

Through the above example, three limitations of the traditional FMEA can be recognized. Firstly, there may be ranking differences among experts, which can lead to the inaccuracy of outcomes. For example, for the same degree of risk impact, expert B1 may score 9 while expert B2 scores 7. Secondly, the approach depends on the experience and knowledge of experts to a large degree and the outcome can be very subjective. Finally, the RPN formula above does not consider the relative importance of three decision factors. The severity of a risk factor could be more important than OF or detectability while in the current approach, three decision factors are treated as equal. As a result, the above RPN may not be able to give accurate risk rankings.

3 Improved FMEA

To reduce the limitations of the traditional FMEA, we propose to improve it using fuzzy set theory and grey relation analysis. The primary procedure of the improved FMEA is as follows.

- Identify the relevant risk factors (regarding the risk categories) and domain experts.
- Reduce the expert bias using Fuzzy Set theory.
- Experts reach consensus for each risk.
- Improve assessment precision through Grey Correlation Analysis.
- Ranking risk factors.

Specially, we use Fuzzy Set theory in step 2 to reduce experts' ranking difference and Grey Correlation Analysis in step 4 to improve assessment precision through applying appropriate weightages to decision factors.

3.1 Fuzzy Set

In the classical discrete sets, an element either belongs to a set or it does not. But for fuzzy sets, their elements have degrees of membership according to certain membership functions (Abdelgawad and Fayek, 2011). There are many types of membership functions based on the graphs, e.g. triangular, trapezoidal, Gaussian, generalized bell, sigmoid, and others. We choose the triangular membership function to improve the traditional FMEA. A triangular membership function is specified by three parameters, a, b, and c in formula (1):

$$M(x) = \begin{cases} 0, & x \le a \\ \frac{x-a}{b-a}, & a \le x \le b \\ \frac{x-a}{b-a}, & b \le x \le c \\ 0, & c \le x \end{cases}$$
(1)

316

where M refers to the membership of score x. The value of the membership ranges from 0 to 1. The value of 1 represents the full membership.

3.2 Grey Relation Analysis

Grey system theory was first developed in 1982 (Deng, 1982). A grey system generally refers to a system lacking certain information, e.g. structure message, operation mechanism, or behavior document. The aims of the grey system theory are to provide theory, techniques, notions and ideas for resolving latent and intricate systems (Deng, 1982). This study adopts the grey relation analysis, which describes the relationships between one main factor and all the other factors in a given system. The degree of correlation among different factors is measured by their grey correlation coefficient. The greater is the value of coefficient, the closer relationship is between the two factors.

4 Case Study

In this section, we use a case study to improve the traditional FMEA through the proposed methodology. Assuming there are three risk factor, e.g. raw material shortage, labour availability, and natural disaster as well as three experts, e.g. B1, B2, B3, the ranking process follows the five steps illustrated in sections 4.1 to 4.5.

4.1 Identify the Relevant Risks List and Experts

Table 3 summarizes the risk factors and experts, which are identified for the ranking process.

Risks identified are raw material shortage (R1), labor availability (R2), and natural disaster (R3) while three experts, B1, B2 and B3 are from supply chain, operation, and R&D departments, respectively.

Potential Risks	Experts	Department
Raw material shortage (R1)	B1	Supply chain department
Labor availability (R2)	B2	Operation department
Natural disaster (R3)	B3	R&D department

Table 3: Risks and experts identified

4.2 Reduce Experts Bias Using Fuzzy Set Theory

The process of reducing expert bias (ranking difference) includes 1) setting up the fuzzy semantic assessment set, 2) determining membership functions for fuzzy semantic items, and 3) calculating specific numbers for fuzzy semantic items through defuzzification.

4.2.1 Set up the fuzzy semantic assessment set (Faisal and Sarah, 2015)

In the study, the fuzzy semantic assessment set is designed to include the sematic items of "very high", "high", "medium", "low", and "very low". The implications of the five semantic items in terms of three decision factors, occurrence frequency, detectability, and severity are illustrated in Table 4.

The definitions of semantic items can guide experts to score a risk factor. In the above table, the semantic item "very high" means "occurs in high frequency" for occurrence frequency, "very hard to detect" for detectability, and "lead to failure of whole supply chain" for severity.

The purpose of the fuzzy sematic assessment set is to provide a few number of semantic items like "very high", "high", etc. to reflect experts' numerical rankings from 1-10. For example, given the same level of risk impact, different experts may give different numerical scores of "7" or "9". But in terms of semantic items, these scores can be translated to either "high" or "very high" with certain degrees of membership. In this way, the ranking difference, e.g. the bias of experts can be reduced.

Semantic items	Occurrence frequency	Detectability	Severity
Very high	Occurs in high frequency	Very hard to detect	Lead to failure of whole supply chain
High	Occurs frequently	Hard to detect	Lead to the failure of critical parts of supply chain
Medium	Occurs occasionally	Can be detected occasionally	Lead to the failure of non-essential parts of supply chain
Low	Occurs in less times relatively	Easy to detect	A little influence on the supply chain
Very Low	Unlikely to occur	Very easy to detect	Mainly no influence on the supply chain

Table 4: Implication of semantic items

4.2.2 Determine membership functions of fuzzy semantic items

The membership functions for the five semantic items, "very low (VL)", "low (L)", "medium (M)", "high (H)", and "very high (VH)" are given in Figure 1 regarding equation (1).



Figure 1: The structure of membership functions

Each membership function is determined by three parameters describing the range of membership. For example, risk factors with scores from a_l to c_l can be generally considered as low risk factors. Score b_l represents full "low risky" membership. For risk factors with scores close to a_l become less "low" but more "very low" risky. Similarly, those with scores close to c_l become less "low" but more "medium" risk. Three parameters determining a membership function are important and need domain experts' inputs to find out their values.

Thus, instead of giving only one number representing absolute "low" risk, an expert should think over and give a range of numbers which can also be considered as "low" risk but with different degrees. In this way, experts are enabled to think thoroughly and give more feasible ranks of risks. Table 5 collects the inputs of three parameters for each semantic item from three experts (B1, B2, and B3). The last row is the average of three scores.

Subsequently, we have three parameters for each membership function through Table 5:

- For semantic item "very low": $a_{vl} = 0$, $b_{vl} = 0$, $c_{vl} = 2.6$;

Expert	Very low	Low	Medium	High	Very high
B1	0, 0, 2.8	1.6, 3.3, 4.8	3.8, 5.8, 7.8	6.8, 8.3, 9.8	8.8, 10,10
B2	0, 0, 2.6	1.2, 3.4, 5.8	3.5, 5.5, 7.6	6.1, 7.8, 9.8	8.6, 10, 10
B3	0, 0, 2.4	1.4, 3.5, 5.1	3.5, 5.5, 8.0	6.6, 8.6, 9.8	8.5, 10, 10
Average	0, 0, 2.6	1.4, 3.4, 5.2	3.6, 5.6, 7.8	6.5, 8.2, 9.8	8.6, 10, 10

Table 5: Experts' inputs regarding three parameters

- For semantic item "low": $a_l = 1.4, b_l = 3.4, c_l = 5.2$;

- For semantic item "medium": $a_m = 3.6, b_m = 5.6, c_m = 7.8$;

- For semantic item "high": $a_h = 6.5, b_h = 8.2, c_h = 9.8$;

- For semantic item "very high": $a_{vh} = 8.6$, $b_{vh} = 10$, $c_{vh} = 10$.

4.2.3 Calculate the specific number of fuzzy semantic item

The aim of the current step is to obtain the specific number, representing a semantic item in one number. Defuzzification is introduced to transfer the values of three parameters into one specific number and the formula (2) adopted is as follow(Chen, 2010).

$$I = \frac{a+ab+c}{4} \tag{2}$$

We then have the specific number of each semantic item as follows.

- For semantic item "very low": $I_{vl} = 0.65$;
- For semantic item "low": $I_l = 3.4$;

- For semantic item "medium": $I_m = 5.7$;
- For semantic item "high": $I_h = 8.2$;
- For semantic item "very high": $I_{vh} = 9.7$;

In summary, the target of the current step is to reduce bias from experts in two aspects. On the one hand, we establish a judgment standard using a fuzzy semantic assessment set so that we can have the same semantic item (very low, low, medium, high, and very high) from the ranges of scores given by experts. On the other hand, in order to establish membership functions, each expert should give three numbers for each fuzzy semantic item describing the extension of membership. This enables them to think thoroughly and subsequently reduces subjectivity of judgment.

4.3 Establish the Judgment Matrix

Now, experts can rank risks R1 to R3 using semantic items ("very low", "low", "medium", "high", and "very high") in terms of their occurrence frequency, detectability and severity. The outcomes of experts' rankings are given in Table 6.

$$x = \begin{bmatrix} x_{o1} & x_{o2} & x_{o3} & x_{o4} & x_{o5} \\ x_{d1} & x_{d2} & x_{d3} & x_{d4} & x_{d5} \\ x_{s1} & x_{s2} & x_{s3} & x_{s4} & x_{s5} \end{bmatrix}$$
(3)

Where "o", "d", and "s" represent decision factors "occurrence frequency", "detectability", and "severity", respectively. The first row represents an expert's rankings of "occurrence frequency" in terms of five semantic items "very high", "high", "medium", "low", and "very low", respectively. Similarly, the second and third rows are the experts' rankings of "detectability" and "severity".

Thus, according to Table 6, we can summarize three experts' judgements on three risks regarding three decision factors in tables 8-10. Each entry records the total counts/percentage of same judgement from experts. For example, in Table 7, 3 out 3 (100%) experts think that the "occurrence frequency" of Risk 1 is "very high"; only 1 out 3 (33%) of them think that the "severity" of Risk 1 is "high".

Expert	Risk	Occurrence frequency	Detectability	Severity
B1	R1	М	L	VH
	R2	н	L	М
	R3	L	М	н
B2	R1	М	М	VH
	R2	М	М	М
	R3	М	L	VH
B3	R1	М	L	н
	R2	Н	М	М
	R3	М	L	Н

Table 6: Judgment of experts

Table 7: Three experts' average judgment for risk 1

	VH	Н	М	L	VL	
Occurrence frequency	0	0	3 (100%)	0	0	
Detectability	0	0	1(33%)	2(67%)	0	
Severity	2(67%)	1(33%)	0	0	0	

	VH	Н	М	L	VL
Occurrence frequency	0	2(67%)	1(33%)	0	0
Detectability	0	0	2(67%)	1(33%)	0
Severity	0	0	3(100%)	0	0

Table 8: Three experts' average judgment for risk 2

Table 9: Three experts' average judgment for risk 3

	VH	Н	М	L	VL
Occurrence frequency	0	0	2(67%)	1(33%)	0
Detectability	0	0	1(33%)	2(67%)	0
Severity	1(33%)	2(67%)	0	0	0

$$x^{1} = \begin{bmatrix} x_{o1}^{1} x_{o2}^{1} x_{o3}^{1} x_{o4}^{1} x_{o5}^{1} \\ x_{d1}^{1} x_{d2}^{1} x_{d3}^{1} x_{d4}^{1} x_{d5}^{1} \\ x_{s1}^{1} x_{s2}^{1} x_{s3}^{1} x_{s4}^{1} x_{s5}^{1} \end{bmatrix} = \begin{bmatrix} 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0.33 & 0.67 & 0 \\ 0.67 & 0.33 & 0 & 0 & 0 \end{bmatrix}$$
(4)

$$x^{2} = \begin{bmatrix} x_{o1}^{2} x_{o2}^{2} x_{o3}^{2} x_{o4}^{2} x_{o5}^{2} \\ x_{d1}^{2} x_{d2}^{2} x_{d3}^{2} x_{d4}^{2} x_{d5}^{2} \\ x_{s1}^{2} x_{s2}^{2} x_{s3}^{2} x_{s4}^{2} x_{s5}^{2} \end{bmatrix} = \begin{bmatrix} 0 & 0.67 & 0.33 & 0 & 0 \\ 0 & 0 & 0.67 & 0.33 & 0 \\ 0 & 0 & 0 & 0 & 0 \end{bmatrix}$$
(5)

$$x^{3} = \begin{bmatrix} x_{o1}^{3} x_{o2}^{3} x_{o3}^{3} x_{o4}^{3} x_{o5}^{3} \\ x_{d1}^{3} x_{d2}^{3} x_{d3}^{3} x_{d4}^{3} x_{d5}^{3} \\ x_{s1}^{3} x_{s2}^{3} x_{s3}^{3} x_{s4}^{3} x_{s5}^{3} \end{bmatrix} = \begin{bmatrix} 0 & 0 & 0.67 & 0.33 & 0 \\ 0 & 0 & 0.33 & 0.67 & 0 \\ 0.33 & 0.67 & 0 & 0 & 0 \end{bmatrix}$$
(6)

Subsequently, the judgment matrices for risks 1, 2 and 3 are listed in matrices of (4) (5) and (6). They represent the rankings from three experts given in terms of five semantic items.

4.4 Improve Precision through Grey System Theory

In the current section, we firstly establish an assessment matrix based on step 3 and then a reference matrix using the most risky semantic item. Subsequently, the degree of relevancy is measured through the grey correlation coefficient. Finally, we apply weights to decision factors (occurrence frequency, detectability, and severity) to differentiate their relevant importance.

4.4.1 Establishment of assessment matrix and reference matrix

The assessment matrix is built based on the judgment matrices of (4) (5) and (6) and specific numbers of sematic items. The assessment matrix for three risks are formed as follow:

$$R = \begin{bmatrix} r_{1o} & r_{1d} & r_{1s} \\ r_{2o} & r_{2d} & r_{2s} \\ r_{3o} & r_{3d} & r_{3s} \end{bmatrix}$$
(7)

Where

 $r_{io}=I_{vh}\times x_{o1}^i+I_h\times x_{o2}^i+I_m\times x_{o3}^i+I_l\times x_{o4}^i+I_{vl}\times x_{o5}^i$ is the score of "occurrence frequency" for risk i.

 $r_{id}=I_{vh}\times x_{d1}^i+I_h\times x_{d2}^i+I_m\times x_{d3}^i+I_l\times x_{d4}^i+I_{vl}\times x_{d5}^i$ is the score of "detectability" for risk i.

 $r_{is}=I_{vh}\times x_{s1}^i+I_h\times x_{s2}^i+I_m\times x_{s3}^i+I_l\times x_{s4}^i+I_{vl}\times x_{s5}^i$ is the score of "severity" for risk i.

With specific numbers for five sematic items, e.g. $I_{vl} = 0.65$, $I_l = 3.4$, $I_m = 5.7$, $I_h = 8.2$, and $I_{vh} = 9.7$, we have:

- For risk 1, $r_{1o} = 5.7$, $r_{1d} = 4.2$, $r_{1s} = 8.9$;
- For risk 2, $r_{2o} = 7.4$, $r_{2d} = 4.9$, $r_{2s} = 5.7$;
- For risk 3, $r_{3o} = 4.9$, $r_{3d} = 4.2$, $r_{3s} = 8.7$;

Then, we have the assessment matrix in (8).

$$R = \begin{bmatrix} r_{1o} & r_{1d} & r_{1s} \\ r_{2o} & r_{2d} & r_{2s} \\ r_{3o} & r_{3d} & r_{3s} \end{bmatrix} = \begin{bmatrix} 5.7 & 4.2 & 8.9 \\ 7.4 & 4.9 & 5.7 \\ 4.9 & 4.2 & 8.7 \end{bmatrix}$$
(8)

Furthermore, we use the specific number of semantic item "very high", e.g. 9.7 to establish the reference matrix in (9).

$$R_f = \begin{bmatrix} r_{fo} & r_{fd} & r_{fs} \end{bmatrix} = \begin{bmatrix} 9.7 & 9.7 & 9.7 \end{bmatrix}$$
(9)

326

This reference matrix represents a very risky situation of a risk factor where the "occurrence frequency", "detectability", and "severity" are all "very high" (Table 4).

4.4.2 Calculating the grey correlation coefficient

Now with the assessment matrix R in (8) and the reference matrix R_f in (9), we can calculate the grey correlation coefficient between them using the grey correlation coefficient function (10) (Du et al., 2011):

$$\lambda(x_{fj}, x_{ij}) = \frac{\min_{i} |x_{fj} - x_{ij}| + v \max_{i} |x_{fj} - x_{ij}|}{|x_{fj} - x_{ij}| + v \max_{i} |x_{fj} - x_{ij}|}$$
(10)

Where

i refers to risk *i*;

j refers to decision factor *j*;

f refers to the reference matrix entry;

 $\lambda(x_{fj},x_{ij})$ refers to the grey correlation coefficient of entries x_{fj} and $x_{ij};$

 υ is the distinguishing coefficient; its value is within [0,1] and normally v=0.5.

Table 10 presents the results of all $|x_{fj} - x_{ij}|$ and the minimal and maximal values are $\underset{i}{\min} |x_f - x_i| = 0.8$ and $\underset{i}{\max} |x_f - x_i| = 5.5$.

Thus, we have

$$\lambda_{1o} = \frac{\underset{i}{\min} |x_{fj} - x_{ij}| + \underset{i}{0.5\max} |x_{fj} - x_{ij}|}{|x_{fj} - x_{ij}| + \underset{i}{0.5\max} |x_{fj} - x_{ij}|} = \frac{0.8 + 0.5 \times 5.5}{4 + 0.5 \times 5.5} = 0.52$$
(11)

No.	Occurrence frequency	Detectability	Severity
$\overline{\Delta_{1j} = x_{fj} - x_{1j} }$	$ x_{fo} - x_{1o} = 4$	$ x_{fd} - x_{1d} = 5.5$	$ x_{fs} - x_{1s} = 0.8$
$\Delta_{2j} = x_{fj} - x_{2j} $	$ x_{fo} - x_{2o} = 2.3$	$ x_{fd} - x_{2d} = 4.8$	$ x_{fs} - x_{2s} = 4$
$\Delta_{3j} = x_{fj} - x_{3j} $	$ x_{fo} - x_{3o} = 4.8$	$ x_{fd} - x_{3d} = 5.5$	$ x_{fs} - x_{3s} = 1$

Table 10: Results of all $|x_{fj} - x_{ij}|$

Similarly, we can get the grey correlation coefficient matrix (12).

$$\lambda = \begin{bmatrix} \lambda_{1o} & \lambda_{1d} & \lambda_{1s} \\ \lambda_{2o} & \lambda_{2d} & \lambda_{2s} \\ \lambda_{3o} & \lambda_{3d} & \lambda_{3s} \end{bmatrix} = \begin{bmatrix} 0.52 & 0.43 & 1 \\ 0.7 & 0.47 & 0.52 \\ 0.47 & 0.43 & 0.95 \end{bmatrix}$$
(12)

Furthermore, assuming the weights of decision factors are given in matrix (13).

$$\omega = \begin{bmatrix} 0.3 & 0.2 & 0.5 \end{bmatrix} \tag{13}$$

Where ω_o is the weight of Occurrence frequency, ω_d the weight of detectability, and ω_s the weight of severity.

We have

$$G = \begin{bmatrix} \omega_o \lambda_{1o} + \omega_d \lambda_{1d} + \omega_s \lambda_{1s} \\ \omega_o \lambda_{2o} + \omega_d \lambda_{2d} + \omega_s \lambda_{2s} \\ \omega_o \lambda_{3o} + \omega_d \lambda_{3d} + \omega_s \lambda_{3s} \end{bmatrix} = \begin{bmatrix} 0.742 \\ 0.564 \\ 0.702 \end{bmatrix}$$
(14)

Matrix G is the final rankings of three risks regarding three decision factors considering three experts' judgement. The final scores of risks 1, 2, and 3 are 0.742, 0.564, and 0.702, respectively. As the 0.742 is the greatest, R1 is the most risky one while R3 is the second and R2, the third. In summary, from the highest to the lowest in term of risk impacts, the ranking of studied risks is R1>R3>R2.

4.5 Compare the Traditional and the Improved FMEA

For the assessment matrix R (8), the RPN applying the traditional FMEA is as follows.

$$RPN = \begin{bmatrix} r_{1o} \times r_{1d} \times r_{1s} \\ r_{2o} \times r_{2d} \times r_{2s} \\ r_{3o} \times r_{3d} \times r_{3s} \end{bmatrix} = \begin{bmatrix} 5.7 \times 4.2 \times 8.9 \\ 7.4 \times 4.9 \times 5.7 \\ 4.9 \times 4.2 \times 8.7 \end{bmatrix} = \begin{bmatrix} 213 \\ 206 \\ 179 \end{bmatrix}$$
(15)

Thus, the ranking of risks is R1>R2>R3, which is different from the outcome of the improved FMEA in matrix (14). The reason is that the improved FMEA considers weights (matrix (13)) for three decision factors.

Furthermore, if we directly include those weights into the assessment matrix (8) without applying the grey correlation coefficient. The result is as follows.

$$RPN^{1} = \begin{bmatrix} 5.7 \times 0.3 \times 4.2 \times 0.2 \times 8.9 \times 0.5\\ 7.4 \times 0.3 \times 4.9 \times 0.2 \times 5.7 \times 0.5\\ 4.9 \times 0.3 \times 4.2 \times 0.2 \times 8.7 \times 0.5 \end{bmatrix} = \begin{bmatrix} 6.39\\ 6.18\\ 5.37 \end{bmatrix}$$
(16)

The new ranking becomes the same as the one from the traditional FMEA, e.g. R1>R2>R3, but different from the improved FMEA. This emphasizes the importance of including the grey correlation coefficient in allocating appropriate weightages to decision factors.

5 Conclusion

In this study, the methodology to improve FMEA for supply chain risk identification is proposed in order to reduce the bias from domain experts and improve the ranking accuracy.

First of all, the subjective bias in ranking from experts can be reduced through establishing semantic items, which are linked to numerical scores through fuzzy membership functions. In this way, even though experts give difference scores for the same level of risk impact, those scores can still represent the same semantic meaning, perhaps in different degrees. In this way, the bias from experts can be reduced.

Furthermore, in order to build a membership function, three parameters are requested to represent the coverage of a semantic item in terms of numerical scores. This enables experts to think thoroughly and further improves the reliability of their judgement.

Finally, in the traditional FMEA, decision factors are treated equally in their roles to determine the impact of a risk. This may not rational. In the improved FMEA, we differentiate the importance of decision factors in ranking risk impacts. The grey correlation coefficient is adopted to extract appropriate weights for decision factors. This further improves the accuracy of the ranking.

References

- Abdelgawad, M. and A. Fayek (2011). "Fuzzy reliable analyzer:quantitative assessment of risk events in the construction industry using fuzzy fault-tree analysis". In: *Journal of Construction Engineering and Management* 137.4, pp. 294–302.
- Berman, K. and D. K. Putu (2011). "SCRIS: a knowledge-based system tool for assisting manufacturing organizations in identifying supply chain risks". In: *Journal of Manufacturing Technology Management* 23.7, pp. 834–852.
- Chen, M. S. (2010). "Evaluating the rate of aggregative risk in software development using fuzzy set theory". In: *Cybernetics and Systems: An International Journal* 30.1, pp. 57–75.
- Chopra, S. and M. S. Sodhi (2004). "Managing risk to avoid supply chain breakdown". In: MIT Slogan Management Review 46.1, pp. 53–62.
- Dakas, I. M., D. A. Karras, and D. C. Panagiotakopoulos (2009). "Fault tree analysis and fuzzy expert systems: early warning and emergency response of landfill operations". In: *Environmental Modeling & Software* 24.1, pp. 8–25.

Deng, J. (1982). "Control problems of grey systems". In: Systems and Control Letter 5, pp. 288-294.

- Du, D. L., J. Qiu, and H. Y. Zhao (2011). "Risk assessment study of manufacturing green supply chain based on grey theory". In: 2011 China located International Conference on Information Systems for Crisis Response and Management.
- Faisal, A. and L. S. Sarah (2015). "A fuzzy-based integrated framework for supply chain risk assessment". In: International journal of production economics 161, pp. 54–63.
- Hauptmanns, U. (2010). "A decision-making framework for protecting process plants from flooding based on fault tree analysis". In: *Reliability Engineering & Sytem Safety* 95.9, pp. 970–980.
- Hichem, B. and C. Pepijn (2007). "Dynamic fault tree analysis using input/output interactive markov chains. dependable systems and networks". In: *Internal Conference on 37th Annual IEEE/IFIP*, pp. 708–717.
- Zhou, R., R. De Souza, and M. Goh (2013). "Risk management of complex supply chains part 1: Supply chain risk and complex systems". In: Vol. 12-Nov-SCI-09.